

Durham E-Theses

Rationality or Irrationality? Deterrence in the Survival Strategy of the North Korean Regime

POWCHAROEN, PHAN-ORN

How to cite:

POWCHAROEN, PHAN-ORN (2015) Rationality or Irrationality? Deterrence in the Survival Strategy of the North Korean Regime, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/11262/

Use policy

The full-text may be used and/or reproduced, and given to third parties in any format or medium, without prior permission or charge, for personal research or study, educational, or not-for-profit purposes provided that:

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Rationality or Irrationality? Deterrence in the Survival Strategy of the North Korean Regime

Phan-orn Powcharoen

Thesis submitted for the degree of Doctor of Philosophy

School of Government and International Affairs

Durham University

2015

Declaration

Signature:
Print Name:
Date:

ABSTRACT

Much of the academic literature portrays North Korea as an irrational rogue state whose behaviour is aggressive and dangerous, as exhibited through the regime's continuous efforts to obtain nuclear weapons and missile delivery systems, and attempts to threaten neighbouring countries with words or provocations. North Korea is viewed as a cause of regional instability since the regime's high level of defence preparedness and acts of provocation drive the other regional powers to search for their own security in an attempt to offset the belligerence of this state. Although International Relations theory has long recognised the problem of the insecurity states experience in living in the system of international anarchy, the case of North Korea seems to shows that states can have an excessive deterrence threshold that arises from a high level of threat perception, which is driven by the environment of the regime and specific internal determinants inside the North Korea system. Nuclear weapons development has some connection with the regime's internal system when it can help support the legitimacy and power of the leader and at the same time is used in a strategic way to cover the regime's failures in supplying essential. Moreover, the strategy of excessive deterrence is used by North Korea for international leverage with the United States and the international community as the nuclear weapons will be used by North Korea in the negotiating process.

The question then arises if North Korea's deterrence strategy conforms to assumptions of rational actor behaviour when it is clear that the situation of high tension benefits North Korea more than peaceful coexistence and the deterrence strategy is used beyond the survival purposes of a normal state. It is the aim of the thesis to find out if North Korea's behaviour and deterrence strategy are rational or irrational when considered through the theory of rational deterrence and the idea of security dilemma. In particular deterrence theory suggests that states should not actively seek a security dilemma with multiple antagonists yet seemingly this is what the DPRK posture of excessive deterrence achieves. It is hoped to explain if North Korea is one of the normal states calculating its own survival and interest in the condition of anarchy or is an irrational actor that uses unpredictable and criminal behaviour in promoting security dilemmas in the inter-state system from its own distorted view of regime legitimacy and survival.

ACKNOWLEDGMENTS

First of all, this thesis would have been impossible without guidance and help of my supervisor, Dr David Kerr. I would like to express my gratitude to him for his dedicated supervision, useful guidance, and continuous support throughout my study. I cherish all the meetings I have had with him since our first meeting in October 2009.

Contributions from all my interviewees also help me complete my thesis. I would like to thank them for their time in sharing knowledge and some useful information with me. Their names are listed later at the end of the thesis.

I am grateful to the Office of the Higher Education Commission, Ministry of Education Thailand, who provides me with a financial support throughout these years.

Last but not least, I would like to thank my family and friends for their love, support, and encouragement throughout my time of study.

ABBREVIATIONS

BDA Banco Delta Asia

CIA Central Intelligence Agency

CVID Complete, Verifiable and Irreversible Dismantlement

DMZ Demilitarized Zone

HEU Highly Enriched Uranium

KCNA Korean Central News Agency

KEDO Korean Peninsula Energy Development Organization

KPA Korean People's Army

KWP Korean Workers' Party

LWR Light Water Reactor

NDC The National Defence Commission

NLL Northern Limit Line

NPT Non-Proliferation Treaty

PSI Proliferation Security Initiative

TCOG Trilateral Coordination and Oversight Group

UNSC UN Security Council

WMD Weapons of Mass Destruction

TABLE OF CONTENTS

Chapter 1: Introduction	1
1.1 North Korea: an Irrational Actor?	1
1.2 North Korea's Security and the Problem of Rational Deterrence	2
1.3 Research Structure	8
1.4 Methodology	10
Chapter 2: Using IR Theory to Define North Korea and Its Behaviour	14
2.1 International Relations Theory: Anarchy and Security	15
2.2 Rational Deterrence in the Cold War and After	22
2.3 Applying Security and Strategic Theory to North Korea's Behaviour	28
2.4 The Establishment of Hypotheses and Research Questions	35
Chapter 3: The North Korean Regime	39
3.1 The Regime of North Korea: the History, the Structure and the System	40
3.2 North Korea's Threat Perception	47
3.3 Threat Perception, Nuclear Weapons, and Deterrence	56
3.4 Conclusion	64
Chapter 4: Inter-Korean Relations	67
4.1 South Korea's Policy towards North Korea and the Inter-Korean Relations	68
4.2 South Korea and the North Korea's First Nuclear Crisis in 1994	77
4.3 South Korea and the North Korea's Second Nuclear Crisis in 2002	80
4.4 Inter-Korean Relations, Unification, and North Korea's Threat Perception	86
4.5 Conclusion	97
Chapter 5: North Korea and the Great Powers in Northeast Asia	100
5.1 China and the North Korea's First Nuclear Crisis	102
5.2 China and the North Korea's Second Nuclear Crisis	106

	5.3 Japan and the North Korea's First Nuclear Crisis	110
	5.4 Japan and the North Korea's Second Nuclear Crisis	113
	5.5 Russia and the North Korea's First Nuclear Crisis	117
	5.6 Russia and the North Korea's Second Nuclear Crisis	118
	5.7 Interaction between the Regional Players and North Korean's Perception	122
	5.8 Conclusion	129
Cł	napter 6: North Korea and the United States	131
	6.1 The First Nuclear Crisis and the US Policy during the Clinton Administration (199: 2001)	
	6.2 The Second Nuclear Crisis and the US Policy during the Bush Administration (200 2008)	
	6.3 The US Policy during the Obama Administration (2009-present)	149
	6.4 US Foreign Policy towards North Korea in the Post-Cold War World	152
	6.5 Conclusion	157
Cł	napter 7: The Attempt at Multilateral Settlement	160
	7.1 The Failure of the Six-Party Talks	162
	7.2 North Korea's Threat Perception and the Six-Party Talks	174
	7.3 Future Prospects for the Six-Party Talks	179
	7.4 Conclusion	183
Cł	napter 8: Conclusion	185
Aı	opendix 1: List of Interviewees	211
Aı	opendix 2: Note to the Interviewees	212
Bi	bliography	215

Chapter 1: Introduction

1.1 North Korea: an Irrational Actor?

The international community has long considered North Korea a threat to international security. This involves a series of acts of violence which North Korea has conducted since the 1960s. The violence by North Korea does not only target the military structures of other countries but also the civilians. Incidents such as the attempted assassination of the South Korean president in 1968, the hijack of a commercial airplane in 1971, the bombing of a hotel in Rangoon in 1983 are among the examples of violence that led some to call North Korea a terrorist country (Y. Kim, 2011, pp. 101-103). The image of North Korea deteriorated further in the post-Cold War period with the first nuclear crisis of 1993 following the regime's declaration that it would withdraw from the Non-Proliferation Treaty. North Korea also conducted a major ballistic missile test in that year. In the 1990s relations between North Korea and the international community appeared to stabilise only to see the total collapse in 2002 of the Agreed Framework achieved after the first nuclear crisis as the US charged North Korea with pursuing a highly enriched uranium programme.

Many literatures portray North Korea as an irrational state. Henriksen (2001, pp. 338-360), for example, says that North Korea is an irrational actor who "acts unpredictably and criminally in the absence of a great power patron". The irrationality of North Korea is judged on various categories. It includes the involvement of terrorist activities, the violation of human rights, the disregard of international agreements, and the pursuit of a nuclear weapons programme. All these characteristics encourage the use of the term "rogue state" in referring to North Korea both in the academic community and in the US foreign policy (Henriksen, 2001, p. 349; Hoyt, 2000, p. 303; Litwak, 2000, p. 26; U.S. Department of State, 2002). The term rogue state reflects the negative perception the international community has towards North Korea (Interview Anonymous, 2011), implying that North Korea is a bad country (Interview Prof. M. Kwon, 2011; Interview Prof. Paik, 2011). The international community perceives North Korea a "bad" and "mad" country which has offensive intent towards the international community and cannot be trusted even after the negotiations (H. Smith, 2000, pp. 597-602). The focus on the irrationality of North Korea is often based on its security

behaviour. North Korea has an unreasonably high level defence and heavily invests in the development of weapons of mass destruction. This effort goes against the economic capability and technical ability of the country to support such development. The regime's high level of defence heightens regional tensions in Northeast Asia and North Korea's decision to increase defence activity provokes other powers to follow the same path. Neither negotiation nor sanction is successful in solving the problem of North Korea's nuclear proliferation. North Korea's nuclear proliferation leads to a condition of near permanent crisis on the peninsula in the post-Cold War period.

North Korea projects an irrational image in terms of its security and defence behaviour, but its irrationality is also perceived to come from the regime's nature being ruled by a dictator under Communist ideology. North Korea trades its security with the welfare of its own people. The system of repressive rule influences the viewpoint of the international community to see North Korea as an irrational country with irrational and unpredictable behaviour (Cha & Kang, 2003, pp. 2-3; Y. Kim, 2008, p. 145). Based on this perception, North Korea is a war-prone country whose irrational and unpredictable behaviour makes it unlikely for the usual operation of deterrence or diplomacy to be useful with this state (Litwak, 2002, p. 56; Nolan & Strauss, 1997, pp. 21-38). The irrationality of North Korea also raises speculation within the international community that the regime will sooner or later collapse.

Therefore, a number of questions arise in relation to North Korea's thinking and behaviour that challenge the assumptions of International Relations about the state as a rational actor particularly in international security.

1.2 North Korea's Security and the Problem of Rational Deterrence

The problem of security of states connects to the concept of rational deterrence. The theory of deterrence assumes that states are rational actors, including in the choices they make in relation to the use of force to settle disputes and the decision to pursue weapons development, including nuclear weapons development. Rational deterrence means that states have to

understand and accept the credibility of the threat that each side imposes on the other before they can make a decision on the choice of action in response to the threat (Achen & Snidal, 1989, pp. 151-152; Knopf, 2009, pp. 37-38; Zagare, 1990, pp. 240-241). Deterrence works when threat is manipulated to make one side convince the other that the cost of war is too high and the end result is too destructive and so the status quo should be preserved (Lebow & Stein, 1989, p. 211). States can face some challenges in carrying out a successful deterrence strategy. Challenges include incomplete information that could impact the decision-making of leaderships or the misreading of the other side's intention on whether the action has defensive or offensive intention (Morgan, 2003, p. 12; Stein, 2009, p. 60).

Based on these calculations of rationality and deterrence, North Korea could be assumed to be a rational actor. North Korea is considered a weak power (Pardo, 2014, pp. 2-4). Although statistics show North Korea as having one of the biggest armies in the world, the military has rather outdated equipment and lack of capable manpower (Interview Prof. M. Kwon, 2011; Interview Prof. Paik, 2011). This has led North Korea to pursue nuclear weapons development for the purpose of adequate deterrence. North Korea claims that,

"The DPRK's nuclear weapons will serve as reliable war deterrent for protecting the supreme interests of the state and the security of the Korean nation from the U.S. threat of aggression and averting a new war and firmly safeguarding peace and stability on the Korean peninsula under any circumstances." (KCNA, 2006a)

The rise of North Korea as a new nuclear state has been primarily a result of the change in the international political structure in the post-Cold War period. The fall of Communism and the end of the socialist bloc led to heightened vulnerability for North Korea. The situation on the Korean peninsula indicates a high level of hostility and tension. Korea remains a divided country and the issue of unification is among the factors influencing North Korea's security perspective. In addition, the Korean War ended with an Armistice agreement between North Korea and the United States and South Korea but not a Peace Treaty. This unresolved issue creates a border dispute between the two Koreas and obstructs the normalisation of relations between North Korea and the United States. Since the Korean War has not been ended, the continued involvement of the United States in the regional affairs and the alliance partnership

of the United States with South Korea and Japan is a permanent source of threat to North Korea.

North Korea was faced with possible nuclear attack from the United States from the time of the Korean War (Engelhardt, 1996, p. 33). But North Korea did not actively pursue nuclear weapons because the strategic deadlock between the Soviet Union and the United States gave it protection. North Korea's fear of the United States heightened greatly in the new system of the post-Cold War. In this new situation North Korea does not have an ally to help it in time of crisis, while the United States security commitment to its allies remains firm. North Korea then seeks to increase its capacity for deterrence by any means, including nuclear development. The security dilemma on the Korean peninsula increases as a result of the security competition. North Korea may be a weak power but its military build-up and nuclear development are considered a serious threat to regional stability. The threat posed by North Korea is credible when combined with its provocative behaviour. North Korea has clearly shown its intention to risk all costs in going to war, which can be seen through many official statements on the regime's willingness to retaliate (Bluth, 2008, p. 173). North Korea's willingness to create security dilemmas for its neighbours can be seen by others as irrational choices.

The discourse on North Korea as irrational has been shaped by the continuing confrontation between North Korea and the United States in the last 25 years, exemplified by the US idea of 'rogue state'. The US use of 'rogue state' shows both the security concern and the norm concern the United States has with North Korea. While the security concern is based on the viewpoint that North Korea is a threat to the non-proliferation regime, the norm concern deals with North Korea's totalitarian system (Harrison, 2002, p. xvi). The United States wants to secure the nuclear non-proliferation regime in preventing the spread of nuclear weapons, which the post-Cold War system provided some opportunity to occur (U.S. Department of State, 1994). The United States demand is for the total nuclear disarmament of North Korea. However, the US policy in achieving this objective has a hidden agenda of coercing change in North Korea. The change includes the behavioural change and the regime change which leads to different use of strategy and policy in dealing with North Korea (Schneider, 2010, pp. 37-43). North Korea understands the US policy. It sees the United States as posing both a military threat and political threat, particularly with the US ambition of a regime change in

North Korea (Bluth, 2008, p. 182). The nuclear ambition of North Korea slowly becomes a long-term goal of the regime, in contrast to when the nuclear project was revealed in the early post-Cold War period as influenced by the fall of Communism (Pritchard, 2003b).

As this shows the irrationality of North Korea is also judged from the internal structure of North Korea. It refers to the viewpoint the United States and the international community has towards the internal system of North Korea. North Korea still adopts Communism as a core system. The confrontation between the US and North Korea includes the identity conflict in a rather similar way as the ideological conflict of the superpowers during the Cold War. The security dilemma created during the Cold War comes from differences of identity which affects the way the two superpowers perceive each other. According to Kennan (1961, p. 190),

"Many people in the Western governments came to hate the Soviet leaders for what they did. The Communists, on the other hand, hated the Western governments for what they were, regardless of what they did."

Tension on the Korean peninsula has also an element of identity problem between the United States and North Korea. Prof. Paik (2011) says that,

"The United States dislikes North Korea because it is a socialist and dictatorial country, and also a human rights abuser. The socialist and dictatorial system North Korea has is different from the democratic political system. Whatever North Korea does and whatever action North Korea has is not something familiar to us. We have an instinctively some antagonistic attitude to whatever North Korea does and whatever North Korea shows." (Interview Prof. Paik, 2011)

The use of the rogue state term reflects not only the US security concern on North Korea but also shows the US belief in the evil nature of North Korea (Moon & Bae, 2003, pp. 19-21). Jae Jung Suh (2006, p. 13) stresses that the conflict on the Korean peninsula comes from differences of identity between the United States and North Korea and this factor affects the way they both perceive and respond to threat. He writes,

"Washington sees in North Korea a rogue state that is hell bent on acquiring weapons of mass destruction, which it depends as a tool of threats, coercion and aggression. Pyongyang views Washington as an "imperialist warmonger" that is constantly looking for an opportune moment to strike. Because each sees the other as an adversary whose identity is the polar opposite of its own, they are predisposed to interpret the other's behaviour as manifesting its aggressive intention. They tend to reduce uncertainty about the other's intention and its likely future behaviour by referring to its antagonistic identity. As a result, they will not merely react to the other's act, but are likely to react more strongly and more quickly than a state that sees its environment as benign." (Jae Jung Suh, 2006, p. 13)

As such, identity is assumed to be also the factor that causes the feeling of fear between states (Booth & Wheeler, 2008, p. 70). The irrationality of North Korea under the term 'rogue state' may be debatable as the term is not an officially acknowledged concept in international relations community and is structured on the western viewpoint (Interview Prof. D. H. Han, 2011; Interview Prof. Koo, 2011). Nonetheless, it reveals concerns about the rationality of North Korea to some extent. It is clear that North Korea does not follow many of the international principles and threatens neighbouring countries with words and behaviours (Interview Prof. S. H. Sheen, 2011). The international community views North Korea as having an irrational system in which the leader disregards the welfare of the people while raising the level of deterrence unnecessarily.

The theories of rational deterrence and the security dilemma both agree that insecurity is created from the international system of anarchy. Since the theory of deterrence assumes that states are rational actors, the deterrence strategy of North Korea should conform to these assumptions. In other words, it should mean North Korea responds to the distributions of external threat resulting from the system of anarchy and the nature of inter-state relations. The deterrence strategy should be created from a sense of insecurity which can in the condition of security competition among states result in the security dilemma. However, there are arguments claiming that the deterrence strategy of North Korea does not conform to these assumptions but is the result of the North Korean regime's nature. The level of

insecurity is created by the totalitarian system which makes it impossible for North Korea to reform their basic system (Interview Prof. S. H. Sheen, 2011). It is the desire of the leader to keep the internal structure and socialist system in place that raises the level of deterrence beyond that related to external threat. The regime factor is clearly seen in the post-Cold War period when there is internal instability caused by the problems of economic deprivation and the leadership succession. The condition of domestic politics motivates the offensive behaviour because the leadership wants to control the situation by drawing attention from inside to outside, such as in the attack on the Yeonpyeong island in 2010 (Interview Prof. D. H. Han, 2011). In addition, North Korea is said to have "a risk-acceptant rationality" because the situation of conflict benefits the regime more than peacetime (Cha, 2012a, p. 228). Nuclear weapons and heightened tension help North Korea draw attention from the United States and gets some concessions out of the negotiations (Bluth, 2008, p. 172; Interview Prof. Paik, 2011).

It seems that North Korea's decision-making and behaviour has both rational and irrational characteristics. North Korea risks provocative behaviour and follows a nuclear ambition that heightens the security dilemma in the Northeast Asian region. At the same time that North Korea claims the military strategy is adopted for the purpose of deterrence. Nonetheless, regardless of the reasons for the deterrence and the high tension that is created, the situation in the Northeast Asian region can be considered stable. Tension has never turned to war since 1953 suggesting that the system of deterrence does fully function. This situation goes rather against the idea that North Korea is an irrational actor who cannot be deterred.

If we are to judge whether North Korea is rational or irrational then it should be done with the use of International Relations theory. The theories of rational deterrence and the security dilemma assume the external factors of an anarchic system and the need to achieve security through deterrence play the major role in shaping the security choices of states. As such, the question to be asked is if North Korea is a rational or irrational actor whose deterrence strategy does or does not conform to international relations theory. The theory assumes that states are driven by the desire for security but the level of insecurity of states varies depending on individual political condition. If we account for all the factors, both internal and external, how rational is the security evaluation and deterrence strategy of North Korea? Is North Korea a rational actor who seeks to attain security in the condition of anarchy or an

irrational actor whose provocative behaviour is beyond the assumptions of rational deterrence?

The thesis will study North Korea's thinking and behaviour on the security perspective. The thesis will assess the security perception and strategic behaviour of North Korea examining both the internal and external factors in order to assess whether North Korea is rational or irrational as based on the assumptions of security studies and strategic studies.

1.3 Research Structure

The deterrence strategy of states deals with how states perceive and respond to threat. There are two main elements in studying North Korea's deterrence strategy. The first is the internal structure which forms part of the regime evaluation of threat. The second is the factor of the regional environment which is a result of the relationship North Korea has with the regional powers and how this affects the perceived need for deterrence by North Korea. The thesis aims at evaluating North Korea's security behaviour since the post-Cold War period where the deterrence strategy has been changed. As earlier outlined, the problem of North Korea also comes from the issue of identity. Therefore, the thesis will use both the realism and constructivism in analysing the case of North Korea in order to find out whether the deterrence strategy is rational or irrational. The international relations theory will appear before other chapters evaluating the internal and external determinants on North Korea's security choices and deterrence behaviours. The structure of the thesis will appear as follows:

Chapter Two focuses on the international relations theory. The chapter will discuss the concepts of anarchy and security through the assumptions of realists and constructivists. After the idea of system of anarchy and inter-state relations are discussed, the chapter will study the concepts of rational deterrence and security dilemma. The political condition North Korea experiences will also be laid out to give some backgrounds for the evaluation of the regime's behaviour from the security and strategic studies. The hypotheses, the conceptual questions, and the analytical questions will be structured at the end of Chapter Two.

Chapter Three focuses on the internal structure of North Korea. The unique nature of North Korea makes it imperative to begin with the study of North Korea since the leadership of Kim Il Sung at the end of the World War II. This is to find out how the internal system is structured and also explore how the historical background, the Juche ideology, and the political system of the regime contribute to shaping the threat perception and the level of deterrence.

Chapter Four focuses on the relationship North Korea has with South Korea. The chapter will explore the inter-Korean relations since the division of the two Koreas from the Korean War. This is based on the historical fact of the forced division of the country. The division makes the issue of unification becomes part of the policy both countries aim at achieving. The relationship between the two Koreas plays a decisive role in the foreign and defence policy of both countries. The chapter will examine how the inter-Korean relations contribute to the threat perception of North Korea and whether or not North Korea's behaviour, including military and other forms of provocations, towards South Korea is for the purpose of deterrence.

Chapter Five focuses on the relationship North Korea has with the three regional powers – China, Japan, and Russia. Since it is assumed that the deterrence is a result of the inter-state relations, it is imperative to study how the bilateral relations North Korea has with these three countries contribute to North Korea's decision making on the deterrence level. The chapter will also examine why the regional powers fail to reassure or deter North Korea despite they share the same perception on the nuclear problem of North Korea.

Chapter Six focuses on the relationship between North Korea and the United States. The United States is believed to have the highest impact in affecting North Korea's deterrence level. The United States has engaged in solving the nuclear problem of North Korea since the first nuclear crisis in 1993 and different methods in policy choice have been applied. Nonetheless, the effort does not result in the denuclearisation of North Korea. The chapter will find out what the primary objective of the US strategy towards North Korea is. It is also aimed at studying whether the US policy towards North Korea is a cause or a means of managing the security dilemma in the region.

Chapter Seven focuses on the Six-Party Talks. The multilateral cooperation in the form of the Six-Party Talks is formed as a means to solve the North Korea's nuclear problem. However, the Six-Party Talks failed in the mission to denuclearise North Korea. North Korea has lost interest in the Talks and eventually declares in 2009 its intention not to attend any future Talks. The aim of this chapter is to find out what causes the failure of the Six-Party Talks and explore if there any possibility to improve the multilateral engagement in persuading North Korea to lower its deterrence threshold.

Chapter Eight is the conclusion of the thesis. Internal and external determinants, as outlined in the previous chapters, will be explored and evaluated to see how they contribute to the security perception of North Korea. The chapter will later conclude whether the North Korea's deterrence strategy is rational or irrational as based on the international relations theory.

1.4 Methodology

The research uses the international relations theory from both the security studies and strategic studies in assessing North Korea's deterrence strategy. It will include the discussion of the assumption of anarchy and states' security by realists and constructivists. The concept of deterrence and the security dilemma will also be analysed based on the assumption that they both are a result of how states live interdependently in the system of anarchy. The use of both the security studies and strategic studies is to assess how North Korea understands the security and how the deterrence strategy is built in response to the political condition North Korea experiences.

In giving more details on how the deterrence strategy of North Korea is formed, the thesis will use two methods. The first is qualitative textual analysis. Data in this method comes from various sources in drawing as much information on North Korea as possible. Apart from textbooks and journal articles, data includes analysis and opinion of experts on North Korea from leading research institutes such as Nautilus Institute for Security and Sustainability, 38 North, the Brookings Institution, the Woodrow Wilson International Center for Scholars. The

Woodrow Wilson International Center for scholars, for example, has the North Korea International Documentation Project which stores a collection of declassified documents on North Korea since the leadership of Kim Il Sung. They are documents which record actions and policy of North Korea based on the report, telegram, conversation North Korea has with the Communist countries or among officials from Communist countries themselves. The documents are translated in English and available online via the website.

Data related to the policy of North Korea comes from the Korean Central News Agency (KCNA), the official news agency of North Korea. The KCNA publishes on a daily basis news and policy of North Korea which is available from 1 January 1997 onwards. They are published in many languages including English and can be retrieved online via the website. Data before this period is retrieved from official speeches made by Kim Il Sung which are complied and published in a series of publications called the Selected Works. The tightcontrolled nature of the regime makes it unlikely to access data beyond what is published at the governmental level. It is also impossible to interview the diplomats or state officials of North Korea. To overcome this constraint, data about North Korea are also retrieved from media websites such as the Daily NK, NK News, the North Korean Economy Watch, the North Korea Tech. These are websites which base outside North Korea but contain insightful data, information, and analysis on the regime. The Daily NK, for example, provides information on North Korea from the perspectives of an ordinary people as the website publishes articles written by the North Korean defectors. The NK News, on the other hand, contains information on internal political structure, news about North Korea, the political structure at the leadership level, and analysis of experts on North Korea. The retrieval of data and information from various sources is imperative in obtaining as much information on North Korea as possible. This is because the leadership style of each leader makes differences in the availability of resources. By contrast to his father who regularly makes speeches in front of the public, Kim Jong II avoids public appearances due to his shy personality. Hence, there are few speeches of Kim Jong II available. Speech and statement made by Kim Jong II are retrieved from memoir of foreign diplomats who met with Kim Jong II. They appear in various sources such as books, journal articles, and analysis by the experts on North Korea.

Apart from the KCNA and the media websites as mentioned above, information on the Juche ideology, internal institutions, and internal political system of North Korea is also retrieved

from the publications by South Korea both at the governmental and public levels. They include the book titled "Understanding North Korea 2012" published by the Ministry of Unification of South Korea and the book named "North Korea Handbook" published by the Yonhap News Agency. They give clear and informative facts on various aspects of North Korea from the political system, the military system, foreign policy and diplomacy, and the hereditary succession.

In analysing the relationship North Korea has with the regional powers, data related to the policy between both sides is also drawn from official statements and declarations each government has on the issues. They are retrieved from, for example, the Ministry of Foreign Affairs, Ministry of Unification of South Korea, the White House, the U.S. Department of State.

News reports at the regional and international levels will also be used. News reports at the regional level are retrieved online from leading newspapers and news agencies. For example, they are Yonhap News Agency from South Korea, RIA Novosti from Russia, Xinhua News Agency from China, and the Japan Times from Japan. News reports at the international level are from, for example, the CNN, the BBC, the AFP.

The second method used in collecting data and information on North Korea is interviewing. The semi-structured interviews were conducted face-to-face with 11 interviewees who are former officials and academicians of South Korea. All the interviews took place between August to September 2011 in Seoul, South Korea. The use of semi-structured interviews is to allow the interviewees to express their own opinions and ideas on the issues of North Korea without being constrained by the fixed questions. Some interviewees have experience working with North Korea, while others have good background knowledge on domestic politics and foreign policy of North Korea. Information shared by the interviewees is deemed perceptive and insightful. This is because although both North Korea and South Korea are two separate countries, the two states are still attached by the history, language, and culture of the Korean nation. Both countries are also bound by the issue of unification which makes the inter-Korean matters and inter-Korean relations become more of a daily topic. Most of the interviews were audio recorded. However, some were not and this is because of the sensitive nature of the issues. The recording of the interviews, the citation of the interviewee's names,

the direct quotation of the conversations on the thesis depend on individual interviewee's agreements. Full details on names and institutions the interviewees are affiliated with are attached in the appendix of the thesis.

Chapter 2: Using IR Theory to Define North Korea and Its Behaviour

North Korea's pursuit of a nuclear capability has raised controversy on whether nuclear acquisition for dangerous and isolated states like North Korea can ever be justified. Under the rogue categorisation North Korea projects the "nuclear nightmare" which involves various aggressive and threatening activities from violating international obligations to the use of nuclear threat for belligerent purposes (Triplett II, 2004, p. 2). Nonetheless, it is clear that states do not develop nuclear weapons for their own pleasure. It has been argued that nuclear weapons enhance a state's military power and international prestige (Alagappa, 2008, p. 85) because the more powerful the states are, the higher the chance for these states to achieve their objectives without having to employ force (Gilpin, 1981, p. 31). Security incentives are assumed to be the major factor pushing states to choose the nuclear path (Epstein, 1977, pp. 16-28; Frankel, 1993, pp. 37-78; Mearsheimer, 1990b, pp. 35-50; Sagan, 1996, pp. 54-86). The expansion of the nuclear club has become even more worrisome with the spread of nuclear weapons among small and middle powers in the post-Cold War period (Bracken, 2012, p. 95). Despite the destructive nature of nuclear weapons, some argue that peace could be guaranteed even with nuclear proliferation. Mearsheimer (1990b, p. 39) writes that "[nuclear weapons] are powerful force for peace" and that states would not choose to fight knowing the "horrendous devastation" that will result from nuclear usage. Waltz (1981, pp. 1-39) says that war with nuclear weapons is costly, and fully aware of this fact, states are more logical and careful when approaching each other as they want to minimise the chance of war. In other words, states would prefer to strategically rather than physically use their nuclear weapons in achieving their objectives.

Nuclear weapons acquisition is strongly connected with the security of states but the prospect of war is changed under nuclear conditions because states are no longer limited to the traditional idea of using force to achieve a policy objective. Nuclear weapons strongly advance the concept of deterrence dissuading states from the use of force when the level of risk appears much higher than with conventional war (Ayson, 2008, p. 562; Brodie, 1978, pp. 65-83). Bull (1968, pp. 593-594) says that the core focus of strategic analysis shifted in the nuclear age from the thinking of "war as instrument of policy" to "the threat of war" driving strategies of "deterrence, crisis management, and the manipulation of risk". These

assumptions point to the correlation between the conceptions of security and strategy. According to Ayson (2008, pp. 558-575), security should be seen as a condition to which strategy is the response: strategy is the way states deal with the security challenges they are facing. Strategy also means the effect the action of one state has on the behaviour of other states. Schelling (1960, p. 3) says that strategy focuses on "the interdependence of the adversaries' decisions and on their expectations about each other behaviours" in which "the best course of action for each player depends on what the other players do". Thus, strategy relates to the problem of how well states adjust their policies and behaviours during experience of threat in order to meet security challenges. By focusing on state's security and the strategy, it is assumed that states would opt for the best choice that would not only help them gain in the conflict but also ensure their survival.

The above-mentioned statements show the interdependence of security and strategy and so the interdependence of security studies and strategic studies. While security refers to "the alleviation of threats to cherished values" (Williams, 2008, p. 1), strategic analysis examines the behaviours of states in responding to the conditions they experience (Ayson, 2008, p. 572). In exploring the nuclear question of North Korea, it is necessary to take examine the security experience and the strategic behaviour of North Korea in order to explain the choices the North Korean leadership has made. This is especially the case in the period after the Cold war as the relationship between security environment, strategic behaviours and decision-making changed decisively at that point. This chapter will review the theory and literatures from both security and strategic studies and connect them to the analysis of North Korean behaviour. It will begin with a review of the International Relations theory focusing on the idea of security as a political condition that states experience in the system of anarchy. Strategic studies will be studied after this with a focus on the rational actor assumptions and the problem of deterrence as this was developed during the Cold War and modified in the post-Cold War period.

2.1 International Relations Theory: Anarchy and Security

The focus on security as a main priority of states is based on the assumption that the international system is anarchic. Waltz (1979, pp. 105, 114) states the international system is

formed by the existence of units that are similar but not identical and in which there is an imperative of self-help that requires each unit to seek a means to protect itself against others. Based on Herz (1962, pp. 39-42), states in the system may recognise each other's sovereignty and independence but states will never feel safe because relations of states are formed on a basis of unequally distributed power that allows the stronger to threaten the weaker. Importantly, anarchy is perpetual in this system as it lacks a central authority or control above states to which they might appeal for protection. This kind of structure causes a condition of permanent insecurity that pushes states to compete for power because power is the only means that could protect states from conquest in a self-help system (Mearsheimer, 2007, pp. 79-80). As well as seeking power for themselves states can also promote predictability through collective efforts to manage the balance of power: they can adjust their relations in a way that equalises the distribution of power and security can be advanced depending on how able states are in maintaining relative distributions of power (Deudney, 2011, p. 17).

A key debate in International Relations has arisen as to how stable anarchy can be when the condition of anarchy itself forces states to strive for more power. Scholars have different viewpoints regarding the amount of power states need, so that contemporary realism has divided into two groups with two views of the consequences of anarchy. Defensive realists such as Waltz believe anarchy shapes the behaviour of states causing them to focus on their security and promoting the balance of power as an ordering principle. Waltz (1979, p. 126) says that power is a means towards the end of achieving security but "the first concern of states is not to maximise power but to maintain their positions in the system". This is because the offensive choice of one state increasing its power could threaten the security of other states forcing them to counterbalance the threat by expanding their power (Lobell, 2009, p. 47). Offensive realists in contrast see the possibility of an imbalance of power caused by the security imperative of states. States would continue to seek for more power until reaching the point of their absolute security, because states find it difficult to calculate an adequate level of power they need in deterring their rivals and in winning a war (Mearsheimer, 2001, pp. 34-35). Therefore the ultimate solution to the security imperative is not balance but hegemony. Despite the differences between defensive and offensive realists as to the way states achieve security, they share the same view that the behaviour of states is fundamentally driven by insecurity under anarchy. A desire for security drives states to compete for survival and the key to competition for security is power. At the same time uncertainty about how much

power is necessary to achieve security can lead states to pursue power beyond the point that is necessary for their security and this may in turn reduce the security available to others (Posen, 1993, p. 28). This then creates the possibility of the security dilemma in which the misperception of one state about adequate security leads other states to respond, undermining and contradicting the initial attempt. Thus the question of how accurately states perceive their security and respond to its needs becomes an important element in understanding balance of power dynamics.

A further issue pointed out by realists is that an international system built on a self-help basis makes it very difficult for states to overcome their fear and cooperate with one another. The motives and intentions of states are seen as unpredictable and changeable over time and this forces states to focus on the ability to attack rather than the intention of their adversaries (C. Glaser, 1994, p. 56). Mearsheimer (1990a, p. 12) argues that states need to have a relative power that matches those of other states and the search for relative power advantage makes it possible for states to behave aggressively seeking more power and weakening their adversaries. Realists therefore support the idea that states need power because power is the only thing that could help them survive the competition among each other; but even though power is the critical factor behind survival states have great difficulty evaluating their relative power: how much power is adequate for their security and how much others should have for their security. Realists themselves have found it difficult to define the power of states leading Lebow to describe such attempts as "imprecise, making it difficult to develop measures of polarity" (Lebow, 1994, p. 255). Nevertheless all realists believe that it is the organisation of relative power among states that gives rise to the international political structure. The nature of this international political structure is important because states do not base their security on individual relationships alone - is A more or less powerful than B? - but on the relative distributions of power among groups of states in a political structure regionally or globally.

The idea that states compete for survival leads most realists to accept that states are rational political actors. Realists believe that the international system comprises of political units, each with a goal of their own, and in securing their own goal, states would rationally choose the most effective means available within the constraints they face with an uncertainty of the environment and an incompleteness of information (Legro & Moravcsik, 1999, p. 12). The assumption about states as rational actors is one of the core ideas of structural realism.

Keohane (1986, pp. 165-167) writes that the rationality assumption explains how strategy and interests of states are set up after a careful calculation states make based on their position in the system, and how costs of action are calculated in a search to increase the expected utility in the environment of uncertainty. Mearsheimer (2011, p. 127) elaborates on the rational actor assumption arguing that it refers to the belief that states are well aware of the external environment and that the interaction they have among each other will effect the action and strategy of each, even as they seek to maximise their chance of survival. The position of Waltz is somewhat different since he questions that states are always rational. States are seen as strategic and logical in their balance of power behaviour and fully understand the need to do well or face falling behind others (Waltz, 1979, p. 118). But he states that,

"the theory requires no assumptions of rationality or of constancy of will on the part of all actors. The theory says simply that if some do relatively well, others will emulate them or fall by the wayside." (Waltz, 1979, p. 118)

It is quite common to find scholars that relate realism with the rational actor assumption. Kahler (1998, p. 923) points out some complexity within the idea of rationality and doubts whether the behaviour of states will always have a rational foundation, but he also notes the pairing of realism with the idea of the state as rational unitary actor. Other scholars, such as Elman (1996, p. 20), do not focus on the nature of the state but on system level features and how these play a role in states' decision-making. Waltz's account of "the third image" focuses on the international system of anarchy, and the relations between anarchy, conflict among states, and behaviour of states that result (Waltz, 2001, pp. 159-186). Elman (1996, p. 20) refers to Waltz's third image as the basis of assumptions that states assess the external environment when making a decision and the strategy of states will depend if it yields an expected gain or an expected loss as rational choice theory suggests.

Realism has been considered one of the oldest traditions of political theory that attempts to explain the behaviour of states both in peacetime and in conflict through using various kinds of analyses from human nature to the structure of the inter-state system. Realism has of course received criticisms which question its assumptions in explaining the organisation and behaviour of states under anarchy. The most concerted criticism involves the realists' focus on material capacity as a basis for explaining how the state system and international political

structure are organised. Realism is criticised as having a materialistic approach with the theory's view of power focusing on the material capabilities and using materialist assumptions such as the influence of human nature on an actor's behaviour (Barkin, 2003, pp. 329-330). Realists consider the international structure a "material structure" as seen in the security competition that pushes states to seek material capabilities of power (Mearsheimer, 1995, p. 91). With a concentration on a material approach, it is pointed out that realism lacks an analysis of the social dimension - factors that differentiate one state from another depending on history, background and culture. Ruggie (1998, p. 865) writes that a political model of neorealism such as that of Waltz can be considered physicalist with little focus on ideational factors. Traditional Realism often starts from the basic assumption that human nature will drive states to seek for power as dictated by the search for survival under anarchy. This emphasis on the relationship between human nature and political structure can be criticised as overlooking the issue of identity formation and its roles in shaping political behaviour (Jepperson, Wendt, & Katzenstein, 1996, pp. 46-47).

The weaknesses of realism in analysing behaviour and organisation in the states' system and the international political structure led to the emergence of a new school of social constructivism in the post-Cold War period. The leading critic of realism Alexander Wendt (1992, p. 397) writes that the realists' conception of anarchy and the distribution of power cannot explain the differences in approach that states have towards friends and enemies in the international system. This is because the realists' assumptions do not consider the conception of a distribution of knowledge in understanding the way states perceive themselves and others (Wendt, 1992, p. 397). Constructivists base their assumptions about political behaviour and organisation on the relationship between social practices and states' identity, and how the interaction of states - each with its own social identity - could effect the formation of the international political structure. In other words, constructivism regards human beings as social beings, whose subjectivity is created and formed as a result of an interaction among them (Dunne, 1995, p. 371). Onuf (1989, p. 36) states the meaning of constructivism is that "people and societies construct, or constitute, each other".

The constructivists do not reject the categories of realism - state, anarchy, competition, war - but the explanations of realism for why these things exist and how they work. So constructivists base their assumptions on the conception of "states under anarchy" in the

same way that the realists do. But constructivists analyse the meaning of anarchy by referring to the perception and judgement each actor may have towards the international system. Anarchy can have several meanings as the interpretation depends on individual understanding, and this reflects the understanding that each state has of its own identity, built up from its own history and culture. It is also social identity that makes states understand each other's behaviour because identity contains preferences of states that influence their consequent actions (Hopf, 1998, pp. 175-176). Wendt (1992, pp. 397-399) emphasises an actor/state's identity because it forms part of the actor/state's interest. The international structure is formed as a result of how states socialise so that anarchy is composed of "the set or structure of identity and interest" (Wendt, 1992, pp. 397-399). As a result anarchy has no fixed form. Copeland (2000, p. 188) elaborates that anarchy has "different cultural instantiations" because it comes from a process of interaction which can be shaped and reshaped according to how the state's self - its identity and interest - is formed from interacting with others. According to Barkin (2010, pp. 27-28), inter-subjectivity is created by actors in the system as they socially interact. Common knowledge and practices actors hold together create the international political structure so that the structure does not and will not "exist outside mutable norms and discourses that define the identities and interests of actors" (Barkin, 2010, pp. 27-28). Identities and interests of states, as such, are built on how states interact among each other under the socialisation process and shared ideas among states from norms and rules that would tell states what behaviours are appropriate (Atkinson, 2006, p. 511).

What can be concluded is that the constructivists focus on three issues - an individual identity, a connection between an actor's identity and actor's interest, and an influence of an actor's identity on the formation of the system. These ideas constitute new assumptions about the actor's perception as affected by each individual interest, and the connection between actors and structure. Constructivists challenge the traditional idea put forward by realists about the dominance of materialist factors by saying that material capabilities could also be viewed through the social context apart from considering them through the issue of necessity alone (Checkel, 1998, pp. 325-326). The "identity-interest" relationship one state has with another - whether friend or foe - plays a fundamental role in threat perception and this turns out to be a better explanation of behaviour than the realists' materialist assumptions. This is shown when shared knowledge comes into effect in explaining a correlation between material

resources and social structure and in providing meanings between material resources and human action (Wendt, 1995, p. 73). It is true that a need for security pushes states to acquire weapons and seek to enhance their military but the constructivists' claim that "threat" and "defense" are social relationships so that weaponry and military hold values beyond national security purposes. Weaponry has a normative nature and holds a symbolic meaning as associated with the social structure. Eyre and Suchman (1996, pp. 86-87) give an example by using the words modernity and efficacy in referring to military equipped with a high technology. Material capabilities provide states with power to not only secure themselves but also the ability to coerce other states to do what they want them to. Nonetheless, the conception of power is more complex than analysed only through the realists idea of physical force. Power is produced based on how actors interact socially. Power is created as actors are socially related and this points to how social practices play a role in shaping the perception of actors on individual identity and ability to exercise power over other actors in the system (Barnett & Duvall, 2005, p. 46).

In consequence, the following arguments can be made as a summary from the assumptions of the dominant school of realism and its main critic constructivism. First of all, security is a political condition. It refers to the condition or the situation that states experience, especially where the interaction between states produces a security condition for one state from the policy/behaviour of another state. Secondly, the system in which states exist is anarchy in that there is no higher authority to which they can appeal to secure their own existence. Anarchy affects the behaviour of states as the desire for security forces states to be strong and powerful. However, anarchy is not a fixed form. It is varied depending on the interpretation of states as based on individual social context, history, and culture, etc. Thirdly, states and the states system complement each other. While relations of states create the system, the system itself affects states' behaviour. The way states behave and perceive threat depend on the experience they have of the system as an international political structure. Fourthly, most realists and constructivists assume that security is what concerns states the most. Nonetheless, they have different views of anarchy and the rational choice of security. To realists, rational responses to security are driven by two factors: anarchy pushes states to compete for security and the balance of power is created from the distribution of power. On the other hand, Constructivists take the issue of identity and interest into consideration. Identity and interest affect a security desire of one state in relation to other states. Thus, the

way states perceive and respond to threat depend on their identities and their understanding of other identities. Lastly, the international political structure needs to be explained through the assumptions of both the realists and the constructivists. For example, the conflict between the United States and the Soviet Union during the Cold War shows the security competition and an attempt to balance each other's power. At the same time, the antagonism of the two superpowers is also a matter of the identity and interest issues. The United States and the Soviet Union do not share the same political ideology and both actors differ in terms of history, culture, and political system and these factors shaped their perception of the identity and interest division between them. As such, analysis of complex security problems should use the assumptions of both schools in explaining the security perception and strategic behaviours of states.

2.2 Rational Deterrence in the Cold War and After

As noted by Bull (1968, pp. 593-594) strategic analysis changed in the nuclear age so that "the studies of actual violence have given place to analyses of "deterrence", "crisis management", "the manipulation of risk". In other words, it changed from the study of the conduct of violence to the study of how to reduce the threat of war by means of deterrence. This assumption gave rise to another vital question — what was an adequate level of deterrence for states to pursue? The system of permanent insecurity pushes a state to seek to guarantee its survival by increasing its defence capacity to a level that it thinks adequate. This idea of adequate defence gave rise to the theory of rational deterrence, which seeks to analyse the strategic behaviours of states as based on the rational choices of their leaderships as they evaluate security realities. It is assumed that these rational choices were practised on a range of objectives from the necessity of survival to more complex gain and loss calculations. However, the questions of security perception and deterrence behaviour among states leaderships produced a new debate. The gap between one state's choice and another state's perception of its intentions gave rise to the idea of the security dilemma, which was first noted by Herz in 1950. Herz (1950, p. 157) writes:

"Wherever such anarchic society has existed--and it has existed in most periods of known history on some level - there has arisen what may be called the "security dilemma" of men, or groups, or their leaders. Groups or individuals living in such a constellation must be, and usually are, concerned about their security from being attacked, subjected, dominated, or annihilated by other groups and individuals. Striving to attain security from such attack, they are driven to acquire more and more power in order to escape the impact of the power of others. This, in turn, renders the others more insecure and compels them to prepare for the worst. Since none can ever feel entirely secure in such a world of competing units, power competition ensues, and the vicious circle of security and power accumulation is on."

A consequence of one state's decision to increase its deterrence level can potentially provoke other states to increase their defence capacity producing a decline not an increase in the security situation of the first state. Booth and Wheeler (2008, pp. 4-5) argue that the security dilemma occurs when one state has to make a choice on how to react to another states behaviour. In other words, a security response is a result of an interpretation of one state's intention by another. Heightened deterrence is a rational response for countering another state's aggressive intention but may not be rational if the state's intention is defensive. What is rational choices for deterrence can be undermined by state's misunderstanding of each other's intentions.

The concept of deterrence becomes a function of inter-state relations in conditions where states are interdependent. One of the most important theorists of rational choice deterrence Thomas Schelling (1960, p. 13) writes that deterrence is demonstrated in the way one state seeks to influence another state's decision-making by implying that its behaviour will affect its own behaviour; and most importantly, deterrence operates in a situation in which the parties involved have both "conflicting and common interests". The essence of deterrence is illustrated further by Knopf (2009, p. 41) who says that deterrence involves anything that dissuades one state from doing something by implying through its manner that there would be negative consequences of either military or non-military kinds if the action is carried out. At its most basic, deterrence is one state's attempt to prevent another from doing something through the use or threat of force, so that this threat creates a feeling of uncertainty and apprehension in the deterred state. It is an attempt to convey that the cost of conflict is greater than any potential gain. A profit-and-loss calculation should bring about increased rationality

in the decision-making process. Rationality is structured on the perception that the optimal policy choice for a state during the approach to, or in the early stages of, conflict, is to avoid employing force. However, if the choice is inclined towards a direct confrontation, the cost of implementing the threat must be lower than other alternatives, and the benefits to be gained through the conflict must also be guaranteed (Morgan, 2003, pp. 44-46). Scholars such as Achen and Snidal (1989, p. 150), George and Smoke (1974, pp. 72-83), Stein (2009, pp. 61-62), and T.V. Paul (2009, pp. 5-6) base their conception of deterrence on state behaviour by assuming that the state in question is a rational actor acting on a calculation of loss and benefit, and this assumes that the state would not choose to engage in conflict if the costs exceed the expected gains. Schelling (1960, p. 6) points out the relations between threat and rationality, saying,

"the rationality of the adversary is pertinent to the efficacy of threat, and that madmen, like children, can often not be controlled by threats. We have recognised that the efficacy of threat may depend on what alternatives are available to the potential enemy, who, if he is not to react like a trapped lion, must be left some tolerable recourse."

Hence, deterrence is more of a "psychological than physical" process as it involves the decision-making of one state when faced with a choice offered by another state (Knopf, 2009, pp. 37-38). Deterrence theory requires some assumptions about the perception of threat and the way states respond to threat. Based on this assumption, it is possible to conclude that deterrence assumes rationality of actors in making a threat credible and in making deterrence work. Nonetheless, there is a possibility that irrationality overcomes rationality and takes control in the decision-making process. Schelling (1960, p. 16) writes about rationality and irrationality as,

"Rationality is a collection of attributes and departures from complete rationality may be in many different directions. Irrationality can imply a disorderly and inconsistent value system, faulty calculation, and inability to receive messages or to communicate efficiently."

Therefore even though deterrence assumes rationality of states Jervis (1979, p. 299) argues that rationality cannot always guarantee peace and security because rationality can also be undermined, for example, if states are faced with a very intense and stressful situation. Irrationality can also happen in a situation when states cannot properly exercise their gain and loss calculations (Morgan, 2003, p. 60).

During the height of a conflict, it is expected that the choice a state makes should be based on a rational calculation of gain and loss, so that the rational choice should be the one that is the least costly, the least risky, but with the most beneficial outcome. Different states may calculate gain and loss differently and may react to a situation differently depending on their perceptions of risk (Lebow & Stein, 1989, pp. 208-209). While some might prefer to focus on minimising losses, others may focus on maximising expected gains. Whatever the preferences are, the choice should be made based on the norm of rationality. What makes the situation more complex is that the uncertainty created by anarchy has an effect on states behaviour: anarchy can cause a sense of insecurity towards neighbours, and states adopt policies that reflect this insecurity. States face a challenge in responding to another's behaviour that it perceives to be aggressive between raising their own deterrence capacity, or acting in a more conciliatory manner. However, states are likely to react to another state's decision to instigate a military build-up by pursuing a similar policy. The reaction one state has to the actions of another thus creates a spiral model, which is a process of mutual and direct military responsiveness driven by insecurity and fear, resulting ultimately in a low level of security at a higher level of destructive power in an insecure environment (Jervis, 1976, pp. 58-113). This scenario frequently occurred during the Cold War period, with the arms race between the superpowers a particularly stark example. The spiral model occurs because states feel uncertain whether and when the situation will turn from peace to war and who their enemy in the conflict will be (Booth & Wheeler, 2008, p. 41). Uncertainty drives states to seek to guarantee its own security, which usually happens at the expense of other states.

Deterrence theory has become the standard explanation for how states interact around threat and conflict. The theory was developed during the Cold War period and faced further change in the post-Cold War period. The arms competition during the Cold War underlined the concept of deterrence. Both sides sought strategic superiority over each other and some of

their behaviour, such as the development of a second-strike capability, appeared to be offensive rather than defensive in nature (Jervis, 2001, pp. 36-60). A most important aspect of deterrence theory in the Cold War and after was the role of nuclear weapons proliferation. The possibility of nuclear warfare transformed deterrence from an occasional practice to a comprehensive and elaborate strategy. The destructive nature of nuclear weapons presented states with only one choice – that of attempting to counter and balance the threat of nuclear weapons by developing their own. The nature of deterrence is fully revealed when states chose to obtain rather than use nuclear weapons in response to other states' nuclear weapons capacity (Freedman, 1989, pp. 40-41). Mutually assured destruction made states realise that there could be no winner in a nuclear war (Jervis, 1999, p. 89) and nuclear weapons, to some extent, can be seen as supporting stability if not peace between great powers. Waltz (1981, pp. 3-5) elaborates on this, stating that a nuclear second-strike capability advanced the quality of deterrence because deterrence is based on "the ability to punish", not "the ability to defend".

With the end of the Cold War assumptions about nuclear proliferation changed also. The sudden dissolution of the Soviet Union in 1991 and the end of its deterrence capacity set the framework for a strategic unipolar world, but some theorists were sceptical about the endurance of this power structure. Waltz (2000, pp. 27-32) writes that unipolarity is not an endurable form as the power of the dominant state will soon be weakened either by involving itself in too many affairs or new great power states will emerge to even out its power. This situation resembles the post-Cold War structure in which there is competition between the dominant power, the United States, and other great powers, notably China and Russia, and also includes the reactions of smaller states to the risks of unipolarity. Nonetheless, Russia and China have tended to focus on their conventional arms modernisation rather than building up nuclear capacity in countering the US power and influence. Proliferation challenges have not come within the politics of the great powers but from other states responding to US dominance or regional problems. Deterrence and proliferation theory changed to take account of these developments. In 1996 Sagan (1996, p. 55) proposed three theoretical models in explaining states' decision on whether "to build or refrain from developing nuclear weapons". These are,

"the security model," according to which states build nuclear weapons to increase national security against foreign threats, especially nuclear threats; "the domestic politics model," which envisions nuclear weapons as political tools used to advance parochial domestic and bureaucratic interests; and "the norms model," under which nuclear weapons decisions are made because weapons acquisition, or restraint in weapons development, provides an important normative symbol of a state's modernity and identity" (Sagan, 1996, p. 55).

Based on Sagan's three models, the security model is the one that reflects the concept of deterrence in the Cold War period. The rise of nuclear weapons states, either in the Cold War period or at the present time, results from the insecurity that drives states to seek the ultimate deterrent (Sagan, 1996, p. 85). However, the security model cannot fully account for the nuclear proliferation strategies in the post-Cold war situation. Bracken (2012, pp. 106-108) points out several other factors influencing states' decision to seek a nuclear capacity. This includes ideological demands such as nationalism and the change in the political structure from the Cold War to the post-Cold War. He argues that the new nuclear states of the post-Cold War period see the possession of nuclear weapons as national accomplishment (Bracken, 2012, pp. 115-116). This in part accounts for the fact that most recent proliferation has been in Asia where many states ideologies are driven by a sense of nationalism or national grievance.

In summary strategic studies has concluded that deterrence is practised on the basis of perception of threat and the gain and loss calculation of states leaderships. Nuclear deterrence has risen as a preference for states leaderships because it is more cost effective than the conventional deterrence: access to nuclear knowledge and technologies makes it cheaper for new nuclear states to "profit from copying experience and knowledge" of the first nuclear states like the United States or the Soviet Union (Bracken, 2012, p. 119). The decision of states to invest in nuclear capacity supports rational actor assumptions about the imperative for survival and on a general calculation of cost and benefit in the making of their security decisions. Nonetheless, strategic studies also concludes that the study of deterrence should not focus only on the rational search for security among states. Domestic politics, ideology, national identity, and international status should be taken into consideration as these effect

states' perceptions of the international political structure and the choices made for strategic success.

2.3 Applying Security and Strategic Theory to North Korea's Behaviour

In assessing North Korea's understanding of security and its deterrence strategy, the assumptions of both security studies and strategic studies will have to be used. Explaining North Korea's behaviour is important not only for developing some responses to the ongoing crisis on the peninsula but because North Korea's behaviour is often seen as a challenge to the assumptions of international analysis that state's decision-making and strategic choices are driven by rational calculations. North Korea's security perception and deterrence strategy are often portrayed as irrational: no attempts to lower North Korea's threat levels have been successful so that the country operates on a condition of permanent belligerence and North Korea has made consistent attempts to escalate deterrence levels, especially through efforts in obtaining nuclear weapons and missile delivery systems. Security studies and strategic studies would address this problem from the following perspectives:

Security studies would suggest:

- 1. A primary issue is the nature and experience of the North Korean regime. The international community sees North Korea as pursuing a high deterrence threshold with its constant increases in defence capability. Given security studies assumptions that threat perception and security concern are shaped by states' identity and interest, the question arises of how the nature and experience of the regime shapes the way it perceives threat, anarchy, and security. In addition, we should ask how the history, culture, and institutions play a role in states' understanding of security as based on the interest-identity context.
- 2. The second point concerns the assumption that the international system is a state of anarchy. Based on this assumption of security studies questions should be asked about the condition of anarchy North Korea has experienced and how the international political structure has affected North Korea's decision-making. These two questions

deal with the complex regional environment which involves a divided nation, several regional great powers, and the United States and its alliance system in Northeast Asia.

On the other hand, different kinds of questions can be drawn when the issue of North Korea is examined through the context of strategic studies:

- 1. Deterrence is a response to how states live interdependently in the system of anarchy. The greater the level of interdependence, the greater the need for states to seek security by deterrence in order to survive. The condition of interdependence also pushes states to make cost calculations in their strategic relations with other states. Based on these assumptions of interdependence, security and strategic calculations what needs to be examined is the changes to the security interdependence of North Korea, especially the changes to the international political structure with the end of the Cold war system.
- 2. The security dilemma is a common condition in the system of anarchy where states have limited means of determining each other's intentions. A spiral model can be created where deterrence actions of one state drive other states increase of defence capacity. The situation on the Korean peninsula suggests that both the security dilemma and the spiral model are in practise. Nonetheless, the continued increase of defence capacity of North Korea raises the question if this is a deterrence strategy for escaping the spiral model it has with other regional powers or whether it has other objectives.
- 3. The first two points interdependence and spiral model assume security to be the highest concern of states. Taking these two points into consideration, focus can be turned to North Korea's acquisition of nuclear weapons. What needs to be answered is whether North Korea conforms to the rational actor model in their deterrence strategy, particularly since the changes to international political structure as the Cold War ended. In addition, we should also ask if the rational actor assumption can be improved when nuclear weapons development is connected to the models proposed by Sagan and Bracken that include domestic coalition, norm taking or norm rejection, and nationalist ideology.

In order to evaluate the relevance of the assumptions of security studies and strategic studies to the North Korean question it is necessary to provide some overview of analysts' assumptions about North Korea's condition and behaviour since the end of the Cold war.

First of all, the Korean peninsula is an area where there is continued influence of other regional powers into Korean affairs. The two Koreas must co-exist with the United States, Russia, China, and Japan who have their own interests in the Korean affairs but also use the Korean question to test their own relations. Prof. Paik (2011) refers them as the "four powers", saying,

"The strategic location of Korea draws the so-called "four powers" to interplay in Korean affairs. They interact cautiously. They are obsessed with one another as they want to counterweigh each other influence on the Korean affairs. The cross-section of their interests gives an opportunity for Korea to be more independent and not entirely depend on one particular nation." (Interview Prof. Paik, 2011)

The interplay of regional powers in the Korean affairs underlines the deterrence problem and the security dilemma. During the Cold War period, the extended deterrence the United States and the Soviet Union granted to their allies created a balance in the level of deterrence between the two camps. The situation changed once the Cold War ended. The fall of Communism deprived North Korea of the extended deterrence of the USSR and made it more vulnerable to the deterrence practises of the other regional states. North Korea was left as a Communist country, struggling to avoid the collapse of its regime. The situation was further exacerbated by the internal economic crisis so that it fell behind other countries in every aspect. North Korea becomes more uncertain and wary of the external environment and develops security concerns with nearly all its neighbours. Nuclear weapons are the method by which North Korea seeks to compensate for the imbalance of power and deterrence. The nuclear programme has been accelerated since the early 1990s and slowly allows North Korea to claim the status of partial nuclear weapons state today.

The outbreak of the first nuclear crisis in 1993 reveals the cycle of a security dilemma. Negotiation fails to reduce the tension on the peninsula. One of the reasons for the failure is

North Korea's sense of fear and insecurity has not been addressed by other states. North Korea considers the overall situation "a state of war" based on its viewpoint that the proper peace treaty has never been negotiated and signed to replace the Armistice that ended the Korean War (D. Kang, 2003, pp. 43-44). Prof. Paik (2011) says that the goals of North Korea are "to officially end the Korean War; to sign the peace treaty; and to get normal relationships with the United States, Japan, and South Korea" (Interview Prof. Paik, 2011). North Korea has not trusted the dialogue on nuclear talks that it has either in the bilateral or multilateral forms. Stephen Bosworth, a former US special representative for North Korea policy, says that North Korea's reluctance to fully disarm comes with the disbelief that the Agreed Framework would help it achieve the normalisation of relations with the United States (PBS, 2003). North Korea has felt intimidated and insecure as a result of internal weakness that makes it fall behind other countries. The feeling of insecurity is further emphasised by the hostile policy of the regional powers. Prof. M. Kwon (2011) says that,

"The external environment is not very friendly to North Korea. The United States threatened North Korea, by saying that North Korea might probably be the next target after Iraq. Also, North Korea feels intimidated by South Korea, particularly when South Korea begins to talk about unification because it would mean the absorption of North Korea by South Korea. They intimidate North Korea but they do not realise it." (Interview Prof. M. Kwon, 2011)

The second point relates to the nature of inter-Korean relations. There have been efforts of both Koreas to reach for reunification of the country, which has been divided since 1953. The issue of unification causes both to compete for their version. A sense of distrust and fear is created but is exacerbated when one side falls behind the other side. Nonetheless, it is claimed that unification is no longer an ultimate goal of North Korea nor has it any connection with the regime's pursuit of nuclear weapons (Interview Prof. D. H. Han, 2011; Interview Dr. E. C. Lim, 2011). The possession of nuclear weapons is rather to serve the regime's threat perception. Prof. I. H. Park (2011) says that,

"North Korea's system is so much different from the ordinary system. The huge gap in terms of political, social, and economic development it has with the outside world and with South Korea are so huge. It feels threatened by this fact even though we do not threaten North Korea." (Interview Prof. I. H. Park, 2011)

The Korean peninsula remains a place where both competitive deterrence strategies and the security dilemma are in practise. In the world of anarchy, it is easy for states to become locked in a cycle of security competition. Fear will drive states to compete with each other, as Booth and Wheeler (2008, p. 1) state:

"Given the stakes involved, the existence of weapons in hands of one state can provoke at least uncertainty and possibly real fear in others even when those weapons are not intended to be used except for self-protection (following an attack, or in the even to of a threat of an attack)."

Hence, it is possible to assume that the security dilemma occurs naturally as a result of how states seek to secure themselves. Nonetheless, the behaviour of North Korea somehow shows that the security dilemma is sought intentionally. It refers to the tactics of provocations and belligerence, which North Korea has been using since the Cold War period. Bracken (2012, pp. 194-195) says that the motive behind provocations is not war, but rather to send a message to the outside world, for example, to show its dissatisfaction to the international pressure and sanction. Such actions appear irrational as they exceed conventional motivations in maintaining military strength for the purpose of deterrence. In contrast to the fact that it has a large conventional army, North Korea is actually a militarily weak country. Although the international community views North Korea as "the most militarised society on earth", North Korea's military is actually rather weak with outdated military equipment and a low spending per capita caused by the economic problem (H. Smith, 2000, p. 599). The theory of deterrence suggests that an increase in security level is usually for the purpose of deterring an aggressor. In addition, states usually choose not to heighten tensions against its rival, particularly if it is highly likely for them to lose in a situation of war (Lebow & Stein, 1989, p. 211). The behaviour of North Korea shows the opposite. North Korea chooses provocations and belligerence despite the fact it stands a high chance of losing if the war breaks out. Prof. M. Kwon (2011) comments that,

"Despite nuclear weapons, North Korea's military is smaller and weaker than other countries, say South Korea. Their military spending is far smaller than that of South Korea. They have very poor weapons in term of conventional arms. They cannot afford big conventional army with their low level of GDP. If South Korea decides to develop nuclear weapons, it would take South Korea a year to possess such weapons." (Interview Prof. M. Kwon, 2011)

Both realists and constructivists talk about how the feeling of fear drives states to compete for security. The regional environment plays a significant role in pushing North Korea toward the development of nuclear weapons as seen from the outbreak of the first nuclear crisis right after the fall of the Soviet Union. Nuclear weapons give North Korea the power to protect itself and deter adversaries particularly in the time North Korea is left with no allies (Interview Prof. S. H. Sheen, 2011). On the other hand, the security dilemma is still sought beyond the deterrence level. There appear two explanations on the motive behind North Korea's use of provocations. The first deals with the internal politics of North Korea. Nuclear weapons and belligerence tactics are connected with the issues of internal stability and leadership consolidation. Prof. D. H. Han (2011) says that,

"Domestic factor plays a role in the decision to belligerence. Both the Cheonan and the Yeonpyeong incidents prove it. The Yeonpyeong shelling in particular made me curious about their intention. When I thought about the peninsula affairs at that time, it was clearly ameliorating. Not only South Korea, but the United States and China also tried to contact or persuade North Korea after the Cheonan incident. The mood was not bad at that time. But North Korea chose to attack Yeonpyeong island. My speculation is the domestic politics outweighs international affairs. I think the North Korean leader tried to satisfy the domestic audience because the political succession process is ongoing and very unstable by their definition. If you have instability in your domestic politics, you need to convert your opposition's attention to outside so that you can solve your domestic problem." (Interview Prof. D. H. Han, 2011)

The second explanation is the use of deterrence to play a game with the United States and the international community. A heightened tension is sought as a means to draw attention from

the United States in particular. It seems that North Korea and the United States are not in the same place in terms of objectives. North Korea's aim is to settle the Korean War issue through a proper treaty, the United States' main focus is the denuclearisation. The situation reaches a deadlock especially if the United States pursues a hardline policy towards North Korea. The US reluctance to negotiate with North Korea, creates a desire to explore tactics to force the United States to the negotiation table. Prof. Paik (2011) considers the North Korea's use of nuclear and missile cards as a policy instrument in order to achieve the regime's policy goal, stating that:

"North Korea, as I understand, wanted to explore a policy instrument which brings the United States to come to the negotiation table. That was the nuclear card and missile card. It is a policy instrument. The policy goal is to end the Korean War, sign the peace treaty, and normalise the relations with the United States. But you cannot achieve that goal if the United States does not come to the negotiation table. So you decided to use a very forceable means to bring the United States to the negotiation table and they used the nuclear card. So the nuclear card is a policy instrument." (Interview Prof. Paik, 2011)

Whenever there is a dialogue, it appears that North Korea achieves some sort of a deal or concession with it. The first nuclear crisis, the second nuclear crisis, and the series of missile provocations create speculation that North Korea purposefully raises tensions for gaining some benefits from negotiations. Cha (2012a, pp. 227-229) says that North Korea does not find it beneficial living in a state of peace and belligerence is conducted as a means to compensate losses by drawing assistance from others, the so-called "Double or Nothing" strategy. The launch of missiles is one of examples of the use of provocations. Bechtol (2007, p. 53) writes that,

"The testing and deployment of large numbers of ballistic and other types of missiles can reasonably be seen as a way of "raising the stakes" when North Korea is seeking concessions from the United States. These types of activities can be used in conjunction with provocations such as DMZ violations, harassment of US reconnaissance flights, and other activities as a way of raising tensions — a tactic that North Korea has used in recent years on several occasions. When preparing for a test-launch of the Taepodong 2 in 2006,

Pyongyang attempted to use the event as a tool for concessions from the United States or others at the Six-Party Talks."

The question of North Korea, its security perception and strategic behaviour both conforms to some of the assumptions of security studies and strategic studies but contradicts others. It conforms to some assumptions of security under anarchy but contradicts others; it conforms to some assumptions of security dilemma but contradicts others; it conforms to some assumptions of deterrence theory but contradicts others. It is hard to say therefore where North Korea stands in relation to rational actor assumptions that play a big part in international relations theory: it is incorrect to say that North Korea's behaviour is entirely rational or entirely irrational (as in the 'rogue state' idea). What this thesis intends to do is evaluate North Korea's security understanding and strategic behaviour to assess the parts of North Korea's behaviour that deserve to be called rational and those to be called irrational. In doing this it also points to some of the limits of applying rational actor assumptions to complex problems in international relations.

2.4 The Establishment of Hypotheses and Research Questions

There seems to be two primary conclusions concerning the case of North Korea as examined through the assumptions of security and strategic studies. The first concerns the external environment which plays a significant part in shaping North Korea's threat perception and level of deterrence activity. The second is about the use of deterrence in a strategic way of covering the regime's weakness and manipulating threats for a wider range of gains. These conclusions lead to the establishment of hypotheses designed to test rationality in North Korea's security understanding and strategic behaviour.

1. North Korea's security perception and strategic behaviour broadly conform to the assumptions of security studies and strategic studies that states leaderships are able to define accurately their security environment and make deterrence responses on a rational calculation of likely gains and losses both in terms of immediate goals and in the ultimate goal of survival. Unable to achieve unification due to internal weakness North Korea's primary goal is maintenance of the status quo and it is prepared to manipulate

the security dilemma if this contributes to short-term gains or the ultimate goal of regime survival. The acquisition of nuclear weapons is a rational choice within this strategy.

2. North Korea's security perception and strategic behaviour contradict many of the assumptions of security and strategic studies. North Korea's understanding of the security environment is distorted by ideology and internal politics; and its deterrence strategy is not a rational response in terms of either short term gains or losses or the ultimate goal of defending the Korean nation. Deliberate provocation of the security dilemma towards South Korea or other neighbours is irrational and cannot provide an exit from the security challenges North Korea faces. North Korea's pursuit of nuclear weapons is not a deterrence strategy but a blackmail strategy which is likely to increase not reduce other states determination to sanction and isolate North Korea.

These hypotheses will be tested by the following research questions and in the following chapters.

Research Questions

Conceptual Questions:

- 1. What are the proper criteria for judging whether states are rational or irrational in terms of their security perceptions or deterrence strategies? How should we evaluate and test the relative influence of internal determinants and external environment in shaping security perception and deterrence behaviour?
- 2. What is an adequate level of deterrence for states? How can states gain security for themselves without making other states feel insecure; and is active provocation of a security dilemma with other states evidence of irrational thinking or behaviour?

Analytical Questions:

1. What are the internal determinants that shape the threat perception of North Korea? To what extent is the high level of threat perception a product North Korea creates in order to keep the legitimacy of its regime and the power of the leader? (Chapter 3)

- 2. How has the relationship of the two Koreas played an effect on the North Korea's perception? Are all the military and other forms of provocations for deterring South Korea or for purposes not connected to deterrence ?(Chapter 4)
- 3. How has the interaction of the major powers in the region contributed to the North Korea's decision making on the deterrence level? Why have the major powers not been able to reassure or deter North Korea? (Chapter 5)
- 4. What have been the primary objectives of the US strategy towards North Korea? How does the US strategy for DPRK interact with broader US objectives for managing its relations in Northeast Asia? Is the US deterrence strategy a cause of the security dilemma in the region or a means of managing it? (Chapter 6)
- 5. Having failed to either co-opt or deter DPRK from its deterrence strategy on a bilateral basis the regional powers proceeded to a multilateral process the Six Party Talks in order to achieve denuclearisation on the peninsula. What does the interaction of the concerned parties in this process tell us about the limits of multilateral diplomacy as a route to normalisation on the peninsula? Is there any prospect that multilateral engagements can persuade DPRK to lower its deterrence threshold? (Chapter 7)

The conclusion to the thesis in Chapter 8 will summarise the answers to the above questions and answer the central question of the thesis: is DPRK deterrence strategy rational or irrational?

Before continuing to the analytical chapters two methodological points should be made. The chapters outlined above have two different timeframes. Chapters 3 and 4 (North Korean regime and inter-Korean relations) take the period from the end of the Second World War to the contemporary era. This is necessary because examination of North Korean internal factors and change on the peninsula must be taken over the whole period since the division of the Korean nation into two states and the struggle for unification that followed. The subsequent chapters on the external environment take the period between the first nuclear crisis on the peninsula 1993-94 and the collapse of the Six Party Talks in 2008. This timeframe is justified because a core argument of the thesis is that the threat environment and deterrence responses

of North Korea changed decisively with the end of the Cold war. The Cold war system and the post-Cold war system are quite different environments for the North Korean regime and its deterrence behaviour changes in consequence. Therefore we should look at the interaction of external environment and strategic behaviour of North Korea only in this period to answer the questions in Chapters 5, 6 and 7.

A further methodology problem is studying North Korea as a political science topic. This is difficult for two reasons. The sources of reliable information on North Korea available for analysts are very much lower than for other topics in political science. Also North Korea is still very much an ideological issue for many analysts in both Koreas and internationally. For these two reasons of limited and biased information studying North Korea poses many problems for the researcher. This thesis has done the best it can to work around these problems but it is necessary to be aware of them when considering research conclusions.

Chapter 3: The North Korean Regime

When discussing North Korea, it seems that the international community sees the country in a very negative light. This is reflected in the rhetoric used, with terms such as 'mad', 'dangerous', or 'unpredictable' employed. Not only is North Korea perceived as a socialist country with a very strange authoritarian internal system, but North Korea is also viewed as an aggressor who threatens international peace and stability through its pursuit of nuclear weapons and missile programmes, and frequent belligerence towards its neighbours and other regional actors. North Korea maintains a high level of defence, a reflection of a national insecurity. North Korea's military build-up, in turn, creates a feeling of insecurity among its neighbours, resulting in a security dilemma in the region. The literature usually views states' insecurity as driven by competition among states who live in an anarchical international system in search of their own security. However, with regards to North Korea, it is argued that the regime's threat perception is created by the personal perception of its leader, and this is most evident as a result of the change in the international structure in the post-Cold war period. This threat perception is exploited by the leader in securing his power and position, using threat and insecurity as propaganda and defence mechanisms. It is widely debated if North Korea's deterrence strategy adequately matches the environment in which the country operates, and whether it is rational given that the threat perception and deterrence strategy of the regime are used to serve to secure the regime's leadership in the Kim family.

North Korea is a socialist country. North Korea may not look that much different from other socialist countries, when judged by its internal systems and the fact that it is controlled by a single party – the Korean Workers' Party (KWP). However, it is argued that North Korean socialism should be regarded as socialism's North Korean style (Interview Prof. Paik, 2011). What distinguishes North Korea from other communist countries is its adoption of Juche ideology, which grants the leader a singular power to put the KWP, the state, and the system under his control (Yonhap News Agency, 2003, p. 87). In addition, North Korea's society has roots in Confucianism. Confucianism has never been replaced by socialism, but instead it has been integrated with socialist ideology in the process of constructing the country's socialist system. Concerning this, T. Kang (1979, p. 99) writes:

"No matter how much change the North Koreans claim their Socialistic or Communistic revolution has wrought in their society, their basic mentality has not changed too much. They have only affected the forms of Communism, but its contents are still of the Korean Confucian tradition. Of course, the contents of Korean Confucianism itself have been changed within its framework. The Korean people's loyalty in the past was expressed in family affairs through Confucian rites such as those for adulthood, ceremonies, marriage, funerals, ancestor's anniversary feast, and worship of family altars. Today some of these rites have become devitalized, but some still survive in an extremely simplified form under the control of the Party."

By claiming that a leader's threat perception has a greater impact on the regime's level of deterrence, rather than external factors, makes it imperative to study the internal dynamics of the regime and their role in contributing to the level of deterrence.

This chapter analyses the historical background, the ideological system, and institutional framing in order to provide some background of the overall internal structure of the North Korean regime, and illustrate how these internal characteristics played a role in supporting the power and perception of its leadership. This will help to analyse the problem of whether North Korea's deterrent strategy is rational or irrational, given the unique internal system as the core structure of the regime; and if the internal determinants are the predominant factors that determine the deterrent strategy, or whether it is the international environment.

3.1 The Regime of North Korea: the History, the Structure and the System

The history of North Korea begins at the end of the Second World War, when an effort by the Allied Powers to liberate Korea from Japanese occupation resulted in the division of the Korean Peninsula into two separate states. The two zones were divided at the 38th Parallel, and the political systems of each zone was shaped and influenced by their two occupiers. Communism was established in North Korea with the arrival of the Soviet army, and the process of Sovietisation began with the appointment of Kim II Sung, who had been trained in the Soviet Union. Both North and South Korea successfully established their own

governments with the support of their respective occupiers, but the desire to unify the country remained strong in the two countries' leadership, despite differences in their political systems. North Korea held advantages in military capability and internal stability, and took the first move intended to achieve this goal by invading South Korea on 25 June 1950, which led to the Korean War. This was to result in a permanent division of the two Koreas. The War concluded with the Armistice Agreement, signed on 27 July 1953.

After the division into two states North Korea adopted the communist system and ideology. The KWP, with Kim Il Sung as the chairman, became the sole and permanent government of North Korea. The Constitution of North Korea clearly states, in Article 11, that "The DPRK shall conduct all activities under the leadership of the Workers' Party of Korea" (Yonhap News Agency, 2003, p. 1013). As written in the KWP charter, the KWP "is guided by the great leader Comrade Kim Il Sung's juche (self-reliance) and revolutionary ideas" (Yonhap News Agency, 2003, p. 87). Kim Il Sung's power in the KWP was evident before the Korean War. As chairman of the party, Kim Il Sung had full control of the Military Committee, which was in charge of preparations for the Korean War, and therefore reflected his consolidated political power (J. K. Park, 1979, p. 117). The years immediately following the Korean War saw Kim Il Sung attempt to consolidate his leadership. Kim purged some party members by charging them with responsibility for the failure of the war, and the purge-process lasted throughout the 1950s and targetted many other political rivals (K. W. Nam, 1974, p. 88).

Kim Il Sung had a vision to rehabilitate North Korea through an economic reconstruction programme. However, such efforts caused dissatisfaction because the programme required the North Korean people to increase their labour output while simultaneously decreasing their rate of consumption (H. S. Lee, 2001, p. 24). In order to secure his leadership amidst opposition and challenges, Kim Il Sung introduced the Juche ideology, which was announced to the public for the first time in December 1955. Juche's principles of self-reliance have three main components: national political independence, economic self-reliance, and national self-defence. These are political tools that influence the internal political structure, and have played a part in the foreign policy of North Korea in the post-Korean War period. Juche also represents Korean nationalism. The self-reliance principle is structured on the idea that "both internal and international affairs of the country should be handled by the North Koreans,

independent of foreign interference, through the policy of self-reliance to the fullest extent possible" (An, 1983, p. 36).

Domestically, Kim II Sung used the Juche ideology to eliminate his opponents and to establish his power as the one-man dictatorship. Internationally, Juche emphasised the sovereignty and independence of North Korea, and was used as a political tool to escape the pressures that arose with the Sino-Soviet dispute. Juche was settled in North Korea until in 1960s, when there appeared to be an attempt to transform Juche into Kimilsungism, or the leader's Cult of Personality. The new interpretation of Juche elevated Kim Il Sung's status from Nation Leader to Supreme Leader, or "suryong". The basic characteristics of Kimilsungism are described as:

"The regime presented Kim Il Sung's childhood as an exemplar for self-reliant communists, and introduced justification for this idolization by claiming that the suryong or Kim Il Sung is the entity that enables socio-political life, which is more important than mere physical existence. The so-called suryong theory is clearly the centrepiece of the personality cult, as it posits the suryong, the key driver that leads revolutions and nation-building, as the intellectual leader at the head of the masses." (*Understanding North Korea 2012*, 2012, p. 32)

Since Kimilsungism specifically focuses on the leader, it laid the foundation for the dynastic succession that justified the elevation of Kim Jong II after the death of his father. The Juche ideology, established by Kim II Sung, was adopted in practice during the leadership of Kim Jong II, with some changes to the concept. The Juche of Kim II Sung emphasised the idea of self-reliance, whereas that of Kim Jong II focused more on the leader and the relationship between the leader and the masses. Juche started to change in the early 1970s, with a focus on man under the "man-centred philosophy" in justifying Kim Jong II's succession. In the

¹ Power competition between China and the Soviet Union caused North Korean discomfort, because both were considered North Korea's allies. To avoid taking a side, North Korea referred to Juche and declared that the regime was a politically independent country (Oberdorfer, 2001, p. 20). North Korea successfully manoeuvred around this competition, making it able to secure economic and military assistance from the Soviet Union and sharing the same political ideology with China – for example, promoting the revolution in Third World countries, and opposing detente with Western countries (Armstrong, 2007, p. 91).

1980s, the idea was expanded to include an absolute loyalty of the masses to the leader (Kihl, 2006, p. 9).

The first signs of a change to military doctrine appeared in December 1962, with the adoption of the "Four-Point Military Guidelines". The Guidelines, considered the bedrock of North Korea's military policy, have the four principles of arming the whole population, fortifying the country, upgrading the military, and training military personnel (Understanding North Korea 2012, 2012, p. 107). The "Four-Point Military Guidelines" shaped North Korea into a fortress, and influenced North Korea to enhance its strategy and tactics focused on surpriseattacks. This lead to the construction of underground tunnels along the Demilitarized Zone (DMZ) in the 1970s (Yonhap News Agency, 2003, p. 677). The fundamental idea to Kim II Sung's militarism came from his personal experience of colonialism, which drove his desire to protect North Korea from foreign influence. This became clearer after South Korea was admitted under the US security umbrella. H. S. Lee (2001, p. 52) interprets the Guidelines as an attempt to militarise in order to achieve the reunification of the two Koreas. The military was given a higher degree of significance in 1964 with Kim II Sung's declaration of "the Three Revolutionary Forces". The "Three Revolutionary Forces" policy is directed at the reunification of Korea and the efforts of revolutionary forces within both Koreas to achieve this (H. S. Lee, 2001, p. 53). The militarism may originate from the personal desire of Kim, but the military pillar gradually acquired a higher level of significance, becoming the primary pillar of the regime, as a result of the changed international environment North Korea faced. The change started with the rift in the Communist camp between the Soviet Union and China during the Sino-Soviet dispute, the strengthening of the US-ROK military alliance but intensified with the fall of the Soviet Union and the failure of the socialist countries in the Cold War political struggle. A final incident to hit the regime was the sudden death of Kim II Sung, in 1994. Kim Il Sung's death led to the succession of Kim Jong Il as the new leader of North Korea.

Apart from Juche, the socialism of North Korea is unique due to the succession of power within the Kim family line. A South Korean expert made this comment on the hereditary succession:

"My analysis of the hereditary succession is that it is the way to ensure the regime's survival. From Kim Il Sung or Kim Jong Il's point of view, the

succession is the safe way to continue the same policy, to keep the system as it is. It is also the best way to block the power from being overthrown by the lieutenants after the death of the leader." (Interview Anonymous, 2011)

Despite the smooth succession from father to son, the control of the military institution appeared to be a challenging task for Kim Jong II. It is possible to say that Kim Jong II's era began in 1998, after the three-year mourning period of the death of Kim II Sung had ended. Kim Jong II was declared the chairman of the National Defence Commission, the highest military organ in charge of the military and defence as stated in the Constitution of North Korea (Yonhap News Agency, 2003, p. 1021). The military underwent new reforms, and old officers were replaced by new officers loyal to Kim Jong II. Kim Jong II's effort to consolidate his leadership is also visible in the introduction of military-first politics, or "Songun Jeongchi". As a result of this slogan, the military became North Korea's first priority. It emphasised the role and power of the leader over military institutions, and called for absolute devotion of the soldiers to the leader (H. Park, 2007, p. 7). I. Kim (2006, p. 71) writes that Kim Jong II used the military-first politics to consolidate his leadership and power, because the slogan seeks the dedication and loyalty of the military to the leader. The basic idea of the relationship between the leader and the military is exhibited in an article published in the Rodong Sinmun² on 21 March 2003, emphasising that the army is "the powerful political force of the society and the revolution only through the great revolution led by a leader [suryong]." (Nautilus Institute for Security and Sustainability, 2003). The power of the Korean People's Army (KPA) was elevated to the highest position over the KWP, giving the KPA an authority to involve itself and decide upon non-military issues (S. Snyder, 2009a, p. 43). The focus on the military reduced the level of significance of economic reform, which Jeung (2007, p. 15) interprets as an attempt to deflect the leadership's responsibility for the country's economic hardships during the 1990s. Despite some changes to the internal military institutions, it has been claimed that North Korea's overall system did not undergo much change after Kim Jong Il took control of the country. This is because,

"North Korea has a unique political system. It is a mixture of Confucianism and Socialism. Most North Korean people believe in Kim Il Sung, even though he is dead. They worship both Kim Il Sung and Kim Jong Il. They are gods. Without

_

² Rodong Sinmun is the official newspaper of the KWP.

them, we cannot live. It is a kind of religion. In other words, North Korean leaders successfully brainwashed their people. I think the North Korean regime was based on this kind of indoctrination. Without changing their ideology, the North Korea regime is unchangeable and it is hard to change the core of the North Korean regime." (Interview Prof. D. H. Han, 2011)

As politics prioritised the military, every aspect of the country came to serve the military, even the economic system. As such, the only main source of revenue was the military industry, which exacerbated the country's economic problems. Any change in the economic system, whether liberalisation or reform, including opening to foreign engagement, became highly unlikely as it would have jeopardised the military's control. In foreign affairs, military-first politics was connected to North Korea's threat perception when the regime was faced with external pressures. Nuclear proliferation and provocation by North Korea under the leadership of Kim Jong II were in part products of the military-first politics. The military was given greatest prominence and it influenced North Korea's choice to rely upon the use of nuclear weapons for coercive bargaining. In return, the military-first politics were strengthened by this policy implementation (DeRochie, 2011, p. 14).

Further to military-first politics, Kim Jong II also held a vision of turning North Korea into a powerful and prosperous nation, or "kangsong taeguk". The phrase first appeared in January 1999. It was published in the New Year Joint Editorial of the KCNA, which says,

"As we built a socialist country, independent in politics, self-supporting in the economy and self-reliant in national defence, through the grand march of Chollima on the ruins under the guidance of the great Kim Il Sung, we should push ahead with a second grand march of Chollima under the leadership of the respected Kim Jong Il. 'Let this year mark a turning-point in building a powerful nation' this is a militant slogan that should be held aloft by the Workers' Party and people of Korea. The editorial calls for glorifying the DPRK as an ideologically powerful socialist state and fully demonstrating its invincible might as a militarily powerful socialist state." (KCNA, 1999a)

The core idea of the kangsong taeguk, apart from strengthening the country in all aspects, was about an absolute loyalty the people must have towards the leader, which could

successfully be achieved with the support of the military (B. C. Koh, 2005). The slogan calls for achievements in three major areas – ideology, economy and military – each of which has to be simultaneously developed in order to reach the goal of a strong and powerful nation (Interview Prof. Y. Y. Kim, 2011). Kim Jong II adopted this slogan as a means to ensure his absolute power as the Supreme Leader, and also to claim his status as the legitimate successor to Kim II Sung. Prof. Moon (2011) notes that,

"North Korea's development of nuclear weapons has something to do with the domestic politics. Kim Jong II has been championing the slogan of strong and prosperous grand nation. The possession of nuclear weapons underscores the strength of North Korea, thereby enhancing political legitimacy of Kim Jong II." (Interview Prof. Moon, 2011)

According to C. Kenneth Quinones (2005a, pp. 283-284), the kangsong taeguk was launched under a short-term objective of reviving the ailing economy, and a long-term objective of safeguarding the regime by strengthening the military in particular. The kangsong taeguk may be a new slogan, but North Korea continued to commit itself to the idea of Juche and anti-US imperialism as emphasised in the 1999 New Year Joint Editorial (KCNA, 1999a). Anti-US imperialism was also used to support the implementation of the slogan with an emphasis on the US threat to the regime. The KCNA says that the nation must be built on the slogan, because the country needed to protect Korean socialism and end the threat of US imperialism (KCNA, 1999c). The strengthening of the military based on the perception that imperialism is a source of insecurity gave kangsong taeguk credibility in North Korean internal politics.

Despite the fact that both North Korea and the United States were bound to each other under the Agreed Framework of 1994 that aimed to resolve the first Korean nuclear crisis, distrust continued as each side suspected the other of breaching the agreement. This resulted in policy implementation in an opposite direction. This factor gave kangsong taeguk credibility in North Korean internal politics, in which imperialism was referred to as a cause for war, and pushed the regime forward with its nuclear and missile delivery system development.

3.2 North Korea's Threat Perception

In the early stages of the post-Korean War period, there were two major challenges that Kim Il Sung had to overcome. The first was the need to consolidate his power as leader by uniting all the domestic factions inside North Korean society. Secondly, he needed to reconstruct the country's economy after the Korean War. To achieve these objectives, policies mainly focussed on the development of a heavy industry, the implementation of the militarisation programme, and the announcement of the Juche principle of self-reliance.

The Juche policy reflected the security perception of the North Korea leadership, because while Juche is a method of seeking internal stability, it is also a political tool for extracting external security from North Korea's two main allies – China and the Soviet Union. Partial independence is actually the policy North Korea pursued during this period, as the economic and political survival of the regime depended on the external assistance from allies. The Sino-Soviet dispute, as such, was a difficult development for North Korea, because the regime found it hard to maintain its neutrality when any slight shift in position could jeopardise the relationship it had with either of its allies. The Juche ideology helped North Korea maintain alliances with both China and the Soviet Union. It was during this period that Kim II Sung started to incorporate Juche and patriotism by defining Juche as the principle of one's own efforts (Jo & Marshall, 1979, pp. 169-170). Nevertheless, the dilemma Kim faced was that, at the same time as needing to maintain the alliance partnerships, there was also the need for minimising the level of influence both China and the Soviet Union had on internal North Korean politics, and to consolidate his own power and that of the Korean Workers' Party.

The benefits North Korea received from its allies assisted economic progress and political stability, which also raised hope for Kim Il Sung to fulfil his goal of Korean reunification. Nonetheless, the situation began to change in the mid-1960s, when the country experienced agricultural shortages caused by the policy that gave too much emphasis on the development of heavy industry. The economy of North Korea fell behind that of South Korea. For the first time since the Korean War ended, North Korea perceived South Korea as a threat. Prof. M. Kwon (2011) observed the situation during that period of time, saying,

"North Korea was far more advanced than South Korea in the 1950s and 1960s but in 1974, the per capita GDP was reverse. South Korea began to go ahead.

All the socialist countries develop very fast in the first one to two decades but in time they are slow down and stagnated. In Europe or other areas, they are just overthrown. But North Korea, for various reasons, it is continuing its regime. It is like riding on the back of the tiger and it is very difficult to get off the tiger. It is running and you have to run. If you run faster, then all the other countries become more hostile to you and you have to be more hostile towards other countries and the more you have to be secretive. Fundamental fact here is they have become economically fast growing and politically democratic in South Korea. That is the main factor that threatens the existence of North Korea." (Interview Prof. M. Kwon, 2011)

The perception of threat from South Korea lowered the regime's security and, in order to compensate for the feeling of economic insecurity, North Korea shifted its focus to strengthening its military capability. The decision resulted in acts of violence³ against South Korea and the United States, which reached a peak in the mid-1960s (Bradshaw, 1979, pp. 188-189). The belligerence had two major consequences: a strengthening of the Juche ideology of self-reliance, and the United States was blamed for being a major obstacle in Korean matters. Juche was given a higher degree of significance when China tried to put pressure on North Korea over its acts of aggression. Kim Il Sung reacted strongly to the Chinese pressure. In an attempt to guarantee its own sovereignty, Juche was again emphasised as the main ideology of North Korea (Jo & Marshall, 1979, pp. 176-177).

North Korea's decision to strengthen its military capability, as compensation for its lack of economic security, produced changes in North Korea's neighbours' level of threat perception. South Korea was most affected by North Korean's military build-up, because most of its acts of aggression were aimed at South Korea. By attempting to match its level of security with its level of threat perception, South Korea finally decided to pursue its own military build-up, which led to an arms race on the Korean peninsula in the 1970s. North Korea's military build-up lowered South Korea's sense of security most keenly during the decline in US

_

³ Major incidents during this period included an attempted assassination of South Korean President Park Chung Hee, in 1968; the capture of the US intelligence ship Pueblo, in 1968; and the shooting down of a US reconnaissance plane, in 1969. In addition, there were also violent intrusions along the DMZ, which significantly increased in frequency in the mid-1960s (Lerner, 2010).

military assistance to South Korea as part of Richard Nixon's military withdrawal plan⁴, in 1969. Although the withdrawal plan was finally suspended after a series of talks and discussions within the US government, it had already had an effect on the South Korean government. South Korea felt uneasy with the US decision, because the South Korean government perceived its security as dependent on the alliance with the United States (Hamm, 1999, pp. 79-80). Unlike the military balance on the peninsula in the 1950s which favoured North Korea by the 1970s and 1980s South Korea was the more advanced economy and military. North Korea may seem to have a large army, but the analysis shows that the manpower and military equipment are of inferior quality, and certainly when compared to that of South Korea (Moon & Lee, 2009).

Apart from the changing relationship with South Korea, North Korea's threat perception was also affected by the changes in the broader international environment of the Cold War. During the 1960s, there were two events that Kim Il Sung exploited for North Korea's own benefit: the Vietnam War and the Chinese Cultural Revolution. The Vietnam War, to Kim Il Sung, was a symbol of Third World resistance against American imperialism. North Korea should have taken this opportunity to adopt a leading role in supporting the worldwide struggle to weaken the power of the United States and to pave the way for a revolution in South Korea (Gills, 1996, pp. 106-109). The Vietnam War took place when China was going through its Cultural Revolution. Kim Il Sung viewed this period as an opportunity to promote himself as the "next leader of the Communist movement among Third World countries", as can be seen in his speeches encouraging nationalist movements in many parts of the world (Cha, 2012a, pp. 47-48). Kim Il Sung's speeches caused some dissatisfaction in both China and Vietnam, but such actions had two major results. The first was the belief Kim Il Sung had in revolutionary power as a means to achieve Korean reunification, and also in becoming the leader of a global Communist movement. The second was the belief in the propaganda campaign in which Kim II Sung exploited the Vietnam War in order to promote the concept of Third World revolution against the imperialists. It was the strategy North Korea pursued in promoting the dangers of imperialism and the need to construct the socialist state. Bradshaw (1979, p. 185) writes that there tends to be no difference between domestic policy objectives

-

⁴ President Richard Nixon announced in July 1969 the so-called "Nixon Doctrine", which planned to cut the US military expenditures in Asia (J. H. Nam, 1986, p. 63). The doctrine also mentioned the security responsibility the allies should be committed to, making the US security assurance more effective (Hamm, 1999, p. 79).

and foreign policy goals, because North Korea still focused on the concept of the power of the revolution in constructing the nation, and in strengthening the solidarity of the socialist state with world revolutionary forces. To achieve these goals, North Korean security needed to be maximised through economic and defence construction, and the revolution in South Korea needed to be encouraged in order to topple the South Korean government and force US withdrawal. North Korea strengthened opposition to imperialism, by framing its propaganda against South Korea and the United States and criticising South Korea's dependence on US imperialism.

The situation on the Korean peninsula experienced more changes as a result of certain transitions and changes in the international system. During the mid-1970s, the bipolar system began to evolve into a multi-polar system – marked by a decline in US power and influence in Asia, and also the normalisation of relations between China and the United States, and detente between the Soviet Union and the United States. The new international atmosphere created a new impetus for the two Koreas to resolve their own issues without foreign interference (Gills, 1996, p. 121). Korean reunification was still North Korea's main objective, but the strategy to achieve this goal was changed to focus on dialogue and negotiation – this led to the organisation of the first North-South dialogue, in 1971.

North Korea's failing economy may have played a part in changing the country's tactics in the 1970s, as the regime was forced to reduce its defence expenditures. At the same time, the change in the policy and tactics were also a result of the regime's desire to gain international recognition and respect. Zagoria and Kim (1975, pp. 1021-1022) have said that North Korea abandoned the policy of supporting worldwide revolution as a means to indirectly weaken the power of the United States, thereby forcing a withdrawal from East Asia, to focus instead on gaining a broader acceptance from the international community by expanding diplomacy. Nonetheless, belligerence against South Korea continued, and it involved the construction of underground tunnels along the DMZ in the 1970s, and the attempted assassination of President Chun Doo Hwan in 1980s. During the 1980s, North Korea suffered from a heightened sense of insecurity, caused by an internal economic decline that made the country unable to match its level of defence spending with its level of threat perception. The threat perception of North Korea was a reaction to the rapid growth in South Korean arms build-up in light of its economic success, and the introduction of the US Air Land Battle doctrine, in 1983. Such threats moved North Korea closer to the Soviet Union and shifted North Korean

strategy from infiltration to terrorism (Hamm, 1999, pp. 84-85). As North Korea compensated for its increased insecurity by raising its defence level, these changes made other countries perceive North Korea as a threat. North Korea's acts of belligerence encouraged the United States and South Korea to consider that a military response was appropriate, which resulted in a strengthening of the US-ROK alliance through an increase in common military exercises (Cha, 2012a, p. 58).

The situation reached its peak in the 1990s, with the collapse of the Communist bloc and the decision of both Russia and China to extend diplomatic relations to South Korea. North Korea felt insecure and isolated particularly after Russia ended relations with communist countries and joined with Western countries in condemning North Korea for human rights violations (Evgeny Bazhanov, 2003, p. 7). The loss of former allies North Korea had an impact on North Korea's sense of security, on which Prof. Paik (2011) elaborates:

"It is quite clear that in the absence of the Soviet Union which collapsed in 1991, North Korea had to find a so-called an exit strategy in order to get out of the difficulties it faced when the Soviet Union and Eastern European socialist states were gone. The first thing they thought of was to end the Korean War, to sign the peace treaty and normalise the relationship with the United States and Japan. Those are the key elements they thought of as part of the strategy of survival in a totally new international environment. The nuclear card and missile card are policy instruments to achieve the policy goal" (Interview Prof. Paik, 2011)

With less Russian engagement, North Korea instead turned more towards self-reliance. The problems North Korea faced in the early 1990s included the regime's lack of confidence and security in the new post-Cold War environment, as well as energy and oil shortages caused by the end of support from China and Russia. Nuclear power increased its appeal as an option for North Korea, because nuclear power can reward a country in terms of both its security and economy. The new post-Cold War environment also forced North Korea to turn itself towards the United States, and an attempt to improve the relationship appeared in the form of dialogue and negotiation between 1992 and 1994. On the one hand, North Korea perceived the United States as the main obstacle to reunification, and advocated withdrawal of US troops from the peninsula; on the other hand, the United States could play an important role

in regime security and regime survival, if it was prepared to offer resources in exchange for restraint. The United States turned out to be an asset in North Korea's strategic calculus. North Korea benefited from the high level talks with the United States, because it raised North Korea's status to coequal with the United States, while at the same time relegated South Korea to a lower status during the negotiations (Koh, 1998, pp. 87-88). However, this did not produce the breakthrough North Korea desired the two sides had different objectives when entering into the dialogue. While the United States sought total nuclear dismantlement, North Korea wanted to a normalisation of relations with the United States. However, North Korea did not believe that the Agreed Framework would help it achieve this aim, and this fact influenced the regime to "have something in reserve in case the Agreed Framework broke down, or was not implemented in a satisfactory fashion" (PBS, 2003).

In addition to the external environment factors that influenced North Korean behaviour in the 1990s, the regime was faced with the new transformation that resulted from the sudden death of Kim Il Sung in 1994. Kim Jong Il, who had been designated the next leader in the 1980s, took control of the country and, under his personal dictatorship, North Korea's internal system underwent some dramatic changes. To begin with, the military was the top priority of the country and its significance was elevated after the announcement of the military-first politics in 1995 (*Understanding North Korea 2012*, 2012, p. 58). The 1995 declaration of military-first politics reflected the perception of threat in North Korea and the need to survive in the post-Cold War environment. From observing the collapse of Communism, North Korea concluded that capitalism was a major cause for the downfall. According to Jeon (2009, p. 183), such understanding pushed North Korea into two things – tighter control of the regime in order to maintain the system, and the strengthening of the military in order to defend the regime against any possible capitalist influence.

Military-first politics became a method for consolidating Kim Jong II's power. The military was given a certain degree of significance, with changes ranging from Kim's field guidance visit to military units and the replacement of old military officers with new officers loyal to him (H. Park, 2007, p. 5). By implementing the military-first policy, the leader's Cult of Personality was further emphasised, and Kim Jong II became the political icon of the regime – equal to his father, who first introduced the Juche ideology to North Korea. Kim Jong II

was promoted as an "eternal chairman of the National Defence Commission of the DPRK⁵", due to his military-first initiative (KCNA, 2012e). Military-first politics may have been implemented for the purpose of preventing the collapse of the regime by using the military to safeguard the country, as well as to divert the attention of the people from economic hardship. However, by implementing this strategy, the military gradually played an evermore significant role in North Korean foreign-policy making. Given the perception of imperialism, coupled with the military-first politics, Kim Jong II sought ways to make North Korea a nuclear state.

The development of nuclear weapons was based on both domestic politics and also external factors, including the strength of South Korea and American involvement in the region. Prof. S. H. Sheen (2011) made this comment about the correlation between nuclear weapons and domestic politics:

"Although the main source of regime's insecurity comes from outside with the isolation North Korea feels after the Cold War, it also comes from inside. That sort of insecurity comes from the nature of the regime itself. The totalitarian nature makes it impossible for North Korea to take a path to reform. They do not want to change their basic systems. This is the fundamental reason why they feel insecure that they seek for the development of nuclear weapons" (Interview Prof. S. H. Sheen, 2011)

Kim Jong II viewed the United States as posing a threat to North Korea both militarily and economically. Given that the regime was facing these threats, Kim Jong II considered it imperative to implement military-first politics, and to push forward with a nuclear weapons programme. I. Kim (2009, pp. 88-89) argues that Kim Jong II aimed to establish diplomatic relations with the United States, and believed the use of tactics with nuclear weapons would help him achieve this goal. North Korea's aggressiveness, to some extent, was also the product of the implementation of military-first politics. The military had the power to intervene in and decide on non-military issues after military institution was elevated to a

⁻

⁵ The National Defence Commission (NDC) is currently the highest organ that controls and commands all military and defence-related matters. Kim Jong II amended the North Korea's Constitution in 1998, which lifted the status of the NDC and gave the NDC power and independence in its control over the military.

higher position than other institutions, as part of the military-first politics (S. Snyder, 2009a, p. 43). As a result, the country's foreign policy and diplomacy were also directed by the military (Ra, 2011). Moreover the regime was fully aware that even the military first policy was not closing the gap on the South Korean military capability. The North Korean army was large but could not match South Korea in other ways. Prof. Paik (2011) comments that,

"Statistically, North Korea is a militarily strong country with the military is ranked fourth in the world. But because of the limited resources and because of their economy, they are experiencing a lot of difficulties in modernising their equipments and weapons. This is the factor influencing North Korea to develop nuclear weapons. Conventional weaponry and the military power have been very much weakened despite statistic number. But if you add nuclear capability to that, its military is powerful. This is the reason why South Korea and other countries try to remove North Korean nuclear weapons." (Interview Prof. Paik, 2011)

North Korea is evidently a state that operates under a high level of threat perception, but also induces high insecurity for other states. This is not only due to its behaviour but because of difficulties in understanding its intentions. Stein (2013, pp. 366-367) writes that the security dilemma usually occurs if the states facing each other are both security seekers, but that threat perception would be problematic if intentions were difficult to read. The nature of the North Korean regime makes it difficult for outsiders to fully understand the objectives and intentions behind its belligerent actions. The secretive nature of North Korea is a result of the regime's sense of insecurity, which has developed since the Korean War. Concerning this, Prof. M. Kwon (2011) observed that,

"Economically, North Korea feels threatened by South Korea. Militarily, North Korea is so afraid of the United States. During the Korean War, the United States bombed all major cities in North Korea. They were devastated so they began to build the military facilities underground. That is not because they are going to attack South Korea, but because they are so afraid of the US air power." (Interview Prof. M. Kwon, 2011)

From the perspective of Kim Jong II, the United States posed a serious threat to North Korea. North Korea perceives the US-ROK joint military exercises "an act of invasion of North Korea", during which the country has to be on alert and prepare to counterattack (Wit, Poneman, & Gallucci, 2004, p. 24). Prof. I. H. Park (2011) explains North Korea's threat perception towards the US-ROK joint military exercises thus:

"North Korea responds very sensitively whenever South Korea and the United States conduct the joint military exercise. This is due to an unsettled maritime territory in the West Sea. It is called the Northern Limit Line. It is a kind of a ceasefire line between North and South Korea. The line is drawn without approval from North Korea. North Korea has opposed and insisted the redrawing of this line." (Interview Prof. I. H. Park, 2011)

Even after the Cold War North Korea lacked a legally binding assurance, such as a peace treaty, from the United States. The joint military exercises, the deployment of patriot missiles in South Korea, and the sanctions from both the United States and the United Nations were considered to be US efforts to strangle the regime (Martin, 2009, p. 61). It appears clear that North Korea wanted to gain diplomatic recognition from the United States, and that matters related to the Korean peninsula should be settled through bilateral dialogue between North Korea and the United States. Nonetheless, failure to keep to the terms of the Agreed Framework of 1994 not only deteriorated the overall situation, but also destroyed the confidence and trust each side had for the other. While the United States was suspicious of North Korea's secret nuclear programme and believed that it was violating the Agreed Framework, North Korea claimed that the United States was insincere in implementing the agreement, as there appeared to be a delay in supplying North Korea with Light Water Reactors. The problem spiralled, as each side blamed the other for noncompliance. North Korea was bound under the Framework to disclose the location of all undeclared nuclear sites, and to allow inspections of such sites upon the completion of a significant portion of the first Light Water Reactor (LWR). However, the delay in the reactor delivery stopped the process of compliance. In the end, the United States suspected North Korea of conducting uranium enrichment activities, and this accusation deepened North Korea's distrust and belief in America's hostile intentions towards the regime.

3.3 Threat Perception, Nuclear Weapons, and Deterrence

Provocations are a popular tactic North Korea uses in its foreign policy. It started with small-scale provocations along the demilitarized zone against South Korea and the United States, but has grown to encompass large scale military activities directed against its neighbours. On the one hand, North Korea's threat perception is a result of external challenges, such as the US-ROK alliance and also inter-Korean competition. On the other hand, the leader's personal experience and worldview play a key role in shaping the level of the regime's threat perception. A tough childhood taught Kim Il Sung the use of violence as a means of success. This experience influenced Kim Il Sung's personal viewpoint towards threat and that the best solution to counter threats is through violence and conflict (Jo & Marshall, 1979, pp. 180-181). North Korea's pursuit of advanced weapons systems is also designed in part to allow North Korea to conduct confrontation, particularly during times when North Korea is faced with a hostile international environment (Y. Kim, 2011, p. 107). Prof. Y. Y. Kim (2011) says that,

"North Korea has many difficulties to act freely in the world due to sanctions from outside. From the North Korean perspective, they want to escape from this situation. The testing of missiles and nuclear weapons are to send message to the United States and South Korea that we want to have a dialogue with you. They want to escape from the sanctions and to have diplomatic relations with the United States." (Interview Prof. Y. Y. Kim, 2011)

During the leadership of Kim II Sung, North Korea was driven by insecurity caused by a fear of the US nuclear threat. Cha (2002, p. 219) writes that Kim II Sung developed an interest in nuclear weapons after witnessing the destructive power and deterrent potential offered by nuclear weapons during World War II. The logic of nuclear deterrence is the official rationale for North Korea's nuclear ambition. This stems from the North Korean belief that the threat of an American nuclear attack is real, and the US inclusion of North Korea as part of an axis of evil and identification as a rogue state confirmed for North Korea the American threat to the regime (Moon, 2007, p. 28). The leader's perception of the insecurity posed by the US threat led to the creation of propaganda encouraging the upgrading of North Korea's defence power in order to counter the threat from the United States. Kim II Sung once referred to the

United States as "the most vicious aggressor and biggest robber in the world", and the main obstacle to Korean reunification (I. S. Kim, 1981, pp. 112-125).

Kim Il Sung's personal preference for expanding military strength after the end of the Korean War pushed North Korea to implement the military build-up programme of the 1960s, and also the development of a nuclear weapons programme in the 1970s. The nuclear ambitions started in the early 1960s, after Kim II Sung sent North Korean scientists to the Soviet Union to research the nuclear fuel cycle. The nuclear research began with the establishment of a nuclear physics laboratory, and developed finally into the creation of the nuclear research complex in Yongbyon, in the mid-1960s. North Korea slowly gained confidence in its nuclear research activity, and it undertook an independent effort to upgrade its nuclear reactor in the 1970s. This lead to speculation that the nuclear weapons programme was secretly launched in the five megawatt reactor in Yongbyon during at this time (Cha, 2012a, pp. 242-243; Zhebin, 2000, p. 31). What arose from this scenario was that, when North Korea built up its military capacity, it created a level of confidence in the regime on both the issues of deterrence against outside threats and also the potential for reunification under North Korean terms. Although it has been recorded that North Korea's military capacity has continued to grow since the mid-1970s, the tactic North Korea would use if a war broke out would be a quick and decisive battle. Part of the explanation comes not only from the fact that North Korea could not afford to sustain a long-term war – a result of its weak economic capacity – but also from the Juche strategy North Korea adopted. To elaborate:

"North Korea's 'Juche strategy' is a mixed strategy combining Mao's guerrilla tactics with old Soviet military strategies in a way that suits the actual situation on the Korean peninsula. The key essence of this strategy is to combine large-scale regular warfare with guerrilla attacks, which will enable the North Korean forces to engage the enemy from every direction." (*Understanding North Korea 2012*, 2012, p. 123)

Guerrilla warfare is a tactic Kim II Sung is familiar with from his experiences during his youth as an active nationalist leader of the anti-Japanese guerrilla unit in Manchuria. The construction of underground tunnels along the DMZ was part of guerrilla warfare tactics aimed at facilitating a surprise attack against South Korea. It is a strategy aimed at achieving reunification during a time when North Korea was better off than South Korea (Prof. M.

Kwon, 2011). D. Kang (1995, p. 262) says that North Korea and South Korea tend to compete with each other, and the more South Korea continues to build up its economy and military, the more aggressive North Korea will become, because it wants to gain any opportunity of ensuring victory in a war.

Kim Il Sung's military programme may have been for the purposes of competing with South Korea by giving North Korea the power to achieve the regime's ultimate goal of reunifying the two Koreas under North Korean terms. The strategic logic behind the high level of militarisation came from the North Korean government's desire to defeat the South Korean army in a potential forceful reunification of Korea. However, a vicious cycle has been created by North Korea's struggle to maintain parity with South Korea's military level, which in turn has caused the North Korean economy to weaken, and the slowdown instead prompts the government to increase military budget allocations (Lankov, 2013, pp. 71-72). North Korea's reunification policy turns out to be mere political rhetoric to preserve the leader's power and regime's existence. Dr. E. C. Lim (2011) notes that,

"North Korea's rhetoric is very different from their real intention. The main objective of North Korea is to maintain current status and to co-exist with South Korea. This is because any reunification would see North Korea absorbed into the South Korean system." (Interview Dr. E. C. Lim, 2011)

North Korea's fear of being absorbed is highlighted every time South Korea mentions the unification of Korea, and is elevated when the South Korean government is run by conservatives. North Korea seemed particularly threatened after President Lee Myung Bak mentioned reunification and his proposed "three-step plan for reunification", including preparing South Korea to bear the burden of the reintegration cost after the collapse of North Korea (Harlan, 2010b). Prof. M. Kwon (2011) comments that,

"President Lee's speech reflected South Korea's intention to absorb North Korea. The anticipation for the North Korean collapse also threatens North Korea. He implies that North Korea will be overthrown or will go bankrupt

bear the cost of reunification (The Chosun Ilbo, 2010b).

_

⁶ The three-step plan calls for the establishment of a "peace community" and "economic community". While the former refers to the denuclearisation of North Korea, the latter outlines the improvement of North Korea's economy through exchanges. The plan also proposes a so-called "unification tax" to

soon. It will just crumble down then the reunification will occur. What he is looking for is not the dialogue but the North Korea's collapse." (Interview Prof. M. Kwon, 2011)

According to Herz states seek ways to obtain or advance their security, but a desire for security would be higher than usual if states were in search of escaping the impact and influence of the power of others (Herz, 1950, p. 157). North Korea seems to compete with South Korea in almost all aspects, and the more it falls behind South Korea, the more it is faced with the threat and insecurity resulting from a fear of being absorbed by South Korea. Nuclear weapons, as such, function as a means to secure the regime's survival amidst inter-Korean competition, and the change in the international security environment. Nuclear weapons, to North Korea, are not for threatening a nuclear first strike (Huntley, 2009, p. 159) or for war-fighting purposes, because the North's military lacks the proper training in the use of nuclear weapons or nuclear war-fighting (Cha, 2002, pp. 218-219). Instead, they are for deterring a regime overthrow or regime absorption strategy (D. Smith, 2006, p. 84).

The concept of deterrence was clearly displayed during the nuclear proliferation crisis in the 1990s. The level of North Korea's threat perception continued to increase after the Agreed Framework was signed, which made total nuclear disarmament highly unlikely. Nuclear weapons were developed primarily to secure the regime as a means of deterring the United States. However, the negotiation process taught North Korea another important value of nuclear weapons when used in negotiating with the United States. According to Prof. Moon (2011) the uses of North Korea's nuclear weapons programme are as follows:

"North Korea has been developing nuclear weapons for national survival. They have been developed as a part of nuclear deterrence. They think they can come up with minimum nuclear deterrence against the United States. This is the most important motive. But in the process, North Korea has learnt the value of nuclear weapons in negotiating with the United States. Therefore, nuclear weapons can be effective way of extracting economic concession from the United States and other countries in the region for their economic benefits." (Interview Prof. Moon, 2011)

It turns out that both the United States and North Korea share some responsibility for the failure of the Agreed Framework. The United States was reluctant to implement the Agreed

Framework because it believed that North Korea could soon collapse. North Korea, on the other hand, also raised tensions in the region in 1998 with its launch of a missile over Japan. The discovery of the suspected, secret underground nuclear site at Kumchangri, near North Korea's border with China, also raised some speculation over North Korea's sincerity on nuclear disarmament.

North Korea's threat perception was further elevated with the inauguration of US President George W. Bush, and the adoption of a hardline policy after 9/11, which saw North Korea named as part of an axis of evil. The outbreak of the second nuclear crisis in October 2002 showed that the sense of insecurity has not been lowered, despite the dialogue and negotiation of the Agreed Framework. Prof. Moon (2011) notes the regime's sense of fear and lack of confidence, and says,

"There is a lack of recognition. In building up North Korea's confidence, you have to recognise North Korea and then engage with North Korea. The United States, Japan and South Korea have always been telling that we are right and North Korea is wrong. But in many cases, it is not true. If you look back to 1994 [first nuclear crisis] up until now, the United States also fail in its mission to deliver the so-called Light-Water nuclear reactors as part of delivery with the Framework. From my point of view, I think the United States breached the promises." (Interview Prof. Moon, 2011)

According to Y. Kim (2011, pp. 113-122), North Korea intentionally projects a belligerent image because a provocative foreign policy is the product of the regime's cost-benefit calculations, using verbal statements to provoke its adversaries whenever it sees that the risks do not place the regime in danger. What makes North Korea successful in applying this strategy is that both North Korea and its adversaries do not want to upset the status quo.

North Korea's provocative strategy appears effective due of two factors: The first is that there is a discrepancy between North Korea's actual behaviour and rhetoric, and the second is the fact that North Korea's adversaries appear to be highly risk-averse. Scholars such as Morgan (2003, p. 44) and Stein (2009, pp. 60-61) say that the best choice for states during a time of conflict is to avoid direct confrontation unless it would bring about benefits based on a rational loss-and-benefit calculation. The United States may not want a war with North

Korea, but it faces challenges in finding an effective approach towards North Korea – ranging from the engagement of the Clinton administration, to the confrontation of the Bush administration. American engagement with North Korea appears on a multilateral platform in the form of the Six-Party Talks, rather than bilateral negotiations, which North Korea has long requested. Multilateral dialogue, in other words, implies an American reluctance to engage with North Korea on its own, the result of which erodes the bilateral relationship and trust.

At present, the internal system of the regime is structured on two main elements: military-first politics and the Juche ideology. Military-first politics continue to be popular in North Korean society, even under the new leadership of Kim Jong Un. The significance of the songun policy was emphasised alongside socialism and the Juche as being key elements of "a far-reaching strategy for final victory of the Korean revolution" during a speech made by Kim Jong Un, at the military parade celebrating the centenary of the birth of Kim Il Sung (KCNA, 2012d).

The implementation of military-first politics reflects two things. The first is the military's control over certain domestic matters which, with the prioritisation of the military institution, shows not only North Korea's insecurity but also its unusual internal systems. Jeung (2007, p. 29) writes the following about the internal system of North Korea after the implementation of the military-first politics:

"Institutionally, the North Korean authorities are likely to opt for the policies designed to expand the roles and functions of the military. The incumbent members on the National Defense Commission are mostly military leaders, which means that these military leaders will rely on 'military logic' in policy discussions and they are likely to have a significant impact on all policy decisions. Consequently, major security policies, including nuclear and missile issues, as well as inter-Korean military dialogue, which are directly related to military values, will probably be decided in favour of the logic and interest of the military. This will also mean that the independence of the military is being enhanced rather than abridged."

The second point reflects the leader's struggle to hold on to a dictator's role and power, and also in ensuring the regime's survival in the new, post-Cold War environment. External factors, such as the inter-Korean conflict, did not have much influence on North Korea's decision to implement the military-first politics, compared to internal factors/considerations. According to Y. Park (2012, pp. 326-327), it was because of a leader's desire to maintain the dictatorship, particularly under the leadership of Kim Jong Un, who was faced with the challenge of continuing the Kim family's ruling style amidst the risks of economic hardship and international sanction.

Concerning the Juche ideology, Kim Jong Un continued to emphasise the significance of self-reliance – as can be seen in his New Year Address of 1 January, 2013. The issues highlighted in the Address included the Juche as a pillar of Kimilsungism-Kimjongilsim, the songun, the economic growth of the nation, and North-South relations. Kim Jong Un said,

"The road of Juche is the only path for our Party and people to invariably follow, and great Kimilsungism-Kimjongilism is the ever-victorious banner of our revolution. We should march forward along the road of independence, the road of Songun, and the road of socialism to the end upholding the banner of Kimilsungism-Kimjongilism. Our Party will triumphantly build a thriving socialist nation, the most powerful country, on this land in our own way" (KCNA, 2013b)

Currently, the ideological system of North Korea is composed of both Juche and songun together: the former is the symbol of Kim Il Sung, and the latter of Kim Jong Il. There is an argument that, although North Korea continues to be a Juche state, the significance of Juche ideology has changed over time. South Korea has observed North Korea and raised the issue of the changes to Juche both at the governmental and the public levels. In a book titled "Understanding North Korea 2012", the Ministry of Unification of South Korea claimed that the Juche has been weakening since the 1994 economic crisis, and especially after Kim Jong Un's adoption of Kimilsungism-Kimjongilism (*Understanding North Korea 2012*, 2012, p. 33). The Chosun Ilbo, one of the leading newspapers in South Korea, speculated about the significance of Juche with particular reference to the heavy emphasis on the "principles of Kim Il Sung and Kim Jong Il" the KWP made in the revised regulations (The Chosun Ilbo, 2012b). The Yonhap News Agency also reported that Kim Jong Un, instead of focusing on

the ideology of Juche like his father, spoke more about the idea of reforms and opening in order to catch up with global trends (Yonhap News Agency, 2012a). The KCNA refers to Kimilsungism-Kimjongilism as a reflection of Kim II Sung and Kim Jong II's ideas, and it is now been adopted to be "the guiding idea of the Workers' Party of Korea" (KCNA, 2013a). Cha (2012a, pp. 58-59) calls the Juche of Kim Jong Un "the neo-Juche revivalism", which calls for the absolute loyalty of the people to the leader and the strengthening of the military, in the same manner as the propaganda of the 1960s and 1970s.

Kim Jong Un's focus on the economy has been visible since his early days in power. For example his speech of 15 April 2012, and its objective of improving the living standards of the North Korean people (KCNA, 2012d). The goal of improving the economy was highlighted again in the 2013 New Year Address, when he emphasised the need for building up the economy and for boosting science and technology to support this economic development (KCNA, 2013b). Dr. E. C. Lim (2011) made this observation on the correlation between the economy and the Juche ideology:

"The Juche ideology still exists in North Korea, and still plays a very important role, even though the power and influence of the Juche ideology has been weakened by the economy reform plan." (Interview Dr. E. C. Lim, 2011)

The purge of Jang Song Thaek, in December 2013, raised some concerns of the regime's internal instability and the future of the regime's economic reform, given Jang's rank as the second most powerful man in the country and his record as the greatest supporter of the Chinese-style economic reform. Jang was executed for various charges, but, among the charges, Jang was accused of being too soft when it came to South Korean policy, and Jang was also against expanding military power in North Korea (O'Carroll, 2013). One of the possibilities for Jang's purge is the disagreement he had with Kim Jong Un. Kim Jong Un and others in the Politburo and the Cabinet sought military prioritisation under the so-called "byungjin line", while Jang's economic reform spoke of a plan to open up North Korea, and move the country closer to China (Mansourov, 2013). Kim Jong Un's byungjin line appears to have been rather similar to Kim Il Sung's 1962 byungjin line, which talked about simultaneous development of the economy and military. The byungjin line of Kim Jong Un, however, emphasised defence and deterrent through the development of nuclear weapons (S. Y. Kim, 2013).

Kim Jong Un and other hardliners viewed Jang's policy as a threat to North Korea's socialism, when his policy of reform threatened to fundamentally change the system. Also the public's favourable view of Jang as "an economically-savvy manager who knew a lot about China", was considered detrimental to the leader's own image (Y. H. Kim & Kang, 2013). However, it is argued that, despite the removal of Jang, the policy of economic reform and opening will nevertheless remain a primary focus. It was reported on 8 December, 2013, the day Jang was dismissed from power, that North Korea would continue with the construction of a high-speed railway and express highway in order to connect the cities of Sinuiju and Kaesong with Pyongyang (Paik, 2013). One of the consequences of Jang Song Thaek's execution was the strengthening of the songun. Kim Jong Un's public appearances throughout December 2013 were mostly at military-related activities, and this was believed to be a way to consolidate the leader's power and strengthen the songun, which had been weakened by Jang's economic development [proposals/policies] (Grisafi, 2013).

3.4 Conclusion

Internal determinants have significant impacts on North Korea's threat perception. First of all, one must look at the nature of the regime. North Korea's system is not exactly a socialist system, but rather a combination of socialist ideology with the Asian culture of Confucianism. When Kim Il Sung came to power, he also introduced the idea of nationalism and the cult of personality of the leader, which turned the system into more of a "North Korean-style socialism" (Interview Prof. Paik, 2011). The internal system is formed on various elements but the most important element is the Juche ideology. North Korea takes pride in its Juche principle. Although the Juche may be weakened over time, it is still seen as the bedrock principle of the country. The Juche helps Kim Il Sung consolidate his power and paves the way for the structure of the internal system to centre on the single leadership. This system makes the leadership security become the regime security. This kind of system made the country strong under the tight control of the leader. However, the negative side is the fact that even a slight change in the system would mean the collapse of the state. As such, North Korea is vulnerable, and its vulnerability was clearly seen when the international political structure changed with the collapse of the Soviet Union and the demise of the Communist

bloc. It is the leader's desire to keep the system and the country as it is which makes North Korea unable to reduce its level of military defence, or give up on its development of nuclear weapons (Interview Prof. S. H. Sheen, 2011). The North Korean socialist system, the Juche ideology, the military politics, and the personality cult of the leader all form the identity of the North Korean regime. It is the identity of North Korea which has a high sense of nationalism. Nationalism does not mean unification of Korea but refers instead to the desire to keep the internal system and overall structure without any change despite the trend of globalisation. North Korea does not want to lose its identity. Although there is a possibility for the existence of the North Korean regime after the major reform, the loss of this identity would mean North Korea has turned into a completely new state. This is a scenario that the North Korean leader does not want and it is the reason why North Korea has a succession within the Kim's family line (Interview Dr. E. C. Lim, 2011).

The internal determinants affect the viewpoint of the leader. Vulnerability causes a sense of insecurity and this insecurity is pushed to a higher level by changes in the international political environment. The first nuclear crisis in 1993 showed North Korea's security sensitivity. The search to obtain nuclear weapons is a result of this sense of insecurity. North Korea also fears the spill-over effect that the international environment could have on the regime, which could lead to regime change. The introduction of Kimilsungism-Kimjongilism, songun, and kangsong taeguk was to justify the existing system and leadership amidst the changes of the post-Cold War period. North Korea could not lower its defence. At the same time the military serves a purpose of defending the state, it also uses to secure the leadership power. The leader has a tight control on the military sector in order to ensure that the chance of regime change by the internal revolution is at the minimum (Interview Prof. I. H. Park, 2011). Therefore, it is possible to see that the North Korean leader is credited every time the military actions are carried out. This is to give the credit to the leader who is the head of state and the leader of the military sector. North Korean leader manipulates the external threat for internal benefit, promoting the legitimacy of the leadership's power and the system.

As discussed in Chapter 2 International Relations Theory speaks of the correlation between a state's identity and its interests. A state's interest is defined by their identity, creating a relationship among states based on their identity. North Korea views and understands the world based on its own perspectives that have not changed greatly since the country was founded. It is a socialistic state with a nationalistic worldview. It is the regime's desire to

preserve itself that shapes North Korea's desire for security and influences its choice of tactics and the way it seeks to secure this interest. As claimed by Legro and Moravcsik (1999, p. 12), states would search for the most effective means to secure their own goals. North Korea chooses to strengthen its defence capacity. It is true that defensive mechanisms North Korea has developed are for state and regime security, but at the same time they also serve to secure North Korea's own identity. The defensive decisions and provocative actions turn out to be more than just for the purpose of security. Nuclear weapons are used beyond the purpose of deterrence. At one point, they were referred to as a result of good leadership. They were used as a means to show the intelligence of the leader in helping the nation survive regardless of the changes in the international political structure. It is this identity that creates the tensions between North Korea and other countries. Wendt (1995, p. 73) stresses the importance of "shared knowledge" among states on the relative security value. In other words, whether one state sees the other as a threat to its security depends on the relationship they have – whether they are friend or foe (Wendt, 1995, p. 73). The case of North Korea shows difficult relations with other regional powers at various times. As a result, mutual threat levels have remained high over time. The unique system that North Korea has is defined in various terms – such as 'authoritarian', 'socialist', 'militaristic'. Other countries may view North Korea's defence level as high, but the defensive measures, however, are developed to serve both the security and interests of the regime. They are developed and adjusted in accordance to the security challenges North Korea faces, both internally and externally, as well as reflecting the needs and ideologies of the regime and its ruling family.

Chapter 4: Inter-Korean Relations

The Korean peninsula was split into two separate states after the United States and the Soviet Union reached an agreement regarding the future of Korea following its liberation from Japanese occupation. The division became starker after the end of the Korean War through the differences in each country's political systems. South Korea adopted a democratic system and experienced several internal changes during the process of democratisation. During the Cold War, the politics of the Korean peninsula were affected by the politics at the international level, as the ideological struggle between the United States and the Soviet Union deepened. The situation became more hostile as the two Koreas saw each other from different political perspectives. A diplomatic breakthrough, in the 1970s, could not lower tensions on the Korean peninsula. It turned out that, as the two countries developed at different rates, the two Koreas become more different and unification as a goal appeared ever harder to achieve. It is noteworthy that, since the division of the Korean peninsula, the two Koreas have been driven by a desire to compete against each other, and to achieve Korean reunification under their own terms.

However, the issue of reunification turned out to be mere policy rhetoric, as the future of Korea will be determined by the factor of economic development. The considerable economic gap makes neither North Korea nor South Korea able to bear the reunification costs. By falling behind South Korea, North Korea is threatened by the possibility that the regime will one day be absorbed by its neighbour. North Korea's threat perception became elevated as time passed, and it became more difficult for the country to secure its communist ideology amidst the changing international political environment. Since the threat perception of North Korea is a creation of the interaction between internal and external determinants, it is worth studying the external factors influencing North Korea's relationships with its neighbours, and with South Korea in particular. Two observations arise from the issue of inter-Korean relations. The first is North Korea's threat perception as driven by South Korea's policies, and how North Korea's impression of the South's intent is reflected in its reaction. The second is North Korea's behaviour, either its provocations or military deterrence, are dictated by purposes other than deterrence, such as coercing South Korea into

reunifying with North Korea. The wider the gap in economic development levels, the more North Korea seeks to achieve reunification in a manner that will avoid its absorption by South Korea.

This chapter will study the relationship between the two Koreas since the end of the Korean War, in order to explore how inter-Korean relations have affected North Korea's threat perception. In particular, it seeks to determine whether North Korea's behaviour towards South Korea, including the military and other forms of provocations, exists in order to deter South Korea or for purposes that cannot be connected to deterrence. It asks whether or not North Korea's provocations towards its southern neighbour fit into a pattern of rational deterrence behaviour.

4.1 South Korea's Policy towards North Korea and the Inter-Korean Relations

The relationship between the two Koreas during the Cold War was generally hostile. The foreign policy of the first Republic of Korea was developed along the anti-Communist policies of President Syngman Rhee, a pro-democratic politician. There were no diplomatic ties established, and each side portrayed the other as an enemy and a "puppet" of either the United States or the Soviet Union (H. J. Kim, 1977, pp. 106-108). Throughout Syngman Rhee's administration, South Korea approached North Korea with a policy of "marching north" or "bukjin tongil" as a means to unite the country through force, a policy born from fervent anti-Communism. Any negotiations with North Korea for reunification would be seen as sympathising with the Communists (K. A. Park & Lee, 1992, p. 430). In the early stages of the post-Korean War period, South Korea struggled to rehabilitate its economy and to make progress in its political development. A failure in democratisation meant the country was faced with internal turmoil in 1960. It began with public demonstrations in big cities in South Korea, such as Seoul and Pusan, and ended with Syngman Rhee's resignation in the same year. The protests not only helped to change the political system in South Korea, but also had an impact on North Korea's perspectives of the North-South dialogue, and US involvement in Korea. According to Shin, Ostermann, and Person (2013):

"After the April 19 Revolution, the North Korean leaders seemed to have believed that reunification was not only possible, but imminent (with the withdrawal of U.S. troops from South Korea). Moreover, because of North Korea's relative political and economic stability, Kim believed reunification could be achieved on North Korean terms."

These perceptions pushed North Korea to reinforce its plans for building up its defence industries, and the pro-industry development strategy that marked the beginning of North Korea's economic collapse (Shin et al., 2013). The policy of 'marching north' was later abandoned by the government in Seoul, under the leadership of Chang Myon of the second republic. The Chang government, though it continued to express anti-Communist policies, declared its preference for an alternative approach to unification; that of holding free elections in both North and South Korea, under the supervision of the United Nations (H. J. Kim, 1977, p. 179). Despite this positive sign from the Chang government on North Korean policy, which inspired increased mutual exchanges, a level of animosity between the two Koreas persisted. Tension rose particularly after Kim Il Sung's proposal for a federal unification strategy, in which he advocated for replacing the South Korean government with Communist principles.

General Park Chung Hee came to power by making a claim that the parliamentary system of the Chang government had failed to establish social order in light of the fact that South Korea was faced with a growing threat from Communist North Korea (H. K. Lee, 1974, p. 44). Under the presidency of General Park, the Constitution was amended to justify the general's accession to power, and to assist in his seizure of absolute control. The amendment took effect in 1962, reintroducing the presidential system of government. In 1969, it was amended again to allow the president to hold a third consecutive term in office. After acquiring these amendments, President Park went further in his attempt to construct a Korean-style democracy by introducing the Yushin system through a 1972 Constitution amendment. The Yushin system, or the Restoration order, was established with President Park's motive for South Korea to achieve a status of "rich nation and strong army". By implementing this system, President Park gained absolute power over the executive, legislative, and judicial branches of the government, which he reassembled as a dictatorial leadership under a democratic system. The Park government focused on three main policies: anti-Communism,

strengthening national power, and building up economic power. North Korea felt threatened by the policies of President Park, particularly after he outlined the necessity for economic and national strength as a prerequisite for a unified Korea. North Korea felt so threatened by this policy that it sought closer security ties with China and the Soviet Union, and increased its own defence capabilities (Koh, 1980, p. 1110).

Inter-Korean relations saw some positive signs of improvement in 1970, with President Park's speech on 15 August 1970, commemorating the 25th anniversary of the Korean liberation. The speech had significant details concerning unification, stating clearly South Korea's intention to remove barriers to North-South relations if North Korea renounced the policy of unification by force, and ceased its hostile acts aimed at overthrowing the South Korean government (C. H. Park, 1978, pp. 27-28). However, President Park's speech increased North Korea's level of threat perception because, to North Korea, the proposal had the political objective of forcing unification on South Korean principles (Koh, 1971, p. 492). North Korea continued to denounce President Park as an obstructer, rather than facilitator of unification. Tensions on the peninsula were further heightened when the two Koreas applied for membership in the United Nations. While President Park stated his desire for the two Koreas to join as separate entities, North Korea preferred the two Koreas be entered as a single state under the name of "Confederal Republic of Koryo" (Choi, 2001, p. 96). North Korea still criticises President Park's policy today. The KCNA released on 10 August 2012 a memorandum denouncing President Park Chung Hee's governance and policies, including "the two-Korea policy". The KCNA reported that "the two-Korea policy" announced in June 1973 was an attempt to divide Korea permanently, and that it was a violation of the July 4, 1972 Joint Communique⁷ planned for the reunification of Korea under three principles (KCNA, 2012b).

-

⁷ The two Koreas reached a breakthrough in diplomatic relations at the governmental level in 1972. The July 4 South-North Joint Communique was issued outlining three major points towards achieving reunification. The first is the peaceful reunification by solely Korean effort. The second calls for a reduction of tension and military confrontation between the two Koreas. The third point calls for an exchange of the two Koreas in non-military areas, such as economic and cultural. The full text of the July 4 South-North Joint Communique can be found on the Ministry of Unification, Republic of Korea website: http://eng.unikorea.go.kr/content.do?cmsid=1806 (Ministry of Unification).

President Park's speech of 15 August 1970 showed South Korea's belief in a potential military assault from North Korea. South Korea's security concern about a North Korean threat in the 1970s was pushed by two incidents: the fall of Vietnam and the US adoption of the Nixon Doctrine. To ensure South Korea's security, President Park agreed with the establishment of the secret defence project, "the Yulgok Operation" in order to strengthen the military. The result, however, pushed the two Koreas into an arms race (H. A. Kim, 2004, pp. 188-190). The military played a significant role in President Park's political posture. President Park not only used the military to suppress those who opposed his rule, but he also used the military as a tool to pursue a self-defence reliance policy when the US security commitment in Asia started to wane. The threat from North Korea appeared real even though North Korea started to suffer from economic decline. South Korea was prompted to upgrade its defence and weaponry by North Korea's heavy investment in its military, which was in turn inspired by a desire to end US imperialism in South Korea (J. H. Kim, 2011, p. 169).

A security threat from North Korea drove President Park closer to a plan to develop nuclear weapons in late 1974. The acquisition of nuclear weapons was considered a means to improve South Korea's self-defence amidst the decline in the US security commitment as part of the Nixon Doctrine. North Korea's provocations, such as an attempted assassination of President Park in January 1968, the so-called "Blue House raid" incident, were also factors that deepened the nuclear ambitions of President Park (Hayes, Moon, & Bruce, 2011). The nuclear weapons project was launched as a means to compensate for a South Korean sense of insecurity resulting from falling behind North Korea in terms of military capability. Hong (2011, p. 488) writes that South Korea relied on the United States to play a deterrent role against North Korea, particularly when North Korea's military was superior to South Korea's in all aspects. Nuclear weapons, therefore, were considered the answer to fulfil a deterrence strategy, especially when the US role began to decline and the conventional military was seen as not enough to deter North Korea. Park Chung Hee's decision to upgrade the military and go nuclear affected North Korea's sense of insecurity, which was described as,

"Park Chung Hee wanted to develop nuclear weapons out of security concern by the United States policy to withdraw its forces from South Korea. But Park Chung Hee's effort to have nuclear weapons sent the wrong signal to North Korea that they wanted to expedite the process of acquiring nuclear weapons. Certainly, Park Chung Hee played the very important role on North Korea's security." (Interview Prof. Moon, 2011)

During the presidency of Park Chung Hee, inter-Korean dialogue was convened at both the public and the governmental levels. It started with a humanitarian project introduced by the ROK Red Cross Society in 1971, aimed at reuniting families torn apart by the country's division. This led to an exchange of delegates of the two countries to each other's capital. The following year, both sides agreed to sign a joint statement – the July 4 South-North Joint Communique. However, problems arose when the two sides developed different interpretations of the provisions listed in the Joint Communique. This put an end to attempts by each side to manipulate the inter-Korean issue to their benefit. North Korea planned to use the inter-Korean dialogue to undermine President Park's government by encouraging more participation from opposition groups, but the plan failed because the Yushin system helped stabilise South Korea's internal system (Shin, 2012). On the other hand, South Korea viewed the reunification issue as dependent on the level of economic strength of the country, leading to President Park's perception that the more South Korea could develop economically, the faster reunification could be achieved. Cha (2004, p. 142) writes that President Park's desire to unify the two Koreas through peaceful means got twisted by North Korea's provocations, such as the Blue House raid incident, and instead influenced a desire to retaliate militarily against North Korea. Therefore, President Park used the issue of reunification and the country's relative weakness against the threat of communism as means to stimulate the development of South Korea's economic, political, and military strength. In his speech "Let Us Turn the Joy of Liberation Into Joy of Unification" of 15 August 1972, President Park said that the only way to achieve peaceful reunification was to strengthen national power, develop a sustained national economy, and maintain substantial growth of the democratic system (C. H. Park, 1975, pp. 19-24). While the policies implemented by the Park government made South Korea more stable, North Korea became less secure (Interview Prof. Paik, 2011).

To North Korea, unification could only happen if there was a socialist revolution in South Korea. Armstrong (2004, pp. 42-44) writes that North Korea had an idea of a revolutionary process called "the United Front Policy", a concept that refers to revolutionary success in one area before another, until the revolutionary reform erupts across an entire country. Since the Korean War failed to unify the country, North Korea moved forward in its attempts to

destabilise the South Korean government through provocations and infiltrations intended to sow public dissatisfaction. North Korea's arms build-up during the 1960s and 1970s did not only serve its deterrence strategy against South Korea, who at the time was enjoying political stability and economic growth, but was also intended to serve North Korea's unification goal. North Korea's use of provocation against and infiltrations of South Korea, including the construction of underground tunnels along the DMZ, were explained thus:

"The underground tunnels were built in the 1960s and 1970s. At that time, North Korea had a better economy. They built the tunnels in an attempt to unify Korea by invading South Korea. But once South Korea's economy began to exceed North Korea's, then there are no new tunnels being built. They do not have the money." (Interview Prof. M. Kwon, 2011)

The fact that North Korea failed to act during the political turmoil in South Korea in the period between Syngman Rhee's resignation and General Park's military coup, inspired North Korea's military build-up strategy, so that it could take advantage if a similar opportunity were to present itself in the future (Eberstadt, 1998, pp. 241-242).

Since South Korea's economic growth overtook that of North Korea in the 1980s, North Korea has appeared reluctant to enter into a dialogue. Inter-Korean dialogue was deadlocked, particularly when external factors influenced the reunification issue – especially the role played by the United States in Korean affairs. President Jimmy Carter's decision to reverse the withdrawal of US forces from the Korean peninsula in July 1979 ended Kim II Sung's hope of uniting the whole of Korea under North Korean terms. North Korea has long stressed that the US troops stationed in the Korean peninsula "infringed the sovereignty of Korea" and obstructed Korean reunification (Kihl, 1984, pp. 210-211). This period, instead, saw Kim II Sung's efforts to engage the United States in order to replace the armistice agreement with a peace treaty. This was a new approach to the US-DPRK dialogue on peace talks in 1973, which was emphasised again in 1977, as a means to avoid a direct confrontation with the United States. The approach hoped to pave the way for the withdrawal of US troops and weapons from South Korea, and to pursue peaceful reunification. However, the change in Carter's withdrawal plan upset Kim II Sung, and he referred to the decision as "deceitful" and

implemented to serve a US plan to continue military exercises and to import more weapons into South Korea (Oberdorfer, 2001, pp. 99-100).

Both the change in the US troop withdrawal plan and North Korea's failure to engage the United States in bilateral peace talks aggravated North Korea's impression and attitudes towards the United States. This negativity gradually affected the North-South dialogue, particularly with regards to reunification. North Korea set the conditions for the inter-Korean dialogue, that the talks could only resume after the withdrawal of US troops from South Korea and the removal of the government of Chun Doo Hwan, who came to power after Park Chung Hee's death (Kihl, 1984, p. 212). During Chun's government, both sides could not reach an agreement on the formula for reunification. While North Korea preferred the confederation formula of two coexisting political systems, South Korea proposed adoption of a constitution of a unified Korea as a prerequisite to the establishment of a national, unified government of Korea. When both sides clearly articulated their objections towards the other's proposal to solve the reunification issue, the North-South dialogue was again faced with deadlock, and no significant progress was made until the end of the Cold War.

During the 1990s, South Korea underwent the process of democratisation. The government of Roh Tae Woo announced on 29 June 1987 the Declaration of Democratisation and Reform. This consisted of eight major points, including the restoration of basic rights and freedom of the people, the amendment of the Constitution to allow for direct presidential elections, and the promotion of the rule of law and anti-corruption policies (J. K. C. Oh, 1999, pp. 98-101). In an attempt to create his own identity, and to distance himself from the shadow cast by Chun Doo Hwan, Roh Tae Woo carried out a series of foreign policy initiatives that were specifically aimed at enhancing relationships with the Communist states and improving the inter-Korean relations. Roh declared a new foreign policy, Northern Policy or "Nordpolitik", on 7 July 1988. The "Nordpolitik" had six main principles under the name "Special Declaration on National Self-Esteem, Unification, and Prosperity", which focused on improving relations between the two Koreas, and establishing relationships with socialist countries, including China and the Soviet Union (Japan Economic Newswire, 1988).

The Northern Policy was implemented in order to adjust South Korea to the new international trend, a result of Gorbachev's plan to establish ties with countries in Asia. Roh's Northern

Policy enhanced a sense of national self-esteem among the South Korean people, and the focus of reunification shifted so that it would be realised through the concept of a single national community (H. J. Kim, 1993, pp. 257-258). The implementation of the Northern Policy helped lower tensions on the Korean peninsula, because South Korea gradually established diplomatic ties with Russia and China in the post-Cold War period. South Korea's attempt to establish relationships with China and the Soviet Union had a strategic implication, and can be thought of as the method South Korea used to counterbalance the North Korean-Soviet and North Korean-China alliances, and to seek to reduce North Korea's aggressiveness over time (S. S. Park, 1993, pp. 221-223). On the one hand, Roh seems to have believed in the engagement strategy with North Korea as a means to end the diplomatic competition and adversarial relationship between the two Koreas. This can be seen in the July 1988 six-point declaration. On the other hand, it appears that Roh also considered the role of the US-ROK alliance and the significance of a deterrent strategy in achieving Korean reunification. The US-ROK alliance was the top priority of the Roh government. The US deterrent is a defence against any possible future military attack from the North, and is a tool for pressuring North Korea to unify with South Korea under the South's unification formula (Wood, 1993, pp. 215-216). The United States is said to have played a significant role in improving inter-Korean dialogue in the post-Cold War years. The desire to lower tensions on the peninsula drove the United States to adopt the role of facilitator in enhancing contacts and exchanges between the two Koreas. However, Cha (2004, p. 143) says that such a role was actually pushed by a demand from Seoul, who started to feel insecure about being left out after seeing a positive development in the US-DPRK relations. As such, the inter-Korean dialogue in the early post-Cold War years also involved the US-ROK alliance, which South Korea used to engage with North Korea. The issue of North-South relations became more significant and it was listed in the 1994 Agreed Framework as a condition for the implementation of the agreement.

South Korea's democratisation process reached a significant level following the presidential election in 1992, which led to the victory of the first civilian president in thirty-two years, Kim Young Sam. After realising that true democracy could not be achieved with the overwhelming power and influence of the military in politics, Kim Young Sam launched reforms to weaken the power of the military. This included a purge of high-ranking military personnel who were members of a secret organisation within the army called the "One

Association" or "Hanahoe", established in 1961 under Park Chung Hee's initiative to privatise military-political controls.

The democratisation process had a significant impact on South Korea's economy and foreign policy. President Kim Young Sam planned to open up South Korea's economy and also sought to increase dialogue with North Korea. One reason for South Korea's decision to embark upon a policy of engagement with North Korea was a change in South Korean public sentiment. South Korea did not consider a threat from North Korea imminent, but rather "a lifestyle threat" due to the problems presented by an influx of North Korean refugees and the economic burden of reunification that South Korea would face (Levin & Han, 2002, pp. 10-11). Levin and Han (2002, p. 11) says, "together with the dramatic process of democratisation that occurred almost simultaneously in the ROK, such developments reinforced the view in the government and public alike that deterrence – while indispensable – needed to be supplemented by some form of engagement". Prof. I. H. Park (2011) commented on the changes to South Korea's policy on North Korea from the Cold War to the post-Cold War period:

"South Korea's democratisation in the 1980s makes some interesting change to North Korea's policy. However, it is the structural change from the Cold War to the post-Cold War rather than the factor of the democratisation that makes the South Korea's new approach to North Korea possible. South Korea's approach to North Korea is a kind of balancing between North-South relations and the international perspectives" (Interview Prof. I. H. Park, 2011)

The atmosphere on the Korean peninsula appeared to be more relaxed during President Kim Young Sam's time in office. The speech President Kim Young Sam made on 15 August 1994, Liberation Day, contained some details about the inter-Korean issue, and focused on enhancing dialogue in order to achieve unification:

"The Government has already made public a three-stage unification formula for building a single national community. It calls first for reconciliation and cooperation between the South and the North, next for forming a Korean commonwealth and lastly for completing a single unified nation-state." (Ministry of Unification, 1994).

Information about North Korea became more publicly available. South Korea even offered food aid to North Korea when the regime was experiencing an economic crisis. A change in North Korean policy made the public perceived North Korea with a dual image: both as an enemy and a partner (S. Snyder, 2004, p. 24). However, North Korea's decision to pull out of the NPT in 1993 led to the first nuclear crisis, and quashed the hope for any improvement in inter-Korean dialogue.

4.2 South Korea and the North Korea's First Nuclear Crisis in 1994

The first nuclear crisis took place during the presidency of Kim Young Sam. In his first year in office, Kim declared that a foreign policy priority would be to hold a summit meeting between the two Koreas. When the crisis developed, Kim Young Sam announced a plan to solve the problem peacefully. As a means of reconciliation, South Korea repatriated North Korean a war prisoner who was captured during the Korean War and imprisoned in South Korea until 1988 (C. S. Lee & Sohn, 1994, p. 8). However, it turns out that Kim's policy was rather inconsistent. When North Korea threatened to withdraw from the NPT, President Kim adopted a hardline policy, declaring that South Korea would not cooperate with "a partner with nuclear weapons" (Yoon, 1996, p. 513). Throughout his term, Kim Young Sam was faced with domestic criticism regarding the effectiveness of his engagement policy with North Korea. After the engagement policy proclamation at his inauguration in February 1993, North Korea declared its withdrawal from the NPT the following month. This was followed by a series of provocations, some of which were directed at South Korea. The Nodong missile launch and a threat to turn South Korea into "a sea of flames" prompted the South Korean government to resort to a hardline policy in early 1996 (Y. Kim, 2011, pp. 177-178). Some South Korean hardliners also expressed their concern that the US-ROK alliance could be damaged by North Korea's nuclear proliferation⁸ (Hayes & Noerper, 1997, pp. 243-244;

_

⁸ Inconsistency in the policy towards North Korea during the time of the crisis made South Korea unable to match pace with the policy of the United States. Kim Young Sam shifted his policy between inter-Korean ties and the US-ROK alliance depending on the situation on the Korean peninsula and

Michishita, 2010, pp. 113-114). In an effort to maintain the US-ROK alliance, the Kim government shifted its focus from engagement with North Korea to maintaining the relationship with the United States. Such a move resulted not only in the delay of an inter-Korean summit, but also led to South Korea increasing its defence budget under the alliance system. Buszynski (2004, p. 98) writes that South Korea's defence spending was for more than self-defence, as it was also aimed at weakening US-DPRK ties which North Korea had been exploiting since the Agreed Framework.

Inter-Korean relations experienced the most progress after Kim Dae Jung took office as the President of the Republic of Korea, in December 1997. Kim Dae Jung believed in the ability of the engagement policy to settle the conflict. He came to this conclusion after witnessing how engagement led to US-Soviet detente in the 1970s (D. J. Kim, 1994, pp. 36-38). On his inauguration day, Kim Dae Jung committed to improving inter-Korean relations through dialogue and cooperation. He said,

"Inter-Korean relations must be developed on the basis of reconciliation and cooperation as well as the settlement of peace. The Cold-War style of South-North relations for over a half century, during which members of separated families could not confirm whether their own parents and brothers and sisters are alive or dead, let alone carry on dialogue and exchanges, must be liquidated as soon as possible." (Ministry of Unification, 1998)

Kim's engagement policy was called the Sunshine Policy. The basic principles of the Sunshine Policy are outlined as follow:

"The Sunshine Policy has an exact background with the story by Aesop. It is about a man wearing a heavy coat. Nothing can persuade him to take off his coat, not until the weather turns bright with sunshine. Kim Dae Jung has a vision that engagement not pressure is the only possible way to make North Korea

public sentiment in South Korea. In addition, Kim Young Sam even showed his resentment towards President Clinton after learning that the United States still had a plan to use force if the negotiation failed to change North Korea.

78

open up. This is the way to achieve the normalisation of relations, the denuclearisation of North Korea" (Interview Prof. I. H. Park, 2011)

The Sunshine Policy brought about the first visit of a South Korean president to North Korea and an Inter-Korean summit was organised in June 2000. At the close of the summit, on June 15, a Joint Declaration was signed, regarding Korean reunification and a commitment to future meetings. Nevertheless, the Sunshine Policy was criticised by some as being ineffective in solving the North Korea's nuclear problem. Concerning this, Prof. I. H. Park (2011) says,

"Kim Dae Jung's administration believes that social exchanges can lessen tension. Humanitarian assistance and economic assistance to North Korea are implemented in hope that they could support with the security and political issues of the two Koreas. This is a nature problem of the policy. South Korea expects a spillover effect from social, economic, and cultural exchanges to political and security issues but this has never happened. North Korea separates the issues of culture and economic from security and political areas." (Interview Prof. I. H. Park, 2011)

Moon (2000, p. 7) argues that the Sunshine Policy was not a weak policy. In fact, it was "an extremely offensive and proactive policy", because an engagement reflected South Korea's initiative in leading the inter-Korean relationship, rather than waiting to respond to the policy of North Korea.

North Korea was suspicious of the Sunshine Policy which it saw as an attempt at ending the Juche policy of North Korea (KCNA, 1999d). North Korea responded with a two-sided policy. The two-sided policy proposed extending the relationship with South Korea for economic aid but without decreasing its own defence (Y. J. Han, Kim, & Kim, 2005). Testimony from North Korean defectors reveals that the Sunshine Policy actually strengthened the North Korean army. Upon receiving South Korea's food aid, the military was the first recipient, and some military officers exploited this benefit by selling the food on the civilian market to line their own pockets (Y. J. Han et al., 2005). The Sunshine Policy reduced tensions on the peninsula, as engagement drew North Korea closer to the

international community, and both Koreas could establish bilateral diplomatic and economic ties. Nonetheless, by the time the Agreed Framework was implemented, the international community was faced with new provocations from North Korea – this time in the form of a missile test, of the Taepodong-1 weapon, fired over Japan in August 1998. The missile test pushed North Korea and the United States into a missile negotiation, resulting in an agreement at the Berlin Conference in 1999. North Korea agreed to suspend future missile tests, and in return the United States would lift sanctions on North Korea.

The US-North Korean relationship deteriorated, however, after the US accused North Korea of violating the Agreed Framework by secretly developing a uranium enrichment programme. The Bush administration criticised the Clinton policy of engagement, claiming that the Framework was akin to rewarding an untrustworthy regime, and a policy of containment and isolation was later chosen as an alternative for dealing with North Korea (Samore, 2003, p. 11). The US-ROK alliance was faced with similar difficulties stemming from differences in policy implementation, as had been experienced under the Kim Young Sam government. Kim Dae Jung's government clearly showed its support for the policy of engagement, which was contrary to what the Bush administration favoured. Different perceptions in security interests made the United States and South Korea adopt different policies towards North Korea. While the United States viewed North Korea's weapons of mass destruction (WMD) and missiles as major concerns, South Korea focused instead on avoiding a military conflict or any other pressures that could cause North Korea to collapse (S. Sheen, 2003, p. 11). The situation deteriorated further after President Bush applied the term 'Axis of Evil' to North Korea in his State of the Union Address of 29 January 2002.

4.3 South Korea and the North Korea's Second Nuclear Crisis in 2002

After its December 2002 general election, South Korea experienced a shift in domestic politics. The election resulted in the victory of Roh Moo Hyun, who favoured continuing the Sunshine Policy. Roh's victory implied not only the continuation of South Korea's engagement with North Korea, but also reflected the public's support for the policy of engagement over confrontation in inter-Korean relations. Roh stressed during the presidential campaign the need to end the old politics and aim for social reform, and also the need to

maintain the Sunshine Policy (H. Y. Lee, 2003, p. 74). In his inaugural address of 25 February 2003, Roh emphasised solving the North Korean nuclear issue through engagement in order to persuade North Korea to give up its nuclear weapons (BBC News, 2003). Roh's North Korea policy, called 'the Peace and Prosperity Policy', was an expanded version of the Sunshine Policy. The policy was aimed at enhancing the relationship between the two Koreas through dialogue and exchanges. C. N. Kim (2005, p. 13) writes that Roh's policy was structured on the viewpoint that engagement would change North Korea's behaviour because this method would make the regime's survival dependent on assistance from the outside world.

Although Roh's compromise and engagement led to various negotiations between the two Koreas, the progress in inter-Korean relations was minimal. A little progress was made by Roh's government, as a result of how the US-ROK alliance partnership has played on the policy of South Korea:

"Before the years 2007 and 2008, the Roh Moo Hyun's administration did their best to improve North-South Korean relations but it was meaningless. It was only possible after the Bush administration decided to use positive approach to North Korea in 2007. South Korea is not an independent actor. South Korea's policy cannot do anything. The policy can have an impact on North Korea only when it is combined with international factors such as the US-ROK relations." (Interview Prof. I. H. Park, 2011)

North Korea continued with its provocations, and South Korea was unable to make any progress in resolving the North Korean nuclear issue. At a first summit meeting between Roh Moo Hyun and George W. Bush, on 14 May 2003, the issue of North Korea was discussed. The two allies agreed to solve the North Korean nuclear problem through peaceful means, but also through multilateral diplomacy in order to achieve "a complete, verifiable and irreversible elimination of North Korea's nuclear weapons programme" (The Chosun Ilbo, 2003). The most striking event that had a negative impact on the improvement of inter-Korean relations during the Roh administration came when a delegation was not sent to Pyongyang on the occasion of the 10th memorial service for the late Kim Il Sung, in 2004. South Korea's failure to recognise the memorial service was a cause for the confrontations

between the two Koreas along the border, and eventually led to the cancellation of the 15th South-North ministerial talks scheduled for 3 August 2004. North-South contacts still existed at the public level, such as at the Kaesong industrial complex,⁹ but dialogue at the governmental level ceased (Y. H. Koh, 2005, p. 211). In the Roh period, efforts to resolve the North Korean nuclear issue appeared in the form of the Six-Party Talks in involving all major regional players. The biggest achievement of Roh's government was the organisation of the second Inter-Korean summit, in October 2007. The summit reaffirmed the commitment to implement the June 15th Joint Declaration of the first summit to enhance inter-Korean exchanges and achieve the unification of Korea. The second summit saw huge successes in the economic arena, when the two Koreas agreed to enhance economic cooperation, such as establishing the economic zone and the maritime peace zone (North Korean Economy Watch, 2007). An economic issue discussed at the summit was Roh's vision to turn the Korean peninsula into the Northeast Asian regional hub (Cumings, 2008).

Lee Myung Bak took office in February 2008, after a campaign characterised by the slogan, 'Anything but the Policy of Roh'. Roh's policy was considered a failure in moving the inter-Korean relations forwards and in easing tensions on the peninsula, because North Korea continued to pose a threat – for example, the October 2006 nuclear test. North Korea's provocations and the continuation of nuclear development raised criticisms in South Korean society about the effectiveness of the engagement policy in solving the problem. Lee's adoption of a hardline policy in an opposite direction from the two previous governments reflected inconsistency in the South Korean government's policy, which was affected by the policy choices of different parties in the democratic system (Interview Prof. Y. Y. Kim, 2011; Interview Prof. Paik, 2011).

Lee Myung Bak denounced Roh's engagement policy and undertook instead a 'Mutual Benefit and Common Prosperity' policy. The policy was structured on two main objectives. The first is to eliminate North Korea's fear of "unification by absorption" and assisting North

-

⁹ The Kaesong industrial complex was established in 1998. It is a project initiated by the Hyundai Group, a private South Korean company, who planned to enhance economic cooperation between the two Koreas. The Kaesong industrial complex is situated in North Korean territory across the DMZ. Throughout, the South Korean government has also been involved in supporting and financing the complex (BBC News, 2013; Manyin & Nanto, 2011, pp. 5-6).

Korea developing its economy (Jae Jean Suh, 2009, p. 9). The Lee's administration also came up with the "Vision 3,000 thru Denuclearisation and Openness" or the "Vision 3000" policy. This policy aimed at increasing North Korea's annual per capita income to US\$3,000 within ten years by helping North Korea enhance economic cooperation with the international community (Korean Institute for National Unification, p 30-31). The implementation of "Vision 3000" policy did not help with the atmosphere in the peninsula. It created more tensions between the two Koreas. North Korea was suspicious of South Korea's intention to destabilise the regime. It was written on the KCNA that Lee Myung Bak's Vision 3000 policy "proves that he is desperately pursuing the confrontation between the north and the south in ideology" (KCNA, 2008d). Apart from official speech denouncing the Lee's government, North Korea also took another step in protesting South Korea by expelling the South Korean officials out of the Kaesong industrial complex. President Lee also set a condition to North Korea that the Vision 3000 policy will be implemented only if there is a progress in the denuclearisation and human rights issues (Moon, 2008b, p. 127). It is assumed that President Lee structured the North Korean policy in a pragmatic way, with rewards granted only when there were signs of improvement in the nuclear deals or North Korean compliance with agreements. In other words, it refers to the expectation the Lee's administration had on North Korea to comply with their policy and the nature of this policy makes it rather difficult to change North Korea's behaviour (Interview Prof. D. H. Han, 2011). Lee's engagement with conditions lowered the chance for contact between the two Koreas.

The situation on the peninsula worsened with the sinking of the South Korea navy ship, Cheonan, in the Yellow Sea near North Korea, on 26 March 2010. North Korea's denial of any involvement in the Cheonan sinking raised a lot of debate in South Korea on the cause of incident (Interview Prof. Koo, 2011). Despite North Korea's insistence that it was not involved, many South Koreans, including President Lee, linked North Korea to the incident. Although he did not directly mention North Korea, President Lee said at a meeting of South Korea's top military commanders that the Cheonan incident was not an accident, and called for strengthening defences against North Korea (Chick, 2010). Nonetheless, a direct confrontation with North Korea was not a viable choice for Lee's government. In an effort to reduce tensions, President Lee shifted the nation's focus from confrontation to the issue of

unification. He came up with a national security strategy containing details on a long-term strategy for achieving Korean reunification (The Korea Times, 2010).

Inter-Korean relations suffered another blow after the shelling of Yeonpyeong island, in November 2010. South Korea condemned the attack. President Lee gave an address to the nation, vowing to take a strong stance and measures against North Korea because South Korea's "endurance and tolerance will spawn nothing but more serious provocations" (Yonhap News Agency, 2010). As a result of this declaration, South Korea appeared to move towards improving its relations with the United States and also its neighbour, Japan, after a period of strained relations under previous administrations. South Korea continued with its plan to conduct military exercises in the Yellow Sea with the United States, five days after the shelling of Yeonpyeong, despite North Korea's warnings of a counterattack in retaliation to a violation of its maritime borders (BBC News, 2010). The US-South Korea relationship was strengthened after the Cheonan incident, and both leaders issued a joint statement condemning North Korea's aggression. The remarks made by both Lee Myung Bak and Barack Obama were issued during the G-20 Summit in Toronto, Canada, on 26-27 June, 2010. The statement contained a number of significant details: as well as criticising North Korea, it stressed the importance of the US-ROK alliance and relationship, and the need for the United States and South Korea to closely cooperate in ending North Korea's and Iran's nuclear weapons proliferation (The White House, 2010a).

These two major provocations from North Korea, in the same year, raised some speculation about the intentions of North Korea behind the incidents. In the end, it was believed that both internal and external factors influence the decision of North Korea. Prof. D. H. Han (2011) elaborates thus:

"We should think about two things – international factor and domestic factor. When you think about year 2010, a political succession process was going on. Kim Jong Un had a clear reason to draw some kind of military support for his side. Also, military side has a clear reason to support Kim Jong Un if he is an heir apparent to Kim Jong II. Probably, some kind of domestic politics caused this kind of assertive North Korea's behaviour." (Interview Prof. D. H. Han, 2011)

Prof. S. H. Sheen (2011) assessed North Korea's intention behind both the Cheonan and Yeonpyeong shelling incidents:

"[The incidents] may have multiple purposes. One thing could be domestic purpose. Kim Jong Un or his military officers try to stable themselves during the succession. In other word, they try to consolidate their own domestic power bases. At the same time they may want to send a signal to South Korea. They may simply want to punish South Korea for their hardline policy or they want to get attention from the United States against the Obama's ignorant policy or whatever. It could be any of them. It could be a combination of all of those." (Interview Prof. S. H. Sheen, 2011)

Park Geun Hye's general election victory over Moon Jae In, on 19 December 2012, had implications for both inter-Korean relations and the US-ROK alliance. First, despite some slight differences in policy between that of Park Geun Hye and Lee Myung Bak, it appears that South Korea's policy towards North Korea would continue to be a hardline policy. Park is a member of the conservative party – Saenuri – whose North Korean policy is structured on a proactive promotion of mutual development as a means for achieving unification, and also the denuclearisation of the Korean peninsula (The Saenuri Party, 2012b). Since the presidential campaign, Park has structured her policy on "Trustpolitik". The policy emphasised the need to strengthen the deterrence strategy while simultaneously enhancing trust between the two Koreas through dialogue and exchange (The Saenuri Party, 2012a). Park's victory came in part because she is the daughter of Park Chung Hee, South Korea's most popular former president. The older generations came out and voted for her, believing that South Korea would be more economically stable and safer from the threat of North Korea (The Chosun Ilbo, 2012a). Secondly, the US-ROK alliance was assured under the leadership of Park Geun Hye. This is because Park's North Korean policy highlights the significance of the deterrence strategy and also solving the problem of nuclear proliferation on the Korean peninsula. South Korea requires the extended deterrence from the United States particularly during times when regional stability is threatened by the problem of a nuclear North Korea (Interview Prof. M. Kwon, 2011). President Parks's policy also implies South Korea's non-tolerance of any future North Korean provocations, and that North

Korea's nuclear development would be a core concern/focus of the Park government. Revere (2012) observed that the US-ROK alliance was strengthened because President Park seemed to share the same perception with the United States, believing that it was highly unlikely that North Korea would give up its nuclear weapons.

President Park's visit to the United States, on 7 May 2013, resulted in a joint declaration in commemoration of the 60th anniversary of the US-ROK alliance. Both sides were determined to bring North Korea out of isolation through the trust-building process proposed by President Park, while at the same time reaffirmed the use of extended deterrence to counter threats by North Korea (The White House, 2013a). South Korea was on the same page as the United States on the issue of North Korea. Cha (2013) writes that,

"Park used very strong words about her willingness to retaliate at any North Korean provocation and President Obama clearly stated that the United States would not give concessions to climb down from any crises created by North Korea. Obama also said that he supported Park's "trustpolitik" which implied that should North Korean behavior moderate in the coming months, then the United States would support any humanitarian assistance that Park might want to provide to the people of North Korea."

4.4 Inter-Korean Relations, Unification, and North Korea's Threat Perception

A part of North Korea's threat perception was developed from the prospect of inter-Korean relations. During the Cold War, it appeared that both North Korea and South Korea sought expansion with the reunification objective. International Relations theorists talk about states existing in a system of anarchy and a sense of competitiveness between states during the search for security. Anarchy makes states rely on one's own effort, and influences states to "identify negatively with each other's security so that ego's gain is seen as alter's loss" (Wendt, 1992, p. 400). As such, states will attempt to manipulate others to satisfy their self-interests. It is possible, in a system of anarchy, that competition occurs between states. According to Mercer (1995, p. 242), states compete for various reasons, ranging from the

basic requirement of security to a need to protect one's own identity, the latter of which influences the belief of states that one is superior to the other.

The two Koreas have been in competition since the division of Korea by the Allied Powers in 1945. The competition began in an effort to be the first to achieve independence and the reunification of Korea. Such strong ambitions led Kim II Sung to start the Korean War in 1950. Apart from the failure of this first attempt to achieve reunification, the differences in the two Koreas' political systems are also factors that have affected the threat perception of leaders in both Koreas. While Kim II Sung started the military build-up, and unification was identified as the primary goal of North Korea, South Korea also appeared aggressive particularly through the 'marching north' policy of Syngman Rhee. Because the two states share the same goals, such as unification and also to blunt the power of the other, a sense of competition has been created, and the need of both sides to secure their security is increased. According on Waltz (1997, p. 915), there are two choices states have, when they are weak: either they seek to ally themselves with stronger states, or they can seek to balance against threats on their own by increasing their own strengths. This could explain why both Koreas sought dialogue with their partners in the early stages of the Cold War, with North Korea attempting to maintain good relationships with both China and the Soviet Union, and South Korea with the United States. When South Korea started to sense a weakening of the US-ROK alliance, as a result of the Nixon Doctrine, it turned towards nuclear weapons development as a means to secure itself. This decision pushed the two Koreas into an arms race, as per the upward spiral model. The spiral was apparent after the Cold War ended and North Korea was left with no allies to whom they could turn. This made North Korea depend more on a policy of self-reliance and nuclear weapons as a means to secure itself and to survive in the new post-Cold War international environment. A South Korean expert explains the impact of inter-Korean relations and the change in the international political structure from the Cold War to the post-Cold War on North Korea's threat perception as,

"I suppose that from the end of the Korean War until the end of 1980s North Korea had thought that South Korea could soon collapse. South Korea's economic development was rather unstable and South Korea's politics was very torment at that time. North Korea did not feel threatened by South Korea until the end of the 1980s. But in the 1990s, North Korea experienced economic

collapse and has also observed South Korea's rising. Probably from that time on, the North Korean regime feels threatened by South Korea's development." (Interview Anonymous, 2011)

During the 1960s, North Korea suffered considerable economic hardship, which made its economic development fall behind that of South Korea. In an anarchic international system, characterised by uncertainty and distrust, states will seek to ensure their own security. Waltz (1979, p. 105) assumes the correlation between anarchy and behaviour of states, saying, "In any self-help system, units worry about their survival, and the worry conditions their behaviour". In inter-Korean relations, North Korea's level of aggression would increase if it experienced a decrease in its regime's perceived level of security. One source of insecurity comes from the military capacity of South Korea, and this factor has a rather significant impact on North Korea because,

"South Korea is modernising its forces every year. In North Korea's view, it is threatening. But the conservative South Koreans do not think that the South Korean army is getting more modernising and intimidating the North Korea's army. They do not see their own threat to the others. It is obvious that North Korea has a far smaller in defence budget. Weapons are so outdated. They are no match to South Korea." (Interview Prof. M. Kwon, 2011)

During the Cold War, terrorist activities were seen as a means to compensate for a sense of insecurity. According to Guillaume (2004, pp. 538-540), terrorism is a method that can provide an outcome far greater than its cost. D. Kang (1998, pp. 257-259) says that terrorism is "politically motivated". North Korea uses terrorist tactics to weaken the South Korean government and create political turmoil in South Korea, in the hope that North Korea can take a chance to intervene and achieve reunification on in its own terms (D. Kang, 1998, pp. 257-259; J. Kim, 2008). North Korea's terrorist activities mostly target South Korea, from which we can conclude that they are driven by the regime's need to compete and address the balance of power with South Korea.

During the Cold War, both North and South Korea had a clear objective of achieving reunification under their own terms. However, as a result of internal changes the two Koreas

have experienced since the division, and also the changing international environment since the fall of the Communist bloc, the reunification issue has begun to lose its significance. Prof. Paik (2011) elaborates on the change of the unification policy of North Korea as affected by the internal and external factors:

"North Korea proposed for the unification formula of federation. In the 1960s and 1970s, North Korea regarded the federal formula a stepping stone to the ultimate unitary state. But in the 1980s, the declaration on the federal formula does not reflect what they had in mind like what many people think North Korea seriously intend to communise South Korea. For the first time, North Korea acknowledged that they have to co-exist with South Korea within the federal state formula. In the early 1990s, Kim Il Sung ratified the federal state formula to accommodate the change in international environment. Kim Il Sung says very clearly that in an effort to unify the country, one side should not devolve or swallow the other side. North Korea is very defensive from recognising the much more powerful South Korea. They are so afraid of being absorbed in the new international environment." (Interview Prof. Paik, 2011)

Reunification may appear in the policy rhetoric of both Koreas, but the more it is mentioned, the greater the feeling of insecurity and distrust that develops between the two countries. Since North Korea has fallen behind South Korea in every aspect, the regime has perceived the reunification issue as a threat to the regime's survival, despite the South Korean government's assurances of peaceful co-existence and non-absorption objectives. Prof. M. Kwon (2011) elaborates on North Korea's sense of insecurity, as affected by the issue of reunification:

"North Korea is afraid of unification. North Korea is worse off than South Korea and does not have ability to unify South Korea even with nuclear weapons. North Korea is fully aware that unification could happen under the South Korean term. To them, it means the possibility of being absorbed by South Korea." (Interview Prof. M. Kwon, 2011)

North Korea has been talking about reunification since the early years of Kim Il Sung's time in power, and continues under the rule of Kim Jong Un. It is possible to conclude from their speeches that North Korea's ultimate objective is the unification of Korea, and the development of nuclear weapons is meant to serve this goal. In fact, North Korea developed nuclear weapons in order to balance South Korea when it did not have sufficient weapons to protect itself. Nuclear weapons, however, are no longer solely connected to the unification issue – especially when North and South Korea have clear differences in unification policies. Prof. Y. Y. Kim (2011) explains that,

"North Korea wants to have a unified nation in a socialist style but in my opinion, they have already given up. What North Korea wants is one nation, one people but two systems and two governments. Economically and socially North Korea wants to integrate with South Korea but militarily and diplomatically they want to separate it. The reason why North Korea chooses a nuclear path is to keep their nation, their socialistic system" (Interview Prof. Y. Y. Kim, 2011)

In particular, unification would appear to be a difficult task to accomplish, given the widening gap between the two Koreas. Prof. Koo (2011) says,

"I sense that the North Korean people always say about a unified Korea. However, I do not think that the North Korean ruling class want to have a unified Korea because there are big gaps between the two Koreas in terms of economy, defence, and political culture. So, I said to you it is just a discourse, not a policy. It is just a saying." (Interview Prof. Koo, 2011)

Considering both the regime's internal weaknesses and external environment, North Korea does not have the capacity to unify the Korean peninsula, even with its possession of nuclear weapons. As such, the possession of nuclear weapons has a defensive purpose unrelated to unification. On the assumptions about the relationship between nuclear weapons and reunification, Prof. S. H. Sheen (2011) concludes,

"A sense of insecurity drives North Korea to develop nuclear weapons. Nuclear weapons are rational choices of North Korea to guarantee its security and

survival against all the domestic and foreign pressures. Nuclear weapons are not for unification." (Interview Prof. S. H. Sheen, 2011)

Nuclear weapons have become political tools for securing North Korea's sovereignty and security during a time of economic decline, and insecurity resulting from the fall of the Communist bloc. When faced with a more skilful South Korean diplomacy, the possession of nuclear weapons has helped North Korea buy itself time to adapt to the new environment (Bracken, 1993, p. 86). Dr. E. C. Lim (2011) elaborates on the connection between nuclear weapons, international environment, and North Korea's domestic politics thus:

"After the collapse of the former socialist countries in Eastern Europe, North Korea strongly feels that it needs nuclear weapons to defend their country. Nuclear weapons are to secure their regime survival. They do not have any intention to attack South Korea or Japan. Many defectors I interviewed unanimously said that Kim Jong Il does not want to lose what he has right now. He is enjoying too many things." (Interview Dr. E. C. Lim, 2011)

Kim Jong Un has mentioned reunification in almost every speech since he came to power in December 2011. In his first speech to the public, on 15 April 2012, celebrating the 100th birthday of Kim Il Sung, Kim Jong Un emphasised reunification as one objective of the nation. Kim Jong Un said,

"It is truly heartbreaking that our brethren are suffering the agony of division for nearly 70 years after living for a long time as a single nation in the same land. Our party and the Republic's government will go hand in hand with anyone who truly desires the country's reunification and the nation's peaceful prosperity and will make responsible and patient efforts to accomplish the historic cause of the fatherland's reunification. Comrades. Our cause is just and the might of Korea that is united with truth is infinite." (North Korea Tech, 2012)

Reunification was emphasised again in the 2013 New Year Address. The Address highlighted reunification as "the greatest national task that brooks no further delay", and this objective could only be achieved by Korean efforts through an improvement in inter-Korean relations

(KCNA, 2013b). It is clear that the factor of the economic gap and differences in the reunification model make it highly unlikely that reunification will occur. Nonetheless, reunification continues to be addressed by the leaders of both Koreas. The mention by the North Korean leader, in particular, cast some doubts on the objective of the regime, the connection between the nuclear weapons and reunification, and the attitude of the regime on reunification. Concerning this, Prof. Paik (2011) explains that,

"The nuclear issue has nothing to do with the unification issue. The hardliners in South Korea are trying to attack North Korea by linking nuclear issue to the unification. In order to reach for unification, the fundamental thing you need is the economy. North Korean economy fails so whatever they say in public or in ceremonial way about unification, actually that does not mean much. In fact, they were defensive as time passes. They recognise the much more powerful South Korea and are afraid of being absorbed in the new international environment. It is a big gap between what they say and what they really thought of in their mind. In politics, you have to take care of some good course, for instances, unify the country, unify the nation. That is something you cannot give up as a politician of divided nation. You have to emphasise the need to unify the country in public speech in all ceremonial things but in your heart and in your mind, you are more realistic." (Interview Prof. Paik, 2011)

North Korea's fear of absorption by South Korea was lessened after the implementation of Kim Dae Jung's Sunshine Policy. Kim structured his Sunshine Policy in an opposite direction from his predecessors, as he called for equal status with North Korea, and rejected the principle of "unification by absorption" (Cumings, 2008).

When Lee Myung Bak took office, the policy of South Korea experienced a dramatic change. Lee refused to continue with the engagement policy, and accused the Sunshine Policy of causing South Korea a decade of economic and diplomatic harm. North Korea had been careful not to criticise Lee Myung Bak during his presidential campaign, but it once again became hostile after the establishment of Lee's administration. According to S. Snyder (2009c, p. 93), what prompted North Korea to begin its criticism of the Lee government are:

"The first was a statement by Unification Minister Kim Ha Joong in his first policy briefing to the president on March 26 in which he said that "the speed and scope of as well as ways to push for any development in inter-Korean relations will be decided according to progress in the North Korean nuclear issue. The second was a statement by the Chairman of the Joint Chiefs of Staff Kim Tae Young at a March parliamentary hearing in which he stated that there were contingency plans for responding to a possible nuclear weapons strike by North Korea"

The threat from South Korea came not only from Lee Myung Bak's foreign policy, but also Lee's remarks on the reunification issue. For example, the KCNA of 17 August 2012 heavily criticised Lee Myung Bak's remarks of 15 August 2012, which celebrated the anniversary of Korea's liberation. The address confirmed South Korea's commitment to the "unification through absorption" principle, with Lee's remarks singling out regime change for North Korea before illustrating that a "unified ROK is a greater springboard for a greater ROK" (KCNA, 2012a). Lee's North Korean policy is criticised as,

"Lee Myung Bak makes speeches on unification on various occasions. He does not only criticise North Korea's choice of going nuclear, but he has a vision that North Korea will soon collapse. He anticipates for a collapse so unification could occur. His speeches and vision threaten North Korea but it seems he does not realise that" (Interview Prof. M. Kwon, 2011)

North Korea's level of threat perception continued to rise after Park Geun Hye came to power. Since her presidential campaign, Park has appeared to structure foreign policy in a similar way to that of Lee Myung Bak. North Korea, therefore, started to attack Park Geun Hye's policy when she was still a candidate. On 5 November 2012, Park Geun Hye announced an outline for North Korean policy. Among issues of Inter-Korean cooperation is a unification plan. The draft unification plan outlined the merger of the two Koreas with a small economic field, before expanding it to include the political field until the two Koreas were completely merged (Yonhap News Agency, 2012d). However, North Korea heavily criticised Park's unification proposal. North Korea voiced its dissatisfaction with Park's unification plan, viewing it as "the commitment to confrontation of the social systems at any

cost" (KCNA, 2012c). It appears that North Korea was cautious with the newly-elected South Korean president. In an unusual manner, North Korea released a short report requesting a clarification from Park Guen Hye on the North Korean policy, and whether it would be structured in line with that of the two former liberal presidents – Kim Dae Jung and Roh Moo Hyun (Yonhap News Agency, 2012c). Tensions on the Korean peninsula continued to escalate after South Korea's 2012 presidential election. The nuclear test conducted on 12 February 2013 drew international condemnation of the regime, and the test marked the third attempt by North Korea to become a nuclear state (Chance & Kim, 2013).

It is possible to explain North Korea's decision to conduct the February 2013 nuclear test through the lens of inter-Korean relations. President Park Guen Hye presented her vision and policy on North Korea on her first day in office, 25 February 2013:

"North Korea's recent nuclear test is a challenge to the survival and future of the Korean people and there should be no mistaking that the biggest victim will be none other than North Korea itself. ... I urge North Korea to abandon its nuclear ambitions without delay and embark on the path to peace and shared development ... Through a trust-building process on the Korean peninsula I intend to lay the groundwork for an era of harmonious unification... I will move forward step-by-step on the basis of credible deterrence to build trust between the South and the North ... It is my hope that North Korea will abide by international norms and make the right choice so that the trust-building process on the Korean Peninsula can move forward." (Office of the President, 2013a)

President Park's inaugural speech implied two things – the regime's instability from focusing too heavily on the nuclear programme, instead of people's well-being and the peaceful reunification of Korea. In the following month, President Park Guen Hye attended the Joint Commissioning Ceremony of the Military Academies and Reserve Officers' Training Corps Units, at which she gave a speech that mentioned the instability of the North Korean regime. President Park said,

"North Korea is pushing ahead with nuclear tests and the development of longrange missiles while threatening to nullify the Armistice Agreement ... No matter how strong a country's weapons are and no matter how advanced nuclear weapons become, resorting to weapons alone cannot protect a nation. If a country just focuses on building up military might while its people are starving, it will not be able to avoid the fate of self-destruction in the end." (Office of the President, 2013b)

President Park's speech was not well-received by North Korea. The North Korean Ministry of People's Armed Forces issued a statement, without mentioning the name of President Park, saying,

"How can they say that a country cannot be protected with weapons alone, and that countries that only concentrate on nuclear weapons and other weaponry are sure to collapse, when they themselves are calling on their own people to put all their energy into maintaining battle readiness?" (The Hankyoreh, 2013)

North Korea had a negative attitude to the newly-elected South Korean government. North Korea criticised South Korea for pursuing an "anti-DPRK confrontation policy" in the hopes of bringing down North Korea through political and military means, before calling for the nullification of the North-South non-aggression agreements (KCNA, 2013d). North Korea declared "a state of war" with South Korea on 20 March 2013 (S. Snyder, 2013).

North Korea has become a hot issue on the agenda of every South Korean government, and most presidential candidates have used North Korean policy to gain public support. The majority of the electorate voted for Park Guen Hye not only because she is the daughter of Park Chung Hee, but also because a conservative candidate is usually more favourable when the threat from North Korea appears high (Choe, 2012). Besides, the South Korean media leaked a secret agreement between Roh Moo Hyu and Kim Jong II on the recognition of the Northern Limit Line (NLL). It was reported that Roh Moo Hyun said, during a closed door meeting between the two leaders at the second inter-Korean summit in 2007, that he did not recognise the NLL as it was "unilaterally drawn by the United States" (Korea JoongAng Daily, 2012). North Korea, aware of South Korea's internal politics, made several comments to the candidate, and the comments reflected North Korea's attempt to intervene in the South

Korean election, to influence the outcome in its favour. Prof. D. H. Han (2011) explained how South Korean domestic politics could have an impact on North Korea:

"I think North Korea's leader always thinks about domestic politics of South Korea. They are well aware of the political event in South Korea. They are very smart. Probably they try to use the situations or changes in South Korea to support their goals or use them in their favour." (Interview Prof. D. H. Han, 2011)

During the South Korean general election, North Korea clearly showed its preference for Moon Jae In. This was expressed in a statement made by the Secretariat of the Committee for the Peaceful Reunification of Korea, on 3 November 2012, that pronounced that inter-Korean relations would suffer if Park Guen Hye came to power (KCNA, 2012f).

An examination of North-South relations reveals that North Korea usually reacts to changes in South Korean politics. In other words, it refers to the way North Korea responds to changes in South Korea's government and policy:

"Either North Korea be cooperative or confrontational depends on the policy of South Korea. When the previous government adopted the Sunshine policy, North Korea responded quite positively. When the Lee Myung Bak's government adopted more aggressive approach, then North Korea became very negative or belligerence." (Interview Prof. S. H. Sheen, 2011)

Provocations are the tactics North Korea uses in order to test the policy of the newly-elected government. Cha and Snyder (2012) said on PBS that,

"The North Koreans tend to do provocations within 16 to 18 weeks of every South Korean election, going back to 1992 ... North Korea always tests the new South Korean leader in various forms. They want to figure out what the parameters are of the relationship and, frankly, what they think they can get out of it." (Sagalyn, 2012).

Thus, it is possible to say that the regime's decision to launch the three-stage rocket on 12 December 2012, a week before South Korea's presidential election, was to test not only the new South Korean president's North Korean policy, but also that of the new US president – also re-elected that year. Nonetheless, there are some statements that claim confrontation between the two Koreas along the NLL could easily happen, for two reasons: the US-ROK joint military exercise along the maritime border, and North Korea's long-time non-recognition of the NLL. North Korea considers the NLL "a ghost line", and calls the US-ROK joint military exercises along the NLL an act of war, because the allies are invading North Korea (Yonhap News Agency, 2012b). The shelling of the Yeonpyeong island is connected to the dispute over the NLL, which Prof. Y. Y. Kim (2011) explains thus:

"The United States and South Korea conducted the joint military exercise near the NLL. North Korea has never accepted this line and tried to have a discussion with South Korea about this line. The military manoeuvre is very near to North Korea's territory and North Korea sees it as a kind of provocation. North Korea warns South Korea many times before not to have such a military drill but we neglect these kinds of warnings. At the end, they attack. I think if this kind of manoeuvre does not exist, they did not attack us. We provoke this kind of action." (Interview Prof. Y. Y. Kim, 2011)

Tensions along the inter-Korean border escalated before the 2012 South Korean election. The number of North Korean intrusions across the NLL is recorded as increasing from October 2012, and it is believed to be a message sent from North Korea to the incoming South Korean government that it did not accept the NLL (S. G. Park & Kim, 2012).

4.5 Conclusion

Inter-Korean relations are among the factors that affect North Korea's threat perception. By reviewing the relationship between North and South Korea, one can see that the two states compete against each other. Based on the assumptions of realists and constructivists, states compete against each other in order to secure their interests. Factors that influence states to

compete can be from security, as argued by realists; or they can include the issue of states' identities, as proposed by constructivists. Competition between the two Koreas is a product of the forced division following the end of World War II. During the Cold War, both sides competed on the assumption that one was better than the other. This perception shaped the leaders' views of policies and system of the other, with the ultimate objective of unification.

The fall of communism highlighted an unequal balance of power between North Korea and South Korea. While South Korea enjoyed the extended deterrence provided by the United States, North Korea lost the support of its allies. Although the internal factor of economic decline influenced the policy of North Korea, the external factor of a changing international political structure pushed and reshaped the interests of North Korea. North Korea's interest shifted from reunification to regime survival. Methods to secure its national interest were also adjusted accordingly. The nuclear crisis, early in the post-Cold War period, showed that North Korea wished to use nuclear deterrence as a strategy to secure its interests. Deterrence was applied based on the concept of retaliatory punishment in order to dissuade South Korea from carrying out any aggressive action (Knopf, 2009, p. 41). North Korea did not want to use nuclear weapons to attack, but rather to deter South Korea from any possible military attack, or to force reunification when it became clear that South Korea was ready and had the ability to achieve it, rather than North Korea. In addition, nuclear deterrence is used to balance against the extended deterrence provided by the US-ROK bilateral relationship.

It is clear that North Korea has used provocations against South Korea since the Cold War period. Provocations are military tactics that serve the regime's interests. During the Cold War, provocations were for reunification purposes. In the post-Cold War world, provocations are conducted for various purposes, as seen in North Korea's desire to test the policy of the South Korean government, or for national security – as seen in the shelling of Yeonpyeong island. The fear and uncertainty North Korea has of the policy of South Korea is clearly illustrated in the new post-Cold War environment. When states feel uncertain, they seek ways to overcome the sense of uncertainty, and usually do so at the expense of other states (Booth & Wheeler, 2008, p. 41). In the case of North Korea, the use of provocations also served the security demand of North Korea and to prepare itself to respond to South Korea's policy.

Competition among states, state interests, and conflict between states are three important factors for analysing international relations, according to both realists and constructivists. As

such, competition and conflict between two Koreas are entirely possible, and even normal and predictable. An increase in the defence capabilities of one can cause an increase in the other's capabilities, based on the assumption that states are uncertain about the level of power they should have in order to secure themselves. Nonetheless, defence build-up in South Korea can have more effect on North Korea than vice versa, because of the power imbalance between the two countries. Nuclear weapons can be used for deterrence purposes, but they still cannot help North Korea reach overall parity with South Korea. Economic weakness and the lack of allies show the inferiority of North Korea compared to South Korea. Competition between North Korea and South Korea in this sense does not only refer to security competition. It can also refer to political, social, and cultural competition, as the two Koreas are considered a divided nation (Interview Dr. E. C. Lim, 2011; Interview Prof. Paik, 2011). States can be made to feel insecure by another state, but it seems the relationship North Korea has with South Korea plays a bigger role in North Korean levels of insecurity, more so than other states. The history of forced division has created a sense of competition and comparison between the two sides. The feeling of threat that has been created by this competition influences North Korean behaviour. Such behaviour appears in the form of nuclear proliferation and provocations as methods by which North Korea chooses to secure itself.

Chapter 5: North Korea and the Great Powers in Northeast Asia

North Korea's geopolitical position is a factor that draws major powers to become involved on the Korean peninsula. During the Cold War, the major powers gained access to the region through two alliance structures: either the Communist camp, or the Western bloc under the US-led system. However, the fall of Communism caused some changes in the structure, as certain players started to lose the power and influence they once enjoyed in the region. Russia, for example, turned out to be a bystander in Korean affairs in the 1990s and struggled to regain its power through involvement in multilateral channels. However, the United States continued to secure its role and power in the region by maintaining good relations with its allies, thereby strengthening its alliance system. The interaction between major powers has an effect on the policy each has in the area, which was particularly evident during North Korea's first nuclear crisis.

The crisis created a sense of insecurity in all regional powers, and led them to attempt to solve the problem through diplomatic means and, at the same time, deter any potential future attack from North Korea. The problem of nuclear proliferation on the Korean peninsula is an undesirable factor to all players, but it seems that different actors have different ways of addressing the problem. First of all, China sees the problem as a threat to regional peace and stability, but China prefers to take the role of a mediator in bringing about the meeting of the key players, rather than taking the lead in coercing North Korea to disarm. Russia, on the other hand, having limited influence in matters relating to North Korea, uses instead power at the international level by joining the international community in pressuring North Korea to comply with the obligations of its international agreements, such as the NPT. Lastly, Japan, who has shown concern over North Korea's nuclear proliferation, uses the alliance partnership it has with the United States to support its allies in solving the problem through diplomatic means. It is clear that cooperation, either within the US alliance system or between the major powers, has not lead to the organisation of a collective defence mechanism in the region, nor has it helped to contain North Korea. Instead, interaction between the major powers has elevated North Korea's level of threat perception, because their efforts to solve the problem – which have led to a loose form of cooperation, including the Korean Peninsula

Energy Development Organization (KEDO) or the Six-Party Talks – creates what the regime perceives as international pressure against the regime. North Korea already felt isolated in the new post-Cold War structure, with the loss of its past alliances with Communist allies. The negative perception it has towards the environment in which it is living is worsened, especially when there is a failure by others to meet the terms of the 1994 agreement.

The new post-Cold War political structure caused major changes in the relationships among the regional powers. Firstly, it allowed for the rise of China as a new superpower. This means its involvement in helping to solve the nuclear problem is significant to regional stability, and became a major factor in a new US-Sino relationship. Secondly, there was the establishment of diplomatic ties between China and non-Communist countries, including South Korea, as a means to support China's rise in the post-Cold War period. Thirdly, relations between Russia and South Korea were normalised in 1991, as a result of the Soviet Union's adoption of a pro-Western policy. These three changes indicated a shift in the region's balance of power, and changed North Korea's views of the great powers.

Although there is an effort from each player to engage bilaterally with North Korea, progress in these relations is faced with obstacles presented by North Korea's provocative behaviour. The lack of progress in bilateral relations, to some extent, has an impact on the multilateral dialogue between North Korea and other players, with hostility and distrust shared by all parties. Interaction between the major powers in the region produces structural change to the post-Cold War system, and North Korea considers this interaction among major powers to be a threat to its regime security.

This chapter will examine the three major powers of Northeast Asia – China, Japan, and Russia – in order to ascertain how the bilateral relations these three countries have with North Korea affect North Korea's decision-making on the deterrence level. In addition, the interaction between these three powers in the post-Cold War period will also be explored, in order to find answers to why, despite sharing the same objective of denuclearisation, they fail to deter or reassure North Korea.

5.1 China and the North Korea's First Nuclear Crisis

North Korea's declaration to withdraw from the NPT in 1993 led to the first nuclear crisis on the peninsula, and raised security concerns for China. China considers the nuclear proliferation on the Korean peninsula a threat because there is a possibility that it could cause a nuclear arms race in the region (Bluth, 2011, p. 186). Instability caused by the North Korean nuclear development shifted China's resources and focus from domestic development to solving the problem (B. Yang, 2006, p. 23). What China advocates is peace and stability on the Korean peninsula, because this kind of environment would support China's concentration on its domestic development and establishment of good relations with its neighbouring countries. Nonetheless, the challenge facing China is the search for a suitable solution to persuade North Korea to abandon its nuclear weapons programme without risking the collapse of the North Korean regime. According to Chinese Foreign Minister Qian Qichen, China's Korean policy during the crisis was based on the desire to have a nuclearfree Korean peninsula, but also that any excessive international pressure to force North Korea to abandon its nuclear ambitions was not acceptable (Agence France Presse, 1991). The survival of North Korea is a priority to China, because a collapse would bring about a flow of North Korean refugees into China. Instability in the region caused by the downfall of North Korea would open an opportunity for South Korea, the United States, or even Japan to intervene in the Korean peninsula.

China's foreign policy towards its neighbouring countries is generally structured on five core issues: "no instability, no collapse, no nukes, no refugees or defectors, and no conflict escalation" (S. Kim, 2003, p. 12). Denuclearisation may be a main objective of China's; but the role of China in solving the problem of nuclear proliferation is minimal, because China does not involve itself in negotiations with North Korea, nor does it take part in the KEDO. According to S. Kim (2004, pp. 83-84), China wanted to distance itself from the problem "from fear that it might get burned if something went wrong".

There are three things that affected China's North Korean policy during the first nuclear crisis: China's perception of North Korea's nuclear capabilities, US power and involvement in Northeast Asia, and the Sino-ROK relationship. First of all, nuclear proliferation on the

Korean peninsula was considered a threat to regional peace and stability, but China did not consider this problem a priority. Aside from the Taiwan Issue, China concentrates on achieving "comprehensive national strength" to serve China's new role as a great power, and also on compensating for the deficits arising from its internal problems, such as economic and religious conflicts (S. Kim & Lee, 2002, p. 110). The main focus of China was economic development in the pursuit of the modernisation of China (Quansheng Zhao, 1997, p. 115). In addition, Moore (2008b, p. 20) observes China's non-involvement in the 1994 nuclear crisis as coming from China's opinion that it does not have the responsibility to solve the United States problems.

Secondly, it concerns the power of the United States in the region, which is highlighted by the fall of Communism. It seems that China is uncomfortable with the power of the United States, and the possible use of military means to solve the North Korean problem. This affects China's policy addressing the North Korean nuclear issue. During the crisis, China strongly opposed the use of sanctions against North Korea by the UN Security Council (Kristof, 1993). The Clinton administration's adoption of an engagement policy was partly a response to China's and Japan's opposition to sanctions (Hughes, 1996, p. 85). China wants stability on the Korean peninsula. China does not want to do much about North Korea's nuclear problem, because China wants to keep North Korea as a buffer state against the competition it has with the United States (Interview Prof. Chung, 2011).

Thirdly, China's policy towards the two Koreas experienced some changes following the normalisation of relations between China and South Korea, in August 1992. Normalisation benefited China's economy. China's decision to normalise relations with South Korea, and China's view of the regional structure of the post-Cold War are described thus:

"China's rapprochement with South Korea has helped confirm its role as an Asian superpower. As one of few countries to have diplomatic ties with both Seoul and Pyongyang, Beijing can substantially enhance its clout on the issue of the divided Korean peninsula. China regards post-Cold War Asia as the site of a three-way power game between itself, the U.S. and Japan, say observers. Some see Beijing's new diplomatic ties with Seoul as an effort to keep the growing influence of Japan in check." (Mitsumori, 1992).

China's normalisation with South Korea, nonetheless, created strain in the relationship between China and North Korea. Although China took it rather slow to normalise relations with South Korea, because it wanted North Korea to save face, the final result was considered a loss for North Korea (Interview Prof. Chung, 2011). The normalisation of relations between China and South Korea meant China saw North Korea through a broader strategic perspective rather than adhering to the Sino-DPRK alliance commitment (H. Lee, 2014, p. 102). In an effort to assure North Korea of the bilateral relations, China sent an envoy to Pyongyang to meet with Kim Il Sung. Kim Il Sung's response was rather cool and distant, and some diplomatic customs were skipped when hosting the Chinese envoy (Funabashi, 2007, pp. 265-266). The strained relationship between North Korea and China made it difficult for China to exercise its influence to stop North Korea from withdrawing from the NPT. North Korea commented on the normalisation of China and South Korea by calling it the "unfaithful actions of some traitors of the revolution" (Chandra, 1993). This is because North Korea viewed normalisation as Chinese recognition of the existence of two separate Korean states on the peninsula. In other words, China's support for South Korea, highlighted North Korean fears of being absorbed into the South Korean system during a time when North Korea was weaker than South Korea (Interview Dr. E. C. Lim, 2011). China's new move created a sense of insecurity in North Korea, which is observed as:

"North Korea has a strong desire to keep the nation and its own system. In the post-Cold War period, the Eastern European countries collapse. Nothing remains. The Soviet Union is changed. China is also changed. North Korea and Cuba are two countries which still have socialistic system. Cuba is very different from that of North Korea. North Korea is the only one country insists its own socialistic system. If you read the works of Kim Il Sung and Kim Jong Il, you can find only one thing. Socialistic system is the most reasonable and the best system. This is why we want to keep the socialistic as our own style. What can they do to keep the nation as socialistic system? They must have to strengthen the military and develop nuclear weapons." (Interview Prof. Y. Y. Kim, 2011)

North Korea's relationship with China worsened after North Korea's decision to go nuclear in 1993. Premier Li Peng explained, in February 1993, China's decision to downgrade its ties

with North Korea by not providing the regime with any advanced military supplies out of its security concern (D. S. Kim, 1994, pp. 40-41). Nonetheless, China realised the need to maintain ties with both Koreas. China tried to maintain the relationship with North Korea for the reason that it views North Korea as a buffer against the United States and wants to assure that, should unification occurr, North Korea would not fall under American or South Korean dominance (Buszynski, 2013a, pp. 27-28). Since a priority of China is to maintain stability on the Korean peninsula, China does not agree with the use of sanctions or any military means to pressure North Korea. S. Kim and Lee (2002, p. 118) write that, "with Chinese help, Pyongyang demonstrated to the entire world that sanctions would be accepted as a declaration of war and, as such, a no-win proposition for all parties concerned".

The relationship may experience some strain, but both China and North Korea are fully aware of the need to maintain a certain level of relations. This is due to the economic benefit both sides expect from the other. The late-1990s was also the period during which North Korea began its economic reform policy. The economic reforms started in 1995, but were accelerated in 1998 after the end of the three-year mourning period following the death of Kim Il Sung. Kim Jong Il called for the economic reform plan under the slogan "the Chollima speed, forced march speed¹⁰", requesting that the North Korean people double their efforts to advance the country's economic development (KCNA, 1998b). The focus on the economy made North Korea lean towards China for advice. This led to frequent visits by Kim Jong Il to China between 2000-2006, in order to observe and study China's economic model. Apart from food aid, North Korea wanted China to assist it with the development of the special economic zone called "Rajin-Sonbong Economic and Trade Zone¹¹," now called "Rason". On the other hand, China also wanted to connect the two provinces in the northeastern part, Jilin and Liaoning, with Rason to gain access to the ports on the East Sea, in order to boost its

-

¹⁰ Kim Jong II adopted Kim II Sung's "the Chollima Movement" idea, launched in 1956. Both Kim Jong II and Kim II Sung's Chollima slogans are similar in that they planned for a rapid development of the economy through workers' hard work and dedication (H. S. Lee, 2001, p. 27). Kim Jong II's adoption of Chollima formed part of his policy objective to achieve the "kangsong taeguk" (K. Oh & Hassig, 2002, p. 87).

¹¹ The Rason was launched under the initiative of Kim Il Sung, in 1991. Rason is situated in the eastern part of North Korea, close to its borders with China and Russia. Kim Il Sung planned Rason to be a special economic zone to attract foreign investment and foreign capital. However, the project did not progress as much as planned, due to a lack of funding support from the North Korean government and poor infrastructure, which make Rason unattractive to foreign investors.

exports (North Korean Economy Watch, 2014; Stangarone & Hamisevicz, 2011, p. 185). Prof. Paik (2011) explains the China-North Korea economic collaboration in post-Cold War period as:

"Both China and North Korea are in cooperation on economic zone. North Korea needs Chinese help for economic development and at the same time China needs Rason. China wants to connect Rason with Changjitu economic zone. For North Korea, they want to introduce market element. They know that they have to feed their own people. They cannot stick to socialist economy all through the years in the future." (Interview Prof. Paik, 2011)

5.2 China and the North Korea's Second Nuclear Crisis

China's role in and policy towards North Korea's nuclear problem during the second nuclear crisis was different from that during the first. China took instead the role of a facilitator who actively pushed forward a multilateral initiative as a means to solving the nuclear issue. China's change was affected by the change in the US administration, the election of George W. Bush, and the adoption of a hardline policy towards North Korea. The agreement China had with the United States to fight the war on terror allowed the United States an opportunity to push China for a more active role in responding to the threat of North Korea (S. Snyder, 2003, p. 95). What China sought was the denuclearisation of North Korea through peaceful means, as outlined in the three principles¹² adopted by President Hu Jintao (Funabashi, 2007, p. 276). China was concerned for the stability of the Korean peninsula, particularly after the US adopted a hardline policy and the US invasion of Iraq in 2003. The hardliners in the Bush administration were also inclined to believe that regime change in North Korea would solve the overall problem of North Korea. The US preference for regime change worried China that it might bring about a collapse of North Korea, leading to an increase of US power on the Korean peninsula (Quansheng Zhao, 2006, p. 58). The replacement of North Korea with a

-

¹² China realised during the crisis that peace and security can only be secured through negotiation. The three principles were outlined as a strategy for how to resolve the conflict, and they are: "the denuclearisation of the Korean peninsula, the maintenance of peace and stability in Northeast Asia, and the settlement through dialogue and negotiation."

new government set up by South Korean and US forces is a worst case scenario for China, because it invites US forces to advance closer to China (Interview Prof. Chung, 2011). When the crisis erupted, China tried to assure North Korea of the ties the two countries had long had, in order to build up North Korea's confidence (Deng, 2008, p. 206). This was due to a fear of abrupt change that could trigger instability on the Korean peninsula. As such, China decided to take the lead in setting up the proposal for the multilateral dialogue. China adjusted the model several times upon North Korea's request. In the end, the Six-Party Talks were created at the height of the crisis, as a means to correct the failure of the previous forum. It was also a method of reconciliation when the United States refused to have a bilateral dialogue with North Korea, as it did during the first nuclear crisis.

The Six-Party Talks raised China's international profile. China cooperated with the United States on an equal basis, and the two countries sought advice from each other in promoting the forum. The United States always turned to China for help whenever the multilateral talks became stalled after North Korea declared that they would not return to the talks. However, China was reluctant to fully comply with the US request to force North Korea back to the negotiation table after North Korea declared that it would not attend any future talks in February 2005 (Ang, 2005). This is because China was well aware that its leverage over North Korea had been weakened (Yonhap News Agency, 2005). The situation worsened with the missile and nuclear tests in 2006 and 2009. The 2009 missile and nuclear tests, in particular, changed China's attitude towards North Korea. China fully realised that North Korea wanted to become a nuclear-armed state, rather than just attempting to increase its leverage against the United States during the negotiations through the development of nuclear weapons (Nanto & Manyin, 2010, p. 11). China's response to both the 2006 and 2009 missile and nuclear tests was the same: after both incidents, it supported the UN Security Council (UNSC) resolutions. Despite China's support for the UNSC resolutions, China emphasised the need to achieve the denuclearisation of the Korean peninsula through diplomatic means. The Ministry of Foreign Affairs, PRC said,

"The Security Council action is not all about sanctions, and political and diplomatic means is the only way to resolve the relevant issues on the Korean Peninsula ... China wishes to reiterate that to bring about denuclearization of the Korean Peninsula, oppose nuclear proliferation and safeguard peace and stability

on the Peninsula and in Northeast Asia is in the common interest of all parties." (Ministry of Foreign Affairs of China, 2009)

According to B. Glaser (2009, p. 3), China's support for the UNSC resolution 1874, on 12 June 2009, was not to punish, but rather to send a message to North Korea that it should reverse its decision and come back to the negotiation table. North Korea reacted negatively to the UNSC resolution. North Korea considered UNSC resolution 1874 to be a "US-led offensive of international pressure", against which the reprocessing of the newly-extracted plutonium and uranium enrichment were countermeasures (KCNA, 2009b). Prof. Y. Y. Kim (2011) elaborates on China's reluctance to exercise pressure on North Korea:

"So far, China is not satisfied with North Korea's nuclear and missile tests. But China wants to have an economically good relationship with North Korea. That is why there were meetings between the two leaders in the past. They do not mix the inner politics. They separate the economy and the political issues." (Interview Prof. Y. Y. Kim, 2011)

The Bush administration's hardline policy supported the rise of China as a new superpower in the new post-Cold War structure. China cooperated with South Korea and the United States in order to solve the crisis. The triangular relationship had implications for the progress and success of the multilateral forum of the Six-Party Talks, and played a role in the power of deterrence against North Korea. What worries China in the wake of the North Korea's nuclear crisis is the possibility of an arms race in the region, which could lead to the pursuit of nuclear weapons as a deterrent against North Korea by both South Korea and Japan. China and South Korea seem to share a similar perception of North Korea's nuclear proliferation as a threat to regional peace and stability, but there are many factors that block progress in bilateral cooperation between the two countries. These factors include South Korea's concern over the rise of China as an economic and military power, China's repatriation of North Korean defectors, and the dispute over the air defence zone between the two. Prof. M. Kwon (2011) explains the relationship between China and South Korea and how this relationship has an effect on the US-ROK alliance:

"South Korea is threatened by the goal of China. Up to one century ago, South Korea was in a sense subject to China. During the Korean War, China joined the North Korean army and also China is not democratic today. So people are worried about China's goals and that South Korea might be in danger by China. In that sense, the alliance with the United States will work to protect South Korea. Many Koreans still believe that South Korea does not have a peaceful environment today which requires the ROK-US alliance." (Interview Prof. M. Kwon, 2011)

North Korea's provocations of 2010, however, brought about Chinese dissatisfaction with the US-ROK alliance. Regarding the Cheonan warship incident, China appeared to side with North Korea. China called for restraint from all parties. China did not consider it unusual that as an incident such as Cheonan would occur given that "an exchange of fire has long occurred between the two, sporadically and accidentally, during the six decades" (S. Lee, 2010a). In addition, there was not enough evidence to prove that North Korea was behind the sinking of the Cheonan (The Chosun Ilbo, 2010a). China did not seem to change its position towards North Korea when they shelled Yeonpyeong Island, in November 2010. China was rather cautious in its response to the Yeonpyeong Island attack. China did not condemn North Korea for the attack, but did express concern for the dispute between the two Koreas. What concerned China most continued to be regional stability, and that the situation might escalate if tensions increased ("Foreign Ministry Spokesperson Hong Lei's Regular Press Conference on November 25, 2010," 2010). North Korea's provocations in 2010 led to an attempt by the allies to strengthen their deterrence against North Korea by conducting the joint military exercises. The United States and South Korea usually conducted joint military exercises as a means to strengthen their defence system against any possible future threats from North Korea. China expressed its opposition to the US-ROK joint military exercises, both in July and November 2010, because the activities would escalate tensions in the region ("Foreign Ministry Spokesperson Qin Gang's Remarks on the Announcement of the ROK-US Joint Military Exercises," 2010; Xinhua News Agency, 2010). Prof. Paik (2011) observed the situation on the Korean peninsula and commented that,

"The United States and China have structured their relations in a confrontation way. One is a rising power and the other is declining hegemony. They are just

confronting each other, sometimes with military power and sometimes with economic power. The Cheonan incident is one of the cases. The way China and the United States deal with the situation of Cheonan is they both try to intervene in order to make a new international order in East Asia much more favourable for themselves and for their own interest." (Interview Prof. Paik, 2011)

According to Song (2011, p. 1137), China would choose to punish North Korea only if its provocations had a high chance of tempting the United States into responding militarily. To some extent, China's position created a strain in its relationships with other regional powers, because others expect China to exercise leverage over North Korea, or take a strong position in response to provocations. China was aware that North Korea had a high level of insecurity, and this made China unwilling to take concrete action against North Korea (Kleine-Ahlbrandt, 2011).

It is important to note that the conflicts China has with South Korea and the United States do not create a positive environment conducive to resolving North Korea's nuclear escalation, considering that trust and full cooperation are required from all parties in order to implement the Six-Party Talks. S. Snyder (2009b, p. 167) writes that the greater the differences among China, the United States, and South Korea, the greater the opportunity for North Korea to exploit the situation. Besides, China's position is believed to have an effect on North Korea's actions and the ability of other states to use a deterrence strategy against North Korea. China's protection of North Korea by disagreeing to the use of international pressure, and China's attempts to maintain ties with North Korea, reduce the ability of other countries to use deterrence against North Korea (Kleine-Ahlbrandt, 2011). China is faced with the dilemma of how to play the role of a responsible regional power in resolving the North Korean problem without damaging its ties with North Korea, while also meeting the demands of other regional players for concrete action on North Korea.

5.3 Japan and the North Korea's First Nuclear Crisis

At the time of the nuclear crisis, Japan showed strong support for the establishment of multilateral cooperation as a means to achieve peace and stability in the region. Japan views

North Korea as a security threat in terms of its military capability and ability to have an impact on the US-Japan alliance by influencing the policy choices in the problem-solving process (Hughes, 2005, p. 42). The Japanese government preferred diplomacy and negotiation to pressure in dealing with the problem of North Korea. Japan's position affected the decision of the Clinton administration on the use of economic sanctions against North Korea, and so the United States sought UN sanctions (Bowman, Collier, Niksch, & Shinn, 1994). Japan regarded the Agreed Framework to be "a specific framework for the realisation of denuclearisation on the Korean Peninsula through dialogue and consultations" (Ministry of Foreign Affairs of Japan, 1995). Japan's contribution to the KEDO showed Japan's willingness to handle a security responsibility equally with the United States (C. S. E. Kang & Kaseda, 2001, p. 53).

Japan turned out to be a key and enthusiastic player of the KEDO. Japan was willing to host the board meeting of members of the KEDO in May 1997, and it committed to contribute \$1 billion to the KEDO, in order to build two light-water nuclear reactors in North Korea. Japan's enthusiasm was based on the opinion that the KEDO could lower tensions and eventually persuade North Korea to disarm (Japan Economic Newswire, 1998). However, Japan's commitment to the KEDO stalled after North Korea test-fired its long-range missile, the Taepodong-1, in August 1998. The incident angered Japan. Japan took some actions in protest against North Korea, including the financial suspension of the KEDO, the postponement of the normalisation talks, and the plan to militarily cooperate with South Korea (Hagstrom & Soderberg, 2006, p. 380; C. S. E. Kang & Kaseda, 2001, p. 54).

With US encouragement, Japan later reversed the decision to suspend financial support for the KEDO. However, North Korea's provocations continued to be a major security threat to Japan. The missile test of 1998 pushed Japan closer to the United States and South Korea. Japan's response to North Korea's 1998 missile launch included trilateral cooperation with the United States and South Korea at all levels, and a review of its defence guidelines (Ministry of Foreign Affairs of Japan, 1998a). Japan's decision heightened tensions on the Korean peninsula. North Korea pointed out that Japan's and other countries' actions increased the regime's insecurity. The KCNA of 2 September 1998 says,

"Japan is zealously developing long-distance vehicles and other uptodate weapons and paving the way for overseas aggression, having worked out "guidelines for Japan-U.S. defence cooperation." Many countries around Japan possess or have deployed missiles. Japanese politicians, however, hurl mud only at the DPRK while being unaware of the background. They emanate from the longstanding hostile policy toward the DPRK." (KCNA, 1998a).

In addition to its provocations, the North Korean abduction of Japanese citizens is an issue that greatly affects the process of normalisation of Japan-North Korean relations, and also Japan's contributions to the denuclearisation process. First of all, there is a broad public debate in Japan about the possibility that North Korea is behind the disappearance of certain citizens following an interview a North Korean agent gave to Japanese media, in 1997. Japan's suspicion of North Korea was confirmed by Kim Jong II himself, during the Kim-Koizumi summit meeting in Pyongyang, on 17 September 2002. Kim Jong Il admitted to North Korea's abduction of 11 Japanese citizens during the 1970s and 1980s, saying "I want to apologise and it will never be allowed to happen again" (Gluck, 2002). The Kim-Koizumi summit ended with a promise by the two leaders to resume normalisation talks, but Kim Jong Il's admission to the abductions angered the Japanese public and caused feelings of hostility towards North Korea, which obstructed progress in the normalisation of relations. Pursuant to the Kim-Koizumi summit, the North Korea-Japan normalisation talks at the ambassador level were organised for 29-30 October 2002, in Kuala Lumpur, but the abduction issue became the main agenda of the talks. Japan demanded explanations from North Korea for the deaths of some of the abductees, and a full investigation of the surviving abductees. Japan even set the abduction and security issues as a precondition for moving forward in the normalisation talks (BBC News, 2002b). The abduction issue slowly took control of Japan's North Korea policy, the more Japan was faced with a lack of progress in settling the issue. Japan even placed the abduction issue on the agenda of the Six-Party Talks in an attempt to marshal international pressure against North Korea. Japan's emphasis on the abduction issue resulted in decreased significance of Japan in the multilateral forum, because it distracted from the main agenda of resolving the nuclear problem. Besides, Japan also placed sanctions on North Korea in order to pressure it into resolving the abduction issue. But, again, the sanctions Japan applied limited the role Japan played in the Six-Party Talks, as Japan showed its

reluctance to participate in providing any aid to North Korea, as required by the Six-Party agreement.

5.4 Japan and the North Korea's Second Nuclear Crisis

North Korea's admission, in October 2002, of the existence of a uranium enrichment programme raised Japan's security concerns to an even higher level, considering it a violation of the 1994 Agreed Framework. This issue affects the normalisation of relations between the two countries. During the second summit meeting between Kim Jong II and Koizumi in May 2004, the nuclear issue was also one of the agenda items for discussion. Koizumi reiterated that dismantling North Korea's nuclear program was a precondition for the normalisation of relations between the two (Brooke, 2004). However, Kim Jong II, apart from affirming that the nuclear issue is a bilateral issue between North Korea and the United States, stated,

"I am of the opinion that nuclear development is good for nothing. We really did not want to engage in nuclear development. But the United States' continuation of hostile policies towards us left us no choice." (cited in Funabashi, 2007, p. 52)

The policy Japan adopted during the second nuclear crisis was a combination of both dialogue and the application of pressure, in line with the policy of the Bush administration. Japan's strong support for the Six-Party Talks upset North Korea, who wished to have a bilateral dialogue with the United States (H. N. Kim, 2006b, p. 13). North Korea accused Japan of supporting the US's hostile policy, which is not in compliance with the provisions stated in the Japan-DPRK Pyongyang Declaration¹³ (KCNA, 2003a). Japan usually sided with the United States when it was faced with a security threat from North Korea. The

¹³ The Japan-DPRK Pyongyang Declaration was a result of a summit meeting between Prime Minister Junichiro Koizumi and Kim Jong II in Pyongyang, in September 2002. The two sides were bound under the Declaration to take necessary steps in reaching for a normalisation of relations. The Declaration clearly indicates that, "both sides would not commit conducts threatening the security of the other side". The full text of the Japan-DPRK Pyongyang Declaration can be found on the Ministry of Foreign Affairs, Japan website: http://www.mofa.go.jp/region/asia-paci/n korea/pmv0209/pyongyang.html (Ministry of Foreign Affairs of Japan, 2002).

strengthened US-Japan security alliance heightened North Korea's sense of insecurity. It appeared on the KCNA as,

"Abe, secretary general of the Liberal Democratic Party of Japan, who is ultra right-wing element, let loose sophism in his speech made in Chiba Prefecture on Nov. 2 that the "Japan-U.S. alliance" is aimed to "let the U.S. strike the north Korean missile base when it launches a missile" and protect Japan from it ... There are in Japan huge U.S. forces including many missiles targeted against DPRK. It is a stark reality of Japan that its moves are being stepped up to emerge a military power." (KCNA, 2003c)

However, Japan later exploited the Six-Party Talks for its own benefit in resolving the abduction issue. Not only did Japan raise the abduction issue at the Six-Party Talks forum, Japan also reiterated that "the abduction issue is part of the overall solution of the North Korean issues" (Ministry of Foreign Affairs of Japan, 18 November 2003).

North Korea strongly criticised Japan, saying it was insincere about solving the nuclear problem because the non-related issue of abduction was raised and became the agenda of the Six-Party Talks (KCNA, 2006d). Later, North Korea claimed that Japan was not qualified to be part of the peace talks (Yonhap News Agency, 2007). North Korea's nuclear proliferation and acts of provocation increased Japan's security concerns, which not only inclined Japan towards adopting a hardline policy against North Korea, but also drove Japan closer to the United States in order to increase Japan's security confidence.

In response to the Cheonan sinking, Japan appeared to take sides with South Korea. Japanese Prime Minister Yukio Hotoyama issued a statement supporting the investigation's report by South Korea, which confirmed North Korea was behind the attack (Ito, 2010). North Korea referred to the Japanese condemnation as a joint effort by the US and its allies to pursue a hostile policy against the regime (KCNA, 2010b). The shelling of Yeonpyeong Island alarmed Japan. In response, it sought to counter the threat of North Korea by conducting joint military exercises with the United States, under the codename 'Keen Sword'. North Korea perceived the joint US-Japan military exercises "a serious threat to the DPRK", because military cooperation with the United States is evidence of "the Japanese intention for

reinvasion of Korea" (KCNA, 2010c). Further to that, Japan even emphasised the need to cooperate closely with its neighbour, South Korea, and to strengthen ties with its allies under the Japan-US-ROK alliance (Prime Minister's Office of Japan, 2010). The trilateral partnership was formed in order to enhance cooperation and dialogue between the three countries, particularly in relation to security threats caused by North Korea's provocative and belligerent behaviour (Ministry of Foreign Affairs of Japan, 2010). In addition to strengthening the US-Japan alliance, Japan also made an effort to improve its defence system. A ballistic missile defence system and a defence protection plan for the citizenry were implemented in preparation to counter a military threat from North Korea (Michishita, 2012, pp. 103-110). Japan's countermeasures show a steady progress by Japan in improving its military capability.

The North Korean nuclear crisis brought about trilateral cooperation between Japan, South Korea, and the United States in the form of the Trilateral Coordination and Oversight Group (TCOG). The TCOG, established in 1999, was a result of a Congressional policy review during the Clinton administration, organised by former Secretary of Defence William Perry. The so-called "Perry Report" emphasises the need for support from Japan and South Korea in order to make "the US policy towards North Korea a success" (Dalton & Snyder, 2004). The United States considered it important that its two allies should share the responsibility of the policy, particularly during a time when the United States wanted to put pressure in isolating North Korea after the Kumchangri incident in 1998 (Schoff, 2005, p. 14). However, differences between the parties' respective North Korean policies gradually affected the overall policy strategy of the TCOG. Another of Japan's priorities, aside from the issue of North Korea's nuclear and missile development, was the abduction issue. The TCOG was later dissolved, as it was undermined by the establishment of the Six-Party Talks.

What was left behind was the significance of Japan's cooperation with its neighbours and allies, both in the form of the US-ROK-Japan trilateral cooperation and also Japan-South Korean relations. First of all, the US-ROK-Japan cooperation was created as a result of North Korea's nuclear proliferation and provocative actions, and because the three countries share the same impression of North Korea as a threat to regional peace and stability. The trilateral cooperation between the United States and its Asian allies reflects an attempt to create a deterrence against North Korea, and to persuade North Korea to comply with its international

obligations. The meeting between the defence ministers of the three countries concluded in strengthening future cooperation in deterring North Korea threats from its nuclear weapons programme, missile development, and provocative acts (U.S. Department of Defense, 2013).

Trilateral cooperation affects North Korea's security and threat perception levels. The cooperation led to joint military exercises between the United States, Japan, and South Korea. North Korea perceived the trilateral military cooperation organised by the US as a "strategy for putting the Asia-Pacific under its control and the strategy toward the DPRK by relying on the triangular alliance" (KCNA, 2012g). To counter the external threat, North Korea must strengthen its deterrent in every possible way with the songun (KCNA, 2012g). The trilateral military exercises also supported a bilateral cooperation between Japan and South Korea, whose relations had been constrained by historical issues (K. J. Kwon, 2012). Secondly, Japanese-South Korean cooperation is considered significant to the peace and stability of the region, particularly when both develop the same policy for North Korea. For example, Japan appears to share South Korea's view on engagement. Following a visit to Japan by President Kim Dae Jung in October 1998, both countries agreed that the solution to the problems of North Korea was to use the dialogue to help North Korea achieve reform and openness (Ministry of Foreign Affairs of Japan, 1998b). However, North Korea did not perceive the mention of reform and openness by its neighbours as an attempt to help. According to the KCNA,

"The Japanese reactionaries are talking rubbish about the 'need' for the DPRK to pursue 'reform and liberalization.' ... We have nothing to open and reform ... Hostile forces' attempt to link the DPRK with any 'change' related to 'reform' and 'opening' is prompted by their sinister aim to tarnish the image of the DPRK." (KCNA, 1999b)

In addition, good relations between Japan and South Korea are important for the United States when it pursues a hardline policy towards North Korea, because the support from these two allies would give the United States the opportunity or excuse to increase its defence and deterrence capabilities against North Korea. Japan's wartime legacy has slowly taken control and obstructed progress in bilateral relations. In addition, a territorial dispute relating to the Dokdo/Takeshima islets has also created conflict and a sense of distrust between the two

countries. The current state of Japan-South Korean relations is said to be at an impasse, and President Park Guen Hye has demanded that Japan resolves the issues of comfort women and other war-related compensation before a summit between the two countries can be convened (Nakano, 2013). That Japanese officials and Prime Minister Shinzo Abe visited the Yasukuni Shrine in October and December 2013, has also cooled relations.

5.5 Russia and the North Korea's First Nuclear Crisis

It appears that Russia shares the same view as other regional powers in viewing the nuclear escalation of North Korea as a threat to regional peace and stability. However, Russia's role on the Korean peninsula was marginalised by the adoption of a pro-Western policy in the 1990s, which made Russia focus on improving its relations with South Korea and begin the normalisation of relations in September 1990. Russia continued to emphasise improvement of its relationship with South Korea for economic purposes, which resulted in weakened ties with North Korea (J. Moltz, 2000, p. 197). North Korea, displeased with the relationship Russia had with South Korea, called it "diplomacy purchased by dollars" (S. Kim, 2007, p. 21). In 1996, Russia also called for an end to the 1961 Treaty of Friendship, Cooperation and Mutual Assistance with North Korea. Instead, Russia proposed a new treaty with North Korea, but with an important condition of the removal of a security commitment in a time of conflict (Agence France Presse, 1996).

Post-Cold War Russia did not see the value in maintaining a relationship with North Korea. The policy of reform Russia undertook in the late 1980s influenced the worldview of Russia, showing that it disagreed with North Korea, who kept emphasising the danger of imperialism (Eugene Bazhanov & Bazhanov, 1991, p. 1128). Russia saw North Korea as a backward country who clung to "a counter-productive strategy" of Juche and cannot take a path to reform (Eugene Bazhanov & Bazhanov, 1991, p. 1127; S. Kim, 2007, p. 21). Russia even expected North Korea to collapse, raising fears in North Korea that Russia would side with South Korea and the United States in an attempt to bring down the regime (Evgeny Bazhanov, 2006, p. 220). During the first nuclear crisis, Russia did not engage in the problem-solving process. Russia lost influence on North Korea and power in the region as a result of the deterioration in its relationship with North Korea. In response to North Korean

nuclear proliferation, Russia joined with the international community in using international pressure to coerce North Korea into complying with its international obligations. For example, President Yeltsin reached an agreement with President Bush, during the Russia-US summit meeting in Washington in June 1992, regarding the issue of North Korea. A joint statement was released with significant details, including the need for North Korea to fully comply with the North-South Joint Declaration on the Denuclearisation of the Korean Peninsula¹⁴ and the obligations of the NPT (US-Russian Joint Statement on Korean Nuclear Non-Proliferation, 1992).

Russia also joined the international community in putting pressure on North Korea regarding its nuclear weapons programme. On 1 April 1993, Russia released a joint statement with the United States and the United Kingdom demanding that North Korea reverse its decision on withdrawing from the NPT, and to comply with its obligations under the Safeguards Agreement (BBC News, 1993a). In October 1993, Russia sided with Japan in pressuring North Korea to rejoin the NPT (Agence France Presse, 1993a). North Korea strongly criticised Russia for taking the side of Japan and having "double standards on nonproliferation", given Russia's lack of involvement on the nuclear proliferation issue in Eastern European countries (BBC News, 1993b). The loss of its alliance with North Korea excluded Russia from engaging in Korean affairs. Funabashi (2007, p. 178) writes that North Korea did not want Russia to be involve in the KEDO, nor participate in the talks as it did not want to have a dialogue with Russia "in the same manner as the United States did".

5.6 Russia and the North Korea's Second Nuclear Crisis

The turning point in Russian foreign policy came with Vladimir Putin's rise to the Presidency, in 2000. At this point, Russia started to consider the Asia-Pacific region as equally important as the West was to Russia. Putin realised the significance of a relationship

¹⁴ The two Koreas signed the Joint Declaration on the Denuclearisation on 31 December 1991. The joint declaration prohibited the two Koreas from developing or possessing nuclear weapons other than for peaceful (*Joint Declaration of South and North Korea on the Denuclearization of the Korean Peninsula*, 1992). The declaration came into effect in February 1992, and was intended to "eliminate the danger of nuclear war and ultimately promote peaceful reunification" (B.-K. Kim, 2001, p. 75).

with North Korea after Russia was excluded from several important negotiations, such as the Agreed Framework and the Four-Party Talks. ¹⁵ A "balanced policy" was adopted in order to balance out the relationship between the two Koreas (C. Moltz, 2003). Putin adjusted the policy out of his concern on an increase of US influence on the Korean peninsula. A decline in power of Russia from a loss in the partnership with North Korea opened an opportunity for the United States to play a more active role in involving in the Korean affairs (Takeda, 2006, p. 192). It turns out to Russia's awareness also that the possibility of the North Korean regime collapse would not happen easily and that a new strategy towards the Korean peninsula needed to be adjusted as a consequence (Bogaturov, 2004, p. 99).

The renormalisation process between North Korea and Russia began in February 2000, with the signing of the Treaty on Friendship, Good Neighbourliness and Cooperation. The new treaty set a new framework for North Korea-Russia relations. The clause on security during a time of crisis was included in the treaty at the request of Russia, who wished to regain its influence over North Korea (Joo, 2003, p. 149). By balancing its policy towards the two Koreas, Russia became more significant, and President Kim Dae Jung requested Russia to take up the role of a mediator in inter-Korean affairs (Alexander Lukin, 2003, pp. 77-79). Security interests and economic benefits were two factors that drove Russia towards supporting a peaceful and stable Korean peninsula. According to Takeda (2006, p. 197), Russia wished to gain access to Asia with the "iron-silk road16" project, which had been outlined during a summit meeting between Putin and Kim Jong II in July 2000. Kim Jong II's meeting with Putin was described as a North Korean play to strategically use Russia to balance out the power of the United States, China, and South Korea (Funabashi, 2007, p. 187). Under the new policy of balancing relations with the two Koreas, Russia's attitudes towards North Korea's leadership, national and political structure, and the issue of missile development began to change. Russia stood up for North Korea during the G-8 summit in

_

¹⁵ The Four-Party Talks were organised between 1997-1999, a result of a proposal made by the United States and South Korea to North Korea in 1996. The four members of the talks were North Korea, South Korea, the United States, and China. The talks were organised to end conflict of the Korean War in the hopes of reaching a new peace agreement (U.S. Department of State, 1997). The Four-Party Talks closed in 1999, after North Korea's refusal to attend.

¹⁶ The iron-silk road project was an economic plan to connect the railways of both Koreas with the Trans-Siberian railway. If successful, the project could boost the economy, as it would reduce the shipment time between South Korea to Europe, and vice versa (Davis, 2003, p. 97).

Okinawa, Japan, in July 2002, by affirming that "North Korea's missile programme has a peaceful purpose", and that it would be worth the effort to negotiate with North Korea (Joo, 2003, p. 155).

The improved Russian-North Korean relations were challenged when the US referred to North Korea as part of an axis of evil, in January 2002, and also the second nuclear crisis in October 2002. Regarding the US identifying North Korea as part of an axis of evil, Russia disagreed, as it viewed the Bush administration as unilateralist, and Bush's statement did not offer any solutions to the non-proliferation issue on the Korean peninsula (Meyer, 2003, pp. 66-67; Takeda, 2006, p. 198). According to S. Kim (2007, p. 24), Putin came up with the so-called "Putin formula" in order to assure North Korea of Russia's commitment to not join in any future hostile acts against North Korea, instigated by either the United States or the international community. Russia's support for North Korea likely came from Russia's need to secure its leverage over North Korea, which was improved after Putin's policy of rapprochement, and also to obtain for Russia economic benefits from the Korean peninsula. North Korea turned to Russia for economic benefits and political purposes, but North Korea was fully aware that Russia did not have much power to help play "the diplomatic game" with the United States and Japan, nor balance them (Joo, 2009, p. 188; The Daily Yomiuri, 2002).

Russia believed Bush's hardline policy towards North Korea created more tension on the peninsula, in a way that would obstruct Russia from realising its economic goals. Although the US's hardline policy brought North Korea closer to Russia, the second nuclear crisis also tested the improved relations of the two countries to some extent. Russia strongly opposed the development of a North Korean nuclear weapons programme, because it would be a threat to regional security. Russia even condemned North Korea when it declared that it would withdraw from the NPT on 10 January 2003. Russia found it difficult to have a nuclear neighbour and, more importantly, to have US forces involved in solving the nuclear issue, because it could create a nuclear catastrophe in the region (Buszynski, 2009, p. 818). However, it turns out that Russia had little leverage over North Korea to make the regime completely dismantle its nuclear weapons programme, and Russia's role was overshadowed by China, who successfully pressured North Korea into attending the Three-Party Talks held in Beijing, on 23-25 April 2003. In an attempt to ensure Russia's involvement in the region

and to block any increase to the United States power, Russia turned to China. The two met and signed the Joint Declaration of the PRC and the Russian Federation on 27 May 2002. The Declaration confirmed an acknowledgement of North Korea's security assurance as an important requirement that could lead to the denuclearisation of the Korean peninsula (World News Connection, 2003).

Negotiation and engagement are Russia's preferred methods for dealing with the problem of North Korea's nuclear proliferation. Russia has been an enthusiastic partner of the Six-Party Talks, and also joined with other regional players in an attempt to increase North Korea's security confidence, with the hope of persuading North Korea to return to the NPT. However, Russia failed in its efforts to improve security in the region when North Korea decided to conduct both missile and nuclear tests in 2006. Putin expressed concern over the missile tests, fearing that tension in the region might escalate if the international community chose to respond to the incident with sanctions (Associated Press World Stream, 2006). Later that year, Russia condemned North Korea for their underground nuclear test on 9 October 2006, believing the test "undermined the non-proliferation process" and set in motion a nuclear arms race in Northeast Asia (RIA Novosti, 2006). The April 2009 missile test and the May 2009 nuclear test elevated Russia's concerns for regional peace and stability. Although Russia did not agree with the missile test, and even condemned North Korea for the nuclear test, Russia opposed the use of sanctions against North Korea by the international community. Russia believed that the use of dialogue and negotiation, not sanctions, were the only productive way to solve the problem of a nuclear standoff, because sanctions would create more tensions in the region and would push North Korea more into isolation (Russian Press Digest, 2009). It turned out that Russia developed a different perception of North Korea from that of the Western countries.

Tensions in the peninsula continued to rise and the situation deteriorated following the sinking of the Cheonan warship and the shelling of Yeonpyeong Island. Regarding the Cheonan sinking, in March 2010, Russia did not share the same conclusion – that the Cheonan was sunk by a North Korean torpedo – as the international community. The investigation report by the Russian navy expert team pointed out that the sinking of the Cheonan was an accident, not caused by a torpedo launched by North (The Hankyoreh, 2010). Russia's declaration came as a surprise to the international community, because

Russia was the first country to claim that North Korea was not behind the incident. As such, Russia's report contradicted the assumptions of other countries, such as the United States, Japan, and South Korea, who strongly believed that North Korea was fully responsible for the sinking of the Cheonan. However, Russia reacted differently in response to the shelling of Yeonpyeong Island. In this case, Russia did not take North Korea's side. On 23 November 2010, Russian Foreign Minister Sergei Lavrov made a statement condemning the attack on Yeonpyeong Island, and urged an end to artillery exchanges between the two Koreas; such actions would only lead to severe damage to the overall situation in the region (Russia & CIS Diplomatic Panorama, 2010). The Yeonpyeong shelling disappointed Russia, because the incident destroyed the hope of resuming the Six-Party Talks. Russia was also concerned that the incident might provoke a military confrontation between the two Koreas. What Russia perceived with regards to the problems of North Korea was that they posed a threat to regional peace and stability, rather than to its own security. Artyom Lukin (2013) writes that missile and nuclear threats posed by North Korea were not considered a serious threat to Russia compared to the missile defence system the United States and its allies had been building to counter North Korea.

5.7 Interaction between the Regional Players and North Korean's Perception

It is clear that all regional players see North Korea's missile and nuclear developments and belligerence as a threat to regional peace and stability. However, different players have different opinions on how serious the problems are to their individual security and to overall regional stability.

First of all, China sees a nuclear-armed North Korea as a problem, but China's policy towards North Korea is likely to be influenced by the reaction of the international community, and also the policy the United States adopts towards North Korea. China does not want to level sanctions or apply pressure on North Korea, because the collapse of North Korea threatens China's security more than having a nuclear-armed neighbour. China is confronted by the expectations of other regional players that it will or should play a larger role in problem-solving – something it believes would negatively affect China's desire to rise as a new superpower. Cha (2012b) says that China faces a tough choice between securing its

national interests and putting pressure on North Korea. Apart from losing an intellectual connection with North Korea, China is also cautious in its role in solving the nuclear problem, because China sees North Korea as a buffer state against the United States. China does not want to use too much pressure on North Korea for fear of a possible regime collapse or regime change (Interview Prof. Chung, 2011). Prof. M. Kwon (2011) explains the relationship between China and North Korea and North Korea's development of nuclear weapons as,

"So far as North Korea determines to develop nuclear weapons and possess nuclear weapons, there is no forceful way to stop it. North Korea has China who is ready to help North Korea. There is no way to force North Korea not to develop its nuclear weapons." (Interview Prof. M. Kwon, 2011)

China wants to maintain ties with North Korea for economic and strategic reasons. At the same time, North Korea wants to have a close relationship with China based on economic benefits and political gain. Concerning this, Prof. Y. Y. Kim (2011) says,

"North Korea has three things in its mind that they want to achieve. They are ideological powerful, military powerful, and economic powerful. These three things are to reach for the goal of kangsong taeguk. North Korea always says that they are already strong in terms of military and ideology. The area they face with the most difficulty is the economy. They think that the sanction from outside must be removed to overcome the economic difficulties. That is the reason why they try to keep a close relationship with China. They want to establish an economically good relationship with China and to balance against the US and South Korea." (Interview Prof. Y. Y. Kim, 2011)

Nonetheless, China's inability to exercise its power to dissuade North Korea to dismantle the nuclear weapons also results from the nature of North Korea itself:

"China has not done much on North Korea's nuclear development. In that sense, China has failed. But North Korea has been an independent country. North Korea has tried desperately to block outside influence into its behaviour." (Interview Anonymous, 2011)

Secondly, it appears that the current, main focus of Japan is to resolve the abduction issue with North Korea. The abduction issue is among the factors that affect the relationship between the two countries. Japan does not have the power or leverage over North Korea even considering the economic assistance the country has extended to North Korea. In addition, North Korea clearly separates the security matter from other matters, and clearly shows that it does not want Japan involved in the nuclear problem-solving process. Kim Jong Il told Koizumi during the first summit between the two leaders in September 2002 that, "The nuclear issue is between the US and North Korea. This is not a subject to discuss with Japan" (Wada, 2012). The fact that Japan is constrained by Article 9 of its Constitution makes Japan leave security matters in the hands of the United States, and this relegates Japan to only a supporting role for the United States. These limits mean that Japan follows US policy on North Korea, in an attempt to maintain US-Japan relations, and to secure a security guarantee from the US. Although the abduction issue is Japan's priority, it is highly unlikely that Japan would push for progress on this issue in a way that could cause irreparable damage to the partnership, because Japan needs the support of the United States in order to pressure North Korea on this issue.

Thirdly, Russia views the problems presented by North Korea through the lens of its national interests. In other words, Russia exploits the situation on the Korean peninsula as an way to regain power in the region. Russia prefers a peaceful and stable Korean peninsula with the existence of North Korea. As a result of its support for diplomacy in dealing with the problems, Russia has been an enthusiastic partner in the Six-Party Talks. Russia's involvement in the Six-Party Talks raised Russia's profile, and the United States turned to Russia during the height of tensions for help in pressuring North Korea to adhere to its international obligations, and also to persuade North Korea to cease its provocations (The White House, 2013b).

The problems posed by North Korea led to not only the strengthening of the alliances between the United States, Japan, and South Korea, but also to the creation of further ties and cooperation between the regional powers. The strength and endurance of the relations

between the US and its regional partners has contributed to an increase in North Korea's level of regime insecurity. The new structure creates an imbalance of power between North Korea on the one hand, and South Korea, Japan, and the United States with its allies on the other. Concerning this, Prof. S. H. Sheen (2011) says,

"One of the main reasons for North Korea's development of missile and nuclear weapons programme is the regime insecurity. Kim Jong Il's insecurity. North Korea feels insecure with the isolation it faces after the Cold War and with increasing gaps between North Korea and South Korea in terms of economy. There are also gaps between North Korea and other powers. North Korea also faces with the US presence in South Korea and Japan. The United States, South Korea, and Japan always say that we have no intention of invading North Korea. But from the North Korea's perspective, it is a kind of serious threat to their national security." (Interview Prof. S. H. Sheen, 2011)

The stronger the ties between the regional powers, the less secure North Korea feels. North Korea considers the US-Japan-South Korean joint military exercises in the West Sea of Japan an attempt to put more pressure on North Korea (KCNA, 2013c). The relationship between the United States and its Asian allies also means the United States is providing security through the form of extended deterrence. The US's extended deterrence emphasised the fear of North Korea of the US, about which Prof. M. Kwon (2011) says,

"North Korea thinks that they have a hostile environment. They think the United States has a hostile policy towards North Korea. To combine the US with South Korea's military is a very strong force. They say if they do not have hostile environment, why would they have to develop nuclear weapons. This is a big incentive for them to develop nuclear weapons obviously." (Interview Prof. M. Kwon, 2011)

China's own perception of a US threat has also risen as a result of the trilateral partnership, and it is believed to be the cause for a closer and stronger partnership between China and North Korea. The relationship between China and North Korea post-Cold War is not as close as it once was during the Cold War period. Nonetheless, China wants to maintain the ties and

wants to keep North Korea as it is, based on the opinion that it still sees North Korea as an asset to the country (Interview Prof. Chung, 2011). The strengthened trilateral partnership would result in a strengthening of military cooperation and also missile defence system between Japan and South Korea, which targets not only North Korea but also China (Moon, 2013).

The post-Cold War changes in the policy and politics of both China and Russia have left North Korea with no allies. China became more pragmatic in its foreign policy making, rather than focusing on maintaining the historical and ideological ties it had with North Korea (B. Glaser & Billingsley, 2012, p. 6). The result of China's post-Cold War policy pushed North Korea further into isolation. A sense of insecurity and the experience of an imbalance of power push North Korea to attempt to balance the threat and protect itself by acquiring nuclear weapons (Interview Prof. Koo, 2011). Prof. I. H. Park (2011) describes the relations between China and North Korea:

"If you take a look at the last two decades, please examine carefully the way how China supports North Korea. China never fully supports North Korea. That means China does not want economic growth and rise of North Korea. But at the same time China never gives up North Korea. It means China is supporting North Korea in one sense. China is supporting North Korea under the conditions that North Korea only survives. China would never let a kind of regime collapse or a sudden type of regime change to happen in North Korea." (Interview Prof. I. H. Park, 2011)

The relationship between North Korea and Russia was also damaged following Russia's decision to normalise relations with South Korea in the early 1990s. North Korea responded to the Soviet Union's decision negatively, and even condemned Gorbachev's visit to South Korea to establish diplomatic ties in June 1990 as "treacherous and unpardonable" (Agence France Presse, 1996). North Korea released a memorandum in 1990 stating that, with the establishment of diplomatic relations between the Soviet Union and South Korea, the USSR-DPRK alliance would effectively discontinue, and North Korea would "have no other choice but to take measures to provide for ourselves some weapons for which we have so far relied on the alliance" (Mack, 1993, p. 342).

Although both China and Russia attempted to restore relations with North Korea for the benefit of their economy and to increase their power and status in the region, their relationships were not restored to the state they had been before. The normalisation of relations with South Korea of both Russia and China, in 1991 and 1992 respectively, caused a reduction in ties North Korea once had with China and Russia (Interview Prof. Chung, 2011). The loss of ties North Korea once has with the two allies, and the change in the international political structure, influence North Korea to take a rather balanced approach towards China and Russia. Concerning this, Prof. Paik (2011) has said,

"Kim Jong II makes a visit to both China and Russia in 2010 and 2011. This is a sort of balancing act on politics of North Korea. They want to cooperate with China economically. They need Chinese help but too much dependence on China is not good. Basically they are independent minded people. They try to play both China and Russia's offer against each other. North Korea tries to get benefits from both countries because China and Russia are very much aware that they need North Korea for their own strategic needs. China and Russia are competitors and North Korea is just using their competitiveness. Their competitions are something they can play off against each other for their own interest." (Interview Prof. Paik, 2011)

China was criticised for manipulating the situation with North Korea for its own end, for example China's reluctance to use its power and influence to pressure North Korea (S. Lee, 2010b). China's response to the North Korea's nuclear development has been described as,

"China's reaction is weak and cautious in dealing with North Korea's nuclear problem. This is because China views North Korea's nuclear development and nuclear test as the matter of North Korea. China cannot and is not in a position to stop the nuclear development of North Korea." (Interview Prof. Y. Y. Kim, 2011).

Both China and Russia share the positive opinion of the use of diplomatic means for solving the North Korean problem, believing that the use of international pressure would risk the loss of trust and influence both countries have on North Korea, and could trigger a war in the region. For not wanting to apply pressure on North Korea, China and Russia have been criticised for being reluctant players in taking the lead on containing North Korea, and their hesitance has resulted in some of North Korea's aggression, because North Korea is well aware that acts of provocation will go unpunished (Cane, 2013). Concerning Russia's involvement in the peace process, Prof. I. H. Park (2011) has commented,

"I am not sure how much serious Russia is about North Korea after the end of Cold War. In particular when Putin became the leader of Russia, Russia is more interested in identifying Russia as a European country. I really do not know how much Russia is interested in North Korea's problem." (Interview Prof. I. H. Park, 2011)

It is important to note that the interaction between the regional powers greatly contributes to North Korea's policy and strategic thinking. North Korea's foreign policy and behaviour are influenced by the interaction of the major powers in the region. Concerning this, Prof. Paik (2011) elaborates,

"The post-Cold War is a new international order. Nobody is clearly in control in international community. The United States, China, Russia, and Japan try to counterweight each other's influence about Korean affairs. If the United States wants to destroy North Korea, then China would come out and defend North Korea. North Korea has an independent army and is independent from Soviet influence and from Chinese influence as well. The interplay of those four respective states gives an opportunity for North Korea to be independent on its own" (Interview Prof. Paik, 2011)

This means that North Korea must search for a way to survive in an environment in which four other players are trying to counter each other. In other words, North Korea must try to survive without falling under another nation's influence (Interview Prof. Paik, 2011). One of the ways to secure the objective of regime survival is with nuclear weapons, which have a low cost but high benefit, and the power of deterrence (Interview Prof. Koo, 2011; Interview Prof. Moon, 2011; Interview Prof. I. H. Park, 2011).

5.8 Conclusion

The issue of North Korea is not about inter-Korean relations alone. It includes also the relationship between North Korea and the regional powers, and the impact of the interaction the regional powers has on North Korea. The fall of Communism created a new international political structure. The structure of the post-Cold War world is no longer shaped by two superpowers confronting each other and representing their own interests. The new system is a rather independent system, in which everyone stands for their own interests. Nonetheless, the interaction between regional powers post-Cold War creates a loose form of confrontation and imbalance of power of the two groups. The first group is the United States and its allies, and this alliance has continued from the Cold-War period. The second group is North Korea and its former-allies, China and Russia. The partnership North Korea has with China and Russia was destroyed and has never been restored to its previous level. China and Russia are interested in the economy and stability of the Korean peninsula, rather than the security of North Korea. China and Russia's ties with North Korea are for economic benefit and political reasons, and the relationship should not be regarded as an alliance. Japan seems to focus on satisfying its domestic interests, while leaving security issues to its alliance with the United States. The structure has become more like an imbalance of power between the two sides.

The structure in Northeast Asia in the post-Cold War world shows not only an imbalance of power, but also an attempt by regional powers to balance against one another. When North Korea uses provocations, their choice in policy to respond also depends on the overall situation. China and Russia lean towards using tough measures on North Korea at the international level, rather than taking their own initiative. Part of this comes from the fact that they both do not have the power to influence North Korea, and another factor is their fear that the situation might become worse. The regional powers try to balance their power against each other by not letting anyone gain more power and influence over the situation. The responses the regional powers had to the Cheonan and Yeonpyeong incidents are among the examples. The Cheonan incident showed the two divided groups between North Korea, China, and Russia on the one hand, and the United States and its allies on the other. On the other hand, the Yeonpyeong incident showed the different approaches China and Russia had towards the incident. They both disagreed with North Korea, and called for restraint and

dialogue to solve the problem. This is because the Yeonpyeong shelling offered a high possibility for destabilising the region, with an exchange of fire between the two Koreas and the strengthening of the US military in the region.

It is clear that regional powers have changed their policies and behaviour in accordance with the changes to the international politics. In the post-Cold War period, China and Russia shifted their focus from political confrontation to reform and their economies. On the other hand, Japan's ties with the United States have been confirmed and strengthened over time with the supporting role Japan plays in the alliance. In light of these changes, North Korea has been left with no allies and is searching to find a way to survive in the new environment. North Korea does not want to change, and is afraid to move itself into a transition period during which external factors could have a greater influence on the regime to change or even collapse, in a similar way to some Eastern European countries. Nuclear weapons are a strategic tool North Korea uses in response to the security challenges it faces. Challenges North Korea faces in the early post-Cold War period include the US nuclear threat, the unequal balance of defence with the United States and its allies, and the loss of its alliances with China and Russia. North Korea acquired nuclear weapons in order to match its security demand and to balance the nuclear threat it faces. They serve the purpose of deterrence, rather than being intended for offensive purposes, like the expansion of North Korean power. The new relationships North Korea has with China and Russia do not help North Korea match the level of external threat it faces, as the relationships are structured on economic rather than security interests. North Korean insecurity is increased by the new political structure, which requires the independence of states. Regional players interact not only to balance each other, but to suit their individual interests. The structure shows the independence of states, and this kind of structure can create a sense of insecurity. As explained by Herz (1962, pp. 39-42) states will never feel secure in a system that allows the strong to destroy the weak. This refers to the need of North Korea to be independent in its search for power and security. The regional powers may acknowledge North Korea as a state based on their preference on the status quo of the Korean peninsula. However, North Korea's security need has not been acknowledged. Regional powers deal with North Korea based on their own interests. The search for security is therefore possible until states find security. The two nuclear crises, one after another, is a reflection of North Korea's insecurity in the new environment.

Chapter 6: North Korea and the United States

The fall of the Soviet Union in the early 1990s gave rise to a new international political structure, as a unipolar system emerged with the United States as the sole remaining superpower. The transitional period created a sense of insecurity and uncertainty among states, which led some to invest more in defence and turn themselves into nuclear states (Millot, Molader, & Wilson, 1993, p. 1). The new structure raised questions regarding the future role of the US, and the durability of its hegemonic power, in the post-Cold War system. According to Cox (1995, pp. 42-44), what the United States faces in the post-Cold War period is a challenge from major regional powers whose interests are opposite to that of the United States. In defining a new strategy in the new era, the Clinton administration conducted the Bottom-Up Review. The Bottom-Up Review pointed out that there are four dangers facing the United States in the post-Cold War world. They are the proliferation of weapons of mass destruction (WMD); aggression by major regional powers; the potential failure of democratic reform in the former Soviet Union and elsewhere; and the potential failure of the US economy (Les Aspin, 1993, pp. 1-107). The Review lists the Korean peninsula as one example of potential dangers given the fact that there is a possibility an arms race could be triggered by North Korea's nuclear ambitions (Les Aspin, 1993, pp. 5-13).

The post-Cold War era also saw increased attention to so-called 'rogue states'. The idea of rogue states is not actually completely new. The term first appeared in the 1970s, and has been used interchangeably with the term 'outlaw state' until acquiring the current meaning of "Third World states who were attempting to acquire weapons of mass destruction" (Miles, 2013, pp. 10-20). During the Clinton administration, foreign policy was structured on a policy of containment focusing specifically on the so-called outlaw states. This was expressed by Anthony Lake, an assistant to the President for National Security Affairs, in a famous article, "Confronting Backlash States". The article talks about need for the United States, as the sole superpower, to take responsibility for dealing with aggressive behaviour of the so-called outlaw states and for transforming them to become part of the international community (Lake, 1994, pp. 45-46). Reducing and limiting the spread of WMD was the focus of the Clinton administration (Walt, 2000, pp. 71-72).

The perception of the rogue states and Clinton's policy of containment were carried over into the George W. Bush administration. The term rogue state appeared more in the wake of the 9/11 incident, referring to states seeking to possess WMD and support international terrorist groups (Tantler, 2003, pp. 16-17). The United States has referred to North Korea as a terrorist state since the Reagan administration, but attempts to contain and counter the challenge posed by rogue states through negotiation appeared during the Clinton administration. Viewing nuclear proliferation as a threat to regional security, the United States under President Clinton made it clear in July 1993 that a nuclear-armed North Korea was unacceptable. US Secretary of Defence William Perry said in May 1994 about the situation of North Korea's nuclear proliferation that, "what the U.S. wants to achieve is a nuclear-free Korean peninsula." (Berry Jr., 1995, p. 41).

After the negotiation and signing of the Agreed Framework in 1994, the Clinton administration continued to face criticism at home that the engagement policy actually encourages North Korea's aggressiveness and brinkmanship. Both the discovery of the secret nuclear underground site at Kumchangri, and the missile test in August 1998, underlined some doubts on the effectiveness of Clinton's policy of engagement. This eventually led to a review of overall US foreign policy towards North Korea by former-Defence Secretary William Perry. The so-called 'Perry Report' highlighted "a two-path approach" of engagement and containment as a strategy for dealing with the problems of North Korea's nuclear weapons and missile programmes (Perry, 1999). Following the Perry Report, the Clinton administration leaned more towards an engagement policy, leading to the visit of Secretary of State Madeleine Albright to Pyongyang, in October 2000 (Miles, 2013, p. 73).

US foreign policy experienced some changes following the election of President Bush, in 2001. The new administration appeared reluctant to continue Clinton's engagement policy. President Bush did not appear to trust North Korea, considering that the regime's secretive nature made it difficult to know whether or not it was adhering to all the terms of its agreements (Sanger, 2001). The attacks on 9/11 were a turning point that completely altered the US view of the world and especially with regard to rogue states. The world was clearly divided between the United States on the one hand, and the terrorists or those supporting the terrorists on the other (The White House, 2001). The United States categorised North Korea as part of an axis of evil, and declared that there would be no dialogue with North Korea unless it changed its behaviour (CNN, 2002). The June 1992 accusation that North Korea was

processing uranium put an end to the 1994 Agreed Framework. At the same time, the United States was disappointed by North Korea's decision to move forwards with its nuclear development programme, which the administration referred to as blackmail.

There arose some expectations to see changes in the US approach towards North Korea after the inauguration of Barack Obama and his incoming administration. When he first took office, President Obama declared a policy of engagement with the rogue states, which gave some hope that relations between the United States and North Korea might improve. However, US policy gradually changed in reaction to a series of North Korean provocations, and a policy of strict engagement became the policy of the administration. The strict engagement strategy is based on a priority of bringing about the total nuclear disarmament of North Korea, in order to halt the spread of nuclear weapons, rather than to improve bilateral relations with North Korea (Chanlett-Avery & Rinehart, 2014, pp. 7-8; Revere, 2013, p. 11).

When it comes to North Korea's nuclear proliferation and military provocations, it is undeniable that the role and policy of the United States are influential in finding a solution. Since the first nuclear crisis in 1994, the United States has pursued both dialogue and sanctions in order to deal with North Korea, but it seems that none of the policies are effective in reducing tensions on the Korean peninsula. While engagement is considered by some a form of appearement that encourages North Korea's provocations and blackmail, the US hardline policy is simultaneously criticised for being a cause of heightened tensions, as conflict and confrontation actually result in increase regime insecurity in North Korea.

The concept of deterrence usually refers to the affect one state's behaviour has on another, which would imply that the policy choices of the United States would inevitably affect the policy choices of North Korea. Since the United States is an important player in the region, this chapter will analyse the US's policy and approach towards North Korea since the start of the first nuclear crisis in 1994. The chapter aims to evaluate how US foreign policy affects North Korea's levels of deterrence, threat perception, and regime insecurity, while also attempting to assess whether US foreign policy towards North Korea is a cause or a solution to the overall security dilemma in the region.

6.1 The First Nuclear Crisis and the US Policy during the Clinton Administration (1993-2001)

By the time President Clinton took office in 1993, the IAEA had accused North Korea of reprocessing plutonium and they were in North Korea to inspect its nuclear facilities. The IAEA report was released in President Clinton's first month in office, and contained some discrepancies with what North Korea had previously declared. Tensions on the peninsula began to rise after North Korea decided to refuse the IAEA inspectors any more access to the two sites at Yongbyon, and declared its intention to withdraw from the NPT. Mazarr (1995, p. 92) writes that the United States structured its post-Cold War policy on non-proliferation in an attempt to end the nuclear challenge of North Korea and extend the scope of the NPT. During the early stages of the crisis, the Clinton administration leaned towards preferring a military option as a means to solve the problem. Nonetheless, negotiation was adopted as a preferred option based on the viewpoint that it could yield a high chance of success and could bring the overall situation under control (Miles, 2013, pp. 49-50). Besides, there was a high probability that the situation could deteriorate into conflict, given North Korea's strong insistence that "sanctions are a declaration of war" (Oberdorfer, 2001, p. 306). In 1992, the Clinton administration announced that it would resume the joint military exercises with South Korea, which were suspended during the George H.W. Bush administration, and also that the US would strengthen its alliance with South Korea. All these US actions threatened North Korea's sense of regime security. A spiral model was created during this period, as a sense of insecurity on both sides was pushed to greater levels because of each other side's desire to increase their level of defence. While North Korea chose to counter the joint military exercises with its declaration to withdraw from the NPT, the United States moved itself closer to South Korea and Japan for consultations on ways to solve the problem (Henriksen, 1996, pp. 30-31).

A series of negotiations led to the signing of the Agreed Framework in 1994, which outlined major details about an exchange between North Korea and the United States as a deal in solving the nuclear issue. Significant points of the deal included a security guarantee in which the United States was asked not to employ any nuclear weapons against North Korea, and an assurance from the United States for future political and economic relations. In return, North

Korea had to agree to freeze its nuclear-related activities and also promise not to construct any new graphite nuclear reactors or reprocessing facilities in the future. The Agreement promised North Korea two light water reactors for alternative energy purposes as a replacement for the graphite moderated reactors. The situation may appear to have stabilised after the Framework was signed, but several problems arose afterwards, which gradually led to criticism of the engagement policy. The first concerned the discovery of an underground nuclear facility in Kumchangri by the US intelligence agencies in the second half of 1996. The report released by the Defence Intelligence Agency in 1997 suggested the possibility that the underground site in Kumchangri could be used for "nuclear production and/or storage", and the site could be "a nuclear weapons-related facility by 2003" (Niksch, 2002, p. 3). The US inspection of the site began following a deal in which the United States promised to supply North Korea with food aid and to assist North Korea grow potatoes in a pilot project (Sanger, 1999). After two visits to the site only to find an empty cave, the inspection team claimed that the site was too small to host nuclear facilities (The Chosun Ilbo, 1999).

The consequences of the Kumchangri suspicion were mistrust between the US and North Korea, and also some criticism at home of US policy towards North Korea. The result of the inspections caused a debate among US intelligence experts on the possibility that North Korea could have removed and hidden some equipment away from the site, given that the inspections had been announced too far in advance. Cordesman (2002, p. 121) writes that, although North Korea was banned from conducting all nuclear activities under the Agreed Framework, the agreement did not mention nuclear programme development at other sites in the country, only at Yongbyon. Not long after the Kumchangri incident, the North Korean decision to test fire a three-stage Taepodong missile in August 1998 presented a new security concern for the United States and its Asian allies. The missile test proved North Korea's missile capability and stage of development, about which the US intelligence agencies had been speculating since 1994 (Niksch, 2002, p. 4). The missile test of August 1998, despite North Korean claims that it was for the purpose of sending a satellite into orbit, greatly alarmed Japan whose security was threatened by a missile crossing over its territory. The United States was anxious about the rapid development of North Korea's missile system, and so it decided to not only offer a new round of dialogue to discuss the issue but also to review its overall policy towards North Korea (Vennet, 2002, p. 3).

Former Defence Secretary William Perry was appointed by President Clinton to undertake the review. Perry later released the so-called "Perry Report". The report pointed out the need to build up North Korea's security confidence with engagement, while containment should only be implemented if engagement failed to encourage North Korea to cooperate (Perry, 1999). However, the Perry Report was criticised as rather weak, reflecting only a certain amount of understanding the United States had of North Korea and the overall situation on the Korean peninsula. C. J. Lee (2006, p. 198) criticised the report as being written from the negative perspective the United States held of the regime's nature and ruling system, and took issue with the fact that the words "blackmail, threats, and provocations" were used throughout. Drifte (2001, pp. 56-58), points out that the report did not outline solutions to problems besetting the work of KEDO, nor list disadvantages of overloading negotiations with North Korea with non-nuclear issues.

The US suspicion of the Kumchangri site was a turning point that greatly affected the engagement policy and the level of confidence between the two countries. While the United States felt uncertain about the regime's nuclear capabilities, North Korea perceived the US suspicion as hostile intent, which was supposed to end in accordance with the provisions listed in the Agreed Framework. North Korea criticised the US accusations regarding Kumchangri as "a fiction" created by the hardliners who wished to replace the engagement of the Agreed Framework with a hardline policy (KCNA, 1998c). According to Niksch (2002, p. 8), the Clinton administration aimed to end all of North Korea's nuclear-related activities at not only the Yongbyon site but also at all other nuclear sites throughout the country. The nature of the Agreed Framework was claimed to be a short-term solution, rather than a solution to fully transform relations through diplomatic and economic engagement (Martin, 2002, p. 53). Suspicion and distrust continued, which led to American accusations that North Korea had violated the Agreed Framework, charging them with the development of a uranium enrichment programme. The uranium enrichment was actually not a new project of North Korea, started during the time the regime froze its plutonium production as per the Framework agreement. North Korea's acquisition of uranium enrichment capabilities could be dated back to 1996. Former secretary of the KWP, Hwang Jang Yap, who defected to South Korea in 1997, testified that North Korea had successfully signed a contract in 1996 with Pakistan in order to acquire Highly Enriched Uranium (HEU) (Y. H. Kim, 2010). The report by Institute for Science and International Security also assessed North Korea's HEU

project to have begun in the late 1990s, but that the project was accelerated in the early 2000s with the supply of raw aluminium tubes from Pakistan (Albright & Brannan, 2010, p. 3). The Clinton administration acknowledged the existence of North Korea's uranium enrichment project in 1998. However, the administration failed to inform the Congress of North Korea's objective for HEU usage, which raised more criticism of Clinton's policy towards North Korea. In other words, there were some clear differences between the Clinton administration and members of the Congress, especially with the Republican majority, on various issues, including North Korea. While the Clinton administration preferred engagement in the hopes of persuading North Korea to change, the Congress was inclined towards adopting a hardline policy rather than one of appeasement (Hathaway & Tama, 2004, p. 715).

The United States seemed to be certain in summer 2002 of North Korea's uranium activity. Information released by the Central Intelligence Agency (CIA) confirmed the construction of a centrifuge facility, but claimed that it may take North Korea a couple of years to upgrade the uranium suitable for producing nuclear weapons (CIA, 2002). The report unavoidably degraded America's trust in North Korea. It also led to the US to declare that the North Korean regime was now violating the Agreed Framework. The US accusation on HEU development can be judged as weak and inadequate. Harrison (2005, pp. 103-104) argues that the report lacked solid evidence because it failed to expose "methods and sources" to support its claim that North Korea was "constructing a centrifuge facility". Pollack (2003, p. 31) says that it was too soon to accuse North Korea, because the CIA report of 19 November 2002 overlooked the fact that, at the time, "North Korea had no operational enrichment facility to declare". The CIA report did not take into consideration North Korea's inability to "build and operate a centrifuge plant" of its own, given that the regime faced some obstacles in terms of finance and know-how (Albright & Brannan, 2010, p. 4). Lastly, the centrifuge plant required a considerable amount of electricity to operate, but North Korea did not have the capacity to produce the required amount of electricity (Harrison, 2005, p. 105).

The uranium suspicion alone did not lead to a total failure of the Agreed Framework. To some extent, it was also a result of US delays in supplying North Korea with the promised light water reactors. This delay made North Korea to question the United States's commitment to the agreement. Cossa (1999, p. 6) writes that, since the Agreed Framework was not a formal treaty, it did not require any ratification process from the Congress. The disadvantage of this lies in its lacks of Congressional financial support that was deemed

necessary for the proper implementation of the agreement. It was clear that the United States and North Korea were not on the same track in committing to the agreement. While North Korea made incremental steps towards complying with the agreement, the United States failed to supply North Korea with heavy fuel oil due to problems of a lack of funding and disagreements within its government (C.K Quinones, 2008, p. 6).

The establishment of KEDO could be said to be an attempt by the administration to ensure the implementation of all provisions listed in the Framework. It later turned out that the KEDO was manipulated as the tool for shifting the financial responsibility from the United States to its Asian allies. Martin (1999, p. 37) claims that South Korea and Japan were the main contributors to the KEDO, both providing the necessary financing for the construction of the LWR project, which amounted to US\$ 4.6 billion. Since the Framework was a US-DPRK bilateral agreement, there was no guarantee for continuous support from either South Korea or Japan in financing the construction of the LWRs. In addition, they were both not subject to control by the United States. As such, it is possible to see how suspension of the LWR project happened during the late 1990s, in response to North Korea's provocations – for example, in 1998, in response to the Taepodong missile launch over Japan. Domestic politics also obstructed the Clinton administration from fully implementing the Agreed Framework. The political and economic normalisation between North Korea and the United States was listed in the Framework, but the Clinton administration was unable to begin the process as it required Congressional approval (Tan, 2014, p. 105). The United States was also uncertain about North Korea. President Clinton wrote in his memoir that, "North Korea had been saying that it wanted peace, and I believed we had to discover whether they were serious about it." (Clinton, 2005, p. 707).

The US's failure to keep to the proposed schedule was not the only factor that led to a dissolution of the bilateral agreement and a deterioration of trust between the two countries. The United States continued to believe that North Korea had undertaken a covert uranium enrichment programme, and this suspicion was finally articulated in the 2002 charge that North Korea was producing uranium. The US charge arose during an official visit by Assistant Secretary of State James Kelly to North Korea, on 3-5 October 2002. The US State Department released the statement detailing the minutes of the meeting between James Kelly and Kang Sok Ju, First Vice Foreign Minister of North Korea. It stated,

"Assistant Secretary James Kelly and his delegation advised the North Koreans that we had recently acquired information that indicates that North Korea has a program to enrich uranium for nuclear weapons in violation of the Agreed Framework and other agreements. North Korean officials acknowledged that they have such a program. The North Koreans attempted to blame the United States and said that they considered the Agreed Framework nullified. Assistant Secretary Kelly pointed out that North Korea had been embarked on this program for several years." (Agence France Presse, 2002a)

Kelly's statements caused heated debates amongst North Korean high-ranking officials, and the decision was made the following morning to respond to the US accusations thus:

"The reclusive nation is 'entitled to have nuclear weapons' to safeguard its security in the face of a growing U.S. threat. After a debate of their own, the Americans interpreted the statement to be an admission that Kelly's charge was true. Now it was the Americans' turn to be stunned by an unexpected declaration and to wonder what to do next." (Oberdorfer, 2002)

North Korea denounced Kelly's accusations. The KCNA referred to the accusations as made with no supporting evidence before pointing out American insincerity to improve relations with North Korea by calling North Korea "part of the Axis of Evil" (KCNA, 2002a). This accusation led to the collapse of the Agreed Framework and, more importantly, it allowed North Korea to use the occasion to push ahead with its nuclear programme (Hecker, Lee, & Braun, 2010). Hersh (2003), a reporter of the *New Yorker* magazine, reported what a former intelligence official said about the Kelly-Kang meeting:

"The Kelly meeting and the subsequent American statement have tipped the balance in Pyongyang. The North Koreans were already terrifically suspicious of the United States. They saw the Kelly message as 'When you fix this, get back to us.' They were very angry. That, plus the fact that they feel they are next in line after Iraq, made them believe they had to act very quickly to protect themselves."

Albright and Brannan (2010, p. 4) claimed that the United States exaggerated North Korea's statement on HEU because, although they did not deny having a centrifuge programme, North Korea never said that it was building a large-scale plant during the meeting.

6.2 The Second Nuclear Crisis and the US Policy during the Bush Administration (2001-2008)

The feeling of distrust played an important role in shaping US-North Korean relations from 1994 until the total breakdown of the Agreed Framework, following the US's uranium charges in 2002. The US policy towards North Korea has changed, influenced by both US domestic politics and also by an event in international politics in early 2000. George W. Bush's victory in the 2000 presidential election resulted in a change in the US's North Korea policy, in the form of the adoption of a hardline policy that was the opposite to President Clinton's policy. Clinton's policy was criticised as being a policy of appeasement, which "left the United States subject to continued blackmail by North Korea." (Martin, 1999, p. 41). North Korea's image in the eyes of the United States has gradually deteriorated, particularly because of the suspicions about uranium development, which the US government has held since the final years of Clinton's second term. Rozman (2007, p. 603) writes that the United States, both at the governmental and public levels, views North Korea as "the most dangerous state of the twenty-first century", due to it being a dictatorial system equipped with WMD.

When Bush came into office, he began to take an offensive stance towards North Korea. First by identifying North Korea as part of the axis of evil in the 2002 State of the Union address (The White House, 2002c), and also including North Korea among targets of possible nuclear pre-emption in a 2002 Nuclear Posture Review (Payne, 2005, pp. 135-151). The 9/11 incident changed the US perception of North Korea. The Bush administration called North Korea "a primary rogue state", implying that the status of North Korea had been lowered from a negotiation partner to a state that posed a serious threat to the international community and the interests of the United States (Pollack, 2003, p. 27). Regardless of the term used, both the terms 'rogue state' and 'axis of evil' shared the same basic assumption that the target engages in evil behaviour, which reflects the US's negative perception of North Korea (Interview

Anonymous, 2011; Interview Prof. Moon, 2011). Prof. Paik (2011) made the following comment on the US use of the term 'rogue state' in relation to North Korea:

"The term rogue state is the term used by hardliners in the United States and other places including Seoul and Tokyo. Basically, it is heavily politically-loaded term. We have some aspects of North Korea which can be described as that of rogue state. But if you characterise a state you are trying to deal with in a negotiation or the diplomatic activity with this term, that means you just tend to regard North Korea not as your partner but as the opposite of doing something. For instance, if you call the Axis of Evil by George W. Bush, then North Korea is regarded just as a target of removal not as a partner in the negotiation." (Interview Prof. Paik, 2011)

The situation deteriorated after President Bush's 2002 State of the Union address. North Korea reiterated that the address reflected US hostility towards North Korea, which confirmed that what the regime had been doing – being vigilant and offensive – was the right thing to do (KCNA, 2002c). Since President Bush took office, the United States has viewed North Korea as a threat to its national security, and this perception was expressed throughout 2002. At West Point on 1 June 2002, President Bush delivered a speech - "Prevent Our Enemies from Threatening Us, Our Allies, and Our Friends with Weapons of Mass Destruction" – in which North Korea was mentioned as a rogue state on equal terms as Iraq (U.S. Department of State, 2002). According to the speech, rogue states do not only support terrorism or acquire WMD, but, more importantly, these states have developed a sense of hostility towards the United States (U.S. Department of State, 2002). The speech was included in the National Security Strategy of the United States of America, released in September 2002, which further confirmed that North Korea was a threat to the security of the United States (The White House, 2002a). In December 2002, the United States also released the National Strategy to Combat Weapons of Mass Destruction, which was claimed to be an addendum to the National Security Strategy of September 2002 (Fuerth, 2003). The document outlined a strategy to control the spread of WMD and missile programmes among other states and terrorist groups by strengthening the US's defence mechanisms (The White House, 2002b).

The adoption of a hardline policy may be suitable in the new environment the United States was faced with after 9/11, and also in light of the nuclear threat from a rising nuclear state like North Korea. The United States is very sensitive about its security post-9/11, and the use of the terms 'rogue state' or 'axis of evil' to identify some countries reflects the US's understanding of international politics at that time (Interview Prof. D. H. Han, 2011; Interview Prof. I. H. Park, 2011). The term 'rogue state' also shows the negative impression the United States has of this type of state, without taking into consideration their specific histories or policies (Interview Anonymous, 2011; Interview Prof. D. H. Han, 2011). The United States views North Korea as posing not only a nuclear threat, but also a missile threat with the regime's long history of exporting missiles and missile components for hard currency. Nonetheless, the hardline policy of the Bush administration was partly the result of Bush's personal attitude towards the North Korean leader, Kim Jong II. In August 2002, President Bush gave an interview with *Washington Post* journalist Bob Woodward, in which Bush responded to a question on North Korea with his own personal dislike of Kim Jong II:

"I loathe Kim Jong II. I've got a visceral reaction to this guy, because he is starving his people. And I have seen intelligence of these prison camps – they're huge – that he uses to break up families, and to torture people." (Agence France Presse, 2002b)

President Bush's personal feelings played a significant role in the formulation of US foreign policy towards North Korea. The policy was drafted by the hardliners who were well aware of the "President's strong personal views" and the hardliners' policy directed Kelly's agenda and his strong reaction to North Korea during his Pyongyang visit in October 2002 (Hersh, 2003). Throughout President Bush's tenure, the United States adopted a hardline policy for dealing with the North Korean nuclear problem. The second nuclear crisis on the Korean peninsula began when North Korea admitted that it possessed a highly-enriched uranium programme in October 2002, and the United States' decision to deal with the problem by stating that dialogue could only occur after the abandonment of the enrichment programme.

To some extent, the nuclear crisis originated from different perspectives and the feeling of mistrust the two sides had towards each other. The United States considered North Korea's possession of highly-enriched uranium a breach of the Agreed Framework and other international agreements – including the Non-Proliferation Treaty, the IAEA safeguard

agreement, and the South-North Joint Declaration on the Denuclearisation of the Korean peninsula (Nikitin, 2013, p. 12). On the other hand, North Korea also blamed the United States of violating the Agreed Framework and of pursuing a hostile policy towards North Korea (KCNA, 2002b). Considering the Agreed Framework a failure, the Bush administration leaned towards a policy of isolation and containment in order to deal with the problem of a nuclear standoff with North Korea. Any solution to the problem should be dealt with through multilateral approaches in involving many countries. However, North Korea was clear in pointing out that the nuclear problem can only be settled through a US-DPRK bilateral discussion, not through other channels like the United Nations or the IAEA (BBC News, 2002a).

The second nuclear crisis clearly illustrated differences between both the United States and North Korea in their perspectives of the origin, nature and solution to the nuclear problem. The United States was faced with a dilemma of either appeasing North Korea through engagement, or isolating North Korea by pursuing a hardline policy. The situation worsened after Kelly's visit to North Korea and it did not show any sign of improvement, particularly after North Korea voiced its security concerns stemming from the joint US-ROK annual military exercises. North Korea perceived the joint military exercises as a US attempt "to settle [the nuclear issue] by military means", justifying the defensive measures North Korea has long been taking (KCNA, 2003b).

As a means to resolve the standstill, China stepped in with a proposal for the three-party talks, involving the United States, North Korea, and China. Since each party had different interests and objectives, the trilateral talks did not bring about a substantial outcome. A failure of the three-party talks made China press North Korea harder to attend the next dialogue and influenced China to also believe that other regional powers should be part of the future negotiations (CNN, 2003). North Korea viewed the US's desire for multilateral dialogue as an attempt to use the international community to help achieve "the DPRK's unilateral disarmament, not a peaceful settlement of the nuclear issue on the Korean peninsula" (KCNA, 2003f). In addition, the United States and North Korea could not agree on the details for disarmament. North Korea proposed "a bold initiative" proposal at the three-party talks, outlining the abandonment of nuclear weapons in four stages coupled with some compensation upon completion of each stage (The International Institute for Strategic

Studies, 2011). The United States was not interested in the proposal, and considered it unacceptable.

This proposal alone did not stall the talks: the responsibility for the failure lies with the North Korean decision to pursue a strategy of brinkmanship. The North Korean delegate, Ri Gun, said during the three-party talks meeting with James Kelly in Beijing that North Korea had not only nuclear weapons but also had 8,000 spent fuel rods completely reprocessed at the Yongbyon nuclear site (Cornwell, 2003). Although it was speculated that the statement was made in order to bluff the United States, Ri Gun's statement worsened the situation. It allowed the United States to charge North Korea with developing a habit for blackmailing. As President Bush said about Ri Gun's statement, North Korea was "back to the old blackmail game." (The New York Times, 2003). C. I. Moon (2012) writes that Ri Gun's statement reflected North Korea's "cheating behaviour" because, in contrast to what he said to James Kelly, Ri Gun denied the existence of nuclear bombs to China.

The three-party talks may have failed in its mission to disarm North Korea, but its failure does not negate the possibility of opportunities for future talks to succeed. The Six-Party Talks, another multilateral forum, were set up in late 2003 through the efforts of both the United States and China, who still share common interests in having a nuclear weapons free Korean peninsula. The role of China as a mediator became clearer when its desire to have a new set of the three-party talks was supported by the United States' suggestion to include more players in the talks. The Six-Party Talks were convened for the first time in August 2003, and also involved the ROK, Japan, and Russia. Substantially, the talks could not bring about a concrete outcome and the more the talks are organised, the more they revealed the feeling of distrust and discrepancies in preferences between the negotiating parties. Wit (2007, p. 55) writes that North Korea skilfully used "a diplomatic strategy" at the Six-Party Talks forum in convincing other countries that the United States was the one to blame for the deteriorated situation, including the fall of the Agreed Framework.

The first breakthrough in negotiations came two years after the fourth round of the Six-Party Talks, with the release of the September 19 Joint Statement.¹⁷ The statement did not lead to

⁻

¹⁷ The September 19 Joint Statement clearly indicated that North Korea committed to totally dismantle all of its nuclear programmes in a verifiable and peaceful manner. In return for such actions, North Korea received a security guarantee, economic cooperation, and energy assistance from

denuclearisation. Instead, it increased levels of distrust and disagreements between the United States and North Korea, which resulted from the issue of the light water reactors. The statement said,

"The D.P.R.K. stated that it has the right to peaceful uses of nuclear energy. The other parties expressed their respect and agreed to discuss, at an appropriate time, the subject of the provision of light water reactor to the D.P.R.K." (U.S. Department of State, 2005b)

In other words, this issue was left open for future discussion without any commitment made. Christopher Hill, the chief US negotiator to the Six-Party Talks, actually disagreed with the North Korean request because "it still raises proliferation risks and cannot be a first step in arranging the nuclear disarmament of the country" (Kahn, 2005b). North Korea was dissatisfied with the US refusal to provide a light water reactor, and said it would return to the NPT and sign the IAEA safeguard agreement only if a light water reactor is supplied (KCNA, 2005b).

Since the agreement was concluded on shaky ground, the process of denuclearisation later faced a major setback arising from the issue of Banco Delta Asia (BDA). The Department of Treasury claimed that North Korea was connected to or involved in money laundering activities with the BDA bank located in Macau (U.S. Department of Treasury, 2005). The Bush administration started to take legal action against the BDA on charges that North Korea had used it to facilitate their criminal activities. According to (Hauben, 2007), the legal actions were undertaken to block the international financial activities of North Korea and also to send a warning message to China, who has been doing business with North Korea. The BDA issue stalled progress on denuclearisation. The gap between the United States and North Korea on this issue widened. The United States preferred North Korea to begin the process of total dismantlement of all of its nuclear programmes regardless of financial sanctions on the BDA (Kahn, 2005a). North Korea, on the other hand, tied the sanctions to the nuclear issue, and requested the United States to resolve the BDA issue before it took any first step towards denuclearisation (Kahn & Weisman, 2007). Some debate on the accuracy of

all parties involved in the talks. The full text of the Joint Statement can be found on the U.S. Department of State website: http://www.state.gov/p/eap/regional/c15455.htm (U.S. Department of State, 2005b).

the US charges against North Korea and the BDA arose following a statement by the BDA's owner, Mr Stanley Au. Mr Au testified that the recent result of an account audit "found insufficient money-laundering controls but no evidence that North Korea used the bank to introduce fake \$100 bills" (Hall, 2007). Mr Au's statement contradicted the findings of the Department of Treasury, raising some questions of whether North Korea's financial activities were illegitimate, and if the regime deserved some financial sanctions over activities unrelated to its nuclear weapons development (Hauben, 2007). According to Chinoy (2010), the US hardliners had high hopes that the US financial pressure would push North Korea to experience financial difficulties that could bring about the collapse of the regime. The result, however, was not regime collapse, but rather missile and nuclear tests in 2006.

North Korea conducted a missile test in July 2006. This was followed by an underground nuclear test, which North Korea declared a success on 9 October 2006. Concerning the nuclear test, the regime stated that it was a response to "the policy of sanctions and blockage", which confirmed for North Korea the US threat against the regime (KCNA, 2006b). The 2006 nuclear test had two major consequences. The first was a change in North Korea's policy and status. North Korea declared itself a nuclear state, and that any future dialogue with the United States would be on arms control and not denuclearisation (Hecker, 2010, p. 50). Secondly, the nuclear test indicated the failure of the United States's policy towards North Korea. The Bush administration failed to understand the predictable "tit-fortat" behaviour of North Korea: that the regime would act in a corresponding to the way the United States structured its policy (Chinoy, 2010). Moore (2008a, pp. 16-18) writes that the Bush administration's North Korea policy was structured by the hardliners, who not only adopted the "Anything But Clinton mentality", but also failed to listen to comments from experts in the field. President Bush condemned the test, calling it a "provocative act", before committing to strengthen the deterrent strategy with its two Asian allies – Japan and South Korea (The White House, 2006). That North Korea has continued with its provocations ever since President Bush came to power raises a question as to how effective the hardline policy is on influencing North Korea's behaviour. Concerning this, Prof. I. H. Park (2011) says,

"Some people say that the US hardline position makes North Korea more irrational or more aggressive. But according to my opinion, whatever American approaches to North Korea, North Korea is only interested in survival. North Korea's response to the United States could be different but fundamentally and

essentially North Korea is only interested in their own survival." (Interview Prof. I. H. Park, 2011)

President Bush showed his strong opposition to North Korea especially after the nuclear test. However, leading up to the end of his presidency in January 2009, the United States relaxed its policy and strategy towards North Korea. Evidence of this can be seen in the decision to unfreeze North Korean accounts at the BDA, in April 2007; and also the decision to consider removing North Korea from the list of state sponsors of terrorism, in September 2007. Nonetheless, US-North Korean relations still had not exhibited any signs of improvement. North Korea voiced its dissatisfaction with the US failure to comply with the "action for action" principle¹⁸ – for example, the US delay to remove North Korea from the list of "state sponsor of terrorism" – in exchange for North Korea's submission to the nuclear declaration (KCNA, 2008b).

It is possible to argue that the United States, to some extent, had softened its policy through engagement, implemented through the multilateral negotiation of the Six-Party Talks. The Six-Party Talks are actually a diplomatic tool the United States employs to use the power of other regional actors to pressure North Korea to disarm. Engagement was actually pursued with the aim of coercing North Korea to change, rather than aiming for a regime collapse (Cha, 2012a, pp. 284-285). However, it appears that the possibility for regime change or regime collapse was widely discussed at the beginning of the Bush administration's first term in office. Secretary of Defence Donald Rumsfeld, wrote a memorandum calling for regime collapse in North Korea, brought about through the use of pressure through the US-Sino diplomatic channel (Sanger, 2003). Regime change usually refers to regime collapse, which could be achieved either through a military invasion or internal political reform. Since North Korea exists in a different geopolitical environment from Iraq, the choice of military invasion to force regime change appeared highly unlikely to succeed. The Bush administration, in the end, did not choose this option from fear that the situation might worsen beyond its control.

¹⁸ The September 19 Joint Statement stated that "the Six Parties agreed to take coordinated steps to implement the aforementioned consensus in a phased manner in line with the principle of "commitment for commitment, action for action"" (U.S. Department of State, 2005b). This principle was North Korea's initiative as the regime wanted to see commitments from other parties. It means North Korea would get some compensation for each action it took towards denuclearisation, and North Korea would "reciprocate only after the other party has done so" (Funabashi, 2007, p. 404).

It is important to note that, at the same time that the United States did not wish to use force to bring about regime change, the US did not clearly provide a security assurance to North Korea. In the end, the Bush administration used international pressure and sanctions against North Korea in order to push for denuclearisation and in an attempt to force regime or behavioural changes. On the US's policy towards North Korea, and how the policy has an effect on North Korea's nuclear strategy, Prof. M. Kwon (2011) explains,

"George W. Bush threatened North Korea, particularly when he invaded Iraq and said that North Korea might probably be the next one. President Bush intimidates North Korea while saying that North Korea should dismantle its nuclear weapons. Basically if the other side has hostility towards you, you would strengthen your own armaments so that you could deter the other. Security of both parties can be guaranteed if there is no hostility between the two sides and then you do not have to develop nuclear weapons. But it is not the case. The United States is very hostile to North Korea which in turn encourages North Korea to develop nuclear weapons." (Interview Prof. M. Kwon, 2011)

The use of international pressure is to force North Korea to make a "cost-benefit analysis" between its choice of going nuclear and the risk of facing deeper diplomatic isolation and economic loss (Schneider, 2010, pp. 47-48). The pressure the Bush administration put on North Korea in order to push for North Korean denuclearisation actually increased North Korea's sense of insecurity. Referring to President Bush's State of the Union Addresses of 2002 and 2005, North Korea declared,

"In his state of the union address late in January 2002, Bush designated the DPRK as part of an 'axis of evil' and, in March of the same years, listed it as a target of the U.S. preemptive nuclear attack ... In his state of the union address on Feb. 2 [2005] he ... once again vociferated about an 'end to the tyranny,' asserting that the U.S. will force north Korea to abandon its nuclear ambition ... U.S. official figures have not expressed any intention to co-exist with the DPRK or make a switchover in its hostile policy toward the DPRK in any recent remarks made by them." (KCNA, 2005a).

Apart from the issue of nuclear and missile proliferation, the Bush administration also put an emphasis on North Korea's human rights abuses, which later became a challenge to US-North Korean relations. The issue of human rights started to take a role in the bilateral dialogue in 2007, when the Bush administration asserted that the normalisation of diplomatic relations would only take place if North Korea made progress on human rights and other issues (Chanlett-Avery, 2008, p. 2). The ultimate goal of the Bush administration was to create a stable and peaceful Korean peninsula in which issues concerning human rights abuses, uranium enrichment, nuclear and missile programmes were resolved. Multilateral diplomacy was the method the United States deemed most suitable for solving these problems, particularly with regards to North Korea's nuclear problem (BBC News, 2008).

6.3 The US Policy during the Obama Administration (2009-present)

President Bush's presidency ended in January 2009, and he left for his successor the unfinished task of creating a written verification protocol on denuclearisation through the Six-Party Talks. After seeing the need to engage with North Korea, the Obama administration planned to continue with the organisation of the Six-Party Talks, in order to realise the goal of the denuclearisation of the Korean peninsula (S. Snyder, 2010, p. 64). A week prior to President Obama's inauguration, North Korea voiced its insecurity based on the US nuclear threat, saying,

"The nuclear issue surfaced on the Korean Peninsula because of the U.S. hostile policy toward the DPRK and its nuclear threat resulting from it ... When the U.S. nuclear threat is removed and South Korea is cleared of its nuclear umbrella, we will also feel no need to keep its nuclear weapons." (KCNA, 2009e).

The Obama administration's selection of Hillary Clinton as Secretary of State was seen as a positive sign for better relations between the two countries, and the likely adoption of an engagement policy. Hillary Clinton was more inclined to adopt an engagement policy towards North Korea, which was clear in her statement that "smart power requires reaching out to both friends and adversaries, to bolster old alliances and to forge new ones." (Choe, 2009). However, the prospect for engagement was later stalled by the North Korean decision

to conduct a missile test, for the Taepodong-2, on 5 April 2009. North Korea instead claimed that this incident was a launch of a satellite for "a peaceful use of outer space" purpose (KCNA, 2009g). Moon (2009) sees the incident as North Korea's attempt to test the Obama administration's policy and the US perception of North Korea.

The situation worsened in the following month, when North Korea decided to conduct a second nuclear test, which put pressure on the United States and its allies in the region. The incident prompted the Obama administration to switch to a hardline policy instead of the policy of engagement outlined by Obama himself during the presidential campaign (Cha, 2012a, pp. 265-266). In addition, Obama's decision to replace US special envoy to North Korea Christopher Hill with Stephen Bosworth, a scholar from Tufts University and former US ambassador to South Korea, in a part-time position caused some concerns over how serious the Obama administration were about addressing North Korean affairs.

There arose some criticisms of the different styles of these two US negotiators. While Christopher Hill was outspoken and was more diplomatic, Stephen Bosworth was a good listener (The Chosun Ilbo, 2009). The North Korean decision to launch a missile and to conduct a nuclear test may be a response to the appointment of Bosworth, to see if the Obama administration was serious about engaging with the regime. The missile and nuclear tests in 2009 were meant to draw the attention of the Obama administration during a time of inaction (C. I. Moon, 2012). This also reflects North Korea's attempt to achieve the kangsong taeguk, and at the same time to strengthen its bargaining position, as the United States inevitably refocused its attention back towards North Korea (Interview Prof. Y. Y. Kim, 2011; Interview Dr. E. C. Lim, 2011). Nonetheless, North Korea's actions did not have a positive result, as the United States began to question the effectiveness of its overall foreign policy. The United States' reaction, particularly to the missile test of April 5, changed the atmosphere entirely because the United States considered the test a provocative act and strongly condemned it. North Korea claimed that the missile test was actually a satellite launch and tough measures like sanctions reflected the US's hostility towards North Korea (KCNA, 2009c). President Obama once made a comment regarding US policy towards North Korea, that he vowed to break the old pattern of rewarding the regime for its belligerent behaviour (The Economist, 2009).

A series of provocations influenced the Obama administration to alter its foreign policy from one of engagement with North Korea to the so-called 'strategic patience', described as being a strict engagement policy. The strategic patience policy outlines the use of pressure and sanctions on North Korea until North Korea changes its behaviour and returns to the negotiating table (Warren, 2014, p. 147). It is the US's effort of bringing the situation under control, not waiting to react to North Korea's provocations like previous administrations have done (Kessler, 2010). For the policy to be successfully implemented, strategic patience also calls for the help of China and other regional powers in exerting pressure on North Korea (J. Kim, Kim, Kim, Kim, & Lee, 2013). By requiring North Korea to change its behaviour, strategic patience sought to isolate North Korea particularly when it challenged regional security with new provocations. A policy that calls for the isolation of North Korea received broad criticisms. For example, Joel Wit said that, "The strategic patience has only allowed Pyongyang's weapons programme to grow and its political system to get more stable" (C. D. Lee, 2012). In addition, Bruce Cumings states that,

"The whole idea of 'strategic patience' is based on Secretary of State Hillary Clinton's assumption that there's a power struggle going on in North Korea, and that the regime is destined to crumble. It's now become clear that this is no strategy at all. There were no problems in the process of power being transferred from Kim Jong II to Kim Jong Un. By just continuing to wait and wait, the US has allowed North Korea to get that much closer to being a nuclear state with strong missile capabilities." (H. Park, 2012).

North Korea's missile test on April 2009 was criticised by the international community, and made the regional powers worried about North Korea's defiance of regional peace and stability. Some, including Japan, wished to see tough punitive actions against North Korea at the UN Security Council level. In response to the missile launch, President Obama called for the strengthening of the US's relationships with the two Asian allies, and to bring this issue to the attention of the United Nations (Obama, 2009). Obama's statement pushed North Korea further into believing in a US threat against the regime (KCNA, 2009a). President Obama did not only confirm the use of pressure as a means to change North Korea, but he also sought to control the spread of nuclear weapons through various forms of international cooperation such as Proliferation Security Initiative (PSI) (The White House, 2009). Since the PSI is not under the control of the United Nations, it has the power to act independently with an

objective of preventing "rogue" nations – e.g., North Korea and Iran – from delivering and trading nuclear weapons (Valencia, 2010). The situation on the Korean peninsula deteriorated after the sinking of the Cheonan warship. Despite North Korea's denial, the United States believed North Korea was behind the attack (The White House, 2010c). It was claimed that the US's policy and strategy towards North Korea were not structured on its interests alone, but were also influenced by the interests of its allies. The United States considers a close cooperation with allies as key to achieving success in solving the North Korean problem, particularly when the progress of the Six-Party Talks could only come with good inter-Korean relations (Manyin, Chanlett-Avery, & Nikitin, 2011, p. 11).

The sinking of the Cheonan led to a strengthening of US-ROK military ties and a series of joint military exercises, including some along the coastal line of the Korean peninsula. To North Korea, the exercises were considered provocative; and the North Korean shelling of Yeonpyeong Island, on 23 November 2010, was an inevitable strike in response to South Korea's actions inside the territorial waters of North Korea during the drill. The consequences of the attack also led to closer security ties between the US and ROK, who also conducted military exercises in the Yellow Sea in preparation for any possible North Korean attack in the future. It also led to a collective security pact between Japan, South Korea and the US for defence against future North Korean provocations.

6.4 US Foreign Policy towards North Korea in the Post-Cold War World

North Korea has long been a primary concern of the United States, given the regime's past record of terrorist activities and their pursuit of a nuclear weapons programme. There are two approaches the US has taken in the hopes of solving the North Korean problem. The first is a policy of containment to limit the spread of nuclear weapons from new nuclear states, with a hope of transforming aggressive behaviour into a constructive part of the international community. The second is a US attempt to maintain bilateral ties with its Asian allies, in order to secure its position as a regional leader. The US's post-Cold War policy towards East Asia is structured on the American belief that alliances and other bilateral partnerships

provide the opportunity for the United States to play an active role in the region (Ikenberry, 2008, pp. 30-31).

During the first nuclear crisis, the United States decided to engage with North Korea to solve the problem through diplomatic means. This showed North Korea's successful use of nuclear weapons as a policy instrument, which Prof. Paik (2011) expands on:

"North Korea uses the nuclear card in the early 1990s and the United States had to come to the negotiation table. There is no other way because of the characters of the nuclear card. If somebody uses a nuclear card against you, then you are not supposed to avoid it. It is a nuclear power. You cannot begin a war. It is impossible to bomb Pyongyang. The only way you can solve the problem is to persuade North Korea to give it up." (Interview Prof. Paik, 2011)

However, it is important to note that engagement was based on the US belief in the possibility of North Korean regime collapse, and this affected the US commitment to the agreements it made with North Korea. The Clinton administration's anticipation of regime collapse was based on two factors.

The first was the international political structure following the collapse of Communism in the Soviet Union and Eastern Europe. The second was North Korea's internal structure itself after the death of Kim Il Sung and the economic collapse. Prof. Y. Y. Kim (2011) makes the following argument on the issue of regime collapse as a result of internal weakness:

"I think it is difficult for North Korea to collapse. North Korea has experienced economic difficulties in the last 20 years but the result is not the collapse of North Korea. North Korea continues on. When Kim II Sung died, we say that it is possible for collapse of North Korea. But Kim Jong II comes to power and now is Kim Jong Un. The military base of North Korea is very strong. The control system is very strong. I do not believe in the collapse of North Korea." (Interview Prof. Y. Y. Kim, 2011)

Harrison (2002, p. 4) writes that the Clinton administration did not take the LWRs construction seriously, believing that internal problems of economic stagnation and internal instability would soon cause North Korea to collapse. The United States felt rather relieved

when the Agreed Framework was nullified, as it put an end to the proliferation risk posed by the LWRs construction project (Kessler, 2005). The belief in an impending regime collapse, which first appeared during the Clinton administration, only strengthened during the Bush administration. This can be seen in high-level official remarks; for example, from Deputy Defence Secretary Paul D. Wolfowitz, who expressed his belief in the possibility of a collapse resulting from the regime's heavy investment in military and defence (Gilmore, 2003). Martin (2007, p. 93) writes that the United States uses the Six-Party Talks to buy time while seeking alternatives to isolate North Korea in the hope of bringing about the total collapse of North Korea. The US expectation of regime collapse also appears in the Obama administration. The report by US think tank the RAND Corporation stated that the collapse of the North Korean regime is inevitable given that North Korea is faced with many challenges, including severe famine and challenges from internal elites (Bennett, 2013). Discussing the possibility of regime collapse as a result of an internal power struggle Prof. Moon (2011) said:

"It is very unlikely but still the end of Kim Jong II and Kim Jong Un does not necessarily mean the end of North Korea. Even if Kim Jong II and Kim Jong Un go, the military can intervene and there could be a military regime or collective leadership between Korean Workers' Party and the military can rule North Korea. I really do not think North Korea would give up its sovereignty. I really do not think the change of regime would necessarily lead to the change of entire sovereign state called the DPRK." (Interview Prof. Moon, 2011)

The Obama administration's adoption of the policy of strategic patience, to some extent, reflected the US wish to see some changes from and in North Korea, or ultimately the collapse of the regime itself. The survival of North Korea, in contrast to the US belief in the possibility of the regime collapse, raises some question about how well the United States understands North Korea, and how possible it is for North Korea to collapse. A South Korean expert on North Korea commented,

"I do not think that North Korea is so unstable to be collapsed. I think North Korea has been stable and will be stable in 5-10 years. We should manage North Korea rather than expect North Korea to collapse. We should manage North Korea's behaviour." (Interview Anonymous, 2011)

Considering that Clinton's policy of engagement was a failure and could not bind North Korea to its agreements, the Bush administration turned towards a policy of containment. During President Bush's second-term, policy changed slightly from containment to engagement following a change in Secretary of State, when Colin Powell was replaced by Condoleezza Rice. The organisation of the bilateral dialogue with North Korea under the Six-Party Talks Framework was a result of Rice's adoption of an engagement policy (C. Kenneth Quinones, 2005b, p. 35). There also appeared a debate in the United States about the degree of progress North Korea could make in producing weapons-grade uranium, raising doubts about the accuracy of the assessment report in 2002 (Sanger & Broad, 2007). The change in the US position, however, was not enough to lower North Korea's level of threat perception, which increased after the signing of the Agreed Framework. There were a number of factors that contributed to North Korea's raised levels of threat perception. These included the delay in the LWR construction project, the uranium accusation, and a lack of progress in normalising US-North Korean relations. Prof. M. Kwon (2011) notes that both parties share the responsibility for the failure of the agreement, but the fundamental reason for the continuing problems is that both sides do not have confidence in each other (Interview Prof. M. Kwon, 2011).

When the Obama administration took office, there appeared the prospect for engagement with North Korea, a result of the confirmation of Hillary Clinton, and also the change in the position of the special representative to North Korea. Engagement was later reversed as a result of North Korea's provocations and the US perception of North Korea's nuclear problem. S. Snyder (2010, pp. 64-65) writes that, among the focuses the Obama administration has on North Korean policy, was the total nuclear dismantlement of North Korea as influenced by its wish to expand the NPT. Besides, the US policy towards North Korea is constrained by the alliances it has with its Asian allies. At the same time that the Asian allies need the United States for security assurance and deterrence purposes, the United States also seeks to maintain a certain level of partnership with its allies to secure its own interests and power in the region. A series of North Korean provocations – including the missile test, the nuclear test, the sinking of the Cheonan warship, and the attack on Yeonpyeong Island – shifted the policy of the Obama administration to one focussed on the use of deterrence and strengthening its alliances in preparation for preventing any possible future North Korean provocation. The US confirmation of its alliances in response to the

Yeonpyeong incident is one example of this (The White House, 2010b). A statement of condemnation by the United States was released for the benefit of South Korea and Japan, who demanded strength and support when they felt threatened by North Korea's provocations. Prof. Koo (2011) says that,

"I think the United States and North Korea are sharing the same idea on security issues. Both think that security can be achieved by the might/military power. The United States invests a lot of money to defence sector. North Korea who does not have enough money tries to develop nuclear weapons. So, it is very similar of the two countries' policies for security." (Interview Prof. Koo, 2011)

The change in US policy from engagement to containment to strategic patience reflected an inconsistency in the US's approach to North Korea. This is partly because of US domestic politics and the change in administration after every general election. North Korea is well aware of how the US governance system works. Concerning this, Prof. Paik (2011) elaborates:

"In the eyes of North Korea, they understand that the Americans change their government through election. They know that the Republicans prefer this and the Democrats prefer that. They understand that any agreement the US government made by far will be reversed by the next government. That is the main difficulty the North Korean negotiators face." (Interview Prof. Paik, 2011)

On the other hand, when looking at the North Korean side, it appears that North Korea's policy was rather consistent, and this is a result of the hereditary succession that helps the regime to continue with the same policy (Interview Anonymous, 2011). Obama's passive and unfriendly policy towards North Korea comes from the government's lack of understanding of North Korea, resulting from the relative lack of experts on Korean affairs in the administration's Asia team (Rogin, 2013).

It is important to note that it is rather difficult for states to totally abandon their nuclear weapons programme once they possess them, particularly if nuclear weapons are a source of security when they are faced with a situation characterised by distrust and misunderstanding. Nuclear weapons provide North Korea with a security guarantee during time when the United States will not recognise the regime (Interview Prof. Moon, 2011). Given the United States

actions and policies, it is rather difficult to see the situation improving. Although there is an argument that North Korea has not changed, even though there is a dialogue between the two countries, it is highly likely that the threat from North Korea will only increase if it is pushed (further) into isolation. Prof. S. H. Sheen (2011) expressed his opinion on the US hardline policy, including strategic patience and its effect on North Korea, as,

"The practical approach to deal with North Korea is somehow we have got to engage rather than isolate. If we try to isolate or if we try to ignore, North Korea tends to react in very dangerous way. Of course one could say that we can even put more pressure or we can apply even bigger stick to such a kind of reaction. But the problem is that confrontation itself involves a lot of damage and that exactly what North Korea knows. In other words, North Korea has nothing to lose. They create the situation which, despite all their difficulties and weaknesses, North Korea tends to have this leverage in dealing with South Korea, the United States, and even China and others." (Interview Prof. S. H. Sheen, 2011)

6.5 Conclusion

The fall of the Soviet Union changed the overall international political structure. The new pattern of the post-Cold War world was a confrontation between the United States and the small and middle powers with nuclear potential. The policy of the United States in the post-Cold War period is structured on the policy of containing nuclear proliferation, particularly among these new nuclear states. The US approach to North Korea is structured based on this policy. The United States wants to achieve the denuclearisation of the Korean peninsula, in line with the US preference for nuclear non-proliferation. Nonetheless, the US policy towards North Korea is also influenced by the US perception of the rise of a new type of nuclear state. The use of various terms such as 'rogue' or 'axis of evil' shows the negative mindset of the United States towards North Korea. The United States sees North Korea as a dangerous and immoral state that will soon collapse as a result of various internal problems it is currently facing. The secretive nature of North Korea makes it important that North Korea be fully

disarmed. The United States is a nuclear power. Nuclear weapons provide the United States with the power to deter North Korea and, at the same time, helps the United States play a role of a security guarantor to its Asian allies. The US alliance system is the product of the Cold War period, and the threat posed by North Korea helps justify the continued US role and involvement in the region in the post-Cold War environment.

The emergence of new nuclear states post-Cold War is a result of the structural change away from nuclear bipolarity. The new structure encourages states to be independent and compete among each other for their own security. Since North Korea decided to increase its defence level, the United States has employed various methods in dealing with North Korea since the first nuclear crisis. The change in approach reflected the US decision to explore different methods in its policy but US inconsistency in policy implementation decreased North Korea's trust. Although the United States does not actually use nuclear weapons in the process to denuclearise North Korea, the United States is fully aware of the power it has over North Korea. The methods of either dialogue or confrontation are attempts to manage the problem, but the result turns out to be the opposite. North Korea continues to elevate its defence level instead of disarm. The United States deals with North Korea based on its policy preference and perception, rather than considering the root cause of the problem. Past negotiations show that dialogue does not actually answer North Korean demands. The security dilemma now is created from a sense of insecurity from North Korea's loss of trust in the United States, which in turn results in a decrease in the United States' trust in North Korea. The security dilemma is a common thing that can happen given states' interdependence in security. The security dilemma can be ameliorated if states feel secure enough that they do not seek to increase their defence capabilities. The case of North Korea does not only show a lack of trust, but also a sense of insecurity created by an unequal distribution of power. North Korea fears an American nuclear attack, and the fear is reinforced by the US hardline policy and the expectation of regime change.

International relations theory talks about the search for power as a means to achieve security in a system of anarchy. The United States emerged as the hegemon in the post-Cold War period, but the status of hegemony does not mean the absolute security of the United States. The emergence of small and middle nuclear powers presents a new threat to the United States. These states become more threatening when there is difficulty in reading the predictability of their actions. The 9/11 incident proves that there is no permanent security in

the system, and that insecurity can happen anytime without advance warning. The objective of preventing the proliferation of nuclear weapons in the system is among the policies the United States pursues in order to secure itself in the new environment. It is clear that there is an imbalance in the distribution of power between the United States and North Korea. The United States is much more powerful than North Korea but there is no guarantee that security can be permanently assured when North Korea is turning itself into a nuclear power. Realists assume that anarchy makes states feel insecure and this insecurity pushes them to increase their defence or risk being overpowered by others. The situation between the United States and North Korea is that deterrence is successfully functioning in keeping one from attacking the other. Successful deterrence does not necessarily mean a decrease in the sense of insecurity. States will continue in their quests for security because they are concerned about their own security above all others'. The United States deals with North Korea with the policy objective of achieving the total nuclear dismantlement of North Korea. However, the process to achieve such an aim lacks a security assurance. Insecurity influences North Korea to seek better security and increase its defence, in the hope of matching deterrence levels with the United States. When North Korea reaches equal deterrence levels with the United States, it believes it will be able to negotiate its primary demands of peace treaty and regime security.

Chapter 7: The Attempt at Multilateral Settlement

The George W. Bush administration made it clear that the United States would not follow the Clinton policy of engagement, because North Korea had developed a habit for blackmail. Despite the adoption of a hardline policy, President Bush still believed diplomacy would provide a solution to the overall problem, and this is one of the reasons the United States remained engaged with North Korea through multilateral channels. It began with the three party-party talks initiated by China, who attempted to achieve a breakthrough from fear that the situation would worsen with a lack of dialogue. The structure of the multilateral forum was later expanded to include more players. This is because the United States believed that the problem of North Korea's nuclear proliferation is a regional issue that requires the involvement of all parties in order to solve the problem (U.S. Department of State, 2003). In addition, the United States also considered it important to bring its Asian allies into the dialogue, believing that international pressure would be more effective than the United States alone in influencing North Korea to accept the complete, verifiable and irreversible dismantlement (CVID) of its nuclear weapons programme. China, who stepped in and acted as a mediator for the Six-Party Talks, had developed major concerns with regards to the situation on the Korean peninsula at that time. Not only did China fear the outbreak of a war resulting from a US invasion; but it was also concerned about the US's close relations with its two main Asian allies, which might lead to the formation of "an anti-China grouping" in the region (Buszynski, 2013b).

The Six-Party Talks were first convened in late 2003, hosted by China. Unfortunately, several rounds of the talks failed to lower tensions on the Korean peninsula, and did not result in much progress towards denuclearisation. All members shared the same objective of achieving a denuclearised Korean peninsula, but it turned out that each participant used the forum as a means for achieving its own aims. While the United States showed strong support for the total dismantlement of all nuclear programmes, as a prelude to discussing other issues, North Korea required some form of deal or proposal before taking the first step towards denuclearisation. Japan was mainly interested in participating in the multilateral forum due to its fear of being left out of any important regional issue. Japan's enthusiasm for multilateral negotiation grew after a bilateral dialogue between Japan and North Korea failed to make any

progress, especially given Japan's dissatisfaction with North Korea on the abduction issue. The United States considered Japan's participation significant for the proper conduct of the talks, especially given the economic support Japan could contribute after the United States entered into a deal with North Korea. Nevertheless, Japan's participation in the Six-Party Talks was a challenge for the United States. The United States had to compromise between progress on the CVID and Japan's interests in order to secure Japan's economic contribution towards any agreements made with North Korea (Okano-Heijmans, 2008).

South Korea, on the other hand, was enthusiastic for the Six-Party Talks, as the initiative coincided with the engagement policy adopted during Roh Moo Hyun's administration. President Roh Moo Hyun believed in the significance of the dialogue as a way to bring North Korea out of isolation, and in the hope of bringing about the regime's total nuclear disarmament (Ministry of Unification, 2004). South Korea later became an active partner in the talks. President Roh's preference for the use of dialogue resulted from the perception that dialogue is the key path towards a definitive peace on the peninsula (S. W. Park, 2010). Russia was the last to join the Six-Party Talks. The aim of Russia, particularly under Putin's government, is to use the Six-Party Talks to counter the United States in the hopes of regaining its power in this area (S. M. Kang, 2006, pp. 87-94).

To some extent, each party's individual preferences weakened the spirit and objectives of the multilateral platform. Differences between the parties surfaced during the second round of the talks, in 2004. It seemed that members were divided, with the US and its Asian allies on the one side and North Korea on the other, on the issue of the peaceful uses of nuclear energy and a uranium enrichment programme. Firstly, North Korea, in contrast to the United States, Japan and South Korea who sought CVID, insisted on pursuing the peaceful uses of nuclear energy and that the dismantlement should be limited to its nuclear weapons development programme. The demand for CVID as a prerequisite to other commitments increased North Korea's concern for other actors' levels of sincerity and commitment to the dialogue, particularly that of the United States (KCNA, 2004). Secondly, North Korea's denial of the existence of a uranium enrichment programme was counter to what the United States, Japan, and South Korea had believed (Ministry of Foreign Affairs of Japan, 2004).

The Six-Party Talks relate to the concept of collective security, which is viewed as being one of the ways that can ameliorate the security dilemma. Collective security usually results in

the formation of a coalition, which wields more power than other mechanisms in balancing and deterring an aggressor (Kupchan & Kupchan, 1991, p. 118). Kupchan and Kupchan (1995, pp. 52-59) elaborate further, that cooperation would create stability when states share the same objective and seek to secure absolute gains for all members, rather than relative gains of their own, which helps reduce the spirals of hostility between states. However, cooperation under the Six-Party Talks failed to stop aggression and create stability in the region. The failure of the Talks to encourage North Korea to disarm raises some questions on the effectiveness of collective security as a means to deter an aggressor, or as a way to ameliorate the security dilemma.

As the Six-Party Talks continued, the members appeared to be ever more concerned with the relative gains for themselves, rather than the absolute gains of the forum, which results in a lack of trust and cooperation between the parties. The talks were eventually faced with a major obstacle, after North Korea declared in 2009 that it would no longer participate in any future dialogue. Since the Six-Party Talks are claimed to be an appropriate solution to the problems on the Korean peninsula, the little progress made raises certain doubts of the effectiveness of multilateral cooperation and, at the same time, challenges the concept of collective security.

This chapter aims to explore the interaction of concerned parties in the context of the Six-Party Talks, before seeking to answer the question of what the limits of multilateral diplomacy are in solving the North Korean nuclear problem. The chapter also seeks to assess whether there is any prospect that the forum can persuade North Korea to lower its deterrence threshold.

7.1 The Failure of the Six-Party Talks

In order to evaluate the failure of the Six-Party Talks, it is best to begin by assessing each party's policy, and the role each contributes to the multilateral forum.

China

China's decision to expand the multilateral forum from three to six parties came from its strong desire for a peaceful regional environment. China facilitates with the organisation of the Six-Party Talks, knowing full well that it is not the main actor of the issue. The Six-Party Talks need to be a continuous process in order to achieve the objective of denuclearisation (Xinhua News Agency, 2004). Although China considers it undesirable that North Korea has a nuclear weapons programme, it is not considered to be the main factor for considering North Korea a major threat to China's security. China's main concerns are the possibility of regime collapse, which could lead to a flood of North Korean refugees into China; and also the possibility of armed conflict between North Korea and the United States, which could provide an opportunity for the United States to move its forces on the peninsula up to the Chinese border.

North Korea's possession of nuclear weapons is considered undesirable, but not a major threat to China's security. To China, North Korea poses a "symbolic threat" not an "actual threat" to international security, because North Korea lacks some resources needed to make its nuclear weapons programme a credible threat (J. Park, 2005, p. 82). However, North Korea's nuclear test in 2006 challenged China's policy and belief. China strongly condemned the test, as it was "flagrantly conducted in disregard of the common opposition of the international community" (Ministry of Foreign Affairs of China, 9 October 2006). Based on China's wish to maintain peace and stability in the region, diplomacy is regarded as the most effective way to deal with North Korea, and China's diplomatic efforts have manifested through both bilateral and multilateral approaches to managing the crisis. Immediately after the nuclear test, China sent a special envoy, Tang Jiaxuan, to meet with Kim Jong Il to discuss the matter. However, the meeting could not secure a commitment from North Korea on the future nuclear test. Kim Jong II said that he could not guarantee that North Korea would not "take further action if it faces bigger and more unfair pressure from outside" (The Chosun Ilbo, 2006). China was able to mediate between North Korea and the United States, and the meeting between the two sides was organised not too long after the test.

There seem to be two things that need to be taken into consideration: China's preference for playing the role of a mediator, and China's priority of preserving the peace and stability of the Korean peninsula. China plays the role of a mediator in balancing the demands of North

Korea and the United States. For example, in an effort to buy North Korea's confidence in the talks, China offered economic assistance and a shipment of a heavy fuel oil to North Korea, with an estimated value of US\$50 million, in February 2004 (Cody & Faiola, 2004). China also attempted to tone down the US's hardline policy on North Korea, as evidenced by China's efforts to convince the United States to put aside the missile issue during the first round of the Six-Party Talks, in August 2003 (Funabashi, 2007, p. 346). China is faced with some constraints in exercising its leverage over North Korea, for example the fear that too much pressure could lead to it losing influence over North Korea, and therefore risk damaging the Sino-DPRK relationship. At the same time that China regarded dialogue and negotiation as the best solution, China also strongly supported the imposition of tougher methods against North Korea, via the United Nations. China approved the adoption of UN Security Council Resolution 1718, condemning North Korea's nuclear test of 2006. This reflected the international community's shared opposition to the test. However, China clearly opposed the use of force or any other tactic that might worsen the situation, and China still believed that diplomatic means were the best method for solving the North Korean nuclear issue (The United Nations, 2006). Nonetheless, the passage of UN Resolution 1718 angered North Korea. North Korea responded harshly, referring to it as "a declaration of war" caused by US hostility, regardless of North Korea's commitment to not using nuclear weapons in a first strike capacity, nor in its promise to not transfer nuclear technology to others (KCNA, 2006c). Kleine-Ahlbrandt (2013) writes that China's support for the UN sanctions were merely rhetorical, because China hoped the use of sanctions would bring North Korea back to the negotiation table, not undermine or weaken the regime. As such, the sanctions implemented must be proportional and moderate. With this kind of policy and perception, China was unable to fully exercise its leverage over North Korea and, at the same time, it was unable to fully commit to bolstering North Korea's security confidence. China views the North Korean nuclear issue as a matter between the United States and North Korea, and improved US-DPRK relations are key to the denuclearisation of the Korean peninsula (B. Yang, 2006, pp. 24-25).

Another important aspect of China's policy worth taking into consideration is that of US-Sino relations, which play a significant role in shaping China's policy towards the two Koreas, the North Korean problem, and the Six-Party Talks. Both China and the United States are concerned about and cautious of each other. At the same time that the rise of China as a

major power in the region is a concern for the United States, China has also developed a sense of anxiety about the US's increased power and influence in the region – a consequence of US efforts in dealing with the North Korea problem.

China's effort to resume the Six-Party Talks in 2005, following the US invasion of Iraq, is a good example to illustrate China's concerns. China seeks to use the Six-Party Talks to encourage the United States to not opt for a military invasion, as it did with Iraq, or applying the Libyan model in solving the problem of North Korea (J. Park, 2005, pp. 84-85). The United States leans towards the Libyan model for North Korea, as a result of its experience in the successful use of "the reciprocal unilateral measures¹⁹" in persuading Libya to denuclearise (Goodby & Gross, 2004; Reuters, 2007). North Korea was strongly opposed to the Libya model, because such a model was considered to be the US's pursuit of North Korean CVID.

The third round of the Six-Party Talks, in June 2004, concluded with the need for all parties to adopt "the principle of words for words, and action for action" as method to achieve denuclearisation. However, differences between the United States and North Korea started to show. While the United States showed its preference for the CVID principle, North Korea offered a "reward for freeze" proposal. The KCNA released a statement saying that,

"[Kelly] insisted on the U.S. assertion that the DPRK dismantle its nuclear programme first, thus rejecting the principles of 'words for words' and 'action for action'. He also totally denied the principle of 'reward for freeze' when he said that the U.S. has no intention to negotiate with North Koreans, there can be no reward for North Korea and the U.S. will not bring any benefit to it ... Since the start of its second term, the Bush administration has not made any trustworthy sincere effort to create conditions for the talks, persistently insisting on the assertion that the DPRK dismantle its nuclear programme first on the basis of CVID." (KCNA, 2005a)

-

¹⁹ The reciprocal unilateral measures require a commitment from all parties to carry out step-by-step actions according to the process of negotiation, until the final step of denuclearisation is reached. The United States successfully negotiated with Libya, and applied this method in persuading Libya to dismantle its nuclear programmes.

North Korea, as such, appeared to lose faith and confidence in the Six-Party Talks early in the negotiations. Although China could persuade North Korea to return for the fourth round of talks in September 2005, which resulted in the release of the September 19 Joint Statement, North Korea still held a negative opinion of the multilateral forum. North Korea reiterated that the talks had not yielded productive results thus far due to differences the members had about the steps to achieve denuclearisation (KCNA, 2005b). China was to be credited for bringing about the first breakthrough of the Six-Party Talks, in September 2005. The United States and North Korea agreed to sign the September 19 Joint Statement, even though they were not content with its contents (Kahn & Sanger, 2005). This was an attempt by China to include North Korea's "word for word, action for action" phrase in the text, and convince the United States to sign on the grounds that it had a group consensus (Kan, 2014, p. 30; S. Kim, 2005, p. 124).

The Six-Party Talks were criticised as ineffective when the talks could not bring about substantive progress on the denuclearisation of the Korean peninsula. Considering that dialogue and consultation are the most pragmatic approaches to achieve a solution to the nuclear issue, China insisted on pushing ahead with the resumption of the talks. President Xi Jinping reaffirmed China's commitment to the use of dialogue in order to achieve the denuclearisation of the Korean peninsula (Zhang, 2013). The significance of the Six-Party Talks lies not only in relation to security matters, but the Talks also play a role in US-Sino relations. At the same time that China used the Talks to contain the US reaction to the nuclear issue, China also considers the Talks an important tool for maintaining a good relationship with the United States. Progress in the Six-Party Talks is significant as the Talks support China's rise by allowing it to adopt the role of a contributor to regional stability (Thompson & Matthews, 2011, pp. 188-189).

Japan

Japan has long been involved in North Korea's denuclearisation process. Since signing the 1994 Agreed Framework, Japan has become one of the largest financiers of the KEDO. According to Auslin (2011, p. 199), there are three main issues that influence Japan's foreign policy: participation in the multilateral dialogue, the abduction issue, and the denuclearisation

issue. When the idea of the Six-Party Talks was first proposed, Japan strongly supported it. One of the reasons for this was because Japan wanted to continue the US-Japan-South Korean trilateral relations, strengthened as a result of the TCOG (Hughes, 2006, p. 165). The United States also counted the inclusion of Japan in the Six-Party Talks as important, because of its perception that the path to striking a deal with North Korea on denuclearisation required Japan's financial contribution (Niksch & Perl, 2007, p. 6).

Japan's role in the region, nonetheless, is said to have been sidelined. Japan was not a member of the Three-Party Talks, nor did Japan take a front seat in the Six-Party Talks. Japan was constrained by domestic pressure to see progress made on the abduction issue. The abduction issue became a priority of Japan after the DNA testing incident²⁰, in November 2004. The policy of the Koizumi government was influenced by the Japanese public, who agreed that the issue of Japanese abduction, not North Korea's nuclear development was what they were most concerned about (H. N. Kim, 2006a, p. 174). The Japanese government started to raise the abduction issue at the Six-Party Talks because it wanted to ease the anger and anxiety of the Japanese public (S. M. Kang, 2006, p. 88). Koizumi's North Korean policy was a combination of engagement through the multilateral platform, and the "tougher measures" of pressure and sanctions, which was in line with the Bush administration's policy (Bloomberg, 2003). Japan's actions raised some criticisms from other members of the talks. Delegates from other nations expressed their dissatisfaction with Japan at the Six-Party Talks, in July 2005, on the grounds that the Talks were bound to fail because the abduction issue distracted from the primary focus (Xinhua News Agency, 2005). The abduction issue has played a considerable role in shaping Japan's North Korea policy. Japan's active push for progress on this issue has affected the role Japan can play in the multilateral forum. Japan does not want to lift sanctions on North Korea and implement the agreement of the Six-Party Talks, because no concrete solution to the abduction issue has been presented (Ministry of Foreign Affairs of Japan, 2007).

²⁰ In December 2004, North Korea returned the remains of one of the abductees to Japan. However, DNA testing showed that the remains belonged to a different person. The incident angered Japan, who accused North Korea of being insincere about solving the issue.

The Six-Party Talks of 3 October 2007 concluded with the Second-Phase Actions of the Joint Statement. The statement outlined the "action for action" principle²¹, through which North Korea receives something in return for taking action towards dismantling its nuclear programmes (U.S. Department of State, 2007). It turns out that Japan was not party to the commitment. Japan refused to take part in providing energy assistance until North Korea made progress in resolving the abduction issue (Manyin & Nikitin, 2013, pp. 5-6). North Korea showed its dissatisfaction for the other members' lack of commitment to the implementation of the October 3 Agreement, as announced through the KCNA:

"It is beyond Dec. 31, 2007, the deadline set in the Oct. 3 agreement. It is regrettable that points agreed there remain unimplemented except the disablement of the DPRK's nuclear facilities. The disablement started early in November last year and all the operations were completed within the 'technologically possible scope' as of Dec. 31. At present, the unloading of spent fuel rods scheduled to be completed in about 100 days is underway as the last process. However, the delivery of heavy fuel oil and energy-related equipment and materials to the DPRK, commitments of other participating nations, has not been done even 50 per cent." (KCNA, 2008a)

The delay to act in time showed that the action for action principle had lost its value, which made it unlikely that North Korea would commit to dismantling its nuclear programme as agreed during the Talks.

Japan's use of sanctions as a means to pressure North Korea on abductions upset North Korea. Japan's refusal to provide North Korea with energy assistance (as promised under the Six-Party framework) made North Korea criticise Japan for being a bad actor in the multilateral forum. Japan's emphasis on the abduction issue in its affairs with North Korea greatly affected North Korea's perception of Japan, and the Six-Party Talks framework as a whole. As part of the February 2007 Six-Party agreement, the United States would take North Korea off "the U.S. list of terrorism-supporting countries" (Niksch & Perl, 2007, p. 4).

²¹ The October 3 Agreement, or "the Second-Phase Actions for the Implementation of the September 2005 Joint Statement", was issued on 3 October, 2007. North Korea is compelled by the Statement to declare all of its nuclear programmes by 31 December 2007. In return, North Korea receives heavy fuel oil, and a commitment from the United States and Japan to begin the process of normalisation of relations.

However, Japan was strongly opposed to North Korea being taken off the list, believing that it would undermine attempts to make North Korea account for the abductions (Alford, 2008). Nevertheless, North Korea was taken off the list. This decision by the United States had an impact on the US-Japan alliance, because the Japanese government was disappointed with the United States for failing to cooperate on solving the abduction issue.

The abduction issue also affects the progress of the Six-Party Talks. The inclusion of the abduction issue in the Six-Party Talks overloads the agenda, and also draws the forum's attention away from its main objective (Interview Prof. Chung, 2011). Japan upholds its position by linking the abduction issue and North Korea's nuclear disarmament. Prime Minister Taro Aso said that, "unless progress is made in Japan-North Korea relations, including on the abductees issue, then we will not participate in the economic and energy aid under the Six-Party Talks, and there is no change in that stance" (The China Post, 2008). The refusal to take part is an attempt by Japan to increase international pressure on North Korea, because, without Japan's economic commitment, the other four nations would have to shoulder Japan's portion of the financial burden (Izumikawa, 2011, p. 41). North Korea voiced its dissatisfaction in return, stating that not only had Japan never tried to implement the agreed-upon actions, but also that it used the Six-Party Talks as a platform to realise its own objectives (KCNA, 2008c). Japan's role in the multilateral forum is increasingly questioned, especially when there appeared to be the possibility for a breakthrough in restarting the Talks after a long break in 2008. It is assumed that Japan may follow United States policy in dealing with North Korea through the multilateral framework. But, with the priority Japan gives to the abduction issue, Japan's role will be diminished. It is said that Japan is constrained by its public's strong reaction to the abduction issue, and that Japan would find it difficult to provide any aid to North Korea if a deal is not reached at the Six-Party Talks.

South Korea

By the time of the second nuclear crisis, in 2002, Roh Moo Hyun was president of South Korea. His administration adopted the Sunshine Policy of his predecessor, Kim Dae Jung, and renamed it the Policy for Peace and Prosperity. Roh aimed to use dialogue to solve the

North Korean nuclear issue, and also to build a peaceful environment on the Korean peninsula. The Policy for Peace and Prosperity was structured on the perception that it was more likely for North Korea to change its behaviour as it was drawn closer to the international community (BBC News, 2003). The policy played a significant role in shaping South Korea's strategy towards North Korea, both in the bilateral dialogue of North-South relations and also the multilateral dialogue of the Six-Party Talks. The Roh government planned to extend invitations for further dialogue with North Korea, believing that inter-Korean relations held the key to reaching denuclearisation. The more relaxed the atmosphere, the easier it could be to convince North Korea to give up its nuclear weapons (H. N. Kim, 2006c, pp. 42-43).

President Roh's strong belief in dialogue and engagement to some extent caused some difficulties for the role South Korea played in contributing to the goal of nuclear disarmament, as outlined in the Six-Party Talks, and also caused some strain on the US-ROK alliance. In contrast to the US's objective of CVID, South Korea supported North Korea's request for a peaceful use of nuclear energy. The Roh government pointed out that CVID was "unfair" to North Korea, given that the use of nuclear power for peaceful activities, such as in the medical and agricultural fields, was rather useful and that, as a result, it should not be banned (Moon, 2008a, p. 86). President Roh expressed his belief that North Korea developed nuclear weapons solely as a result of its security concerns, and that the regime would fully comply with the denuclearisation plan only when it was granted with a security guarantee (Korea JoongAng Daily, 2004). Roh's statement was criticised at home, and also raised concerns that it might jeopardise the US-ROK alliance, which had weakened as a result of Roh's opposition to the US's tough stance and sanctions against North Korea. Armstrong (2009, p. 237) criticised Roh's policy as appeasing North Korea, and that the Roh government's core concern was for maintaining engagement with North Korea rather than focusing on resolving the North Korean nuclear problem. The policy constrained South Korea to act promptly in responding to the missile test of 2006 (BBC News, 2006). Although President Roh condemned the missile launch, he stressed his disapproval for the use of pressure and sanctions imposed on North Korea by other parties (B. M. Lim, 2006). The Roh government structured its policy towards North Korea on the prospect for improved inter-Korean relations, and the belief that inter-Korean relations could play a beneficial role, contributing to a resolution to the nuclear crisis (H. J. Park, 2006, pp. 110-111). Roh's policy created a gap in the US-ROK relations, because the United States at the time preferred the use of pressure and sanctions.

South Korea provided North Korea with half a million tons of rice and other aid in 2003 and 2004. The assistance was supplied as part of a deal to convince North Korea to commit to nuclear disarmament and stay in the Six-Party Talks. The between-the-lines meaning of this action was that the Roh administration wanted to create leverage over North Korea, by making North Korea reliant on South Korea (C. N. Kim, 2005, pp. 15-16). South Korea's provision of energy supplies to North Korea in 2005 was believed to prolong the Six-Party Talks. North Korea agreed to attend the fourth round in July 2005. The Talks continued, and reached a breakthrough in September 2005.

For President Roh, inter-Korean relations and the Six-Party Talks were connected. The problem of North Korea can be solved through the Six-Party Talks, he believed, but it needed inter-Korean dialogue as a supplement to make it function more effectively (Ministry of Patriots and Veterans Affairs of South Korea, 2007). In attempting to reach for peace, South Korea played the role of a facilitator in bridging the gap between the United States and North Korea; and the role of a mediator in assuaging the differences between members of the Talks. However, this particular role of South Korea's disappeared after President Lee Myung Bak came to power and adopted a policy of conditional engagement, under the so-called "Denuke, Open 3,000 initiative". Lee Myung Bak's policy destroyed the positive atmosphere of inter-Korean relations encouraged by the previous administration. North Korea was suspicious of South Korea's intentions, particularly after the Lee administration announced its strategy of grand bargaining, which outlined the condition that economic assistance would only come after nuclear dismantlement (Armstrong, 2010, p. 173). The Lee government also did not play an active role in the Six-Party Talks, and instead passively followed the lead of the United States (C. I. Moon, 2012). Lee Myung Bak's belief was that the involvement of the international community was essential in promoting and achieving the process of denuclearisation (S. U. Nam, 2007). Structured on this vision, the Lee government persuaded the United States not to compromise by taking North Korea off "the U.S. list of terrorismsupporting countries", and also approached Japan to cooperate in increasing pressure on North Korea (S. W. Park, 2010). The hardline policy of Lee Myung Bak damaged the Six-Party Talks. The policy of no-compromise prevented the conditions for the resumption of the Six-Party Talks, as seen after the Cheonan sinking incident when South Korea demanded an

apology as a precondition to dialogue (BBC News, 2011b). The precondition did not bring about a positive outcome for the future of the Talks. Although North Korea regarded dialogue and negotiation as effective ways to end confrontation and solve some difficulties, it insisted that the talks should be organised without any conditions (KCNA, 2011).

Russia

The role of Russia on the Korean peninsula has been minimal, as it was not involved in the denuclearisation process from the 1994 Agreed Framework to the early stages of the multilateral negotiations of the Three-Party peace talks. In addition, Russian exclusion from the Four-Party talks, held between November 1997 and August 1999, was a bitter experience that Russia did not want repeated. This made the Putin government attempt to approach both China and the United States during the groundwork stages for Russian inclusion in the Six-Party Talks.

Russia considered the multilateral framework conducive to its national interests at a time when Russia's power was in decline. Russia entered the Six-Party Talks with the "honest broker strategy" with the hope that, by supporting the multilateral framework on nuclear non-proliferation, it could return to the front of the stage at some point in the future (Funabashi, 2007, pp. 187-188). What Russia preferred was that the Korean peninsula be a nuclear-free zone and that the North Korean nuclear issue should be peacefully resolved through the use of dialogue and negotiation. By engaging all regional players, the Six-Party Talks would also prevent any single regional player from gaining more power and influence and from dominating the Korean peninsula (Joo, 2007, pp. 134-135).

According to S. Kim (2007, p. 25), Russia believed that the multilateral forum would boost North Korea's confidence during a time when the regime did not feel at ease living in an environment occupied by multiple, powerful regional players. As such, Russia appeared to play an active role in the Six-Party Talks, and the forum served a Russian strategic interest of attempting to secure the peace and stability of the region. The result of this was not only to differentiate Russia's position towards North Korea from that of the United States, but Russia's stance is sometimes viewed as appeasing North Korea. Russia made it clear through an official statement at the Six-Party Talks held in February 2004 that Russia would "do

everything necessary for solving the nuclear problem along with ensuring the security of the DPRK" (Nautilus Institute for Security and Sustainability, 2004). In addition, and in contrast to the US preference for CVID, Russia supported North Korea's demand for "compensation for freezing the nuclear programmes", referred to by the Russians as "corresponding measures" (The Korea Times, 2004). Russia emphasised that compensation was important to North Korea, whose economy and society were in need of development (Takeda, 2006, p. 202). Nonetheless, Russia did not totally agree with North Korea's request for peaceful nuclear energy development. Since the September 19 Joint Statement lacked a clear description of a specific timeline, it offered Russia an opportunity to set the conditions regarding this request. Sergei Lavrov, Russian Foreign Minister, said two days after the September 19 Joint Statement was signed, that the LWR would be provided to North Korea only after there was a total dismantlement of its nuclear weapons programme (RIA Novosti, 2005).

Russia strongly supported the Six-Party Talks, believing that the forum would bring about peace and stability in Northeast Asia. Sanctions and pressure were options that Russia was against, believing that the more North Korea was pushed, the higher the chance North Korea would strengthen its military capability to counter any external threat (Vorontsov, 2011). Russia prefers the use of negotiation in solving the North Korean problem, knowing well that North Korea's primary focus is to have bilateral talks with the United States. Although Russia was an enthusiastic partner in the Six-Party Talks, its role was marginalised. Russia wants to achieve the denuclearisation of North Korea through peaceful means. This desire influences Russia's policy on North Korea, and the role Russia plays in the Six-Party Talks. Russia's disagreement over the use of sanctions and pressure against North Korea is criticised as being "supportive" of North Korea (Blank & Kim, 2012, p. 265). President Putin once voiced his concern that the Six-Party Talks would likely be undermined if North Korea is "provoked" to withdraw from the negotiations, or chooses to "respond in an inappropriate manner" (Russia & CIS Diplomatic Panorama, 2006). Ha and Shin (2006, p. 25) say that Russia plays a "buffering role" in preventing the United States from launching a military attack against North Korea in order to bring about a change in the North Korean regime. However, Russia's policy has changed over time, especially when the situation starts to show signs of becoming worse, including increases in tension from both the international community and also North Korea. For example, regarding North Korea's request for LWRs:

Russia's reluctance to comply with North Korea's request for the LWR was later changed from fear that the progress of the Six-Party Talks would be undermined. Russia admitted that commitment by all parties was key to the success of the Talks before committing to the LWR request, saying it "conforms to the agreements reached by the sides and they must be implemented" (BBC News, 2005).

Russia clearly understood North Korea's security concerns. A security assurance was the first step towards denuclearisation, particularly when North Korea developed some fear after observing what happened to Gaddafi of Libya (Toloraya, 2011). Russia also wanted the Six-Party Talks to resume after a long break, since 2009. In 2011, Russia secured a positive response from North Korea on future attendance at the Six-Party Talks during a summit meeting between Kim Jong II and Dmitry Medvedev. However, the resumption of the Six-Party Talks was beyond Russia's control. Russia was unable to convince North Korea to drop its preconditions and return to the Talks, or the United States and South Korea to drop their suspicions of North Korea's sincerity in denuclearisation. Regardless, a summit meeting between the two leaders was organised with the main focus on economic aspects, because Russia wanted to push ahead with a gas pipeline project, which would connect the economies of the two Koreas with Russia's (BBC News, 2011a). The role of Russia continues to be sidelined, despite its enthusiasm for engaging through the talks.

7.2 North Korea's Threat Perception and the Six-Party Talks

Bilateral dialogue with the United States is the main policy goal of North Korea. North Korea entered the Six-Party Talks with the hope of having a bilateral dialogue with the United States on the side, within the framework (KCNA, 2003e). It turned out that the Six-Party Talks could not lower North Korea's threat perception. North Korea revealed its suspicions of the United States in the first round of the talks, in August 2003. North Korea reiterated that, "we can dismantle our nuclear programme if the U.S. makes a switchover in its hostile policy towards us and does not pose any threat to us" (KCNA, 2003d). The September 19 Joint Statement is considered one of the major achievements of the denuclearisation process, but a problem later arose when there were discrepancies in how the parties went about

implementing the statement's provisions. The failure mainly stemmed from the differences between North Korea's and the United States' interpretation of "total dismantlement" in relation to North Korea's nuclear programme. While the United States defined this phrase as CVID, North Korea demanded delivery of a LWR from the international community, claiming that it has the right to use nuclear power and technology for peaceful purposes (Kahn, 2005b). Niksch (2005, pp. 15-18) considers North Korea's demand for a LWR "a stalemate strategy" in order to buy time for itself to denuclearise.

By the end of the fourth round of the Six-Party Talks, in September 2005, no progress had been made on the LWR issue. The issue became more complicated after the United States suggested each country should interpret the Joint Statement's LWR provision clause²² in their own way (Kahn & Sanger, 2005). The Joint Statement did not specify the timeframe for the provision of the LWR. The lack of clarity in the statement made both North Korea and the United States issue their own statements after the Talks were concluded. The United States' chief negotiator, Christopher Hill, declared in Beijing on 19 September 2005 that, "the appropriate time will only come when the DPRK has promptly eliminated all nuclear weapons and all nuclear programmes" (U.S. Department of State, 2005a).

Despite the involvement of all players in the region, the Six-Party Talks are mainly driven by the United States and North Korea, and a failure to reach a compromise between the two greatly affects the future of the forum. It is clear that the two have different agendas when attending the talks, as has been observed by Prof. Paik (2011):

"North Korea presses hard for the agenda of peace settlement and how to get diplomatic normalisation with the United States. These problems are just part of the North Korean nuclear problem. The Americans recognise North Korea's need and objectively speaking they try to press North Korea for something in return. The United States says that North Korea should show seriousness that they are denuclearising because there is a repetition of things that already

²² From the September 19 Joint Statement: "The other parties expressed their respect and agreed to discuss, at an appropriate time, the subject of the provision of light water reactor to the D.P.R.K." (U.S. Department of State, 2005b). This clause was included in the Statement, against the US's wishes, by China who wished to achieve a breakthrough. The word "appropriate" used in this clause was China's effort to find a compromise between the United States and North Korea (Funabashi, 2007, p. 401).

discussed. This is the American kind of understanding. But North Korea understands that the process fails because of the US government's failure to implement it faithfully." (Interview Prof. Paik, 2011)

In a 19 September 2005 statement issued by Christopher Hill, the United States set conditions for the normalisation of relations with North Korea, including progresses on several non-nuclear-related issues. Instead of resolving the problem in a manner preferred by the hardliners in Washington, Hill's statement influenced North Korea to continue with its nuclear development, which lead to a nuclear test in 2006 (Chinoy, 2011).

The nuclear test of 2006 was "a tactical success" for North Korea, as it resulted in a more flexible US position, and American willingness to hold a bilateral negotiation before the next round of the Six-Party Talks (S. Snyder, 2007, p. 165). Nonetheless, the Six-Party Talks in December 2006 made little progress and did not produce any substantive outcome. One of the causes for the lack of progress was the BDA issue. The BDA issue was a serious concern for North Korea. The US sanctions against the BDA can be considered effective. The sanctions not only blocked the financial activities of the BDA, but also influenced other financial institutions in other countries to stop their financial dealings with North Korea from fear of reprisals (Lague & Greenlees, 2007). North Korea connected the BDA sanctions to its nuclear weapons. Kim Gye Gwan, North Korea's chief negotiator, demanded the United States lift the sanctions before any discussion on the nuclear dismantlement:

"The U.S. is using a tactic of both dialogue and pressure, carrots and sticks. We are responding with dialogue and a shield, and by a shield we are saying we will further improve our deterrent." (Kahn, 2006)

The United States and North Korea continued with bilateral talks in January 2007, in Berlin. The Berlin talks showed the US's willingness to hold bilateral negotiations with North Korea. The Berlin talks also changed the pattern of the Six-Party Talks, because the bilateral negotiations came before, not during or after, the Six-Party Talks. Prof. Paik (2011) elaborates on the pattern of negotiation, saying,

"We are interested in whether how soon the Six-Party Talks would be resumed. But what is more important is what is going to take place before between the United States and North Korea. There are two steps of negotiation. The breakthrough or the agreement between the United States and North Korea will be approved or endorsed by the Six-Party Talks. Then we have full official agreement over there. That is the process of it." (Interview Prof. Paik, 2011)

This pattern makes the Six-Party Talks more effective, because the talks can further discuss the details agreed upon during the bilateral negotiations, rather than seeking to resolve the deadlock from previous meetings (Buszynski, 2013a, p. 131). The new approach was rather "realistic", as the United States dealt with North Korea on its own terms, rather than because of the US's position on denuclearisation (Kwak & Joo, 2007, p. 6). Prof. I. H. Park (2011) says that,

"The best solution is a kind of simultaneous progress between multilateral approach such as the Six-Party Talks and North Korea-US bilateral approach. I am not saying that the Six-Party Talks is the best option but it is the only option to continue communication between North Korea and the United States and the international society. What happened insides of the Six-Party Talks is a kind of bilateral negotiation between the United States and North Korea." (Interview Prof. I. H. Park, 2011)

The Agreement reached in Berlin facilitated the organisation of the Six-Party Talks in February 2007. The outcome then was the release of one of the more significant statements in the history of the talks, the Initial Actions for the Implementation of the Joint Statement or the February 13 Agreement²³ (The Washington Post, 2007). Although the February 13 Agreement set out a specific timeframe for each step of action, it did not provide clear details on the type of nuclear projects to be verified and dismantled (C. I. Moon, 2012).

After the February 13 Agreement was signed, North Korea's threat perception still had not lowered, and North Korea remained committed to strict adherence to the term "actions for actions" as a precondition for denuclearisation. President Bush started to adopt a more flexible approach towards North Korea, particularly after the February 13 Agreement. The

_

²³ The February 13 Agreement outlines two phases for denuclearisation. The first phase required North Korea to close its nuclear facility in Yongbyon. In return, North Korea would receive heavy fuel oil and other economic assistance. The actions must be completed within 60 days. The second phase stated that North Korea must disable the Yongbyon nuclear site and declare all of its nuclear programmes. Other members would provide North Korea with heavy fuel oil in exchange.

United States not only agreed to lift the sanctions on the BDA, but also sent Christopher Hill to North Korea, which showed that the United States recognised North Korea as a negotiating partner (Choe, 2007b). Nonetheless, the implementation of the February 13 Agreement was criticised as progressing at a slow pace. There was a delay from North Korea in shutting down the Yongbyon nuclear site within the timeframe, because North Korea demanded the settlement of the BDA issue first (J. A. Yang, 2007). In addition, difficulties in implementing the agreement arose from the US's and North Korea's differences in interpretation of the term used. While North Korea understood that it had to temporarily suspend its nuclear facilities, the United States stressed that the term "disablement" in the agreement referred to a total dismantlement (Choe, 2007a).

Regardless of the progress and breakthrough, the Six-Party Talks were later affected by a change in the administrations of both the United States and South Korea. In addition, progress in the Six-Party Talks were stalled by Japan's refusal not to take part in implementing the Agreement, and Japan's strong request for progress on the abduction issue. North Korea also needed to take some responsibility for the failure of the Six-Party Talks. North Korea failed to submit a full declaration of all of its nuclear programmes by the deadline stipulated in the October 3 Agreement (I. Kim, 2009, p. 93). North Korea's delay was caused by technical problems and for political reasons. S. Y. Kim (2008, p. 109) says that, although there was a technical issue involved in the process of disablement, the delay was mainly motivated by North Korea's dissatisfaction with other parties for not delivering heavy fuel oil within the schedule. Prof. Koo (2011) had this to say about the failure of the Six-Party Talks:

"The US government argues that North Korea has critical responsibility in cancelling the agreement between the two countries. However, North Korea argues that the United States has been false in implementing the Agreed Framework and the September 19 Joint Statement, which is made by the Six-Party Talks in 2005. And by looking at a lot of agreements between the United States and North Korea and also amongst the six parties; however, any agreements have not been abided by the six parties. So, usually western people think that only North Korea violates the agreements. I do not agree with that." (Interview Prof. Koo, 2011)

North Korea's nuclear ambitions grew at the beginning of 2009, when they conducted a missile test in April, followed by an underground nuclear test in May. In response to the UN Security Council's condemnation of North Korea's missile test launch, North Korea declared,

"The DPRK will never participate in such Six-Party Talks nor will it be bound any longer to any agreement of the talks as they have been reduced to a platform for encroaching upon its sovereignty and forcing it to disarm itself and bringing down its system" (KCNA, 2009d)

Despite China's efforts to resume the talks, the Six-Party Talks were at a standstill. Obstacles to the resumption of the talks included the policies of each member state. The Obama administration adopted a policy of strategic patience, waiting for North Korea to return to the table. Japan prioritised the abduction issue over denuclearisation. South Korea's policy was conditional engagement with denuclearisation as a prerequisite for engagement. Lastly, Russia's main focus was the economy and, at the same time, it lacked the political power to influence North Korea.

7.3 Future Prospects for the Six-Party Talks

With the lack of progress in inter-Korean dialogue and in US-North Korean bilateral relations, the Six-Party Talks turned out to be the only available diplomatic channel. Prof. Moon (2011) said of the Six-Party Talks as a means for solving the North Korean nuclear problem,

"The Six-Party Talks is the most effective way in solving the problem of North Korea. This is because since the Second World War, it is for the first time that the United States, Russia, China, and North-South Korea to sit together and call for some kind of negotiation mechanism. The United States alone does not have any finance and resources to revert North Korea. Therefore, the United States needs help from South Korea, China, Japan, and Russia." (Interview Prof. Moon, 2011)

China showed its enthusiasm for reviving the Six-Party Talks, especially with rising tensions in the peninsula following the Cheonan incident in March 2010. The resumption of the Six-Party Talks was on the agenda for meetings between Kim Jong II and Hu Jintao during Kim's two visits to China, in May and August 2010. Kim Jong Il affirmed the significance of the Six-Party Talks (KCNA, 2010a). Fully aware of North Korea's demands, China proposed a plan to organise a US-DPRK bilateral meeting before the Six-Party Talks were convened (Harlan, 2010a). China sees engagement as a method to realise denuclearisation without a sudden change to North Korea, and also a platform that could help China improve its own bilateral relations with the United States (Thompson & Matthews, 2011, pp. 187-188). However, China's efforts failed to revive the Six-Party Talks, and China was criticised for its inability to use leverage to influence North Korea to return to the negotiating table. It was clear that the main players of the talks - namely the United States, South Korea, and North Korea – could not reach a compromise on how best to resume the Talks, particularly when tensions ran high because of North Korea's provocations. South Korea connected the provocations with the resumption of the Six-Party Talks, demanding an apology from North Korea for sinking the Cheonan prior to the talks (S. Kim, 2010). Although the demand was later toned down, South Korea remained tough, and vowed to push back the Six-Party Talks until concrete results were reached (McDonald, 2010). Prof. D. H. Han (2011) expressed his opinion on the prospects of the Six-Party Talks solving the North Korean nuclear problem thus:

"I think the resumption of the Six-Party Talks itself does not change anything. The more important thing is main content. For example, the United States says North Korea should resolve the uranium enrichment programme before the resumption of the Six-Party Talks. But North Korea says the issue should be discussed at the Six-Party Talks. Originally the Six-Party Talks is based on the plutonium nuclear weapons not the uranium." (Interview Prof. D. H. Han, 2011)

Following the Yeonpyeong shelling incident, the United States and Japan joined South Korea in pressuring for a substantial concessions from North Korea (BBC News, 2012). North Korea had requested that the Talks resume without preconditions, but North Korea continued to face demands from the United States for a commitment on nuclear disarmament first. The trilateral cooperation between the United States, Japan, and South Korea was created as a result of the denuclearisation process. It turns out that the allies shared the same perspective

as the United States. The three countries affirmed the need for North Korea to take the talks seriously, to make them an effective negotiation forum (G. L. Moon, 2012).

Talks have yielded some substantive progress towards the peace and security of the region. First, the talks function as a platform for the organisation of bilateral negotiations between the United States and North Korea, as a sideline within the Six-Party framework. Without the Six-Party Talks, it seems unlikely that US-North Korean bilateral dialogue could be convened, particularly with the Obama administration's adoption of the policy of strategic patience. Secondly, the Six-Party Talks also brought China closer to the United States, as both cooperated in attempting to achieve the denuclearisation of the Korean peninsula. Thirdly, the United States plays the role of mediator by connecting South Korea and Japan as part of US-ROK-Japan trilateral relations.

South Korean-Japanese relations were considered the weakest among the bilateral relations in the region, and they had been constrained mainly by Japan's wartime legacy, and also by the dispute over the possession of islets in the Sea of Japan. Nonetheless, it is claimed that the key to reviving the Six-Party Talks lies in two things: Inter-Korean dialogue and US-North Korean bilateral relations. Inter-Korean dialogue, which has lacked any form of progress since 2008, could lead to a breakthrough in the stalemate. Inter-Korean relations are just one of the ways to ease regional tensions because, as the regional atmosphere relaxes, so the chances for resuming the Six-Party Talks would improve (Interview Prof. Y. Y. Kim, 2011). The adoption of the Lee Myung Bak government's policy of grand bargaining pushed North Korea further into isolation. The grand bargaining initiative was structured on the "action for action" principle, and outlined the steps North Korea must take in order to receive compensation from the other five regional powers until it achieves a complete denuclearisation (Ministry of Foreign Affairs of South Korea, 2010). Although the policy outlines the benefits North Korea would receive if it takes steps towards denuclearisation, it also showed a tougher approach for dealing with North Korea. Since the process requires close discussions and consultations among the five participants, it means stronger cooperation between all five parties. Besides, North Korea is also required to make the first move in dismantling its projects before receiving any concessions (The National Institute for Defense Studies, 2010, p. 89). North Korea strongly criticised the grand bargaining policy because it shares the same principles as the "Vision 3000" policy, which sought the end of the North Korean regime (KCNA, 2009f). The grand bargaining did not help in reducing North Korea's sense of insecurity. This is because the policy clearly outlines that North Korea's bilateral relations with the United States will not be part of the concessions (Ministry of Foreign Affairs of South Korea, 2010).

On the other hand, the relationship between the United States and North Korea also played a considerable role in contributing to the progress of the Six-Party Talks. It is clear that the policy adopted by the Obama administration does not facilitate the denuclearisation of the peninsula. The United States was rather "sceptical" of North Korea, as seen in its demand to first see a commitment from North Korea before any steps would be taken in pushing ahead with dialogue (Bosworth & Gullucci, 2013). Obama's passive diplomacy greatly affects the progress of the Six-Party Talks, when the United States makes it clear that it will no longer use concessions as incentives for encouraging North Korea to return to the negotiation table (C. I. Moon, 2012). Prof. Paik (2011) explains the effect the Obama administration's policy has on the Six-Party Talks thus:

"If you put pressure on North Korea, North Korea will not come to negotiation table easily. They rather strengthen defence capability but at the same time North Korea will see the need to come to the negotiation table with the United States and solve the problem so that it can receive less threat from the United States. During the presidential campaign, President Obama promised to have even summit talk to solve the nuclear issue. But Obama did not take a positive step to make his promise true. North Korea was dissatisfied and does not see the need to come to the negotiation table." (Interview Prof. Paik, 2011)

The Six-Party Talks could be resumed if the parties agreed, first, to drop the preconditions they each have set, because they reflect the feelings of distrust between the parties. North Korea in particular believed that the preconditions set by the United States and South Korea are in violation of the spirit of the September 19 Joint Statement. Efforts by China alone might not be enough to persuade North Korea to return to the talks. The current circumstances do not facilitate the resumption of the Six-Party Talks, with the lack of progress in both inter-Korean dialogue and US-North Korea bilateral relations. The Six-Party Talks had already made significant progress towards denuclearisation with the release of the September 19 Joint Statement, which is widely regarded as a major breakthrough that

articulates concrete steps for disarmament. The fundamental question is how best to make all parties commit to implement the provisions listed in the Statement if the Six-Party Talks once again resume.

7.4 Conclusion

The Six-Party Talks cannot persuade North Korea to lower its deterrence level. All parties share the same impression of regional insecurity as caused by North Korea, but their interaction in the forum shows the overlap of individual national interests over the agenda of the forum. The Six-Party Talks show that there are internal structures within the main framework. They are inter-Korean relations, US-DPRK relations, and the US's alliance relationships. These are the result of the national interests that come into play when states try to interact at the international level. National interest can refer to security interest, economic interest, political interest, and identity interest. States do not want to drop their national interests during the times when they cooperate with other states. This is because there is no guarantee that national interests can be secured when the system of anarchy pushes states to act independently. The inability to overcome individual interests leads to the forum becoming ineffective. Regional powers approach the Six-Party Talks in accordance to their interests. The United States' main focus is the total nuclear disarmament of North Korea. The Japanese government wants to see progress on the abduction issue over denuclearisation, because its security is guaranteed through its alliance with the US. South Korea agrees with the need for denuclearisation, but the prospects for handling the issue depends on the policy adopted by each government. China's concern is that denuclearisation should be resolved without altering the status quo. Russia sees the Six-Party Talks as a road to regain power in the region.

The unclearly written clause about the provision of a LWR in the September 19 Joint Statement shows an inability to compromise security among members. The realists assume that there is unequal distribution of power, which causes states to compete out of their security concerns. Although cooperation has formed among the regional powers under the framework of the Six-Party Talks, their behaviour is influenced by the distribution of power. This refers to attempts by both China and Russia to prevent the United States from

dominating the denuclearisation process. This partly explains China's and Russia's emphasis on the use of dialogue and their support for the LWR request, which has led them to be criticised as appearing North Korea. The United States is also concerned about the power of North Korea. The US preference for CVID shows its security concern. Failure to unanimously interpret the clauses in the September 19 Joint Statement and to implement the February 13 Agreement has given North Korea the opportunity to continue developing a nuclear weapons programme.

The Six-Party Talks are an effort to solve the nuclear problem at the international level. However, the forum does not actually address the security needs of North Korea. What North Korea wants is bilateral negotiations with the United States, and diplomatic recognition from the United States is a condition for denuclearisation. The inclusion of other players in the forum means that the power of the United States is diminished. The multilateral platform means other players share the same responsibilities as the United States for dealing and negotiating with North Korea. Although cooperation is assumed possible when confronting an aggressor, the method does not guarantee success when there is the possibility that conflict between states could arise during their interactions (Mearsheimer, 2007, p. 80). The six parties each have different interests. The Six-Party Talks, in the end, are a platform to manage rather than solve the North Korean nuclear problem (Interview Anonymous, 2011). North Korea allows the talks to collapse because it does not see the value of attending. In order to resume the Six-Party Talks, the use of either bilateral dialogue between the US and North Korea, or inter-Korean dialogue to kick-start engagement and trust-building. The assumption of anarchy and insecurity explains states' need for power. It may be impossible for states to fully abandon their defence, but it is more likely that states would lower their defences when their security is assured at a certain level.

Chapter 8: Conclusion

The chapters in this thesis have explored both the internal and external determinants surrounding the North Korean question with the primary objective of finding answers to the problem of whether North Korea's deterrence strategy can be called rational or irrational as international security studies and international strategic studies understands these terms. This concluding chapter has three purposes as set out at the end of Chapter 2: it will summarise the answers to the five analytical questions; it will evaluate the two conceptual questions; and finally it will evaluate the two hypotheses.

Analytical Questions

In an attempt to develop answers for the hypotheses of the thesis, the following analytical questions were proposed:

- 1. What are the internal determinants that shape the threat perception of North Korea? To what extent is the high level of threat perception a product North Korea creates in order to keep the legitimacy of its regime and the power of the leader?
- 2. How has the relationship of the two Koreas played an effect on the North Korea's perception? Are all the military and other forms of provocations for deterring South Korea or for purposes not connected to deterrence?
- 3. How has the interaction of the major powers in the region contributed to the North Korea's decision making on the deterrence level? Why have the major powers not been able to reassure or deter North Korea?
- 4. What have been the primary objectives of the US strategy towards North Korea? How does the US strategy for DPRK interact with broader US objectives for managing its

relations in Northeast Asia? Is the US deterrence strategy a cause of the security dilemma in the region or a means of managing it?

5. Having failed to either co-opt or deter DPRK from its deterrence strategy on a bilateral basis the regional powers proceeded to a multilateral process - the Six Party Talks - in order to achieve denuclearisation on the peninsula. What does the interaction of the concerned parties in this process tell us about the limits of multilateral diplomacy as a route to normalisation on the peninsula? Is there any prospect that multilateral engagements can persuade DPRK to lower its deterrence threshold?

Having concluded a theoretical and empirical examination of the North Korean question this thesis would answer these questions in the following ways:

Question 1

Chapter 3 reviewed North Korea's background, regime structure, ideology and political system. The review showed a connection between the internal characteristics of the regime, its change over time and the deterrence level. In the early stage of the Cold War the first leader of the DPRK Kim Il Sung focused on building up the nation. Kim Il Sung introduced communism into North Korea while maintaining features of the traditional structure of Confucianism. Although Confucianism talks about the respect the younger should pay to the older, the introduction of the Juche ideology and the personality cult of the leader enhanced the strength of Kim Il Sung's leadership role and power. The combination of communism and the leadership cult of personality form the unique system of North Korea so that the leadership's security and the regime's security are united. The connection of these two things makes it very difficult for North Korea to adjust its internal structure when change could possibly lead to a total collapse of the overall system. The political system of North Korea becomes fused with the national identity of North Korea.

North Korea has from the very beginning a security concern about its future survival. The fear of being attacked by the United States and being overtaken by South Korea remains throughout the post-Cold War period. The loss of a primary ally with the fall of the Soviet Union influences North Korea to strive to maintain independence on its own. A sense of nationalism is highlighted because North Korea wants to keep its own identity against all changes in the international system. This identity demand of North Korea influences the behaviour of the state. North Korea remains highly defensive and the nuclear strategy reflects the defensive mechanism of North Korea. Identity also plays a role in structuring the relationship North Korea has with other countries. The mixture of various kinds of political elements makes North Korea has a rather unique behaviour that is different even from the other socialist countries. Some of the elements such as the dictatorship system and the authoritarianism produce the irrational actor image in the eyes of the international community (Interview Prof. Paik, 2011). So North Korea's identity forms part of the relationship structure it has with other countries.

North Korea is sensitive whenever its security is challenged. The post-Cold War structure clearly shows an imbalance of power between North Korea and other countries, which is especially affected by the failure of its ally the Soviet Union. Nuclear weapons become one of the possible ways that help North Korea secure itself and rebalance the defence level with other countries. The political system of North Korea makes it rather unique but at the same time increases its vulnerability. A high threat perception results from how North Korea wants to keep its unique system and its position in the international structure as they are. The system of one-man rule requires the continuous justification of the leadership by maintaining the leader's power and dominance in the ruling system. It is usually seen that the leader is credited whenever the defensive actions are carried out. This is the way North Korea manipulates external threat for internal stability. The use of political ideology and military policy such as Kimilsungism-Kimjongilism, songun, kangsong taeguk are methods to protect the leader, the system, and the nation from challenges of internal change through power transition and external change from the new international political structure. North Korea fears external threat because it has a higher potential than the internal factor in causing change to the internal system. The strengthening of the power and influence of the military sector is for security reasons but at the same time it supports the legitimacy of the leader and the justification of the system. The level of defence operated by the regime cannot be lowered as long as the leadership feels insecure due to the possibility of change to the overall structure of the regime from outside threats. It also ties together the political leadership of the ruling family and the military leadership of the country in the same survival strategy.

Question 2

The review of the inter-Korean relations shows the entrenched competition of the two Koreas since 1950. The security dilemma arising from the competition in inter-Korean relations plays an impact on the behaviour of both countries even more than the influence of other states. This is due to the historical factor of the forced division that makes Korea a divided nation and the resulting competition to achieve reunification by different systems. Competition between the two countries is not about security alone but includes also the elimination of one state by the other under reunification. In the Cold War period both sides sought to undermine the other and competed in reaching for reunification. Although the struggle for reunification is toned down over the course of the Cold war because the United States and the Soviet Union provided guarantees against forced reunification, the issue still affects the policy and perception of the two countries. North Korea is highly affected by this issue. North Korea feels insecure because the disparity it has with South Korea in terms of economic and social developments increases in later decades. The security dilemma between the two Koreas exists throughout the Cold War period but intensifies as the power imbalance between the two countries increases. The change in international politics with the fall of the Soviet Union shifts the balance of power further between North Korea and South Korea. North Korea loses a key ally, while South Korea continues to secure its alliance partnership with the United States and even gains a new relationship with China. Inter-Korean relations and particularly the reunification issue proves a source of fear and insecurity for North Korea. North Korea fears the reunification by absorption as it becomes clear that South Korea has an advantage over North Korea in all aspects in realising the unification objective.

The post-Cold War period shows that North Korea develops a sense of fear from two factors – the reunification by absorption and the US-ROK alliance. The wide development gap North Korea has with South Korea influences North Korea's perception and state's interest. North Korea shifts its focus from achieving reunification to maintaining the status quo. The development of nuclear weapons is part of the strategy to secure these threats - resisting absorption and deterring the US-ROK alliance. Nuclear weapons provide North Korea with

the power of defence and the power to deter. Provocations are also used tactically to serve various objectives ranging from reunification to security and deterrence. Most provocations in the Cold War period are progressive moves North Korea makes in reaching for reunification. Provocations in the post-Cold War are conducted for purposes other than reunification. Some of them are to deter the US-ROK military attack as seen from the shelling of Yeonpyeong island. Some of them such as the missile launch of 2012 are to test the policy of the new South Korean government in preparing its responses. Provocations in the current period serve the security interest of North Korea but do not reflect a desire for expansion or to overturn the status quo. To achieve unification under the North Korean terms has become policy rhetoric rather than policy implementation. North Korea prefers coexistence with South Korea. Provocations sustain the security dilemma but the security dilemma has not resulted in war because the strategy of deterrence, including by nuclear weapons, has worked. As such, North Korea sees the value of provocations as making gains without changing the status quo. Provocations can lead to a result of concessions from the international community but main objective of all actions is for security and survival (Interview Dr. E. C. Lim, 2011). Provocations are conducted as a response to the security demand North Korea has with South Korea, especially during the time of an imbalance of power between the two sides.

Question 3

Chapter 5 reviews the bilateral relations between North Korea and the regional powers in the post-Cold war period and the interaction among the regional powers that influence the Korean affairs. The reviews show an impact of the politics at the regional level on North Korea's perception and decision-making. North Korea has relations with all the regional powers at the bilateral level. However, each player has different interests and objectives in their relations with North Korea. This individual interest plays a role in affecting their perception and interaction among each other when dealing with the question of North Korea. While China and Russia's interest are in economy and regional stability, Japan's interest is to settle the abduction issue. When assessing the defence level, Japan is the weakest among the three regional powers. However, Japan's security is guaranteed under the US alliance partnership and this framework indicates how the role of the United States in the region

shapes the way the regional powers view Korean affairs and drives their efforts at balancing against each other.

The interaction of the regional powers with the objective of balancing shows that they operate independent from each other: any cooperation they may have does not alter their main objective of balancing each other's influence. This independence of action was further highlighted by the new structure of the post-Cold War. China and Russia adjusted themselves to the new environment abandoning the alliance partnership with North Korea. Efforts of Russia and China to revive the relationship have not restored it to the previous level. North Korea lost the support from the allies resulting in an imbalance of power between North Korea on the one hand and the United States and its allies on the other. The new political structure forced North Korea to maintain its independence by its own effort. Nuclear weapons and other defensive measures are developed for this purpose. Although the regional players share the same viewpoint on the level of insecurity in the region arising from the nuclear problem of North Korea, cooperation among them does not reflect their ability to manage the issue at the root cause level. North Korea has long emphasised its focus on security assurances from the United States, beginning with a Peace Treaty. The regional powers may not be the main actors who can respond to this request but their interaction in counterbalancing against each other does not help to move the regional structure in the right direction. Each individual player has their own interest and different views on the level of insecurity the problem of North Korea's weapons programme has caused. As influenced by their own interest, the regional powers manage the situation to prevent it from worsening rather than solving the problem through answering the security needs of North Korea. As such, the regional powers' behaviour of counterbalancing and the shifts of policy towards Korean affairs could not be considered a sign of security assurance. North Korea could not rely on any state particularly during the time when its security has not been assured. The interaction among the regional powers makes the regional environment even more unpredictable for North Korea. North Korea struggles to survive among the interaction of regional power relations and the result of the struggle is an increase in the regime's sense of distrust. The search for independence in the hostile environment is among the rationale for the high level of deterrence of North Korea.

Question 4

Most analysts agree the United States is the key player in the problem of North Korea. The review of the US policy towards North Korea reveals the policy priority of the United States which is the containment of North Korea and the total dismantlement of North Korea's nuclear weapons. Methods to achieve this objective vary from administration to administration. The power of the United States in Northeast Asia is maintained after the fall of Communism through the alliance partnership the United States has with its Asian allies. The US security guarantee is considered essential to Asian allies particularly during the time of threat and insecurity. This is because the military capacity of the countries in Asia in the US alliance system relies heavily on the United States. An example is the military structure of South Korea, which is described as:

"The US security guarantee is still important to South Korea. Historically speaking, South Korea is hugely dependent on the US forces. The United States invest their air and naval powers on the Korean peninsula. Instead we have invested our manpower and army powers. Military structure itself reflects this kind of imbalance of US-Korea relationship." (Interview Prof. D. H. Han, 2011)

The post-Cold War structure influences the policy and perception of the United States. The new structure sees the rise of new nuclear states which the United States perceives as posing a threat to its own position and the system of regimes such as the Non-Proliferation Treaty that it operates. There are various terms used in describing these states but the term 'rogue state' is often used after 2000. Rogue state refers to states that have connections to terrorism or attempt to develop weapons of mass destruction and North Korea is categorised in this group according to the US standard (Interview Prof. I. H. Park, 2011). This perception affects the way the United States approaches and deals with the problem of North Korea. It refers to the US judgement without taking into consideration the background or the situation of North Korea (Interview Prof. D. H. Han, 2011; Interview Prof. Y. Y. Kim, 2011). The United States views North Korea from this perspective of seeing North Korea as an irrational and dangerous country. The unique political system and the characteristics of the regime make it difficult for the United States and the international community to clearly understand North Korea and its behaviour. This gap in understanding of the North Korean regime and the

negative perception from the idea of rogue state form part of explanation for the failure to achieve effective negotiations.

The United States approaches and deals with North Korea through its own understanding and policy preference. Regardless of the methods in achieving total nuclear dismantlement, the United States bases its longterm policy for the North Korean on regime collapse. The anticipation of the regime collapse is a prediction that rather lacks credibility and also shows a lack of knowledge of the United States on the system and nature of North Korea (Interview Prof. Y. Y. Kim, 2011; Interview Prof. Paik, 2011). The US perception of North Korea is further changed by the events of September 11 2001 which heightened the United States concern with the potential of rogue states to mount attacks on the US. After 9/11 the United States named North Korea as part of an axis of evil. The axis of evil term is an extension of the idea of rogue states and shows the evil nature of North Korea by claiming the regime's association with global terrorism (Interview Prof. Moon, 2011; Interview Prof. I. H. Park, 2011). References to North Korea as either rogue state or axis of evil shows the negativity of the United States to North Korea and indicates the US objective of removing the North Korean problem through regime change. The US hardline policy elevates the sense of insecurity of North Korea. North Korea fears an immediate US attack for the first time since the end of the Cold War period and North Korea's demands for security assurances are further away than ever. North Korea perceives the US extended deterrence to its Asian allies as a security threat and the security dilemma between North Korea and the United States is very intense in the years after 9/11. North Korea seeks to reach some equilibrium with the United States through nuclear development in the hope of gaining the power of deterring attack and forcing the United States to negotiate.

Question 5

The Six-Party Talks are the multilateral formula set up as an alternative to solving the North Korean nuclear problem. The forum is initiated on the US belief in the failure of the bilateral negotiation. The involvement of the regional powers in the negotiation process is the way the United States aims to share its responsibility and to use the international cooperation in pressuring North Korea to denuclearise. The regional powers share the same view that the North Korean nuclear problem causes regional instability. However, their cooperation in the

Six-Party Talks could not bring about the denuclearisation of the Korean peninsula. The Six-Party Talks clearly show differences of interest among the members. Each individual has their own interest that they want to secure and it turns out that members concern is with their own national interest rather than the agenda of the talks. In other words, the regional members see the Six-Party Talks as a platform to resolve bilateral issues and secure their national interest. The inclusion of the abduction issue in the agenda of the forum is among the examples of how the Six-Party Talks are manipulated. A failure to reach consensus in interpreting and implementing the agreements causes an ineffectiveness of the Six-Party Talks in dealing with the problem of North Korea. The Six-Party Talks have too many players who do not only fail to reconcile their national interest but also fail to search for an alternative in light of the failure of the forum (Interview Anonymous, 2011; Interview Prof. Chung, 2011).

The progress of the Six-Party Talks is also affected by the relations the regional members have between or among each other. The relationship and interaction the regional members have at the bilateral level are introduced at the multilateral level within the structure of the Six-Party Talks. The Six-Party Talks can be implemented or resumed dependent on the bilateral relations of the members rather than the structure or the agenda of the forum per se. The multilateral negotiation does not help the United States achieve the denuclearisation of the Korean peninsula. The expansion of bilateral to multilateral negotiation by engaging more players complicates the diplomatic effort in solving the nuclear problem. The nature of the nuclear problem requires the bilateral negotiation between North Korea and the United States. The unsettled Korean War issue is the main objective for North Korea because this issue is the bottom line that obstructs the diplomatic normalisation between the United States and North Korea to happen (Interview Prof. Paik, 2011). North Korea does not see the Six-Party Talks as useful because the multilateral forum does not answer the regime's objective of bilateral negotiations with the United States. China is fully aware of North Korea's desire for the direct talks with the United States (C. I. Moon, 2012). However, China does not have power or leverage to influence the United States to have the required dialogue with North Korea. North Korea is not interested in the Six-Party Talks from the beginning and slowly loses trust in the forum particularly after the agreements are not implemented in the time schedule. The Six-Party Talks become a forum that manages but does not solve the North Korea's nuclear problem. This is because of the factor of differences of interest all six

members have. The Six-Party Talks do not provide any security guarantees to North Korea. In addition, the Six-Party Talks need the US-DPRK and the inter-Korean relations to support and complement each other. The North-South relations can help ease tensions in the Korean peninsula but the key to the progress of the Six-Party Talks is the US-DPRK bilateral relations. The regional powers should be involved in the peace process to support the normalisation of DPRK-US relations rather than taking the leading role in solving the nuclear problem. The Six-Party Talks have been widely criticised for their ineffectiveness to denuclearise North Korea. Nonetheless, the forum is a method of engagement that prevents further nuclear proliferation if North Korea is pushed into isolation. The Six-Party Talks, if resumed, require full cooperation and consensus of all parties in implementing all agreements agreed upon in the previous talks, and in realising the denuclearisation goal.

Conclusion to the analytical questions

The review of the North Korean regime and its relations with the regional powers both at the bilateral and multilateral levels shows that the internal and external determinants play a role in the North Korea's deterrence level. First of all, the primary internal determinant is the national identity of North Korea. North Korea has a unique political system with a combination of various elements. The internal system of North Korea was set up by Kim II Sung after the end of the Korean War and has not experienced much change in decades after. Although there is an effort to launch an economic reform initiative more recently, the reform is implemented at a rather slow pace and in a small scale. North Korea fears major change because it is highly likely that it could bring about the total collapse of the overall system. When the international political structure changed following the fall of Communism, North Korea did not adjust itself to the new environment but sought to maintain its own political, social, and economic systems. A desire to keep its identity amidst all changes makes North Korea feel vulnerable towards the international environment. Insecurity leads to the pursuit of all possible defensive mechanisms including the pursuit of nuclear weapons. The outbreak of the nuclear crisis in the post-Cold War period indicates the insecurity of North Korea but when the United States decided to enter into the bilateral dialogue with North Korea, it also showed the successful use of the proliferation issue in drawing attention from the United States. Concerning this, Prof. Paik (2011) says that,

"North Korea wanted to officially end the Korean War and to normalise the relationship with the United States, Japan, and South Korea. The problem was that the United States was not willing to come to the negotiation table. We have to understand the special characters of nuclear weapons development programme and nuclear weapons itself. In other words, if you want to denuclearise North Korea, you are not supposed to begin a war or avoid it. If you avoid, you cannot achieve your security goals and security interest. If you allow North Korea to go nuclear then how the United States can achieve the goal of the world without nuclear weapons. There is no other way if you are trying to take care of security interest of your own. You have to come to negotiation table to deal with all nuclear problems. That is the basic thing the negotiation took place two decades in the past. North Korea used the nuclear card in early 1990s and the United States had to come to negotiation table. So we have 1994 Agreed Framework in Geneva. North Korea used another nuclear card in 2000 and that we have the September 19 Joint Statement." (Interview Prof. Paik, 2011)

The internal determinant of national identity of North Korea plays a role in the perception between North Korea and the international community. The United States in particular has developed a negative viewpoint on North Korea as influenced by the fact that North Korea continues to be a socialistic and dictatorial country despite the change in the international political structure (Interview Prof. Paik, 2011). North Korea is also a human rights abuser according to the western standard and judgment (Interview Prof. D. H. Han, 2011). On the other hand, North Korea understands the world based on its viewpoint as influenced by its own national and political backgrounds. North Korea is structured on a combination between the western idea of Communism and the eastern idea of Confucianism. The combination leads to the unique internal structure under tight control of a single leader drawn from one family. The one-man dictatorship system is the factor that connects the leadership security with the regime security. Prof. I. H. Park (2011) explains that,

"National security is a kind of broad concept based on the agreement and demand of general public. According to western criteria, government supposes to provide a secure condition to their citizens in the name of national security. If the people are not satisfied with condition of national security, they may choose different government, different political leadership and so on. This is the normal

way of the national society. But in the case of North Korea, there is no national security. What is going on in North Korea, they only have leadership security and regime security. They identify leadership security or regime security identical to national security. Since they identify national security to leadership security, whenever North Korea has different leadership or different political power group, they assume such a kind of happening is a loss to their national security. There is no public opinion and there is no ordinary people's demand the way how they define national security. That is why Kim Il Sung chose his son and that is why Kim Jong Il decided to choose his son. North Korea's national security is only possible under the assumption of Kim's family leadership security." (Interview Prof. I. H. Park, 2011)

As such, sustainability of the North Korean regime is assumed under the condition of the leadership security. In other words, a sense of insecurity is caused by the nature of North Korea itself. The regime nature is considered the weakness of North Korea because it makes it difficult for North Korea to take a path to reform in changing its basic system (Interview Prof. S. H. Sheen, 2011). The system and the regime nature form the identity of North Korea. The factor of identity influences the threat perception of North Korea as based on the desire to secure their nation and their system.

Insecurity by the internal factor is pushed further by the international factor. The fall of Communism shifts a sense of insecurity of North Korea from South Korea to the United States. In the Cold War period, North Korea is driven by the desire to achieve unification during the time North Korea is stronger than South Korea. In the post-Cold War period, the desire for unification is overtaken by the desire for regime survival. An imbalance of power between North Korea and the United States is highlighted when the new structure deprives North Korea of allies. North Korea feels uncomfortable with the US presence in Asia and is made insecure by the US power from the alliance system. North Korea develops a belief in the scenario of the US invasion in bringing down its socialist regime following the collapse of the socialist bloc and this is the logic for the development of nuclear weapons to defend their country (Interview Dr. E. C. Lim, 2011). A sense of insecurity towards the United States becomes more apparent following the US adoption of a hardline policy in categorising North Korea as a rogue state and a part of an axis of evil. The US reluctance to have a dialogue with

North Korea also shows the US non-recognition of the legitimacy of North Korea (Bluth, 2008, p. 182).

The post-Cold War structure also shows the independence of states. The interplay of regional powers on the Korean affairs to counterbalance power and influence among one another is rather clear when the new international political structure is not directed and shaped by confrontation of the two superpowers. The new structure instead shows a hegemonic power of the United States in the region to which China and Russia offer resistance. The relationship the United States has with the Asian allies also underlines its power and influence in the region. China and Russia's participation in the multilateral negotiation platform does not come from an objective of solving the North Korea's nuclear proliferation alone. They both also see an opportunity the forum can provide in balancing the power of the United States and also in fulfilling their national interest. In addition, South Korea and Japan also see the need to be involved in the peace process despite the fact that their security is guaranteed under the US alliance system. The nuclear problem of North Korea relates strongly to the security of South Korea. The two Koreas are connected by history and identity and this is the factor which makes the security dilemma between the two so intense. There is asymmetrical level of defence between the two countries. According to the statistics, South Korea's military is smaller than North Korea. However, economic capability and technology know-how make it easier and likely for South Korea to upgrade their military and modernise their weapons (Interview Prof. Paik, 2011). The asymmetrical level is clear when taking the issue of nuclear weapons into consideration. Concerning this, Prof. M. Kwon (2011) elaborates that.

"North Korea has around 1.2 or 1.3 million armed forces but I would say in terms of conventional military, it is not a strong country. Their military spending is far smaller than South Korean military spending. North Korea's GDP is 150th of South Korea's. They have about 20 billion dollars but South Korea's GDP is almost 1 trillion. It is a very small economy and they cannot afford big conventional army. I mean not only in term of soldiers. They have very poor weapons and conventional arms. They are not strong country. However, North Korea tested nuclear devices twice but South Korea does not have one. In that sense, that is what we call asymmetrical in the development of weaponry. In that aspect, North Korea is a strong country. So far as North Korea develops nuclear

weapons and South Korea does not, that makes the US-ROK alliance important because of the extended deterrence." (Interview Prof. M. Kwon, 2011)

Lastly, Japan engages in the nuclear problem from the US desire for the Japan's financial support. Japan later sees an opportunity within the multilateral forum for resolving the abduction issue.

The regional powers interact in counterweighing the power of one another to ensure the status quo in the region. The seeking for status quo plays an effect on North Korea's nuclear development also. In other words, it refers to the desire for regime survival of North Korea which involves with the issue of unification. An internal problem North Korea is experiencing may point out to an inefficiency of North Korea to achieve the unification on its own term. However, an international politics by the interplay of the regional powers is among the factors to the possibility of the unification of Korea. North Korea is fully aware of how the international politics can determine the future of the two Koreas, which Prof. I. H. Park (2011) explains as:

"If anyone asks me what is the most important characteristic factor regarding Korean problem, I would say that Korean problem is a kind of combination between North and South Korea relations and international relations. Unification is only possible under agreements of the United States, Japan, China, and Russia. Unified Korea with nuclear capability is impossible. North Korea knows that as long as they do not give up nuclear capability, unification is impossible." (Interview Prof. I. H. Park, 2011)

The interplay of regional powers on the Korean affairs makes North Korea realise the need to be independent on its own. Independence is rather clear under the condition that North Korea has no allies, has to counter the US military threat, and has to defend itself against the possibility of South Korea's forced unification. The interplay of regional powers together with the hostile policy from the United States and South Korea creates the foreign pressure on North Korea which pushes for a search for defensive mechanisms in guaranteeing its survival in the hostile environment.

It is possible to say that the external determinant has an edge over the internal determinant in influencing the threat perception of North Korea. An increase of North Korea's deterrence

level corresponds to the way North Korea perceives threat. First of all, it is apparent that North Korea faces with an increased threat to its security since the Cold War period. It begins with the nuclear threat from the United States to the threat of unification by absorption from South Korea from the late Cold War period onwards. The new political structure of the post-Cold War underlines the threat perception of North Korea with an imbalance of power between North Korea on the one hand and the United States and South Korea on the other hand. The outbreak of the two nuclear crises one after the other shows North Korea's search for a high level of deterrence which does not lower after the negotiations. The failure of negotiation is based on the reason that it does not directly answer the need of North Korea which is the recognition of its sovereignty and the assurance of its regime security. Negotiation in both bilateral and multilateral models turns out to be the way to manage rather than solve the problem of North Korea. The insecurity North Korea has towards the United States and the regional environment influences North Korea to seek for methods of securing itself. Nuclear weapons turn out to be the effective and cost-cutting methods in helping North Korea reach the balance of power with the United States (Interview Prof. Koo, 2011). Although the development of nuclear weapons for the deterrence purpose is the most important motive of North Korea, a sense of regime survival is also connected with the domestic politics (Interview Prof. Moon, 2011). While the external determinant affects North Korea's threat perception, the internal determinant is the factor that often changes the level of deterrence of North Korea. The unique internal structure distinguishes North Korea from other states. The socialist system which is run by the dictator projects a rogue state image and in the viewpoint of the international community North Korea could soon collapse because it is unable to take a path to reform (Interview Prof. S. H. Sheen, 2011). The negativity North Korea receives from the international community elevates a sense of insecurity for the regime. A desire to survive without changing the regime's nature and identity makes it difficult for North Korea to lower its deterrence level. The deterrence level is likely to increase especially during the time the regime is faced with a hardline policy from the United States.

Before testing the hypotheses and reaching the final conclusion of whether the deterrence strategy of North Korea is rational or irrational under the assumptions of security and strategic studies, the following conceptual questions will be answered:

Conceptual Questions

- 1. What are the proper criteria for judging whether states are rational or irrational in terms of their security perceptions or deterrence strategies? How should we evaluate and test the relative influence of internal determinants and external environment in shaping security perception and deterrence behaviour?
- 2. What is an adequate level of deterrence for states? How can states gain security for themselves without making other states feel insecure; and is active provocation of a security dilemma with other states evidence of irrational thinking or behaviour?

Question 1

The realists and constructivists assume security as the most important thing states seek for in living in the system of anarchy. A desire for security pushes states to search for ways to secure themselves. When states strive for power, it raises the question of the relationship between the security perception and the deterrence strategy of states, and how rational the states are in their choice of increasing the deterrence level. First of all, it is important to take a note on the connection between anarchy, security competition, and power. Realists and constructivists have their own categories in defining the meaning of anarchy but both schools point out that the system of anarchy is the factor that influences states to seek to guarantee their survival. As based on the assumption of anarchy and insecurity, the security dilemma is a common thing that can happen. Booth and Wheeler (2008, p. 12) write that,

"Believing that insecurity is an inescapable feature of the interstate condition, the 'logic of anarchy' is to strive to maximise power, and particularly military power. Sovereign states are seen as the only political organisations that can offer security, and the growth or maintenance of state power is seen as the only sure way to achieve it."

Anarchy conditions states to compete for security. However, the security dilemma can play an impact in influencing the level of threat perception of one state as determined by the structure of interstate relations and the individual background that the state has. As such, the threat perception is subjective and the strategy to serve the threat perception is a choice of the individual state. It is possible to say that states have two choices when faced with what they perceive as an increased capacity of another state and they are either choosing to increase its defence levels or to assume benign intentions. Waltz (2001, p. 188) inclines towards believing in the first choice as "in the absence of a supreme authority, there is then constant possibility that conflicts will be settled by force". Jervis (2001, p. 36) confirms the need for states to increase its own defence levels in time of conflict because,

"The core argument in time of security dilemma is that, in the absence of a supranational authority that can enforce binding agreements, many of the steps pursued by states to bolster their security have the effect – often unintended and unforeseen – of making other states less secure. The anarchic nature of the international system imposes constraints on states' behaviour. Even if they can be certain that the current intentions of other states are benign, they can neither neglect the possibility that the others will become aggressive in the future nor credibly guarantee that they themselves will remain peaceful."

There arises in consequence the debate on how rational states are in their decision to increase efforts to maintain their security. The theory of deterrence does not rule out the possibility of the outbreak of war among states. The theory instead suggests that this option, if it is chosen, should yield the highest outcome. As such, it is possible to say that rationality and irrationality in the decision-making depends on how well states evaluate all variables constituted to the choice and how well states calculate the policy preference of the other side in responding to its choice (Zagare, 1990, p. 239). Rationality in this sense means the final decision states choose should comply with the choice of positive prediction (Lebow & Stein, 1989, p. 211).

The assumption of anarchy suggests that environment as external determinant plays an important role in shaping the security perception and deterrence behaviour of states. Nonetheless, theorists such as Sagan (1996, p. 55) and Bracken (2012, pp. 106-108) point out the internal determinants of domestic politics and a sense of nationalism that could play an effect on the deterrence strategy of states. While Sagan (1996, pp. 63-65) claims the role of general public in influencing the nuclear acquisition of states, Bracken (2012, pp. 115-116)

stresses on the correlation between nationalism and nuclear ambition. According to Bracken (2012, pp. 116-117),

"Nationalism and nuclear weapons are potentially a toxic mixture. The bomb may be viewed as a horrific weapon, but it's also a national accomplishment ... Nationalism can drive regional tensions and rivalry."

The context of domestic politics also plays a major part in the state's identity. Identity is the factor that defines the interests of states and the way states perceive threat (Hopf, 2002, pp. 16-17). Threat perception is varied based on individual background of states as well as external determinants. Stein (2013, pp. 367-368) says that states who have a high sense of nationalism are likely to have a high level of threat perception as they are defensive of their nation and national identity. Although the theory assumes the role of external determinant on the deterrence level and the security perception of states, there is the likelihood that states are influenced by the internal determinant also. Brooks (1997, p. 471) writes that the domestic politics can play a role in influencing the behaviour of states under some circumstances,

"Systemic constraints will often cause states to discern a clear strategy to advance their overriding objectives; this will especially be the case when the international influences on states are very strong and/or when the state is relatively susceptible to these systemic factors. At other times, the strategy to pursue the state's primary objectives may be more ambiguous. In circumstance where the preferred strategy is unclear, the mechanism by which particular policies are selected may be significantly influenced by domestic-level bargaining."

Nonetheless, theorists such as Mearsheimer (2009, p. 246) and Sagan (1996, p. 65) tend to argue that the domestic factor can influence national security and foreign policy of states only at a marginal level. Therefore, the external determinant of environment is the factor that affects the security perception and deterrence behaviour as based on the conception of distribution of power among states within the system of anarchy. The domestic factor has several definitions based on the assumption and understanding of each theorist. The issue of national identity is among the factors that has the highest potential in influencing the behaviour of states. This is based on the assumption of constructivists that identity structures

the relationship of states on the basis of shared knowledge and this is the factor that organises material resources to serve a purpose beyond security assurance.

Question 2

The concept of deterrence talks about the search to balance the security level between states in the system of anarchy. States, when faced with the problem of insecurity, usually incline towards choosing an increase to their own security. The assumption of the correlation between anarchy, states, and power leads to the idea of the security dilemma. The security dilemma occurs when an attempt to increase the security of one state decreases security of another state which tempts the latter to follow the same path in choosing to balance the security level. States would choose to gain rather than to lose security. This is because states are driven by a sense of insecurity that makes them compete for security even without any desire of expansion (Posen, 1993, p. 28). The system of uncertainty influences states to assume the worst by overestimating the hostility of other states and feeling mistrust about intention of other states (Booth & Wheeler, 2008, p. 264; Jervis, 1988, p. 688). Jervis (1978, p. 174) assumes that the higher the degree states value their security, the higher the likelihood that states will incline towards demanding higher levels of deterrence. As such, it is possible for the spiral model of the security dilemma to escalate and eventually results in war.

This raises the question of what the adequate level of deterrence of one state should be. Since the theory assumes insecurity to be the main motivator of states, it is possible to say that the adequate level of deterrence would be at the point when states feel secure. The problem is that states have great difficulty deciding what is the point at which security is assured. Uncertainty about the level of security and insecurity often makes the security dilemma inescapable. The system of anarchy makes states feel uncertain of other states' intentions and this is part of the explanation why states choose to increase their security. According to J. Snyder (1985, pp. 154-155), states would be at risk the more they acknowledge the security of the other because it means "accepting one's own insecurity." Although the security dilemma is a common experience that is inescapable in the system of anarchy, it is argued that the security dilemma can be ameliorated. C. Glaser (1997, p. 181) claims that tension can be eased if there is a communication between states in showing one another that there is no hostile intent. An understanding between states on each other's security could create the

status quo. Concerning this, Jervis (2001, p. 55) elaborates by referring to the competition between the two superpowers during the Cold War,

"The arms competition may also have constituted a security dilemma, at least after the Soviet Union achieved a second-strike capability in the mid-1960s. Security dilemma analysis would be misleading if we found that either side was willing to pay a high price to gain superiority in order to coerce the other into changing the status quo ... Mutual security came within reach only when leaders on both sides became willing to give up the hope for superiority in return for arrangements that precluded the other side from achieving it and that lowered tensions and reduced spending (at least in principle)."

J. Snyder (1985, pp. 154-155), despite his agreement with Jervis on the idea of status quo, also supports the assumption that the security dilemma cannot be resolved. He says,

"Neither deterrent nor spiral axioms can resolve this kind of security dilemma. Unyielding, "deterrent" policies may be stabilising when they defend a balanced status quo, because they induce caution in those who would overturn the balance" (J. Snyder, 1985, pp. 154-155)

As such, it is possible to say that the only possibility for a state to feel secure during the time one state increases its security is when the two sides clearly understand the intention behind such action. A failure to understand and interpret the intention will heighten the security dilemma between the two sides (Jervis, 1978, p. 181).

States seek for security and this is when defensive mechanisms come into play. Defensive mechanisms such as military force development are described by Schelling (2008, p. 2) as having not only "the power of defence" but also "the power to hurt" in which the latter serves the purpose of deterrence. Strategy used to create the means of deterrence includes both conventional arms and nuclear weapons (T.V. Paul, 2009, p. 7). However, there arises some scenario where states use also the method of provocation. This raises a question about the justification of this method because the provocation can easily escalate the security dilemma. The theory of deterrence usually talks about the use of nuclear and conventional weapons as a means to deter. The theory, however, does not rule out the possibility of the use of other defensive means, as Knopf (2009, p. 41) says that "the deterrence can involve anything". The

new environment of the post-Cold War shows the hegemonic power of the United States and the rise of new nuclear states which are small and middle powers. The deterrence is created on an unequal balance between the two sides. Under this circumstance, there is a possibility that provocation is employed. T.V. Paul (2009, pp. 8-9) writes,

"In today's near-unipolar international system, multiple layers of interactions and interconnectedness have emerged. Weak states no longer are all that weak, as asymmetric capabilities and strategies are becoming effective in many situations in constraining the great powers, most prominently the United States."

Provocations can create the security dilemma. At the same time that the security dilemma suggests a sense of insecurity, it can refer to the manipulation of threat for security purposes. According to this, Booth and Wheeler (2008, p. 7) write,

"Security dilemma sensitivity is an actor's intention and capacity to perceive the motives behind, and to show responsiveness towards, the potential complexity of the military intentions of others. In particular, it refers to the ability to understand the role that fear might play in their attitudes and behaviour, including, crucially, the role that one's own actions may play in provoking that fear"

The security dilemma can be the result of how states want to fulfil a security interest. When the security dilemma is set off, the deterrence strategy usually comes into play. The security dilemma and deterrence strategies are structured on the assumption of rationality of states. This is because when states want to increase their security, states usually carefully make a judgement on not only the method to use but also the end result if the decision is taken (Bull, 1968, p. 597). The security dilemma does not usually escalate to the point of war. States hesitate to respond in conflict knowing well the destructive nature of war (Jervis, 1978, p. 177). As such, the strategy states employ in time of threat inclines towards preserving rather than changing the status quo.

Conclusion and the testing of hypotheses

The study of North Korea and its behaviour shows the practice of both deterrence and the security dilemma. North Korea uses deterrence in various forms, which are the conventional forces, nuclear weapons, and provocations. The use of deterrence is influenced by both the internal and external determinants. The internal determinant is the nature of the regime, while the external determinant is the regional environment which is a result of the relationship North Korea has with the regional powers and the interaction among the regional powers on the issue of Korea. When looking at both determinants, the external determinant plays a higher role in affecting North Korea's deterrence. The use of nuclear weapons as a strategy starts in the post-Cold War period. The nuclear strategy shows how North Korea adjusts itself to the new environment in particular an imbalance of power between North Korea and the United States. The role of the United States as a hegemonic power made North Korea believe in the possibility of the US attack on the regime, influencing it to consider obtaining nuclear weapons (Interview Dr. E. C. Lim, 2011). North Korea develops nuclear weapons as part of a strategy of deterrence against the United States. When nuclear strategy is achieved, they provide North Korea with the power to deter. Kim II Sung said in the 1993 New Year Address that.

"Pressure or threat will have no effect on us. Such an attempt may invite catastrophe. Therefore, the nuclear problem on the Korean peninsula must on any account be settled through Korea-U.S. talks." (Agence France Presse, 1993b)

The situation on the Korean peninsula remains in a state of ceasefire. The change in the international political structure influences North Korea with the need to settle the Korean War issue with the United States so that it could overcome all difficulties including the reunification issue (Interview Prof. Paik, 2011). When looking at the nature of the problem, it derives from a sense of insecurity of North Korea. Negotiation in both bilateral and multilateral patterns do not bring about the security guarantee North Korea seeks for. North Korea continues to be faced with a nuclear threat from the United States, a competition with South Korea, and an insecure environment arising from the interplay of the regional powers.

The nuclear strategy provides North Korea with some concessions on various occasions. It is the destructive nature of nuclear weapons that draws the United States to negotiation and to provide North Korea with some concessions. Although North Korea learns the value of nuclear weapons in the negotiation, the main use of nuclear strategy is not for receiving concessions or overturning the status quo. Dr. E. C. Lim (2011) says that,

"It will not be easy to get a sustainable economic assistance from foreign countries as North Korea intend. North Korea themselves do not believe nuclear weapons will be very useful to get more aid/economic aid from foreign countries. They started to admit clear limitations of using nuclear weapons. Even though they have some intention to use nuclear weapons as leverage to get more help, it does not work as they intend anyway." (Interview Dr. E. C. Lim, 2011)

North Korea actually wants to escape the security dilemma with the United States. Ambassador Charles L. Pritchard²⁴ recalled the meeting between Kim Jong II and Madeleine Albright in Pyongyang in October 2000,

"I am struck by what Kim Jong II, North Korea's leader, said to Madeleine Albright, former US secretary of state, in October 2000. He told her that in the 1970s, Deng Xiaoping, the Chinese leader, was able to conclude that China faced no external security threat and could accordingly refocus its resources on economic development. With the appropriate security assurances, Mr Kim said, he would be able to convince his military that the US was no longer a threat and then be in a similar position to refocus his country's resources." (Pritchard, 2003a)

As earlier outlined, Kim II Sung often mentioned the reunification of Korea and the strategy he pursued was to achieve this aim. However, it seems highly unlikely that reunification is a major objective of the regime given the situation of economic crisis the country is experiencing (Interview Prof. M. Kwon, 2011). It is believed that the objective of North Korea changed when Kim Jong II came to power to focus more on achieving the normalisation of relations with the United States. North Korea considered the negotiation during the first nuclear crisis of 1993 a part of a strategy to achieve this aim (Pardo, 2014, p. 25). Kim Jong II stresses in his writing in August 1997 that the aim of North Korea is the

-

²⁴ Ambassador Charles L. Pritchard was the US envoy for negotiation with North Korea in the Bush administration. He accompanied Madeleine Albright to visit North Korea and meet with Kim Jong II in October 2000.

normalisation of relations with the United States and "we do not intend to view the United States as our eternal sworn enemy" (International Business Publications, 2003, p. 58)

Nuclear weapons have become important assets to North Korea. The second nuclear crisis was the period when the threat perception of North Korea was pushed highest by the external factor. President Bush's hardline policy and the US invasion of Iraq in 2003 raised North Korea's insecurity and convinced North Korea to believe in the value of the nuclear deterrent. Hecker (2010, p. 46) notes that this was the period when "Pyongyang was eager to have Washington believe it had the bomb". North Korea learnt from the first nuclear crisis the use of brinkmanship tactics in drawing the United States to the negotiation table and achieved some benefits after the agreement was made. In the second nuclear crisis of 2002 North Korea's showed more confidence in negotiation and used brinkmanship tactics in the multilateral negotiation. For example, North Korea offered to exchange the nuclear dismantlement with the US offer to establish the diplomatic relations with North Korea at the first round of the Six-Party Talks in 2003. In addition, from time to time, North Korea threatened not to attend the Six-Party Talks if the negotiation would not result in its objectives.

Overall, what can be concluded from the case of North Korea as based on the security and strategic studies are the use of nuclear deterrence as a strategy, an influence of history and identity on the deterrence strategy, and an impact of the inter-state system on the deterrence strategy. The deterrence strategy and the security dilemma are results of the insecurity of states. The situation of the Korean peninsula shows that there is a lack of recognition of the security needs of the Korean states. The security needs of North Korea are mentioned during the negotiation process but it has never been acknowledged to the point of reaching a proper Peace Treaty. It is the feeling of fear that influences the behaviour and deterrence strategy of North Korea. The security studies focuses on the security of states, while the strategic studies talks about how states seek to secure their security. Nuclear weapons and all defensive measures are developed as a strategy in response to the insecurity North Korea feels. The deterrence strategy of North Korea is a response to the security dilemma but does not refer to a desire to expand. It refers to how states seek to secure their own objective, which in the case of North Korea is the leadership survival, regime survival and national survival.

The concept of deterrence means how threat is manipulated in order to influence the thinking and behaviour of the other side. North Korea's deterrence strategy is not limited to the use of nuclear weapons but includes also conventional weapons and the use of provocations. The deterrence strategy is driven by an unequal balance of power that makes North Korea want to achieve the same deterrence level as the United States. The missile firing has an objective of testing the policy of the United States and South Korea but at the same time it is for a purpose of deterrence. North Korea has reiterated that the missile firing is intended for the peaceful satellite launch. However, whether the claim can convince the international community or not, the missile firing incident has served the deterrence objective. Bracken (2012, p. 191) comments on the North Korea's missile firing of 2006 as having the deterrence objective,

"The message sent was that a U.S. attack on North Korea would trigger an immediate attack on South Korea and Japan. This hostage strategy says to the United States: "Don't even think about doing something your allies will regret."

The concept of deterrence and the security dilemma do not exactly indicate what the adequate level of deterrence is for a state. The main point that can be concluded from the concept of deterrence is how the national security can be secured at the highest level possible and with a minimum chance that a direct conflict will occur. However, what can be assumed to be the best outcome is a mutual recognition of security which could influence states to value the status quo without seeking to elevate their defence level. The security dilemma will be ameliorated at this point despite the inevitable fact that states have different viewpoints on threat and insecurity. The way states see threat depends on individual perception as influenced by the background and political condition of each state. As such, the way North Korea values its security is different from other states. A high level of deterrence that one state pursues does not necessarily mean that the state in question is irrational. The theory of deterrence and the security dilemma do not suggest a high level of deterrence is connected to irrationality. The theory points out that deterrence occurs in a situation of intense uncertainty about conflict and the role that deterrence has in making states value the status quo. This is when the calculation of gain and loss comes into play. Irrationality can possibly occur under this circumstance if states choose to fight despite knowing that they have a low chance of winning. However, the Korean peninsula shows a high level of security dilemma but the overall situation is rather stable. This is because the mutual deterrence has been in function for several decades and the status quo has been maintained.

These conclusions tend to support Hypothesis One in suggesting that the security perception and strategic behaviour of North Korea conforms to the assumptions of security studies and strategic studies. What North Korea values the most is national security. The use of the deterrence strategy is for a security purpose in response to threats the regime believes it is facing. The case of North Korea shows that a sense of nationalism in the identity of the state is also an important factor. Nationalism makes states value the identity factor in defining their national security but it is not the key factor influencing deterrence strategy. The deterrence strategy is developed to counter the external threats and nationalism shapes the deterrence behaviour to be a symbol of national pride. This allows an opportunity for the manipulation of threat for domestic purposes. The dictatorship system of North Korea shapes the way the leadership security equates to regime security. North Korea does not have the desire to attack its enemies knowing well that nuclear weapons, if used, would mean the start of a comprehensive war (Interview Dr. E. C. Lim, 2011). Nuclear weapons are developed for the purpose of deterrence in maintaining the status quo on the Korean peninsula. North Korea adjusts its objective from unification to regime survival to suit the internal and external conditions the regime faces, especially in the change from the Cold War international structure to the post-Cold War structure. The security dilemma is heightened in this change but does not bring about a total loss to North Korea. The high level of security dilemma North Korea has with the US, South Korea, and other regional powers means that the security of the regime is guaranteed, provided it can sustain deterrence levels. Therefore, the deterrence strategy has served the main objectives of delivering security and maintaining the status quo, which leads to a conclusion that the deterrence strategy of North Korea is rational in accordance with the theory of security studies and strategic studies.

Appendix 1: List of Interviewees

<u>Name</u>	Institution	Date of Interview
Anonymous	Korea Institute for National Unification	16 September 2011
Prof. Chung Jae Ho	Seoul National University	15 September 2011
Prof. Han Dong Ho	Institute of Foreign Affairs and National Security, Ministry of Foreign Affairs and Trade	2 September 2011
Prof. Kim Young Yoon	Korea Institute for National Unification	20 September 2011
Prof. Koo Kab Woo	University of the North Korean Studies	9 September 2011
Prof. Kwon Man Hak	Kyung Hee University	8 September 2011
Dr. Lim Eul Chul	The Institute for Far Eastern Studies, Kyungnam University, The Center for International Cooperation for North Korean Development	9 September 2011
Prof. Moon Chung In	Yonsei University	8 September 2011
Prof. Paik Hak Soon	Sejong Institute	1 September 2011
Prof. Park Ihn Hwi	Ewha Womans University	30 August 2011
Prof. Sheen Seong Ho	Seoul National University	31 August 2011

Appendix 2: Note to the Interviewees

Interviewees	Expertise of the Interviewees
Anonymous at Korea Institute for National Unification	The Korea Institute for National Unification is a research institute under the Ministry of Unification, South Korea. The interviewee works as a senior research fellow in this institute and has experience conducting research in areas of North Korean politics, inter-Korean relations, and the US-ROK relations.
Prof. Chung Jae Ho	Prof. Chung Jae Ho teaches at Department of International Relations, Seoul National University. His research expertise is China and US-China relations. He was once a visiting fellow at the Brooking Institution in the Brookings Center in East Asia Policy Studies conducting research on South Korea-China relations. The questions interviewed with Prof. Chung focuses on China's policy and relations with the two Koreas and the US-China-North Korea relations.
Prof. Han Dong Ho	Institute of Foreign Affairs and National Security is a research institute under the Ministry of Foreign Affairs and Trade, South Korea. Prof. Han Dong Ho works under this institute and his position is a visiting professor. Prof. Han's expertise is the inter-Korean relations and unification issue.
Prof. Kim Young Yoon	Prof. Kim Young Yoon works as a senior research fellow at the Korea Institute for National Unification, the Ministry of Unification, South Korea. His expertise includes the economy, the inter-Korean relations, and North Korean politics.
Prof. Koo Kab Woo	Prof. Koo Woo Kab works as a professor at the University of the North Korean Studies. The University of the North Korean Studies focuses on the study of North Korea. Prof. Koo's expertise is the inter-Korean relations and international politics. Apart from academic publications, Prof. Koo contributes his analysis on Nautilus Institute for Security and Sustainability.

D C 17 3 7 77 1	D CT M TT 1 C C C C
Prof. Kwon Man Hak	Prof. Kwon Man Hak is a professor at the College of International Studies, Kyung Hee University. His expertise is the inter-Korean relations and Northeast Asian security. Apart from academic publications, Prof. Kwon's opinion and analysis on North Korea and peninsula affairs are also referred in the media websites. Among examples are "North Korean Society Matches Chosun Dynasty" appeared on the Daily NK website on 28 March 2011. Prof. Kwon also contributes his analysis on the US foreign policy and Northeast Asia titled "Where does Northeast Asia Fit in the Bush Administration Foreign Policy?". This publication appears on the Korea Economic Institute of America in the publication series titled "Joint U.S Korea Academic Studies 2006" and it is available online via the website.
Dr. Lim Eul Chul	Dr. Lim Eul Chul is a director of the Center for International Cooperation for North Korean Development at Kyungnam University. The Center deals with the issue of the North Korean affairs. Dr. Lim's expertise includes the North Korean politics and economy, and the inter-Korean relations. Dr. Lim has an in-depth knowledge on North Korea from his experience interviewing with the North Korean defectors.
Prof. Moon Chung In	Prof. Moon Chung In is a professor at the Department of Political Science, Yonsei University. Prof. Moon's past experience included his positions as an advisor to the President of the Republic of Korea, an advisor to the Ministry of Foreign Affairs and Trade, an advisor to the Ministry of National Defense. Prof. Moon accompanied President Kim Dae Jung to Pyongyang in an inter-Korean summit meeting in 2000.
	Prof. Moon publishes various books and journal articles and contributes his analysis on North Korea on 38 North, Nautilus Institute for Security and Sustainability, NK News, the Asia-Pacific Journal: Japan Focus, etc. His publications and analysis are also referred in this thesis.

Prof. Paik Hak Soon	Prof. Paik Hak Soon works as a senior fellow at the Department of the Unification Strategy Studies, Sejong Institute, one of the leading research institutes of South Korea. Prof. Paik also works as a policy advisor for Ministry of Unification, Ministry of Foreign Affairs and Trade, and the National Assembly. Prof. Paik's expertise includes the inter-Korean relations, the North Korean politics, the US-North Korean relations, North Korea's nuclear issue. Prof.
	Paik also publishes his analyses on North Korea on the 38 North website which some of it is referred in the thesis.
Prof. Park Ihn Hwi	Prof. Park Ihn Hwi is an associate professor of international relations at the Ewha Womans University. His expertise is the US foreign policy, the US-ROK relations, and Northeast Asian international relations. Prof. Park also takes a position of a member of the advisory committee of the Ministry of Unification.
Prof. Sheen Seong Ho	Prof. Sheen Seong Ho is a professor of international security at Seoul National University. Prof. Sheen's expertise is the US foreign policy, Northeast Asian politics and the Korean peninsula. Prof. Sheen was an advisor to the Ministry of Unification and Committee on Foreign Affairs and Unification.

- Achen, C., & Snidal, D. (1989). Rational Deterrence Theory and Comparative Case Studies. *World Politics*, 41(2), 143-169.
- Agence France Presse. (1991, 14 November). Beijing Opposed to Any Nuclear Presence in Korea: Qian, *Agence France Presse*. Retrieved from http://www.nexis.com
- Agence France Presse. (1993a, 13 October). Japan, Russia Urge North Korea to Dispel Nuclear Suspicions, *Agence France Presse*. Retrieved from http://www.nexis.com
- Agence France Presse. (1993b, 1 January). Kim Il-Sung Warns Against War, Calls for Talks with U.S., *Agence France Presse*. Retrieved from http://www.nexis.com
- Agence France Presse. (1996, 5 July). Russia Wants New Treaty with North Korea: Minister, *Agence France Presse*. Retrieved from http://www.nexis.com
- Agence France Presse. (2002a, 17 October). Text of US Statement on North Korean Nuclear Program, *Agence France Presse*. Retrieved from http://www.nexis.com
- Agence France Presse. (2002b, 16 November). US President Says He "Loathes" North Korea's Kim, *Agence France Presse*. Retrieved from http://www.nexis.com
- Alagappa, M. (2008). *The Long Shadow: Nuclear Weapons and Security in 21st Century Asia.* Stanford: Stanford University Press.
- Albright, D., & Brannan, P. (2010). Taking Stock: North Korea's Uranium Enrichment Program *Korean Peninsula: ISIS Reports 2010* (pp. 1-34): The Institute for Science and International Security.
- Alford, P. (2008, 13 October). Anger as N Korea Delisted Japan Protests as US Softens Line on Pyongyang, *The Australian*. Retrieved from http://www.nexis.com
- An, T. S. (1983). *North Korea in Transition: From Dictatorship to Dynasty*. Westport: Greenwood Press.
- Ang, A. (2005, 22 March). China Tout Six-Party Talks as Only "Real" Way to Solve Nuclear Crisis to Visiting North Korean Leader, *Associated Press International*. Retrieved from http://www.nexis.com
- Anonymous (2011, 16 September). [Interview].
- Armstrong, C. (2004). Inter-Korean Relations: A North Korean Perspective. In S. Kim (Ed.), *Inter-Korean Relations: Problems and Prospects*. New York: Palgrave Macmillan.
- Armstrong, C. (2007). The Koreas. New York: Routledge.
- Armstrong, C. (2009). North Korea's South Korea Policy: Tactical Change, Strategic Consistency. In S. C. Kim & D. Kang (Eds.), *Engagement with North Korea: a Viable Alternative*. Albany: State University of New York Press.
- Armstrong, C. (2010, 20-22 October). South Korea and the Six-Party Talks: The Least Bad Option? Paper presented at the Tomorrow's Northeast Asia.
- Associated Press World Stream. (2006, 6 July). Putin Disappointed by North Korean Missile Tests, Stresses Need for Resumption of Talks, *Associated Press Worldstream*. Retrieved from http://www.nexis.com
- Atkinson, C. (2006). Constructivist Implications of Material Power: Military Engagement and the Socialization of States, 1972-2000. *International Studies Quarterly*, 50(3), 509-537.
- Auslin, M. (2011). Japanese Perspectives on the Six-Party Talks and the North Korean Nuclear Crisis. In N. Finnemann, S. Howe, A. Kim & F. Lowe-Lee (Eds.), *Joint U.S.-Korea Academic Studies: Prospects for Emerging East Asian Cooperation and Implications for the United States* (Vol. 21): Korea Economic Institute.

- Ayson, R. (2008). Strategic Studies. In C. Reus-Smit & D. Snidal (Eds.), *The Oxford Handbook of International Relations*. Oxford: Oxford University Press.
- Barkin, J. S. (2003). Realist Constructivism. *International Studies Review*, 5(3), 325-342.
- Barkin, J. S. (2010). *Realist Constructivism: Rethinking International Relations Theory*. Cambridge: Cambridge University Press.
- Barnett, M., & Duvall, R. (2005). Power in International Politics. *International Organization*, 59(1), 39-75.
- Bazhanov, E. (2003, 4 December 2003). *A CEAS Working Paper: Korea in Russia's Post Cold War Regional Political Context.* Paper presented at the A Seminar on a Russian Perspective concerning the Confrontation on the Korean Peninsula, 4 December 2003, Monterey Institute of International Studies.
- Bazhanov, E. (2006). Korea in Russia's Post-Cold War Regional Political Context. In C. Armstrong, G. Rozman, S. Kotkin & S. Kim (Eds.), *Korea at the Center: Dynamics of Regionalism in Northeast Asia*. New York: M.E. Sharpe, Inc.
- Bazhanov, E., & Bazhanov, N. (1991). Soviet Views on North Korea: The Domestic Scene and Foreign Policy. *Asian Survey*, *31*(12), 1123-1138.
- BBC News. (1993a, 12 April). North Korea Nuclear Issue; DPRK Spokesman Tell Russia to "Mind Its Own Business", *BBC News*. Retrieved from http://www.nexis.com
- BBC News. (1993b, 23 October). Russia Advised to "Mind Its Own Business" Over Nuclear Issue, *BBC News*. Retrieved from http://www.nexis.com
- BBC News. (2002a, 20 November). NKorea: Envoy Blames US for Nuclear Controversy at UN General Assembly Session, *BBC News*. Retrieved from http://www.nexis.com
- BBC News. (2002b, 31 October). Stand-Off over North Korea Kidnap Victims, *BBC News*. Retrieved from http://news.bbc.co.uk/1/hi/world/asia-pacific/2379695.stm
- BBC News. (2003, 25 February). Roh Moo Hyun's Inauguration Speech, *BBC News*. Retrieved from http://news.bbc.co.uk/1/hi/world/asia-pacific/2797053.stm
- BBC News. (2005, 4 November). Putin's Envoy Upholds North Korean Case over US Reactor Pledge, *BBC News*. Retrieved from http://www.nexis.com
- BBC News. (2006, 5 July). South Korea Urges North to End Missile Launches, Return to Talks, *BBC News*. Retrieved from http://www.nexis.com
- BBC News. (2008, 26 June). George Bush North Korea Statement, *BBC News*. Retrieved from http://news.bbc.co.uk/1/hi/7475631.stm
- BBC News. (2010, 28 November). South Korea-US Military Exercises Stoke Tensions, *BBC News*. Retrieved from http://www.bbc.co.uk/news/world-asia-pacific-11855162
- BBC News. (2011a, 24 August). Kim Jong-il in Talks with Russia's Dmitry Medvedev, *BBC News*. Retrieved from http://www.bbc.co.uk/news/world-asia-pacific-14600366
- BBC News. (2011b, 3 January). South Korea Stands Firm against North's 'adventurism', *BBC News*. Retrieved from http://www.bbc.co.uk/news/world-asia-pacific-12106396
- BBC News. (2012, 29 December). South Korea Backs Six-Party Talks with North, *BBC News*. Retrieved from http://www.bbc.co.uk/news/world-asia-pacific-12087986
- BBC News. (2013, 3 April). Q&A: Kaesong Industrial Complex, *BBC News*. Retrieved from http://www.bbc.co.uk/news/business-22011178
- Bechtol, B. E. (2007). *Red Rogue: The Persistent Challenge of North Korea*. Dulles: Potomac Books, Inc.
- Bennett, B. (2013). Preparing for the Possibility of a North Korean Collapse (pp. 1-346): The RAND Corporation.
- Berry Jr., W. (1995). North Korea's Nuclear Program: The Clinton Administration's Response *INSS Occasional Paper No.3* (pp. 1-48): USAF Institute for National Security Studies, U.S. Air Force Academy.

- Blank, S., & Kim, Y. (2012). Restarting the Six-Party Talks: Russia's Dilemmas and Current Perspectives. *Korea Observer*, 43(2), 253-277.
- Bloomberg. (2003, 23 May). Bush Says U.S., Japan United on North Korea Policy (Update4). Bloomberg. Retrieved from http://www.bloomberg.com/apps/news?pid=21070001&sid=aHFEbwOF.2VU
- Bluth, C. (2008). Korea. Cambridge: Polity Press.
- Bluth, C. (2011). Crisis on the Korean Peninsula. Washington, D.C.: Potomac Books.
- Bogaturov, A. (2004). Russia's Priorities in Northeast Asia: Putin's First Four Years Brookings Northeast 84 Asia Survey 2003–04: The Brookings Institution.
- Booth, K., & Wheeler, N. (2008). *The Security Dilemma: Fear, Cooperation and Trust in World Politics*. New York: Palgrave Macmillan.
- Bosworth, S., & Gullucci, R. (2013, 27 October). Reasons to Talk to North Korea, *The New York Times*. Retrieved from http://www.nytimes.com/2013/10/28/opinion/reasons-to-talk-to-north-korea.html
- Bowman, S., Collier, E., Niksch, L., & Shinn, R. S. (1994). North Korea's Nuclear Weapons Program: U.S. Policy Options. In R. Cronin (Ed.), *CRS Report for Congress, CRS94-470F* The Federation of American Scientists.
- Bracken, P. (1993). The North Korean Nuclear Program as a Problem of State Survival. In A. Mack (Ed.), *Asian Flashpoint: Security and the Korean Peninsula*. New South Wales: Allen & Unwin.
- Bracken, P. (2012). *The Second Nuclear Age: Strategy, Danger, and the New Power Politics*. New York: Henry Holt and Co.
- Bradshaw, R. (1979). Major Determinants of North Korean Foreign Policy. In J. K. Park & J. G. Kim (Eds.), *The Politics of North Korea*. Seoul: The Institute of Far Eastern Studies.
- Brodie, B. (1978). The Development of Nuclear Strategy. *International Security*, 2(4), 65-83.
- Brooke, J. (2004, 8 June). Japan Leader Calls on North Korea to Dismantle Nuclear Program, *The New York Times*. Retrieved from http://www.nytimes.com/2004/06/08/international/asia/08japa.html
- Brooks, S. (1997). Dueling Realisms. *International Organization*, 51(3), 445-477.
- Bull, H. (1968). Strategic Studies and Its Critics. World Politics, 20(4), 593-605.
- Buszynski, L. (2004). Asia Pacific Security-Values and Identity. London: Routledge.
- Buszynski, L. (2009). Russia and North Korea: Dilemmas and Interests. *Asian Survey*, 49(5), 809-830.
- Buszynski, L. (2013a). *Negotiating with North Korea: The Six-Party Talks and the Nuclear Issue*. London: Routledge.
- Buszynski, L. (2013b, 9 August). The Six-Party Talks Have Had Their Day: Time for an Expanded Dialogue. *East Asia Forum*. from http://www.eastasiaforum.org/2013/08/09/the-six-party-talks-have-had-their-day-time-for-an-expanded-dialogue/
- Cane, P. (2013, 3 April). Beijing and Moscow: Pyongyang's Silent Partners. *International Policy Digest*. from http://www.internationalpolicydigest.org/2013/04/03/beijing-and-moscow-pyongyangs-silent-partners/
- Cha, V. (2002). North Korea's Weapons of Mass Destruction: Badges, Shields, or Swords? *Political Science Quarterly*, 117(2), 209-230.
- Cha, V. (2004). The U.S. Role in Inter-Korean Relations: Container, Facilitator, or Impeder? In S. Kim (Ed.), *Inter-Korean Relations: Problems and Prospects*. New York: Palgrave Macmillan.

- Cha, V. (2012a). The Impossible State: North Korea, Past and Future. London: The Bodley Head
- Cha, V. (2012b, 6 April). Why China Can't Cut Off North Korea, *The Huffington Post*. Retrieved from http://www.huffingtonpost.com/victor-cha/china-north-korea-relations_b_1404178.html?view=screen
- Cha, V. (2013, 8 May). South Korean President Park Geun-Hye's First Visit to Washington, D.C., from http://csis.org/publication/south-korean-president-park-geun-hyes-first-visit-washington-dc
- Cha, V., & Kang, D. (2003). Introduction: the Debate over North Korea *Nuclear North Korea: A Debate on Engagement Strategies*. New York: Columbia University Press.
- Chance, D., & Kim, J. (2013, 12 February). North Korean Nuclear Test Draws Anger, Including from China, *Reuters*. Retrieved from http://www.reuters.com/article/2013/02/12/us-korea-north-idUSBRE91B04820130212
- Chandra, R. (1993, 15 March). North Korea: Soured Ties with China Hamper Bid to Defuse Crisis, *IPS-Inter Press Service*. Retrieved from http://www.nexis.com
- Chanlett-Avery, E. (2008). Congress and U.S. Policy on North Korean Human Rights and Refugees: Recent Legislation and Implementation *CRS Report for Congress*: Congressional Research Sevice, CRS Report for Congress.
- Chanlett-Avery, E., & Rinehart, I. (2014). North Korea: U.S. Relations, Nuclear Diplomacy, and Internal Situation *CRS Report for Congress* (pp. 1-27): The Federation of American Scientists.
- Checkel, J. (1998). Review Article: the Constructivist Turn in International Relations Theory. *World Politics*, 50(2), 324-348.
- Chick, K. (2010, 4 May). South Korea's Leader Calls Cheonan Warship Sinking 'No Accident', *The Christian Science Monitor*. Retrieved from http://www.csmonitor.com/World/terrorism-security/2010/0504/South-Korea-s-leader-calls-Cheonan-warship-sinking-no-accident
- Chinoy, M. (2010, 29 November). Bush on North Korea: Wrong Again. *38 North*. from http://38north.org/2010/11/bush-on-north-korea-wrong-again/
- Chinoy, M. (2011, 10 February). Six Party Talks: The Least Bad Alternative. *38 North*. from http://38north.org/2011/02/six-party-talks/
- Choe, S. H. (2007a, 16 July). Korean Reactor Is Shut, U.N. Confirms, *The New York Times*. Retrieved from http://www.nytimes.com/2007/07/16/world/asia/16cndkorea.html?r=2&hp=&pagew anted=all&
- Choe, S. H. (2007b, 26 June). North Korea Receives Funds and Says It Will Shut Down Its Main Nuclear Reactor, *The New York Times*. Retrieved from http://www.nytimes.com/2007/06/26/world/asia/26korea.html
- Choe, S. H. (2009, 4 January). North Korea Tries New Tack with Obama, *The New York Times*. Retrieved from http://www.nytimes.com/2009/01/14/world/asia/14ihtnorth.1.19345227.html?r=0
- Choe, S. H. (2012, 19 December). Ex-Dictator's Daughter Elected President as South Korea Rejects Sharp Change, *The New York Times*. Retrieved from http://www.nytimes.com/2012/12/20/world/asia/south-koreans-vote-in-closely-fought-presidential-race.html?pagewanted=all&_r=0
- Choi, S. (2001). Divided States: Reunifying without Conquest. In W. Zartmen (Ed.), *Preventive Negotiation: Avoiding Conflict Escalation*. Lanham: Rowman & Littlefield Publishers, Inc.

- Chung, J. H. (2011, 15 September). [Interview].
- CIA. (2002). Unclassified Report North Korean Nuclear Weapons: CIA Estimate for Congress: The Federation of American Scientists.
- Clinton, B. (2005). My Life. London: Arrow Books.
- CNN. (2002, 17 February). Colin Powell Discusses Situation in Afghanistan; Joe Biden and Henry Hyde Discuss Bush Policies, *CNN*. Retrieved from http://edition.cnn.com/TRANSCRIPTS/0202/17/le.00.html
- CNN. (2003, 1 August). Bush 'Optimistic' about North Korea Talks: Pyongyang Agrees to Six-Nation Talks on Nuclear Standoff, *CNN*. Retrieved from http://edition.cnn.com/2003/ALLPOLITICS/08/01/bush.n.korea/
- Cody, E., & Faiola, A. (2004, 20 April). N. Korea's Kim Reportedly in China for Talks; Meetings With Leaders Said To Focus on Nuclear Standoff, *The Washington Post*. Retrieved from http://www.nexis.com
- Copeland, D. (2000). Review: the Constructivist Challenge to Structural Realism: A Review Essay. *International Security*, 25(2), 187-212.
- Cordesman, A. (2002). Strategic Threats and National Missile Defenses: Defending the U.S. Homeland. Westport: Praeger Publishers.
- Cornwell, R. (2003, 26 April). Bush Accuses North Korea of Blackmail After Talks Fail, *The Independent (London)*. Retrieved from http://www.nexis.com
- Cossa, R. (1999, 9 July). The U.S.-DPRK Agreed Framework: Is It Still Viable? Is It Enough? *PacNet No. 27 Pacific Forum CSIS.* from http://csis.org/publication/pacnet-27-july-9-1999-us-dprk-agreed-framework-it-still-viable-it-enough
- Cox, M. (1995). US Foreign Policy after the Cold War: Superpower without a Mission? London: Pinter.
- Cumings, B. (2008). Korea's Two North-South Summits and the Future of Northeast Asia: Back to the Future. from http://www.japanfocus.org/-Bruce-Cumings/2873
- Dalton, T., & Snyder, S. (2004). Ties that Bind? Culture, Values, and Ideation in U.S.-Korea-Japan Security Cooperation. In T. H. Kim & B. Glosserman (Eds.), *The Future of U.S.-ROK-Japan Relations: Balancing Values and Interests*. Washington, D.C.: Center for Strategic and International Studies.
- Davis, S. (2003). Russian Far East: The Last Frontier? London: Routledge.
- Deng, Y. (2008). *China's Struggle for Status: the Realignment of International Relations*. Cambridge: Cambridge University Press.
- DeRochie, P. (2011). The Driving Factor: Songun's Impact on North Korean Foriegn Policy. *International Affairs Review*, 20(1), 1-20.
- Deudney, D. (2011). Anarchy and Violence Interdependence. In K. Booth (Ed.), *Realism and World Politics*. Abingdon: Routledge.
- Drifte, R. (2001). The Perry Report and US-North Korea Relations. Asien, 79, 50-61.
- Dunne, T. (1995). The Social Construction of International Society. *European Journal of International Relations*, 1(3), 367-389.
- Eberstadt, N. (1998). North Korea's Unification Policy: 1948-1996. In S. Kim (Ed.), *North Korean Foreign Relations in the Post-Cold War Era*. Hong Kong, Oxford University Press.
- Elman, C. (1996). Horses for Courses: Why nor Neorealist Theories of Foreign Policy? *Security Studies*, 6(1), 7-53.
- Engelhardt, M. (1996). Rewarding Nonproliferation: the South and North Korean Cases. *The Nonproliferation Review*, *3*(3), 31-37.
- Epstein, W. (1977). Why States Go -- And Don't Go Nuclear. *Annals of the American Academy of Political and Social Science*, 430(1), 16-28.

- Eyre, D., & Suchman, M. (1996). Status, Norms, and the Proliferation of Conventional Weapons: An Institutional Theory Approach. In P. Katzenstein (Ed.), *The Culture of National Security: Norms and Identity in World Politics*. New York: Columbia University Press.
- Foreign Ministry Spokesperson Hong Lei's Regular Press Conference on November 25, 2010. (2010). [Press release]. Retrieved from http://dk.china-embassy.org/eng/fyrth/t772580.htm
- Foreign Ministry Spokesperson Qin Gang's Remarks on the Announcement of the ROK-US Joint Military Exercises. (2010). [Press release]. Retrieved from http://www.fmprc.gov.cn/eng/xwfw/s2510/t718571.htm
- Frankel, B. (1993). The Brooding Shadow: Systemic Incentives and Nuclear Weapons Proliferation. *Security Studies*, 2(3-4), 37-78.
- Freedman, L. (1989). The Evolution of Nuclear Strategy. Basingstoke: Palgrave Macmillan.
- Fuerth, L. (2003, 1 January). Outfoxed by North Korea, *The New York Times*. Retrieved from http://www.nytimes.com/2003/01/01/opinion/outfoxed-by-north-korea.html
- Funabashi, Y. (2007). *The Peninsula Question: A Chronicle of the Second Korean Nuclear Crisis*. Washington, D.C.: The Brookings Institution.
- George, A., & Smoke, R. (1974). *Detterence in American Foreign Policy: Theory and Practice*. New York: Columbia University Press.
- Gills, B. (1996). Korea Versus Korea: A Case of Contested Legitimacy. London: Routledge.
- Gilmore, G. (2003, 31 May). North Korea's Nuclear Program Threatens Regional, Global Stability. U.S. Department of Defense. *U.S. Department of Defense*. from http://www.defense.gov/News/NewsArticle.aspx?ID=28928
- Gilpin, R. (1981). War and Change in World Politics. Cambridge: Cambridge University Press.
- Glaser, B. (2009). China's Policy in the Wake of the Second DPRK Nuclear Test. *WSI China Security*, *5*(2), 1-11.
- Glaser, B., & Billingsley, B. (2012). Reordering Chinese Priorities on the Korean Peninsula *A Report of the CSIS Freeman Chair in China Studies* (pp. 1-24): Center for Strategic and Internatinal Studies.
- Glaser, C. (1994). Realists as Optimists: Cooperation as Self-Help. *International Security*, 19(3), 50-90.
- Glaser, C. (1997). The Security Dilemma Revisited. World Politics, 50(1), 171-201.
- Gluck, C. (2002, 18 September). North Korea Admits Kidnap Plot, *The Times (London)*. Retrieved from http://www.nexis.com
- Goodby, J., & Gross, D. (2004, 3 September). Step by Step: The 'Libya Model' Could Help Disarm North Korea, *The New York Times*. Retrieved from http://www.nytimes.com/2004/09/03/opinion/03iht-edgross_ed3_.html
- Grisafi, J. (2013, 3 January). Regime Activity Limited and Conservative in December. *NK News*. from http://www.nknews.org/2014/01/regime-activity-limited-and-conservative-in-december/
- Guillaume, G. (2004). Terrorism and International Law. *International and Comparative Law Quarterly*, 53, 537-548.
- Ha, Y. C., & Shin, B. S. (2006). Russian Nonproliferation Policy and the Korean Peninsula: The Strategic Studies Institute.
- Hagstrom, L., & Soderberg, M. (2006). Taking Japan-North Korea Relations Seriously: Rationale and Background. *Pacific Affairs*, 79(3), 373-385.

- Hall, K. (2007, 16 May). Bank Owner Disputes Money-Laundering Allegations, *McClatchy Newspapers*. Retrieved from http://www.mcclatchydc.com/2007/05/16/16191/bank-owner-disputes-money-laundering.html
- Hamm, T. Y. (1999). *Arming the Two Koreas: State, Capital, and Military Power*. London: Routledge.
- Han, D. H. (2011, 2 September). [Interview].
- Han, Y. J., Kim, Y. S., & Kim, K. C. (2005, 10 April). How We Saw the Sunshine Policy in North Korea: Analysis by Three North Korean Defectors. from http://www.dailynk.com/english/read.php?cataId=nk00100&num=108
- Harlan, C. (2010a, 1 September). China Pushes Six-Party Talks with North Korea, but Others Remain Skeptical, *The Washington Post*. Retrieved from http://www.washingtonpost.com/wp-dyn/content/article/2010/08/31/AR2010083101909.html
- Harlan, C. (2010b, 16 August). South Korean Leader Has Reunification Plan, *The Washington Post*. Retrieved from http://www.washingtonpost.com/wp-dyn/content/article/2010/08/15/AR2010081502100.html
- Harrison, S. (2002). *Korean Endgame: A Strategy for Reunification and U.S. Disengagement*. Princeton: Princeton University Press.
- Harrison, S. (2005). Did North Korea Cheat? Foreign Affairs, 84(1), 99-110.
- Hathaway, R., & Tama, J. (2004). The U.S. Congress and North Korea During the Clinton Years: Talk Tough, Carry a Small Stick. *Asian Survey*, 44(5), 711-733.
- Hauben, R. (2007, 5 June). Behind the Blacklisting of Banco Delta Asia. *NAPSNet Policy Forum 07-044*. from http://nautilus.org/napsnet/napsnet-policy-forum/behind-the-blacklisting-of-banco-delta-asia/#axzz2dg24PPH9
- Hayes, P., Moon, C. I., & Bruce, S. (2011). Park Chung Hee, the US-ROK Strategic Relationship, and the Bomb. *Annals of the American Academy of Political and Social Science*. from http://japanfocus.org/-scott-bruce/3630
- Hayes, P., & Noerper, S. (1997). The Future of the U.S.-ROK Alliance. In Y. W. Kihl, R. Scalapino & P. Hayes (Eds.), *Peace and Security in Northeast Asia: the Nuclear Issue and the Korean Peninsula*. Armonk: M.E. Sharpe, Inc.
- Hecker, S. (2010). Lessons Learned from the North Korean Nuclear Crises. *Daedalus*, 139(1), 44-56.
- Hecker, S., Lee, S., & Braun, C. (2010, 13 July). North Korea's Choice: Bombs over Electricity. *NAPSNet Policy Forum 10-038*. from http://nautilus.org/napsnet/napsnet-policy-forum/north-koreas-choice-bombs-over-electricity/#axzz2dCJ9DMt3
- Henriksen, T. (1996). *Clinton's Foreign Policy in Somalia, Bosnia, Haiti, and North Korea*. California: Hoover Institution on War, Revolution and Peace.
- Henriksen, T. (2001). The Rise and Decline of Rogue States. *Journal of International Affairs*, 54(2), 249-371.
- Hersh, S. (2003, 27 January). The Cold Test: What the Administration Knew about Pakistan and the North Korean Nuclear Program. *The New Yorker*.
- Herz, J. (1950). Idealist Internationalism and the Security Dilemma. *World Politics*, 2(2), 157-180.
- Herz, J. (1962). *International Politics in the Atomic Age*. New York: Columbia University Press.
- Hong, S. (2011). The Search for Deterrence: Park's Nuclear Option. In B. K. Kim & E. Vogel (Eds.), *The Park Chung Hee Era: the Transformation of South Korea*. Cambridge: Harvard University Press.

- Hopf, T. (1998). The Promise of Constructivism in International Relations Theory. *International Security*, 23(1), 171-200.
- Hopf, T. (2002). Social Construction of Foreign Policy: Identities and Foreign Policies, Moscow, 1955 and 1999. Ithaca: Cornell University Press.
- Hoyt, P. (2000). The "Rogue State" Image in American Foreign Policy. *Global Society*, 14(2), 297-310.
- Hughes, C. (1996). The North Korean Nuclear Crisis and Japanese Security. *Survival*, 38(2), 79-103.
- Hughes, C. (2005). *Japan's Re-emergence as a 'Normal' Military Power*. Abingdon: Routledge.
- Hughes, C. (2006). Japan and Multilateralism in the North Korean Nuclear Crisis: Road Map or Dead End? . In L. Hagstrom & M. Soderberg (Eds.), *North Korea Policy: Japan and the Great Powers*. Abingdon: Routledge.
- Huntley, W. (2009). Bucks For the Bang: North Korea's Nuclear Program and Northeast Asian Military Spending. *Asian Perspective*, 33(4), 147-182.
- Ikenberry, G. J. (2008). The Political Foundations of American Relations with East Asia. In G. J. Ikenberry & C. I. Moon (Eds.), *The United States and Northeast Asia: Debates, Issues, and New Order*. Plymouth: Rowman & Littlefield.
- International Business Publications. (2003). *North Korea: General Secretary Kim Jong Il Handbook* (Vol. 1). Washington, D.C.: International Business Publications, USA.
- Ito, M. (2010, 21 May). Hatoyama Slams North Over Cheonan Sinking, *The Japan Times*. Retrieved from http://www.japantimes.co.jp/news/2010/05/21/national/hatoyama-slams-north-over-cheonan-sinking/
- Izumikawa, Y. (2011). North Korea Problems and US-Japan Relations: A View from Japan. In Y. Tatsumi (Ed.), *North Korea: Challenge for the US-Japan Alliance* (pp. 26-43). Washington, D.C.: The Henry L. Stimson Center.
- Japan Economic Newswire. (1988, 7 July). Asian News-South Korea; South Korean President Roh Unveils Plan to Promote Inter-Korean Exchanges. Retrieved from http://www.nexis.com
- Japan Economic Newswire. (1998, 3 June). Japan to Pay 1 Bil. Dlrs for KEDO Project as Planned, *Japan Economic Newswire*. Retrieved from http://www.nexis.com
- Jeon, M. (2009). The Kim Jong-il Regime's "Military-First Politics": Structure and Strategy of Discourse. *The Review of Korean Studies*, *12*(4), 181-204.
- Jepperson, R., Wendt, A., & Katzenstein, P. (1996). Norms, Identity, and Culture in National Security. In P. Katzenstein (Ed.), *The Culture of National Security: Norms and Identity in World Politics*. New York: Columbia University Press.
- Jervis, R. (1976). *Perception and Misperception in International Politics*. Princeton: Princeton University Press.
- Jervis, R. (1978). Cooperation under the Security Dilemma. World Politics, 30(2), 167-214.
- Jervis, R. (1979). Deterrence Theory Revisited. World Politics, 31(2), 289-324.
- Jervis, R. (1988). War and Misperception. *The Journal of Interdisciplinary History*, 18(4), 675-700.
- Jervis, R. (1999). The Utility of Nuclear Deterrence. In R. Art & K. Waltz (Eds.), *The Use of Force: Military Power and International Politics*. Lanham: Rowman & Littlefield.
- Jervis, R. (2001). Was the Cold War a Security Dilemma? *Journal of Cold War Studies*, 3(1), 36-60.
- Jeung, Y. T. (2007). *North Korea's Civil-Military-Party Relations and Regime Stability*. Seoul: Korea Institute for National Unification.

- Jo, Y. H., & Marshall, R. (1979). Linkage Sources of North Korean Foreign Policy. In J. K. Park & J. G. Kim (Eds.), *The Politics of North Korea*. Seoul: The Institute of Far Eastern Studies.
- . Joint Declaration of South and North Korea on the Denuclearization of the Korean Peninsula. (1992). The Federation of American Scientists Retrieved from http://fas.org/news/dprk/1992/920219-D4129.htm.
- Joo, S. H. (2003). Russia and the Korean Peace Process. In T. H. Kwak & S. H. Joo (Eds.), *The Korean Peace Process and the Four Powers*. Aldershot: Ashgate Publishing, Ltd.
- Joo, S. H. (2007). Russia and the North Korean Nuclear Crisis. In S. H. Joo & T. H. Kwak (Eds.), *North Korea's Second Nuclear Crisis and Northeast Asian Security*. Aldershot: Ashgate Publishing Limited.
- Joo, S. H. (2009). North Korean Policy toward Russia. In T. H. Kwak & S. H. Joo (Eds.), *North Korea's Foreign Policy Under Kim Jong Il: New Perspectives*. Farnham: Ashgate Publishing Limited.
- Kahler, M. (1998). Rationality in International Relations. *International Organization*, 52(4), 919-941.
- Kahn, J. (2005a, 12 November). North Korea and U.S. Spar, Causing Talks to Stall, *The New York Times*. Retrieved from http://www.nytimes.com/2005/11/12/international/europe/12korea.html?red
- Kahn, J. (2005b, 19 September). North Korea Says It Will Abandon Nuclear Efforts, *The New York Times*. Retrieved from http://www.nytimes.com/2005/09/19/international/asia/19korea.html?_r=0
- Kahn, J. (2006, 23 December). Talks End on North Korea's Nuclear Weapons, *The New York Times*. Retrieved from http://www.nytimes.com/2006/12/23/world/asia/23korea.html?r=0&pagewanted=print
- Kahn, J., & Sanger, D. (2005, 20 September). U.S.-Korean Deal on Arms Leaves Key Points Open, *The New York Times*. Retrieved from http://query.nytimes.com/gst/fullpage.html?res=9905E4DA1630F933A1575AC0A9639C8B63&pagewanted=1
- Kahn, J., & Weisman, S. (2007, 15 March). North Korea Is Said to Tie Nuclear Accord to Freeing of Funds, *The New York Times*. Retrieved from http://www.nytimes.com/2007/03/15/world/asia/15iht-web-0315korea.4915291.html
- Kan, S. (2014). China and Proliferation of Weapons of Mass Destruction and Missiles: Policy Issues *CRS Report for Congress* (pp. 1-74): The Federation of American Scientists.
- Kang, C. S. E., & Kaseda, Y. (2001). Korea and the Dynamics of Japan's Post-Cold War Security Policy. *World Affairs*, 164(2), 51-59.
- Kang, D. (1995). Rethinking North Korea. Asian Survey, 35(3), 253-267.
- Kang, D. (1998). North Korea: Deterrence Through Danger. In M. Alagappa (Ed.), *Asian Security Practice: Material and Ideational Influences*. Stanford: Stanford University Press.
- Kang, D. (2003). Threatening, but Deterrence Works. In D. Kang & V. Cha (Eds.), *Nuclear North Korea: A Debate on Engagement Strategies*. New York: Columbia University Press.
- Kang, S. M. (2006). Six-Party Talks: "The Other Four" *SAIS U.S.-Korea Yearbook 2006* (pp. 87-94): The U.S.-Korea Institute at SAIS, the Paul H. Nitze School of Advanced International Studies, Johns Hopkins University.

- Kang, T. (1979). Changes in the North Korean Personality from Confucian to Communist. In J. K. Park & J. G. Kim (Eds.), *The Politics of North Korea*. Seoul: The Institute for Far Eastern Studies, Kyungnam University.
- KCNA. (1998a, 2 September). Japan Warned to Act with Prudence, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (1998b, 26 May). Rodong Sinmun Calls for Creating New Chollima Speed, Forced March Speed, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (1998c, 13 October). U.S. warned not to provoke DPRK, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (1999a, 1 January). Joint Editorial Published on New Year, *Korean Central News Agency* Retrieved from http://www.kcna.co.jp
- KCNA. (1999b, 23 October). KCNA on Interference in DPRK's Internal Affairs, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (1999c, 5 January). Papers Call for Pushing Ahead with General Onward March for Building Powerful Nation, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (1999d, 6 November). S. Korean Authorities "Sunshine Policy" Flayed, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2002a, 25 October). Conclusion of Non-Aggression Treaty between DPRK and US Called For, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2002b, 22 November). DPRK Foreign Ministry Spokesman on U.S. Decision to Stop Supplying Heavy Oil, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2002c, 31 January). Spokesman for DPRK Foreign Ministry Slams Bush's Accusations, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003a, 3 January). Japan Urged to Drop U.S.-Toeing Policy, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003b, 21 March). KCNA Dismisses Large-Scale U.S. War Maneuvers as Alarming Development, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003c, 18 November). KCNA Refutes Japan's Talk about "Military Threat", *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003d, 29 August). Keynote Speeches Made at Six-Way Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003e, 1 August). Spokesman for DPRK Foreign Ministry on Recent DPRK-U.S. Contact, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2003f, 2 June). U.S. Urged to Accept DPRK-Proposed Format of Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2004, 29 February). DPRK Foreign Ministry Spokesman on Six-Way Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2005a, 3 March). Memorandum of DPRK Foreign Ministry, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2005b, 21 September). Spokesman for DPRK Foreign Ministry on Six-Party Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2006a, 3 October). DPRK Foreign Ministry Clarifies Stand on New Measure to Bolster War Deterrent, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2006b, 11 October). DPRK Foreign Ministry Spokesman on U.S. Moves Concerning Its Nuclear Test, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp

- KCNA. (2006c, 17 October). DPRK Foreign Ministry Spokesman Totally Refutes UNSC "Resolution", *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2006d, 6 December). KCNA Slashes at Japan's Intention to Play Bad Role at Six-Party Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2008a, 4 January). DPRK Foreign Ministry Spokesman on Issue of Implementation of October 3 Agreement, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2008b, 26 August). Foreign Ministry's Spokesman on DPRK's Decision to Suspend Activities to Disable Nuclear Facilities, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2008c, 23 October). Japan Accused of Disturbing Six-Party Talks, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2008d, 30 May). S. Korean Authorities' Traitorous "Pragmatism" under Fire, *Korean Central News Agency*. Retrieved from www.kcna.co.jp
- KCNA. (2009a, 7 April). Aggressive Nature of U.S. Moves to Build Missile Shields Disclosed, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009b, 13 June). DPRK Foreign Ministry Declares Strong Counter-Measures against UNSC's "Resolution 1874", *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009c, 4 May). DPRK Foreign Ministry Spokesman Blasts U.S. Misinterpretation of Satellite Launch as Ballistic Missile Launch, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009d, 14 April). DPRK Foreign Ministry Vehemently Refutes UNSC's "Presidential Statement", *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009e, 13 January). DPRK Foreign Ministry's Spokesman Dismisses U.S. Wrong Assertion, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009f, 30 September). KCNA Dismisses S. Korean Chief Executive's "Proposal" as Rubbish, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2009g, 5 April). KCNA on DPRK's Successful Launch of Satellite Kwangmyongsong-2 *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2010a, 8 May). Kim Jong Il Makes Unofficial Visit to China, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2010b, 20 May). National Defence Commission Issues Statement *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2010c, 19 December). Rodong Sinmun Warns of Danger of Japan's Militarism, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2011, 8 January). Rodong Sinmun Calls for Unconditional and Early Opening of Negotiations, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2012a, 17 August). Anti-Unification Remarks of Traitor Lee Myung Bak Flayed, Korean Central News Agency. Retrieved from www.kcna.co.jp
- KCNA. (2012b, 10 August). DPRK History Society Discloses Truth about "Yushin" Dictatorship, *Korean Central News Agency*. Retrieved from www.kcna.co.jp
- KCNA. (2012c, 12 November). KCNA Commentary Slams park Geun Hye's Commitments to Confrontation, *Korean Central News Agency*. Retrieved from www.kcna.co.jp
- KCNA. (2012d, 15 April). Kim Jong Un Speaks at Military Parade, *Korean Central News Agency*. Retrieved from www.kcna.co.jp

- KCNA. (2012e, 13 April). Korean People Hold Kim Jong II in High Esteem as Eternal NDC Chairman, *Korean Central News Agency*. Retrieved from www.kcna.co.jp
- KCNA. (2012f, 3 November). S.Korea's "Saenuri Party" Accused of Burlesque for Confrontation with Fellow Countrymen, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2012g, 21 June). U.S.-Japan-S. Korea Triangular Military Alliance Censured: Rodong Sinmun, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2013a, 8 January). Kimilsungism-Kimjongilism, WPK's Guiding Idea, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2013b, 1 January). New Year Address Made by Kim Jong Un, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2013c, 5 June). Rodong Sinmun Slams Moves for U.S.-Japan-South Korea Triangular Military Alliance, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- KCNA. (2013d, 18 March). S. Korea Is Violator of North-South Agreements on Nonaggression: Rodong Sinmun, *Korean Central News Agency*. Retrieved from http://www.kcna.co.jp
- Kennan, G. (1961). *Russia and the West under Lenin and Stalin*. London: Hutchinson & Co. (Publishers) Ltd.
- Keohane, R. (1986). Theory of World Politics: Structural Realism and Beyond. In R. Keohane (Ed.), *Neorealism and Its Critics*. New York: Columbia University Press.
- Kessler, G. (2005, 13 July). South Korea Offers to Supply Energy if North Gives Up Arms, *The Washington Post*. Retrieved from http://www.washingtonpost.com/wp-dyn/content/article/2005/07/12/AR2005071200220.html
- Kessler, G. (2010, 27 May). Analysis: North Korea Tests U.S. Policy of 'Strategic Patience', *The Washington Post*. Retrieved from http://www.washingtonpost.com/wp-dyn/content/article/2010/05/26/AR2010052605047.html
- Kihl, Y. W. (1984). *Politics and Policies in Divided Korea: Regimes in Contest.* London: Westview Press, Inc.
- Kihl, Y. W. (2006). Staying Power of the Socialist "Hermit Kingdom". In Y. W. Kihl & H. N. Kim (Eds.), *North Korea: The Politics of Regime Survival*. Armonk: M.E. Sharpe.
- Kim, B.-K. (2001, 23 July). Step-by-Step Nuclear Confidence Building on the Korean Peninsula: Where Do We Start?
- Kim, C. N. (2005). The Roh Moo Hyun Government's Policy Toward North Korea *The East-West Center Working Paper: Politics, Governance, and Security Studies No. 11, August 2005* (pp. 1-28): The East-West Center.
- Kim, D. J. (1994). *Korea and Asia: A Collection of Essays, Speeches, and Discussions*. Seoul: The Kim Dae Jung Peace Foundation Press.
- Kim, D. S. (1994). China's Policy Toward North Korea and Cooperation Between South Korea and China. *The Korean Journal of International Studies*, 25(1), 29-46.
- Kim, H. A. (2004). Korea's Development under Park Chung Hee: Rapid Industralization, 1961-79. London: RoutledgeCurzon.
- Kim, H. J. (1977). *The Unification Policy of South and North Korea: A Comparative Study*. Seoul: Seoul National University Press.
- Kim, H. J. (1993). The Republic of Korea's Northern Policy: Origin, Development, and Prospects. In J. Cotton (Ed.), *Korea Under Roh Tae Woo: Democratisation, Northern Policy, and Inter-Korean Relations*. New South Wales: Allen & Unwin.

- Kim, H. N. (2006a). Japanese-North Korean Relations under the Koizumi Government. In Y. W. Kil & H. N. Kim (Eds.), *North Korea: The Politics of Regime Survival*. Armonk: M.E. Sharpe.
- Kim, H. N. (2006b). The Koizumi Government and the Politics of Normalising Japanese-North Korean Relations *East-West Center Working Papers: Politics, Governance, and Security Series No. 14* (pp. 1-47): The East-West Center.
- Kim, H. N. (2006c). South-North Korean Relations Under the Roh Moo-Hyun Government. *International Journal of Korean Studies*, *10*(1), 37-59.
- Kim, I. (2006). Kim Jong Il's Military-First Politics. In Y. W. Kihl & H. N. Kim (Eds.), *North Korea: The Politics of Regime Survival*. Armonk: M.E. Sharpe.
- Kim, I. (2009). Kim Jong II and Denuclearization of the Korean Peninsula. In T. H. Kwak & S. H. Joo (Eds.), *North Korea's Foreign Policy under Kim Jong II: New Perspectives*. Burrington, VT: Ashgate Publishing Limited.
- Kim, I. S. (1981). On a Historic Victory in the Fatherland Liberation War and the Tasks of the People's Army, Speech before the Officers and Men of the Unit No. 256 of the Korean People's Army, October 23, 1953 Kim II Sung Selected Works Vol. 8, (pp. 112-125).
- Kim, J. (2008). *The Source of North Korean Terrorism: Analyses at Three Levels.* (PhD), University of Hawai'i at Manoa.
- Kim, J., Kim, D., Kim, H. W., Kim, Y., & Lee, W. (2013). *The Second Term Obama Administration's Policy towards the Korean Peninsula*. Seoul: Korea Institute for National Unification.
- Kim, J. H. (2011). The Armed Forces. In B. K. Kim & E. Vogel (Eds.), *The Park Chung Hee Era: the Transformation of South Korea*. Cambridge: Harvard University Press.
- Kim, S. (2003). China and North Korea in a Changing World *Asia Program Special Report No. 115: Uneasy Allies: Fifty Years of China-North Korea Relations*: Woodrow Wilson International Center for Scholars.
- Kim, S. (2004). The Changing Role of China on the Korean Peninsula. *International Journal of Korean Studies*, 8(1), 79-114.
- Kim, S. (2005). China and the Six-Party Talks: the New Turn to Mediation Diplomacy. *International Journal of Korean Studies*, 4(2), 117-144.
- Kim, S. (2007). North Korean Foreign Relations in the Post-Cold War Period (pp. 1-123): Strategic Studies Institute: United States Army War College.
- Kim, S. (2010, 7 May). (4th LD) N. Korea's Kim Reaffirms Commitment to Denuclearized Korean Peninsula: Report, *Yonhaps News Agency*. Retrieved from http://english.yonhapnews.co.kr/search1/2603000000.html?cid=AEN2010050700520
- Kim, S., & Lee, T. H. (2002). Chinese-North Korean Relations: Managing Asymmetrical Interdependence. In S. Kim & T. H. Lee (Eds.), *North Korea and Northeast Asia*. Oxford: Rowmand Littlefield Publishers, Inc.
- Kim, S. Y. (2008). The Torturous Dilemma: The 2008 Six-Party Talks and U.S.-DPRK Relations *SAIS U.S.-Korea Yearbook 2008* (pp. 107-123): The U.S.-Korea Institute at SAIS, the Paul H. Nitze School of Advanced International Studies, Johns Hopkins University
- Kim, S. Y. (2013, 1 April). Byungjin Lives as Kim Seeks Guns and Butter. from http://www.dailynk.com/english/read.php?cataId=nk01700&num=10453
- Kim, Y. (2008). North Korea: a Perpetual Rogue State? In G. J. Ikenberry & C. I. Moon (Eds.), *The United States and Northeast Asia: Debates, Issues, and New Order*. Plymouth: Rowman & Littlefield Publishers, Inc.

- Kim, Y. (2011). *North Korean Foreign Policy: Security Dilemma and Succession*. Plymouth: Lexington Books.
- Kim, Y. H. (2010, 8 August). North Korea Obtained HEU from Pakistan. *Daily NK*. from http://www.dailynk.com/english/read_print.php?cataId=nk02200&num=6680
- Kim, Y. H., & Kang, M. J. (2013, 4 December). Jang's Economic Policy Hangover. from http://www.dailynk.com/english/read.php?cataId=nk01500&num=11235
- Kim, Y. Y. (2011, 20 September). [Interview].
- Kleine-Ahlbrandt, S. (2011, 21 January). Despite Reports, China's North Korea Policy Stays the Same. from http://www.crisisgroup.org/en/regions/asia/north-east-asia/china/Ahlbrandt-North-Korea-Despite-Reports-Chinas-North-Korea-Policy-Stays-the-Same.aspx
- Kleine-Ahlbrandt, S. (2013, 2 July). China's North Korea Policy: Backtracking from Sunnylands? *38 North*. from http://38north.org/2013/07/skahlbrandt070213/
- Knopf, J. (2009). Three Items in One: Deterrence as Concept, Research Programe, and Political Issue. In T.V. Paul, P. Morgan & J. Wirtz (Eds.), *Complex Deterrence: Strategy in the Global Age*. Chicago, : The University of Chicago Press.
- Koh, B. C. (1971). Dilemmas of Korean Unification. Asian Survey, 11(5), 475-495.
- Koh, B. C. (1980). Inter-Korean Relations: Seoul's Perspective. *Asian Survey*, 20(11), 1108-1122.
- Koh, B. C. (1998). North Korean Policy towards the United States. In D. S. Suh & C. J. Lee (Eds.), *North Korea after Kim Il Sung*. London: Lynne Rienner Publishers, Inc.
- Koh, B. C. (2005, 14 April). Policy Forum 05-32A: "Military-First Politics" And Building A "Powerful And Prosperous Nation" in North Korea. *NAPSNet Policy Forum, Nautilus Institute for Security and Sustainability* from http://nautilus.org/napsnet/napsnet-policy-forum/military-first-politics-and-building-a-powerful-and-prosperous-nation-in-north-korea/
- Koh, Y. H. (2005). Two Years of the Roh Moo-Hyun Administration's North Korea Policy. *Sejong Policy Review*, *2*, 201-219.
- Koo, K. W. (2011, 9 September). [Interview].
- Korea JoongAng Daily. (2004, 14 November). Roh Calls for Softer Stance against North, *Korea JoongAng Daily*. Retrieved from http://koreajoongangdaily.joins.com/news/article/article.aspx?aid=2492321
- Korea JoongAng Daily. (2012, 9 October). Rep Says Roh Didn't Recognize NLL, *Korea JoongAng Daily*. Retrieved from http://koreajoongangdaily.joinsmsn.com/news/article/article.aspx?aid=2960518&cloc=joongangdaily%7Chome%7Cnewslist1
- Kristof, N. (1993, 24 March). China Opposes U.N. Over North Korea, *The New York Times*. Retrieved from http://www.nytimes.com/1993/03/24/world/china-opposes-un-over-north-korea.html
- Kupchan, C., & Kupchan, C. (1991). Concerts, Collective Security, and the Future of Europe. *International Security*, *16*(1), 114-161.
- Kupchan, C., & Kupchan, C. (1995). The Promise of Collective Security. *International Security*, 20(1), 52-61.
- Kwak, T. H., & Joo, S. H. (2007). Introduction. In S. H. Joo & T. H. Kwak (Eds.), *North Korea's Second Nuclear Crisis and Northeast Asian Security*. Aldershot: Ashgate Publishing Limited.
- Kwon, K. J. (2012, 21 June). North Korea Slams Joint Drill among U.S., Japan, and South Korea, *CNN*. Retrieved from http://edition.cnn.com/2012/06/21/world/asia/north-korea-drills/

- Kwon, M. (2011, 8 September). [Interview].
- Lague, D., & Greenlees, D. (2007, 18 January). Squeeze on Banco Delta Asia Hit North Korea Where It Hurts Asia Pacific International Herald Tribune, *The New York Times*. Retrieved from http://www.nytimes.com/2007/01/18/world/asia/18ihtnorth.4255039.html?pagewanted=all
- Lake, A. (1994). Confronting Backlash States. Foreign Affairs, 73(2), 45-55.
- Lankov, A. (2013). *The Real North Korea: Life and Politics in the Failed Stalinist Utopia*. Oxford: Oxford University Press.
- Lebow, R. (1994). The Long Peace, the End of the Cold War, and the Failure of Realism. *International Organization*, 48(2), 249-277.
- Lebow, R., & Stein, J. (1989). Rational Deterrence Theory: I Think, Therefore I Deter. *World Politics*, 41(2), 208-224.
- Lee, C. D. (2012, 10 December). Obama's 'Strategic Patience' Fuels N.Korea's Weapons Program: Joel Wit, *Yonhaps News Agency*. Retrieved from http://english.yonhapnews.co.kr/national/2012/12/10/56/03010000000AEN201212100 01200315F.HTML
- Lee, C. J. (2006). *A Troubled Peace: U.S. Policy and the Two Koreas*. Baltimore: The Johns Hopkins University Press.
- Lee, C. S., & Sohn, H. S. (1994). South Korea in 1993: the Year of the Great Reform. *Asian Survey*, *34*(1), 1-9.
- Lee, H. (2014). Rising China and the Evolution of China-North Korea Relations. *The Korean Journal of International Studies*, 12, 99-111.
- Lee, H. K. (1974). Constitutional Development in South Korea. *Hong Kong Law Journal*, 4(1), 41-51.
- Lee, H. S. (2001). North Korea: A Strange Socialist Fortress. Westport: Praeger.
- Lee, H. Y. (2003). South Korea in 2002: Multiple Political Dramas. *Asian Survey*, 43(1), 64-77.
- Lee, S. (2010a, 18 July). China has Different View on Cheonan, *The Korea Times*. Retrieved from http://www.koreatimes.co.kr/www/news/nation/2010/07/116_69702.html
- Lee, S. (2010b, 29 July). Post-Cheonan Dilemma: How to Deal with China. Korea Times, *The Korea Times*. Retrieved from http://www.koreatimes.co.kr/www/news/nation/2010/07/113_70422.html
- Legro, J., & Moravcsik, A. (1999). Is Anybody Still a Realist? *International Security*, 24(2), 5-55.
- Lerner, M. (2010). "Mostly Propaganda in Nature:" Kim Il Sung, the Juche Ideology, and the Second Korean War. *North Korea International Documentation Project Working Paper* #3. from http://www.wilsoncenter.org/sites/default/files/NKIDP_Working_Paper_3_Kim_Il_Sung_Juche_Ideology_Second_Korean_War_web_0.pdf
- Les Aspin. (1993). *Report on the Bottom-Up Review*. Department of Defense, United States of America Retrieved from http://www.dod.mil/pubs/foi/administration and Management/other/515.pdf.
- Levin, N., & Han, Y. S. (2002). Sunshine in Korea: The South Korean Debate over Policies Toward North Korea. from The RAND Corporation http://www.rand.org/pubs/monograph_reports/MR1555.html
- Lim, B. M. (2006, 19 July). South Korean President Says North Korea's Missile Tests Could Spark Arms Race, *The Associated Press*. Retrieved from http://www.nexis.com
- Lim, E. C. (2011, 9 September). [Interview].

- Litwak, R. (2000). Rogue States and U.S. Foreign Policy: Containment after the Cold War. Washinton, D.C.: Woodrow Wilson Center Press.
- Litwak, R. (2002). The New Calculus of Pre-Emption. Survival, 44(4), 53-80.
- Lobell, S. (2009). Threat Assessment, the State, and Foreign Policy: a Neoclassical Realist Model. In S. E. Lobell, N. M. Ripsman & J. W. Taliaferro. (Eds.), *Neoclassical Realism, the State, and Foreign Policy*. Cambridge: Cambridge University Press.
- Lukin, A. (2003). Russian Policy towards the Korean Peninsula. *The Korean Journal of International Studies*, 30(1), 75-97.
- Lukin, A. (2013, 3 March). Russia Shows Little Concern Over North Korean Nukes (for now). *East Asia Forum*. from http://www.eastasiaforum.org/2013/03/03/russia-shows-little-concern-over-north-korean-nukes-for-now/
- Mack, A. (1993). The Nuclear Crisis on the Korean Peninsula. Asian Survey, 33(4), 339-359.
- Mansourov, A. (2013, 9 December). North Korea: The Dramatic Fall of Jang Song Thaek. from http://38north.org/2013/12/amansourov120913/
- Manyin, M., Chanlett-Avery, E., & Nikitin, M. (2011). U.S.-South Korea Relations *CRS Report for Congress* (pp. 1-35): U.S. Department of State.
- Manyin, M., & Nanto, D. (2011). The Kaesong North-South Korean Industrial Complex *CRS Report for Congress* (pp. 1-25): The Federation of American Scientists
- Manyin, M., & Nikitin, M. (2013). Foreign Assistance to North Korea *CRS Report for Congress* (pp. 1-25): the Federation of American Scientists.
- Martin, C. (1999). Lessons of the Agreed Framework for Using Engagement as a Nonproliferation Tool. *The Nonproliferation Review*, *6*(4), 35-50.
- Martin, C. (2002). Rewarding North Korea: Theoretical Perspectives on the 1994 Agreed Framework. *Journal of Peace Research*, 39(1), 51-68.
- Martin, C. (2007). U.S. Policy Toward North Korea Under George W. Bush: A Critical Appraisal. In T. H. Kwak & S. H. Joo (Eds.), *North Korea' Second Nuclear Crisis and Northeast Asian Security*. Aldershot: Ashgate Publishing Limited.
- Martin, C. (2009). North Korea's Negotiating Position During Fifteen Years of Chronic Crisis: Continuities and Discontinuities. In T. H. Kwak & S. H. Joo (Eds.), *North Korea's Foreign Policy under Kim Jong II: New Perspectives*. Burrington, V.T. Ashgate Publishing Limited.
- Mazarr, M. (1995). Going Just a Little Nuclear: Nonproliferation Lessons from North Korea. *International Security*, 20(2), 92-122.
- McDonald, M. (2010, 8 November). South Korea Drops Its Call for Apology from North, *The New York Times*. Retrieved from http://www.nytimes.com/2010/11/09/world/asia/09korea.html? r=0
- Mearsheimer, J. (1990a). Back to the Future: Instability in Europe after the Cold War. *International Security*, 15(1), 5-56.
- Mearsheimer, J. (1990b). Why We Will Soon Miss the Cold War. *The Atlantic Monthly*, 226, 35-50.
- Mearsheimer, J. (1995). A Realist Reply. *International Security*, 20(1), 82-93.
- Mearsheimer, J. (2001). *The Tragedy of Great Power Politics*. New York W.W. Norton & Company, Inc.
- Mearsheimer, J. (2007). Structural Realism. In T. Dunne, M. Kurki & S. Smith (Eds.), *International Relations Theories: Discipline and Diversity*. Oxford: Oxford University Press.
- Mearsheimer, J. (2009). Reckless States and Realism. *International Relations*, 23(2), 241-256.

- Mearsheimer, J. (2011). Reckless States and Realism. In K. Booth (Ed.), *Realism and World Politics*. Abingdon: Routledge.
- Mercer, J. (1995). Anarchy and Identity. *International Organization*, 49(2), 229-252.
- Meyer, P. (2003). Russia's Response to the 2002-2003 North Korean Nuclear Crisis. *International Journal of Korean Studies*, 7(1), 57-78.
- Michishita, N. (2010). *North Korea's Military-Diplomatic Campaigns*, 1966-2008. Abingdon: Routledge.
- Michishita, N. (2012). Japan's Response to Nuclear North Korea. In G. Rozman (Ed.), *Joint U.S.-Korea Academic Studies: Asia at a Tipping Point: Korea, the Rise of China, and the Impact of Leadership Transitions* (Vol. 23, pp. 99-112): Korea Economic Institute.
- Miles, A. (2013). US Foreign Policy and the Rogue State Doctrine. London: Routledge.
- Millot, M., Molader, R., & Wilson, P. (1993). "The Day After...": Nuclear Proliferation in the Post-Cold War World: The RAND Corporation.
- Ministry of Foreign Affairs of China. (9 October 2006). Statement of the Ministry of Foreign Affairs of the People's Republic of China. from http://www.fmprc.gov.cn/eng/xwfw/s2510/t275508.htm
- Ministry of Foreign Affairs of China. (2009). Foreign Ministry Spokesperson Qin Gang's Statement on the Adoption of the UN Security Council 1874 on the DPRK Nuclear Test [Press release]. Retrieved from http://www.mfa.gov.cn/ce/ceil/eng/fyrth/t567565.htm
- Ministry of Foreign Affairs of Japan. (18 November 2003). Press Conference 2003: Questions Concerning Six-Party Talks on North Korean Issues [Press release]. Retrieved from http://www.mofa.go.jp/announce/press/2003/11/1118.html
- Ministry of Foreign Affairs of Japan. (1995). *The International Community in 1994*. Ministry of Foreign Affairs, Japan Retrieved from http://www.mofa.go.jp/policy/other/bluebook/1995/chp1.html.
- Ministry of Foreign Affairs of Japan. (1998a). Announcement by the Chief Cabinet Secretary on Japan's Immediate Response to North Korea's Missile Launch [Press release]. Retrieved from http://www.mofa.go.jp/announce/announce/1998/9/901-2.html
- Ministry of Foreign Affairs of Japan. (1998b). Japan-Republic of Korea Joint Declaration A New Japan-Republic of Korea Partnership towards the Twenty-First Century [Press release]. Retrieved from http://www.mofa.go.jp/region/asia-paci/korea/joint9810.html
- Ministry of Foreign Affairs of Japan. (2002). Japan-DPRK Pyongyang Declaration [Press release]. Retrieved from http://www.mofa.go.jp/region/asia-paci/n korea/pmv0209/pyongyang.html
- Ministry of Foreign Affairs of Japan. (2004). The Second Round of the Six-Party talks (Overview and Evaluation). from http://www.mofa.go.jp/region/asia-paci/n_korea/6party0402.html
- Ministry of Foreign Affairs of Japan. (2007, 9 October). Press Conference, 9 October 2007. from http://www.mofa.go.jp/announce/press/2007/10/1009.html
- Ministry of Foreign Affairs of Japan. (2010). Trilateral Statement Japan, Republic of Korea, and the United States of America [Press release]. Retrieved from http://www.mofa.go.jp/region/n-america/us/juk_js1012.html
- Ministry of Foreign Affairs of South Korea. (2010, 31 May). The Grand Bargain Initiative. North Korea Nuclear Issue/ 6-Party Talks. from <a href="http://www.mofa.go.kr/ENG/policy/kpen/nknuclear/documents/index.jsp?menu=m_2_0_20_10&tabmenu=t_3&sp=/webmodule/htsboard/template/read/engreadboard.jsp%_3FtypeID=12%26boardid=318%26seqno=309478

- Ministry of Patriots and Veterans Affairs of South Korea. (2007, 17 August). Address by President Roh Moo-Hyun on the 62nd Anniversary of the National Liberation of Korea.

 from http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd=CM">http://english.mpva.go.kr/basic/notice/view.asp?sgrp=&siteCmsCd=&topCmsCd
- Ministry of Unification. *Key Points in the July 4 South-North Joint Communique*. Ministry of Unification, Republic of Korea Retrieved from http://eng.unikorea.go.kr/content.do?cmsid=1806.
- Ministry of Unification. (1994, 15 August). President Kim Young Sam's 1994 Liberation Day Speech. *Ministry of Unification*. from http://www.unikorea.go.kr
- Ministry of Unification. (1998). Inaugural Address by Kim Dae Jung the 15th-Term President of the Republic of Korea. from Ministry of Unification, Republic of Korea http://eng.unikorea.go.kr/content.do?cmsid=1837&mode=view&cid=31906
- Ministry of Unification. (2004, 1 March). Address by President Roh Moo Hyun on the 85th March First Independent Movement Day. from http://eng.unikorea.go.kr/CmsWeb/viewPage.req?idx=PG0000000568&curNum=2
- Mitsumori, K. (1992, 5 September). Rapprochement Strengthens China's Hand in Region; Taming North Would Win IOUs from Japan, US, *The Nikkei Weekly (Japan)*. Retrieved from http://www.nexis.com
- Moltz, C. (2003). DPRK Briefing Book: Russian Policy on the North Korean Nuclear Crisis. from http://nautilus.org/publications/books/dprkbb/russia/dprk-briefing-book-russian-policy-on-the-north-korean-nuclear-crisis/
- Moltz, J. (2000). The Renewal of Russian-North Korean Relations. In J. Moltz & A. Mansourov (Eds.), *The North Korean Nuclear Program: Security, Strategy, and New Perspectives from Russia*. London: Routledge.
- Moon, C. I. (2000). The Sunshine Policy and the Korean Summit: Assessments and Prospects. *East Asian Review*, 12(4), 3-36.
- Moon, C. I. (2007). *The North Korean Nuclear Dilemma and the Six-Party Talks: A South Korean Perspective*. Paper presented at the International Symposium on Security Affairs 2007: The 10th Symposium (FY 2007), December 13, 2007, "North Korea's Nuclear Issues: Toward Peace and Security in Northeast Asia".
- Moon, C. I. (2008a). Diplomacy of Defiance and Facilitation: the Six Party Talks and the Roh Moo Hyun Government. *Asian Perspective*, 32(4), 71-105.
- Moon, C. I. (2008b). South Korea in 2008: From Crisis to Crisis. *Asian Survey*, 49(1), 120-128.
- Moon, C. I. (2009, 5 July). Obama's North Korea Policy and the June 15 South-North Joint Declaration. *East Asia Forum*. from http://www.eastasiaforum.org/2009/07/05/obamas-north-korea-policy-and-the-june-15-south-north-joint-declaration/
- Moon, C. I. (2011, 8 September). [Interview].
- Moon, C. I. (2012, 24 April). The Six Party Talks and Implications for Northeast Asia Nuclear Weapons Free Zone. *NAPSNet Special Report*. from http://nautilus.org/napsnet/napsnet-special-reports/the-six-party-talks-and-implications-for-a-northeast-asia-nuclear-weapons-free-zone/#axzz2j4RTfnst
- Moon, C. I. (2013, 18 July). Regime Survival is Primary Goal-Moon Chung-In Interview Part 2. *NK News*. from http://www.nknews.org/2013/05/regime-survival-is-primary-goal-moon-chung-in-interview-part-2/
- Moon, C. I., & Bae, J. Y. (2003). The Bush Doctrine and the North Korean Nuclear Crisis. *Asian Perspective*, 27(4), 9-45.

- Moon, C. I., & Lee, S. (2009). Military Spending and the Arms Race on the Korean Peninsula. *The Asia-Pacific Journal: Japan Focus* from http://www.japanfocus.org/-Chung_in-Moon/3333
- Moon, G. L. (2012, 19 January). 3 Allies Hope 6-Party Talks Resume, *Korea Joongang Daily*. Retrieved from http://koreajoongangdaily.joins.com/news/article/Article.aspx?aid=2947282
- Moore, G. (2008a). America's Failed North Korea Nuclear Policy: A New Approach. *Asian Perspective*, 32(4), 9-27.
- Moore, G. (2008b). How North Korea Threatens China's Interests: Understanding Chinese 'duplicity' on the North Korean Nuclear Issue. *International Relations of the Asia-Pacific*, 8(1), 1-29.
- Morgan, P. (2003). Deterrence Now. Cambridge: Cambridge University Press.
- Nakano, A. (2013, 5 November). S. Korea Park: Summit with Japan Pointless Unless 'Comfort Women' Issue Settled, *The Asahi Shimbun*. Retrieved from https://ajw.asahi.com/article/asia/korean_peninsula/AJ201311050056
- Nam, J. H. (1986). *America's Commitment to South Korea: the First Decade of the Nixon Doctrine*. Cambridge: Cambridge University Press.
- Nam, K. W. (1974). The North Korean Communist Leadership, 1945-1965: A Study of Factionalism and Political Consolidation. Alabama: University of Alabama Press.
- Nam, S. U. (2007, 28 August). [Column]Lee Myung-bak's North Korean policy initiatives, *The Hankyoreh*. Retrieved from http://english.hani.co.kr/arti/english_edition/e_opinion/232017.html
- Nanto, D., & Manyin, M. (2010). China-North Korea Relations *CRS Report for Congress* (pp. 1-22): The Federation of American Scientist.
- Nautilus Institute for Security and Sustainability. (2003). DPRK Briefing Book: DPRK 'Military First' Doctrinal Declaration, February 26, 2003. from http://nautilus.org/publications/books/dprkbb/military/dprk-briefing-book-dprk-military-first-doctrinal-declaration/#top
- Nautilus Institute for Security and Sustainability. (2004). DPRK Briefing Book: Official Statements for Second Round of Six-Party Talks. from http://nautilus.org/publications/books/dprkbb/multilateraltalks/dprk-briefing-book-official-statements-for-second-round-of-six-party-talks/#axzz2iV94MuTD
- Nikitin, M. (2013). North Korea's Nuclear Weapons: Technical Issues *CRS Report for Congress*: U.S. Department of State.
- Niksch, L. (2002). Issue Brief for Congress: North Korea's Nuclear Weapons Program *CRS Report for Congress* (Issue Brief for Congress ed.): The Federation of American Scientists.
- Niksch, L. (2005). The Six-Party Statement: Is It a Viable Roadmap or a Road to Nowhere? In L. Niksch, J. C. Park, H. S. Paik, L. G. Flake, Y. H. Park, W. Drennan & Y. K. Park (Eds.), *Implementing the Six-Party Joint Statement and the Korean Peninsula*. Seoul: Korea Institute for National Unification.
- Niksch, L., & Perl, R. (2007). North Korea: Terrorism List Removal? *CRS Report for Congress* (pp. 1-22): U.S. Department of State.
- Nolan, J., & Strauss, M. (1997). The Rogues' Gallery. *Brown Journal of World Affairs*, 4(1), 21-38.
- North Korea Tech. (2012, 15 April). English Transcript of Kim Jong Un's Speech, 15 April 2012. from http://www.northkoreatech.org/2012/04/18/english-transcript-of-kim-jong-uns-speech/

- North Korean Economy Watch. (2007, 4 October). 2nd South-North Korean Summit Joint Statement. *NK Brief No. 07-10-4-2.* from http://www.nkeconwatch.com/2007/10/04/2nd-south-north-korean-summit-joint-statement/
- North Korean Economy Watch. (2014, 7 May). Who uses Rason's Ports? Lease Confusion Explored (UPDATED). from http://www.nkeconwatch.com/2014/05/07/rason-port-lease-confusion/
- O'Carroll, C. (2013, 3 December). Jang Song Thaek Dismissed, Confidents Executed-S. Korean Lawmaker, NK News. from http://www.nknews.org/2013/12/jang-song-thaek-dismissed-confidents-executed-s-korean-lawmaker/
- Obama, B. (2009, 5 April). Obama's Statement on the North Korean Missile Launch, April 2009. from http://www.cfr.org/proliferation/obamas-statement-north-korean-missile-launch-april-2009/p19043
- Oberdorfer, D. (2001). The Two Koreas: A Contemporary History. New York: Basic Books.
- Oberdorfer, D. (2002, 10 November). My Private Seat at Pyongyang's Table, *The Washington Post*. Retrieved from http://www.nexis.com
- Office of the President. (2013a). The 18th Presidential Inaugural Address: "Opening a New Era of Hope" [Press release]. Retrieved from http://english1.president.go.kr/activity/speeches.php?srh%5Bboard_no%5D=24&srh%5Bdetail_no%5D=2617&srh%5Bdetail_no%5D=1
- Office of the President. (2013b). Remarks by President Park Geun Hye at the Joint Commissioning Ceremony of the Military Academies and ROTC Units [Press release]. Retrieved from <a href="http://english1.president.go.kr/activity/speeches.php?srh%5Bboard_no%5D=24&srh%5Bpage%5D=4&srh%5Bview_mode%5D=detail&srh%5Bseq%5D=2619&srh%5Bdetail_no%5D=3
- Oh, J. K. C. (1999). Korean Politics: the Quest for Democratization and Economic Development. Ithaca: Cornell University Press.
- Oh, K., & Hassig, R. (2002). Korea Briefing 2000-2001: First Steps Toward Reconciliation and Reunification. Armonk: M.E. Sharpe,Inc.
- Okano-Heijmans, M. (2008, 21 October). Japan as Spoiler in the Six-Party Talks: Single-Issue Politics and Economic Diplomacy towards North Korea. from http://www.japanfocus.org/-Maaike-Okano_Heijmans/2929
- Onuf, N. (1989). World of Our Making: Rules and Rule in Social Theory and International Relations. Columbia: University of South Carolina Press.
- Paik, H. S. (2011, 1 September). [Interview].
- Paik, H. S. (2013, 13 December). Jang Song Thaek's Execution. from http://38north.org/2013/12/hpaik121313/
- Pardo, R. (2014). North Korea-US Relations under Kim Jong II: the Quest for Normalization? Abingdon: Routledge.
- Park, C. H. (1975). *Major Speeches by President Park Chung Hee*. Seoul: Samhwa Publishing Co.
- Park, C. H. (1978). Towards Peaceful Unification: Selected Speeches & Interviews by Park Chung Hee. Seoul: Kwangmyong Publishing Co.
- Park, H. (2007) Military-First Politics (Songun): Understanding Kim Jong Il's North Korea. *Academic Paper Series 2007*: Korea Economic Institute of America.

- Park, H. (2012, 18 December). Bruce Cumings Says Sanctions Have Failed, Time for Dialogue with North Korea, *The Hanhyoreh*. Retrieved from http://www.hani.co.kr/arti/english_edition/e_northkorea/565936.html
- Park, H. J. (2006). The ROK's Role in Northeast Asia in the Wake of North Korea's Nuclear Test. In K. Y. Park, K. Oh, R. Hassig, B. G. Jun, D. Strab, H. J. Park & A. Romberg (Eds.), *The North Korean Nuclear Test and the Future of Northeast Asia*. Seoul: Korea Institute for National Unification.
- Park, I. H. (2011, 30 August). [Interview].
- Park, J. (2005). Inside Multilateralism: The Six-Party Talks. *The Washington Quarterly*, 28(4), 75-91.
- Park, J. K. (1979). Power Structure in North Korea. In J. K. Park & J. G. Kim (Eds.), *The Politics of North Korea*. Seoul: The Institute for Far Eastern Studies, Kyungnam University.
- Park, K. A., & Lee, S. C. (1992). Changes and Prospects in Inter-Korean Relations. *Asian Survey*, 32(5), 429-447.
- Park, S. G., & Kim, D. S. (2012, 2 November). NLL Nonsense Means Nothing At Home. Daily NK. from http://www.dailynk.com/english/read.php?cataId=nk00400&num=9989
- Park, S. S. (1993). Northern Diplomacy and Inter-Korean Relations. In J. Cotton (Ed.), *Korea Under Roh Tae Woo: Democratisation, Northern Policy, and Inter-Korean Relations*. New South Wales: Allen & Unwin.
- Park, S. W. (2010, 17 March). Strategic Posture Review: South Korea. from http://www.brookings.edu/research/articles/2010/03/17-korea-park
- Park, Y. (2012). The Structural Elements of North Korea's Insecurity Applying the "Regional Security Complex Theory". *The Korean Journal of Defense Analysis*, 24(3), 321-333.
- Payne, K. (2005). The Nuclear Posture Review: Setting the Record Straight. *The Washington Quarterly*, 28(3), 135-151.
- PBS. (2003, 21 February). Interview Stephen Bosworth/Kim's Nuclear Gamble, 21 February 2003, *PBS Frontline*,. Retrieved from http://www.pbs.org/wgbh/pages/frontline/shows/kim/interviews/bosworth.html
- Perry, W. (1999). *Review of United States Policy Toward North Korea: Findings and Recommendations, October 12, 1999*. U.S. Department of State Retrieved from http://1997-2001.state.gov/www/regions/eap/991012_northkorea_rpt.html.
- Pollack, J. (2003). The United States, North Korea, and the End of the Agreed Framework. *Naval War College Review*, *56*(3), 11-49.
- Posen, B. (1993). The Security Dilemma and Ethnic Conflict. Survival, 35(1), 27-47.
- Prime Minister's Office of Japan. (2010). Excerpt from the Press Conference by Mr. Yoshito Sengoku, Chief Cabinet Secretary, Held in the Evening of Wednesday, November 24, 2010 (Provisional Translation) [Press release]. Retrieved from http://www.kantei.go.jp/foreign/kan/topics/20101124 chief cabinet secretary e.pdf
- Pritchard, C. (2003a, 10 October). A Guarantee to Bring Kim into Line. from http://www.brookings.edu/research/opinions/2003/10/10northkorea-pritchard
- Pritchard, C. (2003b, 16 November). North Korean Nuclear Brinkmanship: Testing the Nuclear Nonproliferation Regime. *Presented at the Monterey Nonproliferation Strategy Group Conference November 16*, 2003. from http://www.brookings.edu/~/media/research/files/papers/2003/11/16northkorea-pritchard20031116.pdf
- Quinones, C. K. (2005a). Chapter 14: Kim Jong Il's "Strong and Great Nation" Campaign and the DPRK's Deterrence of the U.S. "Imperialist" Threat. In A. Mansourov (Ed.), *Bytes*

- and Bullets: Information Technology Revolution and National Security on the Korean Peninsula, Asia-Pacific Center for Security Studies. Honolulu: The Asia-Pacific Center for Security Studies.
- Quinones, C. K. (2005b). Dealing with Pyongyang In Search of a More Effective Strategy. *International Journal of Korean Studies*, 9(1), 27-49.
- Quinones, C. K. (2008, 14 March). The US-DPRK 1994 Agreed Framework and the US Army's Return to North Korea. from http://www.ncnk.org/resources/publications/Quinones_US_Army_in_DPRK.pdf
- Ra, J. Y. (2011, 26 November). North Korea's 'Military First' Politics Are Behind Recent Attacks, *The Huffington Post*. Retrieved from http://www.huffingtonpost.com/rajongyil/north-koreas-military-fir_b_788592.html
- Reuters. (2007, 10 February). North Korea Plan Based on "Libya Model": NY Times, *Reuters*. Retrieved from http://www.reuters.com/article/2007/02/10/us-korea-north-usa-idUSN1024155720070210
- Revere, E. (2012, 19 December). Park Geun Hye's Electoral Victory: A Sigh of Relief from Washington? , from http://www.brookings.edu/blogs/up-front/posts/2012/12/19-south-korea-president-revere
- Revere, E. (2013). Facing the Facts: Towards a New U.S. North Korea Policy (C. f. N. A. P. Studies, Trans.) (pp. 1-24). Washington, D.C.: The Brookings Institution.
- RIA Novosti. (2005, 21 September). N.Korea should end its nuclear program before getting reactor-Russia, *RIA Novosti*. Retrieved from http://www.nexis.com
- RIA Novosti. (2006, 9 October). Russia Condemns N. Korean Nuclear Test, *RIA Novosti*. Retrieved from http://www.nexis.com
- Rogin, J. (2013, 12 February). Few Korea Hands on Obama Administration's Asia Leadership Team. *The Cable, Foreign Policy.* from http://thecable.foreignpolicy.com/posts/2013/02/12/few korea hands in obama administration_s_asia_leadership_team#.URrjC2UF9gI.twitter
- Rozman, G. (2007). North Korea Nuclear Crisis and U.S. Strategy in Northeast Asia. 2007, 47(4), 601-621.
- Ruggie, J. G. (1998). What Makes the World Hang Together? Neo-Utilitarianism and the Social Constructivist Challenge. *International Organization*, *52*(4), 855-885.
- Russia & CIS Diplomatic Panorama. (2006, 12 July). Steps Provoking Pyongyang to Pull Out of Talks Should Be Avoided PUTIN, *Russia & CIS Diplomatic Panorama*. Retrieved from http://www.nexis.com
- Russia & CIS Diplomatic Panorama. (2010, 23 November). Interfax Russia & CIS Diplomatic Panorama: Attack on South Korean Island Must Be Condemned Lavrov, *Russia & CIS Diplomatic Panorama*. Retrieved from http://www.nexis.com
- Russian Press Digest. (2009, 29 May). Russia Still Opposes Sanctions Against North Korea, RusData Dialine-Russian Press Digest. Retrieved from http://www.nexis.com
- Sagalyn, D. (2012, 21 December). Analysts: South Korean Election May Trigger North Korean Provocation, *PBS*. Retrieved from http://www.pbs.org/newshour/rundown/2012/12/north-korea-2.html
- Sagan, S. (1996). Why Do States Build Nuclear Weapons?: Three Models in Search of a Bomb. *International Security*, 21(3), 54-86.
- Samore, G. (2003). The Korean Nuclear Crisis. Survival, 45(1), 7-24.
- Sanger, D. (1999, 17 March). N. Korea Consents to U.S. Inspection of a Suspect Site, *The New York Times*. Retrieved from http://www.nytimes.com/1999/03/17/world/n-korea-consents-to-us-inspection-of-a-suspect-site.html?pagewanted=all&src=pm

- Sanger, D. (2001, 8 March). Bush Tells Seoul Talks with North Won't Resume Now, *The New York Times*. Retrieved from http://www.nytimes.com/2001/03/08/world/bush-tells-seoul-talks-with-north-won-t-resume-now.html?pagewanted=print
- Sanger, D. (2003, 21 April). After Effects: Nuclear Standoff; Administration Divided Over North Korea, *The New York Times*. Retrieved from http://www.nytimes.com/2003/04/21/world/aftereffects-nuclear-standoff-administration-divided-over-north-korea.html
- Sanger, D., & Broad, W. (2007, 1 March). U.S. Had Doubts on North Korean Uranium Drive, *The New York Times*. Retrieved from http://www.nytimes.com/2007/03/01/washington/01korea.html?pagewanted=all&r="http://www.nytimes.com/2007/03/01/washington/01korea.html">http://www.nytimes.com/2007/03/01/washington/01korea.html?
- Schelling, T. (1960). The Strategy of Conflict. Cambridge: Harvard University Press.
- Schelling, T. (2008). Arms and Influence. New Haven: Yale University Press.
- Schneider, J. (2010). The Change toward Cooperation in the George W. Bush Administration's Nuclear Nonproliferation Policy toward North Korea. New York: Peter Lang.
- Schoff, J. (2005). Tools for Trilateralism: Improving U.S.-Japan-Korea Cooperation to Manage Complex Contingencies. Dulles: Potomac Books, Inc.
- Sheen, S. (2003). Grudging Partner: South Korea. Asian Affairs, 30(2), 96-103.
- Sheen, S. H. (2011, 31 August). [Interview].
- Shin, J. D. (2012). DPRK Perspectives on Korean Reunification after the July 4th Joint Communiqué. from Woodrow Wilson International Center for Scholars, http://www.wilsoncenter.org/sites/default/files/NKIDP_eDossier_10_DPRK_Perspectives on Korean Reunification after the July 4th Joint Communique 0.pdf
- Shin, J. D., Ostermann, C., & Person, J. (2013). North Korean Perspectives on the Overthrow of Syngman Rhee, 1960. from Woodrow Wilson International Center for Scholars http://www.wilsoncenter.org/sites/default/files/NKIDP_eDossier_13_North_Korean_Perspectives_on_the_Overthrow_of_Syngman_Rhee.pdf
- Smith, D. (2006). *Deterring America: Rogue States and the Proliferation of Weapons of Mass Destruction*. Cambridge: Cambridge University Press.
- Smith, H. (2000). Bad, Mad, Sad or Rational Actor? Why the 'Securitization' Paradigm Makes for Poor Policy Analysis of North Korea. *International Affairs Review*, 76(3), 593-617.
- Snyder, J. (1985). Perceptions of the Security Dilemma in 1914. In R. Jervis, R. Lebow, J. Stein & with contributions by Patrick M. Morgan and Jack L. Snyder (Eds.), *Psychology and Deterrence*. Baltimore: The John Hopkins University Press.
- Snyder, S. (2003). China-Korea Relations: Regime Change and Another Nuclear Crisis. In B. Glosserman & V. Fritschi (Eds.), *Comparative Connections: A Quarterly E-Journal on East Asian Bilateral Relations* (Vol. 5, pp. 93-100). Honolulu: Center for Strategic and International Studies.
- Snyder, S. (2004). Inter-Korean Relations: A South Korean Perspective. In S. Kim (Ed.), *Inter-Korean Relations: Problems and Prospects*. New York: Palgrave Macmillan.
- Snyder, S. (2007). U.S.-North Korean Negotiating Behavior and the Six-Party Talks. In S. H. Joo & T. H. Kwak (Eds.), *North Korea's Second Nuclear Crisis and Northeast Asian Security*. Aldershot: Ashgate Publishing Limited.
- Snyder, S. (2009a). Assessing North Korea's Strategic Intentions and Motivation. In T. H. Kwak & S. H. Joo (Eds.), *North Korea's Foreign Policy Under Kim Jong Il: New Perspectives*. Farnham: Ashgate Publishing, Ltd.

- Snyder, S. (2009b). *China's Rise and the Two Koreas: Politics, Economics, Security*. London: Lynne Rienner Publishers, Inc.
- Snyder, S. (2009c). Lee Myung Bak's Foreign Policy: a 250-Day Assessment. *Korean Journal of Defense Analysis*, 21(1), 85-102.
- Snyder, S. (2010). Korean Peninsula Policy of the Obama administration: Focus on Security Issues. Paper presented at the The 13th Symposium (FY 2010), December 2, 2010: "Deterrence and Dialogue-The Korean Peninsula after the Cheonan Incident.
- Snyder, S. (2013, 29 March). Why North Korea Regime is Scary, *CNN*. Retrieved from http://edition.cnn.com/2013/03/28/opinion/snyder-north-korea
- Song, J. (2011). Understanding China's Response to North Korea's Provocations. *Asian Survey*, 51(6), 1134-1135.
- Stangarone, T., & Hamisevicz, N. (2011). The Prospects for Economic Reform in North Korea after Kim Jong II and the China Factor. *International Journal of Korean Unification Studies*, 20(2), 175-197.
- Stein, J. (2009). Rational Deterrence against "Irrational" Adversaries? No Common Knowledge. In T.V. Paul, P. Morgan & J. Wirtz (Eds.), *Complex Deterrence:* Strategy in the Global Age. Chicago: The University of Chicago Press.
- Stein, J. (2013). Threat Perception in International Relations. In L. Huddy, D. Sears & J. Levy (Eds.), *The Oxford Handbook of Political Psychology (Second Edition)*. Oxford: Oxford University Press.
- Suh, J. J. (2006). Producing Security Dilemma out of Uncertainty: The North Korean Nuclear Crisis *Working Paper Series No. 8-06*.
- Suh, J. J. (2009). The Lee Myung-bak Government's North Korea Policy: A Study on its Historical and Theoretical Foundation. Seoul: Korea Institute for National Unification.
- T.V. Paul. (2009). Complex Deterrence: An Introduction. In T. V. Paul, P. Morgan & J. Wirtz (Eds.), *Complex Deterrence: Strategy in the Global Age*. Chicago: The University of Chicago Press.
- Takeda, Y. (2006). Putin's Foreign Policy toward North Korea. *International Relations of the Asia-Pacific*, 6(2), 189-207.
- Tan, E.-W. (2014). *The US Versus the North Korean Nuclear Threat: Mitigating the Nuclear Security Dilemma*. Abingdon: Routledge.
- Tantler, R. (2003). Classifying Evil: Bush Administration Rhetoric and Policy toward Rogue Regimes *Policy Focus No. 44* (pp. 1-40). Washington, D.C.: The Washington Institute for Near East Policy.
- The China Post. (2008, 15 October). Japan Nixes North Korea Aid Despite U.S. Delisting, *The China Post*. Retrieved from http://www.chinapost.com.tw/asia/japan/2008/10/15/178728/Japan-nixes.htm
- The Chosun Ilbo. (1999, 24 June). US Clears Kumchangri Suspicions, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/1999/06/24/1999062461522.html
- The Chosun Ilbo. (2003, 15 May). Roh-Bush Joint Statement (text), *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2003/05/15/2003051561001.html
- The Chosun Ilbo. (2006, 25 October). Kim Jong-il Plans No More Tests 'for Now', *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2006/10/25/2006102561008

- The Chosun Ilbo. (2009, 13 May). How Bosworth Differs from Hill, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2009/05/13/2009051300341.html
- The Chosun Ilbo. (2010a, 19 May). China Signals Displeasure with Cheonan Incident, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2010/05/19/2010051901203.html
- The Chosun Ilbo. (2010b, 16 August). Lee Lays Out 3-Stage Master Plan for Reunification, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html dir/2010/08/16/2010081600756.html
- The Chosun Ilbo. (2012a, 21 December). How Older Voters Swung Election for Park, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2012/12/21/2012122100789.html
- The Chosun Ilbo. (2012b, 13 April). Is N.Korea Abandoning Its Quest for Self-Reliance?, *The Chosun Ilbo*. Retrieved from http://english.chosun.com/site/data/html_dir/2012/04/13/2012041301329.html
- The Daily Yomiuri. (2002, 26 August). Editorial; Russian crutch failing North, *The Daily Yomiuri*, (*Tokyo*). Retrieved from http://www.nexis.com
- The Economist. (2009, 18 June). Obama, Iran, and North Korea: Meeting Thuggery with Coolness, *The Economist*. Retrieved from http://www.economist.com/node/13871987
- The Hankyoreh. (2010, 29 July). Russian Navy Expert Team's Analysis on the Cheonan Incident, *The Hankyoreh*. Retrieved from http://www.hani.co.kr/arti/english_edition/e_northkorea/432230.html
- The Hankyoreh. (2013, 14 March). North Korea Refers to President Parks as a "Venomous Swish of Skirt", *The Hankyoreh*. Retrieved from http://www.hani.co.kr/arti/english_edition/e_northkorea/578036.html
- The International Institute for Strategic Studies. (2011). North Korean Security Challenges: A Net Assessment *An IISS Strategic Dossier* The International Institute for Strategic Studies.
- The Korea Times. (2004, 13 May). Allies Trying to Persuade NK on Nuke Programs, *The Korea Times*. Retrieved from http://www.nexis.com
- The Korea Times. (2010, 2 June). Is Seoul Softening Stance on Pyongyang?, *The Korea Times*. Retrieved from http://www.nexis.com
- The National Institute for Defense Studies. (2010). Chapter 3: The Korean Peninsula Strengthened Military-First Politics and "Grand Bargain" Proposal. In Y. Sakaguchi (Ed.), *East Asian Strategic Review 2010* (1 ed., Vol. 69-95): The National Institute for Defense Studies, Japan.
- The New York Times. (2003, 25 April). Full Text of Brokaw's Interview with Bush, *The New York Times*. Retrieved from http://www.nytimes.com/2003/04/25/international/worldspecial/25BUSH-TEXT.html
- The Saenuri Party. (2012b). Platform: the Whole Statement [Press release]. Retrieved from http://www.saenuriparty.kr/web/eng/partyPlatform%20/party.do

- The United Nations. (2006). Security Council Condemns Nuclear Test By Democratic People's Republic of Korea, Unanimously Adopting Resolution 1718 [Press release]. Retrieved from http://www.un.org/News/Press/docs/2006/sc8853.doc.htm
- The Washington Post. (2007, 13 February). Joint Statement: Six-Party Talks on N. Korea Disarmament, *The Washington Post*. Retrieved from http://www.washingtonpost.com/wp-dyn/content/article/2007/02/13/AR2007021300508.html
- The White House. (2001). Address to a Joint Session of Congress and the American People [Press release]. Retrieved from http://georgewbush-whitehouse.archives.gov/news/releases/2001/09/20010920-8.html
- The White House. (2002a). *The National Security Strategy of the United States of America, September* 2002. The White House Retrieved from http://www.state.gov/documents/organization/63562.pdf.
- The White House. (2002b). *National Strategy to Combat Weapons of Mass Destruction*, *December 2002*. Retrieved from http://www.fas.org/irp/offdocs/nspd/nspd-wmd.pdf.
- The White House. (2002c, 29 January). President Delivers State of the Union Address. from http://georgewbush-whitehouse.archives.gov/news/releases/2002/01/20020129-11.html
- The White House. (2006, 9 October). President Bush's Statement on North Korea Nuclear Test, 9 October 2006. from http://georgewbush-whitehouse.archives.gov/news/releases/2006/10/20061009.html
- The White House. (2009, 5 April). Remarks by President Barack Obama in Prague as Delivered. from http://www.whitehouse.gov/the_press_office/Remarks-By-President-Barack-Obama-In-Prague-As-Delivered
- The White House. (2010a). Remarks by President Obama and President Lee Myung-bak of the Republic of Korea after Bilateral Meeting [Press release]. Retrieved from http://www.whitehouse.gov/the-press-office/remarks-president-obama-and-president-lee-myung-bak-republic-korea-after-bilateral-
- The White House. (2010b, 23 November). Statement by the Press Secretary on North Korea Shelling of South Korean Island. from http://www.whitehouse.gov/the-press-office/2010/11/23/statement-press-secretary-north-korean-shelling-south-korean-island
- The White House. (2010c). Statement by the Press Secretary on the Republic of Korea Navy Ship the Cheonan [Press release]. Retrieved from http://www.whitehouse.gov/the-press-office/statement-press-secretary-republic-korea-navy-ship-cheonan
- The White House. (2013a). Joint Declaration in Commemoration of the 60th Anniversary of the Alliance between the Republic of Korea and the United States of America [Press release]. Retrieved from http://www.whitehouse.gov/the-press-office/2013/05/07/joint-declaration-commemoration-60th-anniversary-alliance-between-republ
- The White House. (2013b, 8 April). Press Briefing by Press Secretary Jay Carney, 4/8/2013. from http://www.whitehouse.gov/the-press-office/2013/04/08/press-briefing-press-secretary-jay-carney-482013
- Thompson, D., & Matthews, N. (2011). Six-Party Talks and China's Goldilocks Strategy: Getting North Korea Just Right. In N. Finnemann, S. Howe, A. Kim & F. Lowe-Lee (Eds.), *Joint U.S.-Korea Academic Studies: Prospects for Emerging East Asian Cooperation and Implications for the United States* (Vol. 21, pp. 179-194): Korea Economic Institute.

- Toloraya, G. (2011, 1 September). Policy Forum 11-29: Russia's Stake in a Denuclearized Korean Peninsula-Will the Ulan-Ude Summit Help? . *NAPSNet Policy Forum*. from http://nautilus.org/napsnet/napsnet-policy-forum/russias-stake-in-a-denuclearized-korean-peninsula-will-the-ulan-ude-summit-help/#axzz2iV94MuTD
- Triplett II, W. (2004). Rogue State: How a Nuclear North Korea Threatens America. Washington DC: Regnery Publishing, Inc.
- U.S. Department of Defense. (2013). Joint Statement of the Japan, Republic of Korea, the United States Defense Ministerial Talks [Press release]. Retrieved from http://www.defense.gov/releases/release.aspx?releaseid=16054
- U.S. Department of State. (1994, 26 September). Address by President Bill Clinton to the UN General Assembly. *Remarks to the 49th Session of the United Nations General Assembly*. from http://www.state.gov/p/io/potusunga/207377.htm
- U.S. Department of State. (1997). Secretary of State Madeleine K. Albright Statement on "North Korea: Four Party Peace Talks" [Press release]. Retrieved from http://www.state.gov/1997-2001-NOPDFS/statements/970630.html
- U.S. Department of State. (2002). U.S. National Security Strategy: Prevent Our Enemies from Threatening Us, Our Allies, and Our Friends with Weapons of Mass Destruction. Retrieved from http://2001-2009.state.gov/r/pa/ei/wh/15425.htm.
- U.S. Department of State. (2003, 6 March). President Bush Urges Multilateral Effort on North Korea. from http://2001-2009.state.gov/p/eap/rls/rm/2003/18521.htm
- U.S. Department of State. (2005a). North Korea-U.S. Statement by the Head of the U.S. Delegation to the Six-Party Talks, Christopher R. Hill [Press release]. Retrieved from http://2001-2009.state.gov/r/pa/prs/ps/2005/53499.htm
- U.S. Department of State. (2005b, 19 September). Six-Party Talks, Beijing, China. from http://www.state.gov/p/eap/regional/c15455.htm
- U.S. Department of State. (2007, 3 October). Six Parties October 3, 2007 Agreement on "Second-Phase Actions for the Implementation of the Joint Statement". from http://2001-2009.state.gov/r/pa/prs/ps/2007/oct/93223.htm
- U.S. Department of Treasury. (2005). Treasury Designates Banco Delta Asia as Primary Money Laundering Concern Under USA Patriot Act [Press release]. Retrieved from http://www.treasury.gov/press-center/press-releases/Pages/js2720.aspx
- Understanding North Korea 2012. (2012). Seoul: Institute for Unification Education, Ministry of Unification.
- US-Russian Joint Statement on Korean Nuclear Non-Proliferation. (1992). "Joint Statement on Korean Non-Proliferation." US-Russian Joint Statement on Korean Nuclear Non-Proliferation Supporting the Efforts of the International Community to Counter the Proliferation of Nuclear Weapons on the Korean Peninsula [Press release]. Retrieved from http://www.fas.org/news/dprk/1992/920618-dprk-usia.htm
- Valencia, M. (2010, 29 July). 38 North Special Report: North Korea and the Proliferation Security Initiative (PSI). *Special Report 4*. from http://38north.org/wp-content/uploads/2010/07/38north_SR_PSI.pdf
- Vennet, E. (2002). Incorrect Assumptions: a Critical Review of U.S. Policy Toward North Korea (pp. 1-14). Washington, D.C.: National War College, National Defense University.
- Vorontsov, A. (2011, 22 April). Moscow's Views on the Korean Peninsula. *38 North*. from http://38north.org/2011/04/vorontsov042211/
- Wada, H. (2012, 27 February). Kim Jong II and the Normalization of Japan-North Korea Relations. from http://www.japanfocus.org/-Wada-Haruki/3699/article.html
- Walt, S. (2000). Two Cheers for Clinton's Foreign Policy. Foreign Affairs, 79(2), 63-79.

- Waltz, K. (1979). *The Theory of International Politics*. Reading: Addison-Wesley Publishing Company.
- Waltz, K. (1981). The Spread of Nuclear Weapons: More May Be Better: Introduction. *The Adelphi Papers*, 171, 1-39.
- Waltz, K. (1997). Evaluating Theories. *The American Political Science Review*, 91(4), 913-917.
- Waltz, K. (2000). Structural Realism after the Cold War. *International Security*, 25(1), 5-41.
- Waltz, K. (2001). *Man, the State, and War: a Theoretical Analysis*. New York: Columbia University Press.
- Warren, A. (2014). The Obama Administration's Nuclear Weapon Strategy: the Promises of Prague. Abingdon: Routledge
- Wendt, A. (1992). Anarchy is What States Make of It: The Social Construction of Power Politics. *International Organization*, 46(2), 391-425.
- Wendt, A. (1995). Constructing International Politics. *International Security*, 20(1), 71-81.
- Williams, P. (2008). Security Studies: An Introduction. In P. Williams (Ed.), *Security Studies: An Introduction*. Abingdon: Routledge.
- Wit, J. (2007). Enhancing U.S. Engagement with North Korea. *The Washington Quarterly*, 30(2), 53-69.
- Wit, J., Poneman, D., & Gallucci, R. (2004). *Going Critical: The First North Korean Nuclear Crisis*. Washington, D.C.: The Brookings Institution Press.
- Wood, P. (1993). The Strategic Equilibrium on the Korean Peninsula in the 1990s. In J. Cotton (Ed.), *Korea Under Roh Tae Woo: Democratisation, Northern Policy, and Inter-Korean Relations*. New South Wales: Allen & Unwin.
- World News Connection. (2003, 27 May). XNA: Hu Jintao, Putin Issue PRC-Russia Joint Declaration Bilateral Ties, Iraq, DPRK, World News Connection. Retrieved from http://www.nexis.com
- Xinhua News Agency. (2004, 26 June). China's Role in Six-Party Talks: Promoting Peaceful Talks, *Xinhua News Agency*. Retrieved from http://www.nexis.com
- Xinhua News Agency. (2005, 28 July). Stressing Abduction Issue Will Distract Six-Party Talks, *Xinhua News Agency*. Retrieved from http://news.xinhuanet.com/english/2005-07/28/content/3279535.htm
- Xinhua News Agency. (2010, 26 November). China Opposes any Military Acts in Exclusive Economic Zone without Permission, *Xinhua News Agency*. Retrieved from http://news.xinhuanet.com/english2010/china/2010-11/26/c_13624036.htm
- Yang, B. (2006). China and North Korea-The View from Beijing. In M. Nohr (Ed.), *Asia Program Special Report No. 134: Six-Party Stall: Are South Korea and China Part of the Problem or Part o the Solution?*: Woodrow Wilson International Center for Scholars.
- Yang, J. A. (2007, 23 May). 100 Days after Feb 13 Agreement, Has the U.S. Fallen Into a Ditch? *Daily NK*. from http://www.dailynk.com/english/read.php?cataId=nk00400&num=2102
- Yonhap News Agency. (2003). North Korea Handbook. London: M.E. Sharpe.
- Yonhap News Agency. (2005, 23 March). Hu Commits China's Role for Resuming Six-Party Talks, *Yonhap News Agency*. Retrieved from http://www.nexis.com
- Yonhap News Agency. (2007, 31 March). N. Korea Accuses Japan of Attempt to Scuttle Six-Party Talks, *Yonhap News Agency*. Retrieved from http://www.nexis.com
- Yonhap News Agency. (2010, 29 November). Full Text of President Lee's National Speech, *Yonhap News Agency*. Retrieved from

- http://english.yonhapnews.co.kr/national/2010/11/29/32/0301000000AEN20101129006400315F.HTML
- Yonhap News Agency. (2012a, 8 July). Kim Jong-un Calls on Nation to Catch Up with "Global Trends", *Yonhap News Agency*. Retrieved from http://english.yonhapnews.co.kr/national/2012/07/08/12/0301000000AEN201207080 02400315F.HTML
- Yonhap News Agency. (2012b, 8 November). North Korea Newsletter No. 235, *Yonhap News Agency*. Retrieved from http://english.yonhapnews.co.kr/northkorea/2012/11/07/93/0401000000AEN2012110 7006100325F.HTML
- Yonhap News Agency. (2012c, 27 December). North Korea Newsletter No. 242, *Yonhap News Agency*. Retrieved from http://english.yonhapnews.co.kr/northkorea/2012/12/26/23/0401000000AEN20121226007100325F.HTML
- Yonhap News Agency. (2012d, 5 November). Park Calls for Inter-Korean Summit, Cooperation Centers in Seoul, Pyongyang, *Yonhap News Agency*. Retrieved from http://english.yonhapnews.co.kr/northkorea/2012/11/05/31/04010000000AEN2012110 5008700315F.HTML
- Yoon, S. (1996). South Korea's Kim Young Sam Government: Political Agendas. *Asian Survey*, 36(5), 511-522.
- Zagare, F. (1990). Rationality and Deterrence. World Politics, 42(2), 238-260.
- Zagoria, D., & Kim, Y. K. (1975). North Korea and the Major Powers. *Asian Survey*, 15(12), 1017-1035.
- Zhang, J. (2013, 7 October). China Sticks to Denuclearization of Korean Peninsula, Chinese President Says, *CCTV*. Retrieved from http://english.cntv.cn/20131007/102485.shtml
- Zhao, Q. (1997). Chinese Foreign Policy in the Post-Cold War Era. World Affairs, 159(3), 114-129.
- Zhao, Q. (2006). Moving toward a Co-Management Approach: China's Policy toward North Korea and Taiwan. *Asian Perspective*, 30(1), 39-78.
- Zhebin, A. (2000). A Political History of Soviet-North Korean Nuclear Cooperation. In J. Moltz & A. Mansourov (Eds.), *The North Korean Nuclear Program: Security Strategy, and New Perspectives from Russia*. London: Routledge.